PEAKS Online 11 User Manual

PEAKS Online Team

PEAKS Online 11 User Manual

PEAKS Online Team

Publication date 2023

Table of Contents

1. Overview	1
1. How to Use This Manual	. 1
2. What is PEAKS Online ?	. 1
3. What is New in PEAKS Online 11 ?	
4. Supported Data Types	. 4
5 Ouick Walkthrough	4
5.1 PEAKS Online Main GUI	5
5.1.1 PFAKS Online User Ontions	5
5.2 Creating a PEAKS Online Project	. 5
5.2. Conducting an Analysis	. 0
5.4. Project Summery Monitoring and Desults	/
5.5. Opening an Existing Project	. 0
5.6. Patch Operation in Project listing	. 🦻
5.0. Datch Operation In Project listing	9
5.7. Analysis Properties	10
2. Management Configuration	11
1. Database Management and Configuration	11
2. Library Management and Configuration	12
2.1. Create a New Library	13
2.2. Import a New Library	14
3. Modification Management and Configuration	15
3.1. Modifications	15
3.1.1. Modification Types	16
3.2. Create a New Modification	16
4. Enzyme Management and Configuration	17
4.1. Create a New Enzyme	17
5. Instrument Management and Configuration	18
5.1. Create a New Instrument	18
6. TMT/iTRAQ and custom Label Q Methods	19
7. SILAC Q Methods	21
8. Workflows	23
9. Account Settings	24
9.1. My account	24
9.2. Algorithm Parameters	25
9.2.1. PEAKS PTM Algorithm Parameters	25
9.2.2. DIA DeenNovo Pentidome	26
9.2.3 De Novo Engine	26
3 Archive Project	27
1 Overview	27
2 Archive Project	27
2.1 Create a new folder in Archive Repository	27
2.1. Create a new forder in Archive Repository	$\frac{27}{28}$
3 Import Archived Project	28
3.1 Import Archived Projects Page	20
A Data View	29
1 Overview	31
2. Accessing the Date View	21
2. Accessing the Data View	21
5. Navigaining the Data view	22
J. Ourminer	33
1. Overview	33
1.1. Standalone <i>DeepNovo/De Novo</i> Analysis	33
2. DeepNovo/De Novo Analysis Workflow and Parameter Settings	34
2.1. Precursor and Fragment Error Tolerance	36
2.2. Enzyme Settings	36
2.3. PTM settings	37
3. Understanding PEAKS De Novo Sequencing Results	37

3.1. Result Filtration	37
3.2. Exporting <i>De Novo</i> Results	38
3.3. Summary View	38
3.3.1. Result Statistics	38
3.3.2. Low Quality Data Warnings	39
3.4. Peptide Table	39
3.5. Spectrum Annotation	40
3.6. Error Map	41
3./. Ion Table	41
4. FAIMS Data	41 12
0. Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER)	+5 //3
2 Database Search Workflow and Parameter Settings	43 43
2. Database Search Workhow and Farameter Settings	46
2.2. Precursor and Fragment Error Tolerance	46
2.3. Enzyme Settings	47
2.4. Database Settings	47
2.5. PTM Settings	47
2.6. Find Unspecified PTMs with PEAKS PTM	48
2.6.1. PTM signature Ions	48
2.7. Find More Mutations with SPIDER	49
2.8. PSM Filter	49
3. Understanding PEAKS Search Results	49
3.1. Result Filtration	50
3.1.1. Confident PTM Site Filter	50
3.1.2. Show Results for All Samples	50
3.1.3. Show Results for Specific Sample	50
3.2. Result Exporting	51
3.3. Summary View	52
3.3.1. False Discovery Rate (FDR) Curve	53
3.3.2. Result Statistics Tables	55
3.3.3. Summary Figures	55 55
5.5.4. Low Quanty Data warnings	55 56
3.4.1 Protein View	56
3.4.2 Protein Table	57
3 4 3 Global Protein Comparative View	58
3.4.4. Protein Coverage	58
3.4.5. Supporting Peptides	59
3.4.6. PTM Profiling	60
3.5. Peptide View	63
3.5.1. Peptide View Filters	63
3.5.2. Peptide Table	64
3.5.3. Peptide Global Comparative View	65
3.5.4. Peptide Menu	65
3.5.5. Peptide-Spectrum Matches	66
3.6. De Novo Only View	66
4. Running PEAKS PTM and SPIDER Separately	66
5. FAIMS Data	66
1. Our Streamlined Identification	68
2. Adding a Spectral Library from the Peremeters Page	00 68
2. Adding a Spectral Library Search Workflow and Parameter Settings	68 68
3.1 Precursor and Fragment Error Tolerance	69
3.1.1. Optimize Tolerance	70
3.1.2. Optimize for Short Gradient	70
3.2. Spectral Library	70
3.3. Protein Inference Settings	70

70
70
71
71
71
71
72
72
73
73
73
74
74
74
74
74
74
75
75
76
76
77
78
78
78
. 79
80
81
81
81
83
83
84
85
85
86
86
86 87
80 86 87 87
86 87 87
80 86 87 87 87 88
86 87 87 88 88 88
80 86 87 87 87 88 88 88 88
80 86 87 87 88 88 88 88 88 88
86 87 87 88 88 88 88 88 88 88
80 86 87 87 88 88 88 88 88 88 89 89
80 86 87 87 88 88 88 88 88 89 89 89
80 86 87 87 88 88 88 88 88 88 89 89 89 90
80 86 87 87 88 88 88 88 88 89 89 89 90 90
80 86 87 87 88 88 88 88 88 89 89 89 90 90 90
86 86 87 87 88 88 88 88 88 89 89 90 90 91 91
86 86 87 87 88 88 88 88 88 89 89 90 90 91 91 91
86 86 87 87 88 88 88 88 88 88 89 89 90 90 91 91 91 91 92
86 86 87 87 88 88 88 88 88 88 88 89 90 90 91 91 91 91 92 92
86 86 87 87 88 88 88 88 88 88 89 89 90 90 91 91 91 92 92 92
86 86 87 87 88 88 88 88 88 88 89 89 90 90 90 91 91 91 92 92 92 92 92
86 86 87 87 88 88 88 88 88 88 88 89 90 90 91 91 91 91 92 92 92 92 92 92
86 86 87 87 88 88 88 88 88 88 88 89 90 90 91 91 91 91 92 92 92 92 92 92 92 92
86 86 87 87 88 88 88 88 88 88 88 88 89 90 90 91 91 91 92 92 92 92 92 92 92 92 92 92 92

6.4.1. Protein View Filters	. 94
6.4.2. Protein Table	. 95
6.4.3. Global Protein Comparative View	. 95
6.4.4. Protein Coverage	. 95
6.4.5. Supporting Peptides	. 97
6.5. Peptide View	. 97
6.5.1. Peptide View Filter	. 98
6.5.2. Peptide Table	. 98
6.5.3. Peptide Global Comparative View	. 99
6.5.4. Peptide Menu	100
6.5.5. Peptide-Spectrum Matches	100
6.5.6. DIA Database Search Mirror Plot	100
6.5./. Error Map	101
6.5.0. Procursor Profile	101
6.5.9. Precursor Profile	101
6.5.10. Fragment Ion AIC	101
0.5.11. TIMSTOF IN Frame Mode	101
7. DIA De Novo Analysis worknow and Parameter Settings	102
7.1.1 Optimize Telerance	102
7.2 Enzyme Settings	103
7.2. Enzyme Settings	103
7.4 Da Nova Filter	103
8 Understanding PEAKS DIA De Novo Sequencing Results	103
8.1 Result Filtration	104
8.2 Exporting De Novo Results	104
8.3. Summary View	105
8.3.1. Result Statistics	105
8.3.2. Low Quality Data Warnings	106
8.4. Peptide Table	106
8.5. Spectrum Annotation	107
8.6. Spectrum View	108
8.6.1. Raw Scan	108
8.6.2. Mirror Plot	108
8.6.3. Pseudo Scan	109
8.7. Error Map	109
8.8. Ion Table	109
8.9. Precursor Profile	109
8.10. Fragment Ion XIC	109
8.11. TimsTOF In Frame Mode	110
9. FAIMS Data	110
8. Label Free Quantification (LFQ)	111
1. Overview	111
2. Setting Parameters	112
3. Understanding the LFQ Results	115
3.1. Protein Filters	116
3.1.1. Protein Filter	116
3.2. Summary View	117
3.2.1. Statistics of Filtered Results	117
3.2.2. The voicano Piot for Proteins	117
3.2.3. Protein Profile Heatmap	117
3.2.4. KI SHILL & WI/L SHILL	117
3.2.5. reterinage of missing values in each sample with or without ID transfer	117
3.2.0. FIORENT AREA COEFFICIENT OF VARIANCE.	11/ 11Q
3.2.7. Low Quality Data Wallings	110 119
3.3.1 Protein Table	118
3.3.2. Protein Volcano Plot	119
5.5.2. Hotem volumo Hot management and the second s	11/

3.3.3. Coverage view	119
3.3.4. Peptides view	120
3.3.5. Protein View Filter	121
3.4. Peptide View	122
3.4.1. Peptide Table	122
3.4.2. Feature Details	122
4. Ion Mobility Spectrometry (IMS) Data	123
4.1. Result Exporting	124
9. PEAKS Quality Control	68
1 Overview	68
2 OC Parameters	126
2. Qe l'arameters	120
2.1. Quality Control Attributes 2.11 DDA OC Attributes	127
2.1.2. DIA OC Attributes	127
2. DEAKS OC Depute	127
5. PEAKS QC Results	127
3.1. QC Summary Tab	127
3.2. QC Data Tab	128
3.2.1. Data QC Figures	129
3.3. QC Identification Tab	130
3.3.1. Identification QC Figures	131
3.4. QC Quantification Tab	133
3.4.1. Quantification QC Figures	134
3.5. QC Standard Peptides Tab	135
3.6. QC Results Exporting	136
3.6.1. Result Page Exporting	136
3.6.2. Analysis Export Page	137
10. Isobaric Labelling Quantification (TMT/iTRAQ)	138
1. Overview	138
2. Setting Parameters	139
2.1. Spectrum Filter:	139
2.2. Protein Filter:	139
2.3 Experiment Setting	139
3 Understanding the Results	141
3.1 Result Filtration	1/12
3.2 Result Exporting	1/2
3.2. Rusmary View	142
2.2.1 Statistics of Filtered Decults	143
2.2.2. The Malassa Dist for Destring	143
3.5.2. The volcano Piol for Proteins	143
3.3.3. Protein Profile Heatmap	143
3.3.4. Low Quality Data Warnings	143
3.4. Protein View	144
3.4.1. Protein Table	145
3.4.2. Protein Volcano Plot	145
3.4.3. Protein Coverage	146
3.4.4. Support Peptides	146
3.4.5. Protein View Filter	147
3.5. Peptide View	148
3.5.1. Peptide Table	148
3.5.2. Ion Match	148
3.5.3. Reporter Ion	148
4. Ion Mobility Data	149
11. Stable Isotope Labeling by Amino acids in Cell culture (SILAC)	150
1. Overview	150
2. Setting Parameters	151
3. Understanding the SILAC Results	154
3.1. Result Filtration	155
3.1.1 Protein Filter	155
3.2. Summary View	156

3.2.1. Statistics of Filtered Results	156
3.2.2. Protein Profile Heatmap	156
3.2.3. Ratio Distribution	156
3.2.4. The Volcano Plot for Proteins	156
3.2.5. Low Quality Data Warnings	156
3.3. Protein View	157
3.3.1. Protein Table	157
3.3.2. Protein Volcano Plot	158
3.3.3. Protein Coverage	158
3.3.4. Support Peptides	159
3.3.5. Protein View Filter	159
3.4. Peptide View	160
3.4.1. Peptide Table	160
3.4.2. Peptide View Filters	161
3.4.3. Feature Details	161
3.5. Result Exporting	163
12. Fraciton Assisted ID Transfer SILAC	164
1. Overview	164
2. Setting Parameters	165
3 Understanding the SILAC Results	167
3.1 Protein Filter	168
3.1. Floteni Fliter 3.2. Summary View	169
3.2. Summary View	169
3.2.1. Statistics of Thered Results	160
3.2.2. Protein Profile Heatman	160
2.2.4 Low Quality Data Warnings	160
3.2.4. Low Quality Data Wallings	109
5.5. Piolelli View	170
2.2.2. Duotoin Volcono Diot	170
3.3.2. Protein Volcano Plot	1/1
3.3.5. Protein Coverage	1/1
3.3.4. Support Peptides	172
3.3.5. Protein View Filter	172
3.4. Peptide View	173
3.4.1. Peptide Table	173
3.4.2. Peptide View Filters	174
3.4.3. Feature Details	174
3.5. Result Exporting	176
13. DeepNovo Peptidome Workflow	178
1. Overview	178
2. DDA Peptidome Workflow and Parameter Settings	178
2.1. Data Refinement Parameters	180
2.2. Precursor and Fragment Error Tolerance	181
2.3. Enzyme Settings	181
2.4. PTM Settings	181
2.5. Peptide Search Parameters	181
2.6. Report Filter	181
3. DIA Peptidome Workflow and Parameter Settings	178
3.1. Precursor and Fragment Error Tolerance	184
3.2. Enzyme Settings	185
3.3. PTM Settings	185
3.4. Peptide Search Parameters	185
3.4.1. Target Library	185
3.4.2. Target Database	185
3.5. Report Filter	185
4. Understanding Peptidome Results	186
4.1. Result Filtration	186
4.1.1. Show Results for All Samples	186
4.1.2. Show Results for Specific Sample	186

4.2. Result Exporting	187
4.3. DeepNovo Summary View	187
4.3.1. Result Statistics	187
4.4. DeepNovo Sequence Table	189
4.4.1. Peptide View Filters	189
4.4.2. DeepNovo Table	190
4.4.3. Annotated Spectrum	191
4.5. Peptide Search Summary View	192
4.5.1. Result Statistics	194
4.5.2. Summary View Figures	194
4.5.3. Low Quality Data Warnings	195
4.6. Peptide View	195
4.6.1. Peptide View Filters	196
4.6.2. Peptide Table	198
4.6.3. Peptide Global Comparative View	199
4.6.4. Annotated Spectrum	200
14. Instrument Daemon	201
1. What is an Instrument Daemon?	201
2. How to add Data using an Instrument Daemon	201
2.1. Data Page	201
2.2. Data Copy	204
2.3. Instrument Daemon Samples	205

Chapter 1. Overview

Welcome to PEAKS Online 11.

1. How to Use This Manual

This chapter provides an overview of PEAKS Online's features and describes a typical data analysis work flow in PEAKS Online. It is strongly recommended to read this chapter to get a big picture of what PEAKS Online provides and how to use PEAKS Online.

2. What is PEAKS Online ?

PEAKS Online is a distributed computing platform for data dependent acquisition (DDA) and data independent acquisition (DIA) shotgun proteomic mass spectrometry data analysis. PEAKS Online users get detail-oriented results with the advantage of using a powerful, shared resource. This solution is designed to harness the hardware of a powerful server to analyze data using sophisticated algorithms on a high-volume of mass spectrometry data simultaneously, all while providing an easy-to-use and easy-to-access interface without the need to install a local client.



The following is a brief overview of the distinctive features included in PEAKS Online :

Complete Analysis

PEAKS Online provides a complete solution for shotgun proteomics data analysis. It is a vendor neutral computing platform capable of reading in raw mass spectrometry data and public data formats. *De novo* sequencing, PEAKS DB (database searching)-based protein identification, PEAKS PTM (post translational modification) analysis, and SPIDER homology search are then used to identify the most likely peptide sequence that match the given spectra. Relative quantification by label-free quantification (LFQ) can also be performed. Intuitive result visualization tools are provided at every stage of analysis and results can be exported.

• Built-in Result Validation

Statistical tools are provided at every stage of analysis. *De novo* peptide sequencing results are confirmed by local confidence scores at the amino acid level. The "decoy-fusion" method is used to validate the peptide identifications automatically and a false discovery rate (FDR) is estimated and provided. Quantification results can also be analyzed by significance tests such as ANOVA or PEAKS Q.

• Result Visualization

Peptides are mapped to the identified proteins using an interactive protein coverage view. Clicking on peptides will bring up the peptide spectrum match that best explains the identification. Sample comparative coverage

view visualizes the frequency of supporting peptides of a protein from selected samples. Modification and mutations can be visualized and filtered based on certain thresholds. Quantification results can be visualized using heat maps.

• Accuracy and Sensitivity

The superiority of the PEAKS algorithms has been demonstrated by publications and third-party evaluations. By combining four complementary algorithms, including *de novo* sequencing, PEAKS DB, PEAKS PTM and SPIDER, the sensitivity is further improved.

• Multiple-Enzyme Project

To maximize a protein's sequence coverage, it is advantageous to use different proteolytic enzymes to digest the protein sample. PEAKS Online supports the use of different enzymes in different samples of a single project. The software will automatically use the enzyme specified in each sample and then combine all sample results together.

Label Free Quantification

Proteins are quantified based on peptide features detected from LC-MS/MS data by integrating the area under the curve (AUC) of the precursor ion or maximum feature intensity. An Expectation-Maximization (EM) based algorithm is used for feature detection, deconvolution and refinement. PEAKS Online uses an optimization model for simultaneous feature matching and retention time alignment. This approach allows PEAKS Online to quantify peptide features independent of MS/MS identification.

Reporter Ion Quantification

Reporter ion quantification with chemical labels at the MS2 (MS3) level is one of the two quantification modes that are supported by the optional PEAKS Q module of PEAKS Online. It is based on the relative intensities of the fragment peaks at fixed m/z values within an MS2 (MS3) spectrum. In this mode, isobaric chemical labels are introduced to several samples. The samples are then analyzed together in an LC-MS/MS experiment.

• SILAC (Stable Isotope Labeling by/with Amino acids in Cell culture)

In SILAC, the isotope labels with different mass values are introduced as two or more samples. The samples are then analyzed together in an LC-MS/MS experiment. The same peptide from different samples is recognized by a set of precursor ion peaks with similar retention time and mass differences within the retention time window. The ratio is calculated from the intensities of those peaks. PEAKS Online supports the analysis of Super-SILAC experiments that uses a mixture of SILAC-labeled cells as a spike-in standard for accurate quantification of other unlabeled or labeled samples.

• ID Trasnfer LFQ/SILAC (Stable Isotope Labeling by/with Amino acids in Cell culture)

Compared to the regular SILAC, ID transfer LFQ/SILAC can provide quantification work flow with high throughput and cov- erage. The unidentified precursors/pairs can be transferred from pre-run fractions in ID Transfer group.

Multiple Instrument Vendor Support

PEAKS Online supports the raw file formats of most major instruments, which avoids file format conversion. More importantly, the algorithms have been tuned for different instrument types to ensure optimal accuracy and sensitivity.

• Exporting Results

Analysis result can be exported to a variety of formats. The .csv format make it easy to view the results in spreadsheets and to analyze with in-house scripts or other third-party software. PEAKS Online also supports the ability to export data to standard result formats such as pepXML and mzIdentML.

3. What is New in PEAKS Online 11?

PEAKS Online 11 is a high throughput solution for proteomics allowing concurrent access from multiple users to support parallelism at project and data levels. It uses a distributed database to yield higher I/O performance and better fault tolerance. PEAKS online is ready to scale vertically and horizontally, add new worker nodes or database nodes to the system and see performance improvements right away.

PEAKS Online 11 offers the following improvements and added features:

• Flexible User allocation.

PEAKS Online now allows flexible user allocation through deletion of previous users and username editing.

• Increased Data Support.

We now support analysis of mgf files and Bruker .d folders containing .tsf files from timsTOF instruments operating in timsOFF mode.

• Enhanced Password Security.

We now have optional enhanced password security features including required reset for temporary passwords, password history uniqueness, password expiry and password similarity strictness.

• Quality Control.

PEAKS Online now has an automated quality control tool for DDA and DIA identification and label-free quantification workflows. Compared to the summary page from the database search of quantification results, the QC module in PEAKS Online provides more metrics to assess the quality of each LC-MS run.

• DDA Deep Learning Boost.

New optional Deep learning boost for DDA data analysis can increase peptide identification over 10%.

• Streamlined DIA Workflow.

PEAKS Online now offers a powerful and flexible solution for DIA analysis by allowing peptide identification using a spectral library search with or without a protein inference database direct database search or a combination of both for optimal peptide coverage. Then, unmatched spectra from the database search are analysed using *de novo* sequencing. By combining identification strategies we can provide increased sensitivity and and increased identification rate.

Enhanced Deep learning-based technology for predicting spectra, retention time, and collision cross-section values for Direct database search to improve sensitivity and accuracy including DIA with short gradient optimization.

• DeepNovo Peptidome.

The immunopeptidome is an attractive avenue for peptide-based vaccine and cancer immunotherapy development. As such, PEAKS Online now offers a specialized DeepNovo Peptidome workflow for both DDA and DIA data acquisition. De novo sequencing for these workflows is powered by a deep learning model trained specifically with large HLA peptide data set for retention time, fragmentation ion and ion mobility predictions. Furthermore, tailored second-round searches allow FDR to be estimated for *deep novo* peptides to increase accuracy of identification.

• DeepNovo Standalone Workflow.

PEAKS Online now offers a standalone Deep Novo sequencing workflow

• Spectral Library Generation improvements.

In PEAKS Online 11, we have expanded spectral library generation to support more workflow types. Spectral libraries can now be generated from De Novo, DB, and Peptidome search results for both DDA and DIA acquisition modes.

• PEAKS PTM improvements.

PEAKS provides comprehensive tools for PTM identification and PTM profiling. In PEAKS Online 11, we now offer two options for users to determine confident modification sites: minimal ion intensity and AScore. We also offer a new PTM profiling tool for direct visualization of quantitative information (eg. Abundance of modified and unmodified forms of the PTM sites identified).

PEAKS PTM searches also contain a new "Modified Peptide" tab for direct comparison of modified and unmodified peptide identification parameters.

• LFQ Normalization Improvements.

PEAKS Online 11 now uses a local RT normalization algorithm for TIC and internal standard protein normalization.

• User-specific Algorithm Parameters.

PEAKS now allows individual algorithm parameters to be controlled on the user-level. These parameters include confident signature ion requirements for PEAKS PTM, DIA DeepNovo parameters and default *de novo* search engine.

4. Supported Data Types

- **mzXML** : Any instrument type converted to this format is accepted. MSConvert from ProteoWizard is suggested for conversion.
- **mzML** : Any instrument type converted to this format is accepted. MSConvert from ProteoWizard is suggested for conversion.
- **mgf**: Any instrument type converted to this format is accepted.
- Thermo raw files : Thermo .raw files are accepted.
- Waters raw folders: The master node must be on a Windows operating system. Otherwise, conversion to mzXML is necessary. If loading directly from your computer, compress the folder into a zip file before uploading.
- Waters Mse: Waters Mse files are now accepted. The master node must be on a Windows operating system. Otherwise, conversion to mzXML is necessary.
- **Bruker .d folders:** Folders containing .baf, .tdf and .tsf files from Impact, Maxis, or timsTOF instruments are accepted. If loading directly from your computer, compress the folder into a zip file before uploading.
- Thermo DIA: Thermo DIA files are now accepted.
- Bruker DIA: Bruker DIA files are now accepted.
- WIFF DDA/DIA: WIFF DDA/DIA files are now accepted.

5. Quick Walkthrough

This section presents a quick walkthrough of a typical data analysis work flow. For step-by-step instructions and detailed explanation of the results, refer to Chapters 2-6.

5.1. PEAKS Online Main GUI

The main graphical user interface (GUI) of PEAKS Online can be accessed from a Web Browser. Upon entering PEAKS Online, you must sign in first. There are two types of users: Administrator or User. The user manual for Administrators is provided separately.

5.1.1. PEAKS Online User Options

PEAKS Online User accounts can only be created by the Administrator(s). After logging in, PEAKS Online users can access their completed projects or create a new project from the home page. For convenience, Import Project, Create New Project and Download an be accessed from the top left. Once logged in as a PEAKS Online user, the following items can be accessed from the top menu:

PROJECT ^	CT ✔ SETTINGS ▲ ACCC	IIN 🗸 ACCOUNT 🔨 HELF	HELP 🔨
New Project	Databases	My account	About
My Projects	Spectral Libraries	Algorithm Parameters	Contact
Shared Projects	Modifications	Log out	
	Enzymes		
	Instruments		
	TMT/iTRAQ Q Methods		
	SILAC Q Methods		
	Workflows		

1. Project

- a. New Project: Users can create a new project by clicking here to upload data and set parameters.
- b. My Projects: Users can access all projects that were created and run on their user account.
- c. Archived Projects: Users can archive projects that were created by their account.
- d. Shared Projects: Users can access all projects that were shared to or by their account.
- 2. Settings
 - a. Databases: Users can access all databases uploaded on their user account and databases configured by the administrator.
 - b. Modifications: Users can view default modifications included in PEAKS Online and can also, access all custom modifications configured on their user account and modifications configured by the administrator.
 - c. Enzymes: Users can view default enzymes included in PEAKS Online and can also, access all custom enzymes configured on their user account and enzymes configured by the administrator.
 - d. Instruments: Users can view default instruments included in PEAKS Online and can also, access all custom instruments configured on their user account and instruments configured by the administrator.
 - e. TMT/iTRAQ Q Methods: Users can view the default Label Q Methods included in PEAKS Online and can also, access all custom Label Q Methods configured on their user account and Label Q Methods configured by the administrator.

- f. SILAC Q Methods: Users can view the default SILAC Q Methods included in PEAKS Online and can also, access all custom SILAC Q Methods configured on their user account and SILAC Q Methods configured by the administrator.
- g. Workflows: Users can access all custom workflows made on their user account and workflows made by the administrator.
- h. Libraries: Users can generate or import spectral libraries.Users can access all libraries configured on their user account and libraries configured by the administrator.
- 3. Account
 - a. My Account: Users can manage their account name, password and associated email and set priority.
 - b. Algorithm Parameters: Users can manage their individual algorithm parameters for confident signature ion requirements for PEAKS PTM, DIA DeepNovo parameters and default *de novo* search engine.
- 4. Help
 - a. About: Users can view End-User PEAKS Online Software License Agreement.
 - b. Contact: Links to Bioinformatics Solutions Inc. contact information.

5.2. Creating a PEAKS Online Project

To create a new PEAKS Online project from raw data files, perform the following steps:

- Select the a icon from the top left or select "New Project" from the menu. This will prompt the "New Project" page where the project can be named and parameters set.
- 2. Name the new project and/or the description. Select either an existing work flow or create a project from scratch. Click Next button.
- 3. Select "Local" tab and click 💿 to bring up a browser window where files can be selected for uploading. If the administrator has specified folders on the server that can be accessed for data uploading they will be listed under the "Remote" tab. Click "In Project" tab to bring in data from the existing project if this is a new analysis for an existing project.
- ^{4.} Place the selected data from the list into samples: use \checkmark to place all files in a new sample as fractions; use
 - \Rightarrow L to put them in an existing sample; or put them in individual samples for each file using \rightarrow .
- 5. For each sample, specify the name and sample details, including "Enzyme" name, "Activation Method", and "Instrument" type, or specify the sample details on the top row if all samples have been prepared and acquired with the same method.

USER MANUAL DATA									
							CANCEL	АСК	NEXT
Select Data			Enzyme		Activation Meth	od	Instrument		
Local Remote In Project		Sample Name	Specified by S	•	HCD	Ŧ	Orbitrap (Orbi-Orbi)	•	
Select All Q	ן ו	Sample 1	Trypsin	*	HCD	~	Orbitrap (Orbi-Orbi)	*	×
		BSA-Trypsin-1.RA	W				\uparrow	\checkmark	×
Click to upload data	.O .	Sample 2	Lys C	*	HCD	•	Orbitrap (Orbi-Orbi)	*	×
BSA-GluC-1.RAW X	⇒≞∡	BSA-LysC-1.RAW					\uparrow	\downarrow	×
	*}₽								
	⇒≞								
	→ ^(*) 且								
	J								
							UPLOAD SAMPLES	RE	MOVE ALL

- 6. Click the Next button to select an Analysis for the project.
- 7. Populate the parameters for the analysis. If desired you can save this work flow for future projects.
- 8. Submit Project

5.3. Conducting an Analysis

To conduct a PEAKS Online Search, a Label-free quantification (LFQ) work flow for a complete identification plus quantification analysis is shown as an example. Refer to Chapter 8, *Label Free Quantification (LFQ)* for more details about conducting an LFQ analysis. Refer to Chapter 5, *Peptide De Novo Sequencing* and Chapter 6, *Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER)* for more details about conducting identification analyses.

Select All Q			
		Group	Color
	→°	▼ Group 1	× *
		Sample 1	×
	→ œ.	Sample 2	×
		Group 2	×
	-	Sample 3	×
	- 14	Sample 4	
	.(7)		
	* ‱		-
tch Between Runs			REMOVE ALL
tch Between Runs Aass Error Tolerance: 20 Tolerance Unit:	PPM • Ret	ention Time Shift Tolerance(min):	REMOVE ALL
tch Between Runs Mass Error Tolerance: 20 Tolerance Unit: Feature Intensity 2: 0 ;	PPM - Ret	ention Time Shift Tolerance(min):	REMOVE ALL
tch Between Runs Mass Error Tolerance: 20 Tolerance Unit: Feature Intensity ≥: 0 : RT Range: 0 : ≤ RT ≤	PPM - Ret	ention Time Shift Tolerance(min): 1 Base Sample: Sample	REMOVE ALL
tch Between Runs Mass Error Tolerance: 20 Tolerance Unit: Feature Intensity ≥: 0 : RT Range: 0 : ≤ RT ≤ ptide feature p* Avg. Area ≥:0, Quality ≥:0, 1 ≤ Charge ≤10, 1 ≤ Charge ≤10, 1	PPM Ret Max Peptide ID Counts 0 per a	ention Time Shift Tolerance(min): 1 Base Sample: Sample group, and detected in at least samples	REMOVE ALL Auto Detect

The different options when setting up a full LFQ work flow are described below (see screenshot above):

- 1. Depending on the instrument used for data acquisition, set the parent and fragment error tolerance.
- 2. If the proteolytic enzyme was specified for each sample at the project creation step, users can select "specified by each sample". This makes it possible to use multiple enzymes in a single project and a single search.
- 3. Select a protein sequence database and the taxonomy for the database search. Select a contaminant database to be included in the database search. A decoy database will be created based on the target database by default. Decoy-fusion is an enhanced target-decoy method for result validation with FDR. Decoy-fusion appends a decoy sequence to each protein as the "negative control" for the search. Please see BSI's web tutorial (http://www.bioinfor.com/fdr-tutorial/) for more details.
- 4. Specify the fixed PTMs and a few (no more than 5) common variable PTMs expected in the sample. To prevent long search times, select only the most frequent PTMs in the sample for PEAKS DB, and check the PEAKS PTM option to search for unspecified modifications.
- 5. *De novo* sequencing and PEAKS DB search will be performed using the same parameters listed above. See Chapter 5, *Peptide De Novo Sequencing* and Chapter 6, *Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER)* for more details on parameter setting for *de novo* sequencing and database searching.
- 6. Specify parameters and grouping for LFQ analysis. See Chapter 8, *Label Free Quantification (LFQ)* for more details.

5.4. Project Summary, Monitoring and Results

When uploading data using a browser, after the submit button is clicked, the Upload Fractions page will appear. Be sure to keep the browser open as the data loads to server.

Upload Fractions	
Do not close the website during uploading Severe_SMA_1.raw uploading	
Severe_SMA_2.raw uploading	
Severe_SMA_3.raw uploading	
Wild_Type_1.raw uploading	
Wild_Type_2.raw uploading	
Wild_Type_3.raw uploading	

Once the data upload is complete, the progress page will appear, where the user can view the entire project analysis and monitor each individual step.

• User can change the priority for each task. The highest number means the highest priority. And the task with a high priority will finish before the task with a low priority.

	DONE	IN PROGRESS	FAILED	CANCELLED	Priority 3 💌	CANCEL	VIEW/CHANGE
Data Loading							PROCESSING
Data Refinement							PENDING
De Novo							PENDING
Database Search							PENDING
LFQ							PENDING

When the project is processed and tasks are completed, results will populate and can be viewed.

5.5. Opening an Existing Project

After launching PEAKS Online, existing projects can be accessed in "My Projects". Alternatively, projects that have been shared by the administrator or other users can be accessed from "Shared Projects".

From "My Projects" page, users can observe all their projects that have been completed and the projects currently being processed. Clicking on a project shows all analyses under this project. Click 💿 to archive the project.

Click 💄 to share the project with others. 🚢 means that the project is shared with other users. To edit the project

name, click 🧪	• . To delete a	project, click					
My Projects	Search		CLEAR				
Project Name 🗸	Description	Created	Last Updated	#MS Runs	Progress	Actions 💽	Select All
WildLFQ		May 15 2020, 08:29:00 pm	May 19 2020, 09:37:30 am	6	Done		

5.6. Batch Operation in Project listing

In the project list page, batch operations can be conducted. Multiple projects or all projects can be selected using the check box.

Actions 💽 🧻	Select All
0 1 / 1	

• User can achieve, share, edit or delete the selected projects suing the action buttons

When the projects are processed and tasks are completed, results will populate and can be viewed.

5.7. Analysis Properties

Open any project and click one analysis, the "Analysis Properties" will show up and display the parameters used in the analysis.

The following tables will show up according to the different types of analysis:

- Table 1: Data samples used in the analysis
- Table 2: Data Refinement
- Table 3 : DeNovo Parameters
- Table 4 : ID Parameters
- Table 5 : Q Parameters

Analysis Properties

Table 2: Data Refinement	
ection	
ime Correction	
sociation	
novo Parameters	

Table 2: Data Refinement	
tem	Value
Mass Correction	Yes
Retention Time Correction	N/A
Chimera Association	Yes
Table 3: Denovo Parameters	
tem	Value
Mass Error Tolerance	Precursor:15 PPM; Fragment: 0.5 DA
nzyme	Specified by each sample
Additionations	Carbamidomethylation(F); Deamidation (NQ)(V);
viouncations	Oxidation (M)(V);
Table 4: ID Parameters	
tem	Value
Mass Error Tolerance	Precursor:15 PPM; Fragment: 0.5 DA
inzyme	Specified by each sample; Semi
4 - 1161 41	Carbamidomethylation(F); Deamidation (NQ)(V);
viouncations	Oxidation (M)(V);
Database	NEW_uniprot
able 5: Q Parameters	
tem	Value
) Method	ID-directed LFQ
W and Outlier Removal	No: Vee

Chapter 2. Management Configuration

From the top menu, within the Settings drop-down menu, PEAKS Online users can access options to configure databases, modifications, enzymes, and instruments for use in their PEAKS analyses. Users can only edit and delete the ones they create.

1. Database Management and Configuration

To use the PEAKS DB, PEAKS PTM and SPIDER functionalities to identify proteins, PEAKS Online must have access to a protein database in FASTA format. To add, edit or delete a database, users can access the "Manage Database" page by selecting "Database" from the Settings drop-down menu. Users can also add a database from within an analysis workflow by selecting the "Create New Database" option from the Target Database drop-down menu on the analysis parameters page.

• Add a database

Open the "Add database" dialog by clicking the + on the "Manage Databases" page, or from the "Create New Database" option within a workflow. Give the new database a name, choose the format to match where the file came from (e.g. UniProtKB, Ensembl, etc.), and then click on the "Choose Database File button" to specify the file path of the database's FASTA file. Finally, click Upload. The database will then be uploaded and added to the list of configured databases.

Note

Once uploaded, the FASTA file can't be changed. The database must be deleted and a new one created with the updated FASTA file

Swiss-Prot_Human_Cano	onical	Format UniProtKB (Swiss-Prot, TrEMBL,) 🔹
Accession Pattern	Description Patte	m
Copy Protein Header Here (Start >sp L0R819 ASURF_HUN	ing with >) IAN ASNSD1 upstr	ream open reading PARSE HEADER
Accession Result		Description Result
LOR819 ASURF_HUMAN		ASNSD1 upstream open reading frame pro
Comments		
Comments Database File		CHOOSE DATABASE FILE

• Edit a database

From the "Manage Databases" page, click
button in the Action column of a database to open the "Edit Database" dialog. Change the necessary parameters (e.g. name, format) and click Save.

• Delete a database

From the "Manage Databases" page, click 🧵 button in the Action column to delete a database permanently.

• View database Information

From the "Manage Databases" page, click \checkmark button in the Action column to view database information. By default all species are selected. But You can also select the species you want by clicking.

Human_Ecoli	UniProtKB (Swiss-Prot, TrEMBL,)	93078	Human_Ecoli.fasta	admin	Done	6 / Î	^
Accession Pattern ^>((tr\) (sp\))?(\S+)	Description Pattern ^\S+\s+(.*)\$			Taxonomy	,	#Amino Acids	
A0AUZ9-2 KAL1L_HUMAN	lsoform 2 of KAT8 reg sapiens GN=KANSL1	gulatory NSL com L	plex subunit 1-like protein OS=Homo	Select Taxo	nomy	73147848	
A0AUZ9-3 KAL1L_HUMAN	Isoform 3 of KAT8 reg sapiens GN=KANSL1	gulatory NSL com L	plex subunit 1-like protein OS=Homo	93078 fast	a sequences	_	
A0AUZ9-4 KAL1L_HUMAN	Isoform 4 of KAT8 reg sapiens GN=KANSL1	gulatory NSL com L	plex subunit 1-like protein OS=Homo				
A0AV02-2 S12A8_HUMAN	Isoform 2 of Solute c GN=SLC12A8	arrier family 12 m	ember 8 OS=Homo sapiens				
A0AV02-3 S12A8_HUMAN	Isoform 3 of Solute c GN=SLC12A8	arrier family 12 m	ember 8 OS=Homo sapiens				

2. Library Management and Configuration

In order to complete a Spectral Library Search Workflow, users must first create or import a spectral library. Configuration of spectral libraries is accessed by selecting the "Spectral Libraries" option from the Settings dropdown menu, which will open the Manage Spectral Libraries page below. For more information on the use of Spectral Libraries please see the Chapter 7, DIA Streamlined IdentificationSection 2, "Adding a Spectral Library from the Parameters Page".

Manage Spectral	Libraries	+ -	Search			CLEAR			
Name	#Entry	Database	Names	Created time	Owne	er ↑	RT	State	Actions
ABRF-2017	46111	02-28-IPRG2 species)	2017 (all	Feb 28 2021, 11:06:44 pm	admin		Map to iRT	Done	🛆 O 🗡 🔋
1% psm 24 Fractions TimsTOF	483263	uniprot- human_prot 21032020 (a	eome_reviewed_ all species)	Feb 02 2021, 12:30:49 pm	admin		Map to iRT	Done	60/1
1% psm ABRF	45145	Human_Non species)	n_Redudant (all	Feb 02 2021, 12:30:27 pm	admin		Map to iRT	Done	6 0 / Î
1% psm Monash HLA	21765	uniprot- human_prot 21032020 (a	eome_reviewed_ all species)	Feb 02 2021, 12:31:13 pm	admin		Map to iRT	Done	60/1
1% psm MS1_MS2 Thermo	283244	uniprot_spro species)	ot_49 (all	Feb 02 2021, 12:31:43 pm	admin		Map to iRT	Done	🛆 O 🖍 🔋
1% psm westlake	837666	Guo_swatch	(all species)	Feb 02 2021, 12:32:11 pm	admin		Map to iRT	Done	60/1

From the Manage Spectral Libraries page, PEAKS Online Users can perform following actions:

• Create a new library

Click + next to "Manage Spectral Libraries" to open the "Create Libraries" dialog. Configure the Libraries accordingly and click Save. See Create New Library section below for more details on choosing settings.

• Download a Library

Click 🚯 button in the Action column of a customized library to download library.

• Edit a Library

Click
Vertice the state of the state of

• Delete a Library

Click 📋 button in the Action column of a library to delete the library permanently.

2.1. Create a New Library

Libraries can be created using pre-existing analysis results from a PEAKS De Novo DDA, PEAKS DB DDA, or a PEAKS DB DIA, PEAKS De Novo DIA, Peptidome and Deep Novo workflow.

In order to create a Spectral Library from an existing analysis follow these steps:

- 1. Select the "Spectral Libraries" option from the Settings drop-down menu.
- ^{2.} Click the + button to open the "Create Library" dialog.

lect Froject	S/Allalysis.		_		Adding Projects/Analysis	
Search DA De Novo	CLEAR		*		Spectral Library Name Spectral Library	
DDA De Novo	io Data Test	~				
DDA DB	OF Data1	~)			
DIA De Novo	Using Daemon	~		-		
DIA DB	oject 2022-11-30,15:00	~		⊉		
Peptidome	iject 2022-12-05,11:29	~				
DeepNovo		~)			
Copy Thermo D	ata	~)			
Copy TimsTOF I	Data	~				

- 3. Select the workflow type from the workflow drop-down menu below the search bar. These workflow types include DDA De Novo, DDA DB, or DIA De Novo, DIA DB, Peptidome, and DeepNovo.
- 4. Find the analysis you wish to create a library from in the list below. You can use the search bar to further filter the list of analyses.
- 5. Click the \boxdot button to move the analysis to the right side of the pane.
- 6. Enter a name for the new Spectral Library. This should be a descriptive name as it will be used to identify the library during project creation.
- 7. Set the parameters for library creation. These parameters will be different depending on the workflow being used to create the library.

For DDA Denovo workflows the parameters are:

• ALC Threshold: Only denovo candidates with an average local confidence score equal to or greater than this value will be used for library generation.

Spectral Library Name			
Denovo Library			
ALC Threshold >	70	:	

For DDA DB Search the parameters are:

- Include DE NOVO Only: If this is checked, denovo only candidates will also be included in library generation.
- ALC Threshold: This parameter is only used if the "Include DE NOVO Only" checkbox is checked. Only denovo candidates with an average local confidence score equal to or greater than this value will be used for library generation.
- Retention time: The map to iRT function maps experimental retention times for peptides to an indexed RT (iRT) to account for chromatographic differences between runs (e.g. different gradient lengths).

:
Map to iRT

• Retention time: The map to iRT function maps experimental retention times for peptides to an indexed RT (iRT) to account for chromatographic differences between runs (e.g. different gradient lengths).

	Spectral Library Name			
	DIA DB Library			
		Use original PT		Map to iPT
1		Ose original Ri	U	Map to IKI

8. Click the "Start" button to begin library generation.

2.2. Import a New Library

From the Manage Spectral Libraries page, click the import button 5 to open the "Import Library" dialog. The following information can be configured for a new custom library:

- Name: This name will appear in the library list for future use after it is saved.
- Spectral Library File: The spectral library file in TSV format.
- **Spectral Library File Information:** The meta information for the spectral library in the format of an INFO file (plain text information). This information is optional but can be added to describe the contents of the library for future reference.

Import Spectral Library	
Name	
Spectral Library File	CHOOSE SPECTRAL LIBRARY FILE
Spectral Library File Information (Optional)	CHOOSE META INFO FILE
	CANCEL IMPO

3. Modification Management and Configuration

To view PTM details for default and customized modifications, select the "Modifications" option from the Settings drop-down menu.

From the Manage Modifications page, PEAKS Online Users can scroll through lists of default modifications using the "All", "Common", "Custom", etc., options at the top of the page. Users can also perform the following actions:

• Create a new modification

Click + next to "Manage Modifications" to open the "Create Modification" dialog. Configure the PTM accordingly and click Save. See the Create New Modification section below for further details.

• Edit a modification

Click *integrability* button in the Action column of a customized modification to open the "Edit Modification" dialog. Change the necessary parameters and click Save. Default modifications cannot be edited.

• Delete a modification

Click **i** button in the Action column of a customized modification to delete the PTM permanently. Default modifications cannot be deleted.

• View a modification

By clicking \checkmark button in the Action column of any PTM, the parameters configured for that modification can be viewed.

3.1. Modifications

By default, PEAKS Online includes modifications in the Unimod database as built-in modifications. Unimod modifications are categorized into four lists:

• Artificial

The "Artificial" list contains modifications that can only be artificially induced for a specific purpose, including various chemical and isotopic tags for quantification or other experiments.

• Common

The "Common" list contains commonly observed modifications, including naturally occurring PTMs, modifications induced by standard sample preparation, and common artifacts.

• Uncommon

The "Uncommon" list contains less commonly observed modifications, including rare PTMs and artifacts.

• Glycosylation

The "Glycosylation" list contains commonly observed glycosylations.

Additional modification categories offered are "All" and two other lists of modifications:

• Recent

The "Recent" list keeps track of previously selected modifications. By default, modifications are sorted by the last time they were selected. You can conveniently select the modifications that are most frequently used for your analysis from this list.

• Customized

The "Customized" list shows all the user-defined modifications. User-defined modifications can be modified and/or deleted.

Note

Built-in modifications cannot be modified or deleted.

3.1.1. Modification Types

In PEAKS online, modifications can be classified as one of three types:

- PTM
- Isotopic
- Artifact

The new PTM categories Isotopic and Artifact describe types of artificial modifications such as those used for chemical labelling. For modifications which do not change the chemical properties and retention time of the unmodified peptide, Isotopic PTM should be chosen. Otherwise, the Artifact type should be chosen.

3.2. Create a New Modification

Click + button to open the "Create Modification" dialog.

Note: The other way is at the parameter page when submitting a new analysis, click "ADD/REMOVE MODIFI-CATION" then click "CREATE MODIFICATION".

The following information can be configured for a new custom, user-defined modification:

Name	Abbreviation	Fo	ormula	
Hexose	Hex			
Mass (Monoisotopic)	Average Mass (Optional)	Ru	ule (Optional)	
162.052824				
Туре	Source	Ca	ategory	
PTM	✓ USER	▼ C	ustomized	Ŧ
Residues can be modified: ((X = all amino acid residues)			
Peptide	O Protein			
C	O HIOLEIN			
0				
Anywhere	N-term	C-	term	
Anywhere	N-term X	C-	term	
Anywhere KRNSTWCY Neutral losses	NtermX	C-	term	
Anywhere KRNSTWCY Neutral losses Residue	N-term X Mass Loss (Monoisotopic)	° — — Г	term]Alwavs⊥oss →	
Anywhere KRNSTWCY Neutral losses Residue K	N-term X Mass Loss (Monoisotopic) 54.031694	C	term] Always Loss X	
Anywhere KRNSTWCY Neutral losses Residue K + PATTERN	N-term X Mass Loss (Monoisotopic) 54.031694	C 	Term	
Anywhere KRNSTWCY Neutral losses Residue K + PATTERN Signature lons	N-term X Mass Loss (Monoisotopic) 54.031694	~ 	term] Always Loss X	
Anywhere KRNSTWCY Neutral losses Residue K + PATTERN Signature lons Name	N-term X Mass Loss (Monoisotopic) 54.031694 Mass	~ C	term	

- Name: This name will appear in the modification list to specify the custom PTM for future use after it is saved.
- Abbreviation: The shortened, abbreviated form for the custom PTM (optional).
- Formula: The chemical formula of the modification (optional). This should correspond to the mass listed for the modification.
- Mass (Monoisotopic): The monoisotopic mass that a residue will gain or lose as a result of the modification.
- Average mass : The average mass that a residue will gain or lose as a result of the modification (optional).
- Rule: This field can be used to enter a comment about the PTM, to be used for your own reference (optional).
- **Type:** This field is used to set modification type. Type can be set to PTM, isotopic, artifact (see section 3.1.1 above for more information on modification types).
- **Residues can be modified**: This section is used to indicate which residues can be modified. Users must list the amino acids that can be modified anywhere, at the N-terminus, or at the C-terminus, and whether the modification occurs on the peptide or protein level. Residues are added using their single letter code without spaces or commas. See the example in the screenshot above.
- Neutral Losses : Users can define neutral loss patterns observed for the custom PTM (optional).
- **Signature Ions :** Users can define signature ion patterns observed for the custom PTM (optional). Through rescoring based on signature ions, these can increase the confidence of your PTM identification.

Note

Once a custom modification is saved, the name of that PTM cannot be changed.

Leucine (L) and Isoleucine (I) cannot be directly distinguished using traditional mass spectrometry techniques, since the two residues are isobaric, meaning they have exactly the same mass. For that reason, PEAKS Online uses L in *de novo* sequencing to represent either L or I. If a user-defined modification modifies I, it should be defined on L instead. When I is defined as a modification site, the modification site will be ignored by the *de novo* sequencing algorithm.

4. Enzyme Management and Configuration

To view and configure enzymes in PEAKS Online, select "Enzymes" from the Settings drop-down menu. This will bring you to the "Manage Enzymes" page where you will be able to view all built-in enzymes and create/modi-fy/delete customized enzymes.

• Built-in enzymes

All of the built-in enzymes included with PEAKS Online will be visible on this page. Built-in enzymes cannot be modified or deleted.

• Edit enzymes

Click *in an enzyme Actions column to open the "Edit Enzyme" dialog.*

• Delete enzymes

Click **i** button in an enzyme Actions column to delete an enzyme.

4.1. Create a New Enzyme

Click + button next to "Manage Enzymes" to open the "Create Enzyme" dialog.

Note: The other way is at the data loading page, click the drop-down window at "Enzyme", then click "Create a new Enzyme". This will open the same "Create Enzyme" dialog.

Provide the name of the new enzyme in the "Enzyme Name" field. Specify how the custom enzyme will cleave the protein between two amino acids to create peptides in the "Patterns" section. The letter, X, denotes any amino acid at this position. Add multiple amino acids to indicate that cleavage happens before or after any of the stated amino acids. For example, RK means after R or K, not R and K.

Choose where the cleavage sites are by specifying amino acids in the "cut set" and restricting the locations with the "restrict set". Set the side of the cut with the "cut side", which must be C or N. For example, "Cut Set: RK, Restrict Set: P, Cut Side: C" would denote a cut after R or K but not before P.

nzyme Name vsC + Trypsin			
,,,,			
Patterns:			
(X = all amino acid)			
Cut Set	Restrict Set	Cut Side	
К		С	
Cut Set	Restrict Set	Cut Side	
R	P	С	
+ PATTERN			
		C	ANCEL S

5. Instrument Management and Configuration

Instruments can be configured in PEAKS Online by selecting "Instruments" from the Settings drop-down menu. This will bring you to the "Manage Instruments" page where you will be able to view all built-in instruments and create/modify/delete customized instruments.

• Built-in instruments

All of the built-in instruments included with Peaks Online will be visible on this page. You can view the instrument details by hovering over icon in the instrument Actions column. Built-in instruments cannot be modified or deleted.

• Edit custom instruments

Click *i* button in an instrument Actions column to open the "Edit Instrument" dialog.

• Delete custom instruments

Click i button in a custom instrument Actions column to delete an instrument permanently.

5.1. Create a New Instrument

To create a new instrument, click + button next to "Manage Instruments" to open the "Create Instrument" dialog. The following information can be added for a new instrument:

- Instrument Type : This name will appear in the instrument list for future use after it is saved.
- Manufacturer: Details can be added for future reference.
- Model: Details can be added for future reference.

- **Ionization Source** : Select the ion source that was used: MALDI/SELDI or ESI (nano-spray). This will help the PEAKS Data Refine tool determine the charge of the ions.
- Lock Mass : set "-1" if instrument doesn't have lock mass. Enter the lock masses if used.
- MS Precursor Scan Mass Analyzer : PEAKS Online supports a single MS1 analyzer. Select the type and set the resolution and error tolerance. Check "Centroided" if applicable.
- MS/MS Product Scan Mass Analyzer : PEAKS Online supports multiple MS2 analyzers. Click "Add" to add as many as required. Fill in the information similar to MS1 analyzers.

6. TMT/iTRAQ and custom Label Q Methods

To run a labelled quantification analysis, the labelled quantification method must be setup beforehand. The Manage Label Q Methods window can be accessed by users and the administrator in the Settings drop-down menu.

PEAKS Online comes with a number of predefined labelled quantification methods for TMT and iTRAQ quantification. Custom label Q methods can also be defined by users.

Users are able to create their own label Q methods, which will only be available to them.

Manage Label Q Meth	nods + Search	CLEAR	
Method Name 个	Modification(s)	Label(s)	Actions
aminoxy TMT-6plex (CID/HCD)	TMT6plex	TMT6-126, TMT6-127, TMT6-128, TMT6-129, TMT6-130, TMT6-131	/ 🗍 🗐 🗸
aminoxy TMT-6plex (ETD)	TMT6plex	TMT6-126, TMT6-127, TMT6-128, TMT6-129, TMT6-130, TMT6-131	Image: Contract of the second seco
iodo TMT-6plex (CID/HCD)	iodoTMT6plex	TMT6-126, TMT6-127, TMT6-128, TMT6-129, TMT6-130, TMT6-131	Image: Contract of the second seco
iodo TMT-6plex (ETD)	TMT6plex	TMT6-126, TMT6-127, TMT6-128, TMT6-129, TMT6-130, TMT6-131	Image: Contract of the second seco
iTRAQ-4plex	iTRAQ 4plex (K, N-term)	iTRAQ4-114, iTRAQ4-115, iTRAQ4-116, iTRAQ4-117	/ 🗇 📋 🗸
iTRAQ-8plex	iTRAQ 8plex (K, N-term)	iTRAQ8-113, ITRAQ8-114, ITRAQ8-115, ITRAQ8-116, ITRAQ8- 117, ITRAQ8-118, ITRAQ8-119, ITRAQ8-121	\[\begin{aligned} & \leftilde{\begin{aligned} & \begin{ali

Built-in Label Q Methods. The built-in label Q methods within PEAKS Online come standard with all installations and can not be modified by users. They will be available to all users on the server.

To modify a built-in label Q method click the copy button in the label Q method's actions column. This will create a new label Q method (it will *not* modify the original method) with the same settings as the built-in method, which can then be modified.

Note

Built-in label Q methods cannot be deleted or edited. They can only be copied.

Create a new Label Q Method. Click the + button at the top of the Manage Label Q Methods page to create a new label Q method. The label Q method can then be configured in the "Create Label Q Method" dialog (see below). Click the save button to save the instrument for use on the server.

Note

This label Q method will be available to all users on the server.

Edit Label Q Method		
Method Name TMT-9plex (CID/HCD)		
Method Type		
reporter ion quantification		
РТМ		
Fixed PTM		
• TMT10plex		
ADD/REMOVE PTM		
Label Name	Reporter Ion Mass (Da)	
TMT10-126	126.1277	Î
TMT10-127N	127.1248	Î
TMT10-127C	127.1311	Ē
TMT10-128N	128.1281	Î
TMT10-128C	128.1344	Î
	1	1 1

- Method Name: This is the name that will appear in the labelled Q methods list for future use after it is saved. *This can not be modified after saving.*
- Method Type: At this time PEAKS Online only supports Reporter Ion Quantification (TMT / iTRAQ).
- **PTM:** The modifications used as labels for this label Q method. Click the Add/Remove PTM button to expand the PTM section (see below) and set the modifications for this label Q method.
 - **Fixed PTM:** PTMs to be used as fixed labels for this label Q method. Select PTMs from the right side, the search and headers can be used to locate specific PTMs, and then click the left arrow beside the Fixed PTM box to add them to the label Q method.
 - Variable PTM: PTMs to be used as variable labels for this label Q method. Select PTMs from the right side and then click the left arrow beside the Variable PTM box to add them to the label Q method.

Search CLEAR				
🔾 Recent 💿 Common 🔘	Uncommon O Gl	cosylation () Customized () Artif	icial 🔘 All	
xed PTM		NAME .	MONO MASS	RESIDUE SITE
NAME	→	4-hydroxynonenal (HNE)	156.11503	[СНК]
TMT10plex	4	Acetylation (K)	42.010565	[K]
		Acetylation (N-term)	42.010565	[X]@N
		Acetylation (Protein N-term)	42.010565	[X]@N
		Amidation	-0.984016	[X]@C
ariable PTM		Ammonia-loss (C@N-term)	-17.026549	[C]@N
NAME		Ammonia-loss (Protein N-term)	-17.026549	[TS]@N
	\rightarrow	Beta-methylthiolation	45.987721	[C]
	~	Biotinylation	226.077598	[K], [X]@N
		Carbamidomethylation	57.021464	[C]
		Carbamylation	43.005814	[K], [X]@N
		Carboxylation (E)	43.989829	[E]

- Label Options: The labels used by this label Q method. Each label must include a:
 - Name: (e.g.: iTRAQ-114). This will be the name associated with that channel in the labelled Q parameter page and on labelled Q result pages.
 - Reporter Ion Mass: (e.g.: 114.11 for iTRAQ-114). The mass for this channel's reporter ion.

Delete a custom Label Q Method. The user can delete the custom Label Q Methods, but not built-in label Q methods. To delete a label Q method find it in the list on the Manage Label Q Methods page and click the delete button in it's action column.

Edit a custom Label Q Method. To edit a label Q method find it in the list on the Manage Label Q Methods page and click the edit button in it's action column. All fields besides the method name are editable.

7. SILAC Q Methods

To run a SILAC quantification analysis, the SILAC quantification method must be setup beforehand. The Manage SILAC Q Methods window can be accessed by users in the Settings drop-down menu.

PEAKS Online comes with a number of predefined SILAC quantification methods for SILAC quantification. Custom SILAC Q methods can also be defined by users.

Users are able to create their own SILAC Q methods, which will only be available to them.

Manage SILAC Q Metho	ods + Search		CLEAR	
Method Name 1	Modification(s)	Label(s)		Actions
180 labelling 3plex	O18 Labeling (X@C-term), O18 label at both C-terminal oxygens	Light, Medium, Heavy		/ 🗋 🗎 🖌
Dimethylation 3plex	Dimethylation, DiMethyl-CHD2, Dimethylated arginine	light, medium, heavy		/ 🗇 🖹 🖌
Full 180 labelling	018 label at both C-terminal oxygens	Light, Heavy		1 🗐 🕈 🖌
ICAT-C	ICAT light, ICAT heavy	light, heavy		/ 🗊 🗎 🖌
ICAT-D	Applied Biosystems original ICAT(TM) d0, Applied Biosystems original ICAT(TM) d8	d0, d8		/ 🗇 🗎 🗸
ICAT-H	N-iodoacetyl, p-chlorobenzyl-12C6- glucamine, N-iodoacetyl, p-chlorobenzyl 13C6-glucamine	- C12, C13		/ 🗇 🖹 🗸
ICPL	ICPL light, ICPL medium, ICPL heavy	light, medium, heavy		/ 🗊 🗎 🖌
SILAC - Ile6	13C(6) Silac label	light, heavy		/ 🗊 🗎 🖌
SILAC-2plex (R10,K6)	SILAC K6, 13C(6) 15N(4) Silac label	Light, Heavy		/ 🗊 🗎 🖌
SILAC-2plex (R10,K8)	13C(6) 15N(2) Silac label, 13C(6) 15N(4) Silac label	Light, Heavy		/ 🗇 🖹 🖌
SILAC-2plex(R10,K6)	SILAC K6, 13C(6) 15N(4) Silac label	Light, Heavy		🖍 🗊 🖹 👻
SILAC-3plex (R10,K8 R6,K4)	4,4,5,5-D4 Lysine, 13C(6) Silac label, 13C(6) 15N(2) Silac label, 13C(6) 15N(4) Silac label	Light, Medium, Heavy		[•] [•] [•] [•] [•] [•] [•]
SILAC-C(6)	SILAC K6	Light, Heavy		/ 🗊 🗎 🖌
SILAC-C(6)N(4)	13C(6) 15N(4) Silac label	Light, Heavy		/ 🗇 📋 🖌

Built-in Silac Q Methods. The built-in SILAC Q methods within PEAKS Online come standard with all installations and can not be modified by users. They will be available to all users on the server.

To modify a built-in SILAC Q method click the copy button in the SILAC Q method's actions column. This will create a new silac Q method (it will *not* modify the original method) with the same settings as the built-in method, which can then be modified.

Note

Built-in SILAC Q methods cannot be deleted or edited. They can only be copied.

Create a new SILAC Q Method. Click the + button at the top of the Manage SILAC Q Methods page to create a new SILAC Q method. The SILAC Q method can then be configured in the "Create SILAC Q Method" dialog (see below). Click the save button to save the method for use on the server.

Note

This SILAC Q method will be available to all users on the server.

Modi	Modi Detail		1	
SILAC K6 🔀	6.0201[K]	Î		
13C(6) 15N(4) Silac label	10.008269[R]	Î		
		1	1	
	Modi SILAC K6 [2] 13C(6) 15N(4) Silac label [2]	Modi Modi Detail Image: Constraint of the second sec	Modi Modi Detail Image: Constraint of the second sec	Modi Modi Detail Image: Constraint of the second sec

- Method Name: This is the name that will appear in the SILAC Q methods list for future use after it is saved. *This can not be modified after saving.*
- Label Options: The labels used by this label Q method. Each label must include a:
 - Name: (e.g.: Light). This will be the name associated with condition in the SILAC Q parameter page and on SILAC Q result pages.
 - Modi: The modifications used as a condition for this SILAC Q method. Click the 🔀 button to open the Select Modifications dialog (see below) and set the modifications for this silac Q method.
 - Modi Detail: Shows the selected modification's monoisotopic mass and residue site.

Search	С	LEAR	
🔿 Recent 🔿 Common 🔿 Uncom	mon 🔘 Glycosylation	🔿 Customized 💿 Artificial () all
NAME _	MONO MASS	RESIDUE SITE	
13C(1) 2H(3) Silac label	4.022185	[M]	
13C(4) 15N(2) Lysine glygly	120.050415	[K]	
□ 13C(6) 15N(1) Silac label	7.017164	[IL]	
13C(6) 15N(2) (D)9 SILAC label	17.07069	[K]	
13C(6) 15N(2) Lysine glygly	122.05713	[K]	
13C(6) 15N(2) Silac label	8.014199	[K]	
□ 13C(6) 15N(4) Silac label	10.008269	[R]	
13C(6) Silac label	6.020129	[R]	
13C(6) Silac label - IL	6.020129	[IL]	
13C(6) Silac label - ILR	6.020129	[ILR]	

Delete a custom SILAC Q Method. The user can delete the custom SILAC Q Methods, but not built-in SILAC Q methods. To delete a SILAC Q method find it in the list on the Manage SILAC Q Methods page and click the delete button in it's action column.

Edit a custom Silac Q Method. To edit a SILAC Q method find it in the list on the Manage SILAC Q Methods page and click the edit button in it's action column. All fields besides the method name are editable.

8. Workflows

To view or edit your current saved analysis work flows, you must go to the workflows tab. This section can be accessed by users and the administrator in the Settings drop-down menu.

Users are able to save workflows by selecting "Save Workflow As" on the parameters page when they are setting up an analysis within a project. These workflows will only be available to them and the Admin unless they share their workflow for other users.

Manage Workflows	Search	CLEAR		
Workflow Name 个		Owner	Туре	Actions
Brenton WildLFQ		admin		1 🖌 🗐 👻
Comparative PTM TEST PLAN		evan	*	1 🖌 🖉 🗐 🗡
Comparative Spider - Scott		scott	₩	1 🖍 🗐 🕥
CPTAC TMT 10 - Tracy		jiaqi	q	1 🖌 🖉 🗐 🗡
fullDB Orbi		admin	₩	1 🖌 🖉 🗐 🗡
INJ TMT-4-plex		jiaqi	ď	1 🖍 🗐 🕥
orbi db workflow rsahar		rsahar		1 🖌 🖉 🗐 🗡
Orbi-DB-Tracy		jiaqi		1 🖍 🗐 🕥
PeptideLengthRange1		admin		1 🖍 🗐 🗸
scott LFQ test		scott	ď	1 🖍 🗐 🕥
timstof		admin	Q	1 🖍 🗐 🕥
				1-11 of 11 < >

To view the settings and parameters of a saved workflow click \checkmark

Manage Workflows Sea	rch	CLEAR		
Workflow Name ↑		Owner	Туре А	ctions
Project WildDB		admin	*	* 🔳 🔨
Data Information	Enzyme Trypsin	Activation Method CID	Instrument Orbitrap (Orbi-Trap)	^
De Novo Precursor Mass Error Tolerance: 15 PPM Fragment Mass Error Tolerance: 0.5 DA	Max Variable PTM Per Peptide: 3 Enzyme: Specified by each sample	Fixed Modifications: Carbamidomethylation	<u>Variable Modifications:</u> n Deamidation (NQ) Oxidation (M)	^
<i>DB Search</i> Precursor Mass Error Tolerance: 15 PPM Fragment Mass Error Tolerance: 0.5 DA Missed Cleavage: 2 Taxonomy: all species PSM FDR (%) ≤: 1 Contaminant Database: Not set	Max Variable PTM Per Peptide: 3 Enzyme: Specified by each sample Denovo Only ALC (%): 50 Target Database: uniprot_sprot Digest Mode: Semispecific Peptide Length Range: 4 to 60	Fixed Modifications: Carbamidomethylation	<u>Variable Modifications:</u> n Deamidation (NQ) Oxidation (M)	^

To edit workflow settings click the *i* button. This will take you to a dialog to change the parameters

9. Account Settings

The account settings can be accessed from the top menu.

9.1. My account

The user name and email address of the account currently logged in will be displayed. Click \checkmark button to edit the account. This will bring up a page where the password, email address, and full name of the account can be changed. The user name and email address of the account currently logged in will be displayed.

- Receive email for task updates: Check this box, the user will receive the email when the analysis is finished.
- **Default Priority:** Setting the default priority of the task when the user create a new analysis.

Edit Account			
Username			
admin			
Password (leave it empty to keep current password)			
Confirm Password			
Full Name			
Email			
Select a Language			
English			*
Figure Exporting			
● PDF ○ HTML ○ PNG			
	Default Priority		
Receive email for task updates	1	-	
		CANCEL	SAVE

9.2. Algorithm Parameters

User defined Algorithm Parameters are PEAKS PTM Algorithm Parameters and DIA DeepNovo Peptidome parameters.

	RESET SAVE ALL
PEAKS PTM Algorithm Parameters	C
lonfident Signature Ion Requirements Peptide Tag Length ≥: 3 Ion Intensity ≥: 2	
Confident Potential Signature Ion Requirements Use Top: 150 Peaks in Spectra Peak Ion Intensity 2: 2 % Peaks Exists In: 30 % of Modified PSMs	
Modified PSM Count ≥ : 50 Difference between Modified and Unmodified PSM Count ≥ : 30	
DIA DeepNovo Peptidome	c
Peotidome SL	
Select Library: POL Built in Spectral Library 2023-06-27	
teptidome Candidates Deep Novo ALC(%) ≥: 80 Delta RT <: 10 mins	
De Novo	(
ingine	

9.2.1. PEAKS PTM Algorithm Parameters

Confident Signature Ions Requirements:

• Peptide Tag Length: peptide tag length of 3 or more amino acids is recommended for a confident PTM assignment. Peptide Tag Length $\geq X$: The number of X consecutive ions at the modification site in the peptide that have ion intensity above the value Y set in the following algorithm parameter. If the PTM takes place at the N-term or C-term, the requirement is a number of consecutive ions + 1 (default would be 3 + 1)

Ion Intensity: percent ion intensity for the fragment ions within the peptide tag compared to base peak for a confident PTM assignment. Ion Intensity ≥ Y: Defines the minimum ion intensity Y that is used in considering whether the PTM is confident

Confident Potential Signature Ion Requirements:

- Use Top # Peaks in Spectra: This top number of peaks (in intensity) in the spectra is considered when searching for potential signature ions. It is used in conjunction with the peak ion intensity parameter in determining how many peaks are considered when searching for potential signature ions.
- Peak Ion Intensity: The ion intensity for a peak must exceed this value for it to be considered as a potential signature ion. This is used in conjunction with the Use top # peaks in the spectra parameter to only select # peaks with ion intensity \geq than a set value. If a signature ion does not meet these criteria, it will not display in the result
- Peak Exists in $\ge X$ % of modified PSMs: For the list of Modified PSMs, if the number of PSMs with this signature ion is greater than X % of the total, then the potential signature ion will be reported. Potential signature ions that do not pass this threshold will still be in the all potential signature ion export which is useful in determining how to tune this parameter for future runs.
- Modified PSM Count: minimal number of modified PSMs with the potential signature ion required to be reported in the result.
- Difference between Modified and Unmodified PSM count: percent difference between the numbers of modified and unmodified PSMs. Both the list of Modified PSMs with the modification of interest that has the detected signature ion as well as the list of Unmodified PSMs that has the detected signature ion are recorded. A % of the Modified PSMs will have the potential signature ion detected, and a % of Unmodified PSMs will have the potential signature ion detected. If the % of Modified PSMs is greater than X % compared to the % of Unmodified PSMs, then this signature ion is reported as a potential signature ion. Potential signature ions that do not pass this threshold will still be in the all potential signature ion export which is useful in determining how to tune this parameter for future runs.

9.2.2. DIA DeepNovo Peptidome

Peptidome SL:

• Select Library: Spectral library used for internal pre-search. PEAKS Online installations include a default Spectral Library generated by PEAKS Research team.

Peptidome Candidates:

- Deep Novo ALC(%) minimum deep novo score cut off to be included in second round Peptide Search.
- Delta RT minimum delta RT for the deep novo peptide to be included in the second round Peptide Search.

9.2.3. De Novo Engine

Each user has the option to set their default de novo search engine for the Deep Novo/De Novo standalone workflow. DeepNovo searching uses curated deep learning models. The speed of a DeepNovo search can be accelerated if a GPU is enabled with PEAKS Online.

Chapter 3. Archive Project

1. Overview

After the project is finished processing, users and admin can click subtraction button to archive the project or click button to delete the project to save the cassandra memory. When the user or admin need to view an archived

project, the user or admin can click button on the top left of the menu to import the archived project from the Archived Repository back to my project list.

My Projects	Search		CLEAR				
Project Name	Description	Created	Last Updated	#MS Runs	Progress ↑	Actions 💽 🧻	Select All
Comparison Project		May 20 2020, 03:55:04 pm	May 21 2020, 10:49:00 am	1	Done		
Orbitest1		May 21 2020, 10:16:07 am	May 21 2020, 10:18:36 am	5	Done		

2. Archive Project

Click the archive **s** button on the selected project to start archiving a project that has a Done progress status.

2.1. Create a new folder in Archive Repository

User and Admin can create a new folder under the archive repository to distinguish the archived project.

- 1. Go to the archive repository and click create a new folder **Create a new folder** button.
- 2. Fill in the folder name and press enter button on the keyboard.
- 3. The new folder will be created and you can click the folder to archive the project into this folder.

Archiv	X	
Save As Comparison Project	Archive Raw Scans	
Sources / Archives		START
CREATE A NEW FOLDER		CANCEL
🔨 ABRF 2017 DIA (no scan)	A	
🔨 ABRF 2017 DIA (no)		
📉 Inj		
LFQ_MaxQuant_1082_2Samples		
LFQ_MaxQuant_2samples_online1.2		
🔊 Stanford-1		
🔊 Stanford-2		
🔊 Stanford-3		
Stanford1074	•	RETRY
2.2. Archive Project in Archive Repository

- 1. Click the archive **S** button on the selected project to start archiving the project that has a Done progress status.
- 2. The archive Project dialog pop up.
- 3. Choose the archive repository from the archive repository list.
- 4. User and Admin can create a new folder in the archive repository or use an existing folder.
- 5. User and Admin can edit the names of archived projects in the Save As name field.
- 6. Click the start **START** button to start archiving project.



- 7. The archive progress bar will occur on the bottom. It displays the archiving details, which include the finished tasks and the total tasks.
- 8. User and Admin can click the × button to hide archive project dialog and let the archiving run in the back-ground.

9.	Archiving		Failed	
There are three archive progress status: Archiving	rushing	, Failed Archive	Archive	and Done
Done				

- Archiving status: The project can not do any actions: open project, edit project, delete project. The user or Admin can click the archive button to view the archiving progress details.
- Failed Archive status: The project can not do any actions: open project, edit project, delete project. The user or Admin needs click the archive button and see the following information:



- Done status: The project can be viewed and edited. It also can be found in Archived project Page, if it done by the archiving.
- 10. The archive progress bar will occur on the bottom. It displays the archiving details which include the finished tasks and the total tasks.

3. Import Archived Project

After archiving the projects in the Archive Repository, if the user or admin want to import the archived project back, they should delete the archived project in my project list. Otherwise they will get an error message that reads "This project is already in the project list".

3.1. Import Archived Projects Page

- 1. Using the button on the top left menu to go to Import Archived Project Page.
- 2. Select an Archived Project from the repository.
- 3. Click the import button in the middle of the import page.
- 4. The user or admin can edit the import project name if you needed.
- 5. Click import button to start import archived project or cancel button to cancel the importing.

	mport Projects		
Walkthrough Denovo			
	_	_	

- 6. The importing project will show on the right side with progress details, which include project name, finished tasks and total tasks.
- 7. The importing project has three status: importing, failed and finished.
 - The importing status: You can click cancel **CANCEL** button to cancel the importing project.
 - The failed status: User can click retry button to re-import the project.
 - The finished status: The importing project will show in my project list.
- 8. The user or admin can use clear all button to clear all finished importing project in the importing projects list. Or using clear button for each finished importing project to clear it in the importing projects lit.

PEAKS Online	🔁 🎝
--------------	-----

ŧ

Import Archived Projects		
Select Project	Importing Projects	CLEAR FINISHED TASKS
Sources / Archives Select All Search Archives Archives ABRF 2017 DIA (no) Inj LFQ_MaxQuart_1082_2Samples LFQ_MaxQuart_2samples_online1.2 Stanford-1 Stanford-1 Stanford-2 Stanford-3 Stanford-3 Stanford-3 Walkthrough Walkthrough	LFQ_MaxQuant_1082_2Samples Importing (1/45) Starford-1 Importing (0/0)	CANCEL

9. User can click PeaksOnline logo on the top left of menu to leave the importing archived project page.

Chapter 4. Data View

1. Overview

The data view gives a visualization of the mass spectrometry data loaded into the project. It allows for the inspection of LC-MS/MS details in an intuitive way for quality control and troubleshooting issues.

2. Accessing the Data View

The data view can be accessed by following the procedure below:

- Open an existing project/analysis. Refer to Section 5.5, "Opening an Existing Project" in Chapter 1, Overview for details.
- On the bottom left hand side of the project page, select "Data" to open the list of samples.
- At the top of the screen will be a summary listing "Sample(s) in Project", "Sample(s) in Analysis", "MS Run(s) in Project", and "MS Run(s) in Analysis".
- Click a sample row, and then click the 💿 button to expand the details of the sample.
- After you click the 🚯 button, click the 📳 button to see data view.
- After you click the 🚯 button, click the 💆 button to download the MGF file.
- Type in a search word in the search box to list any samples with a matched sample name.
- Click the delete button beside the status icon to delete any samples in that status.
- Click the retry button below "Action" to have the failed samples start re-uploading.

nline 🗈 🖉	± 6-6					project 🗸	SETTINGS 🗸	admin \checkmark	account \checkmark	HELP 🗸
Orbitest1 +	<	Project Description 🖌					MODI	FY DATA		
		5	iample(s) in Proje 3	ct MS Run(s) ir 5	n Project					
		Search CLEAR	Done	3 Processing 0	Pending 0 🛢	Failed 0	Actions:			
		Sample Name	No. of Fraction	Enzyme	Activation Method	Instrument	Progress			
		🕒 Sample 1 🧪	1 Fraction	Trypsin	HCD	Orbitrap (Orbi-Tra	p) Done			
		😋 Sample 3 🥜	1 Fraction	Trypsin	HCD	Orbitrap (Orbi-Tra	p) Done			
		🕒 Sample 5 🧪	3 Fractions	Trypsin	HCD	Orbitrap (Orbi-Tra	p) Done			
🚍 Data Search 🔍 🗙 🖍										
Sample 1 Sample 3 Sample 5										

3. Navigating the Data View

The data view is shown in three panes:

- 1. The TIC curve: shown vertically along the left side of the page. The total ion chromatogram (TIC) gives the total intensity of all signal acquired at each retention time. Use the Scan (or Frame Index with timsTOF) text box above the TIC display to search the data based on scan number. Also, it is possible to click on any point in the TIC curve to show details at that retention time. The arrow buttons beside the TIC graph can be used to scroll through the data based on retention time. The position of the red line indicates which retention time is currently being displayed.
- 2. The MS Scan: The MS scan at the given retention time is shown in the top right pane. The retention time (RT), scan number (Scan), and TIC intensity values are given in the top right corner of the MS scan display. The y-

axis of the spectrum can be switched between observed and relative intensity values using the # and % buttons located above the axis respectively. Hold the cursor within the spectrum and use your mouse wheel to zoom in and out on the m/z axis. Hold your cursor to the left of the y-axis to zoom in and out on the intensity axis. Use the 1:1 button to return to the default scale.

3. The MS/MS Scan: In the middle pane, m/z values of the precursors that were selected for MS2 and associated with the MS1 scan shown on top are displayed. Users can select the m/z and view its MS2 spectrum in the bottom right pane. The retention time (RT), scan number (Scan), TIC, mass over charge of the precursor ion (MZ, corrected by PEAKS mass correction), and charge (Z) are given in the top right corner. TimsTOF data will additionally show Ion Mobility and PrecursorID. The y-axis of the spectrum can be switched between observed and relative intensity values using the # and % buttons respectively. Hold the cursor within the spectrum and use your mouse wheel to zoom in and out on the m/z axis. Hold your cursor to the left of the y-axis to zoom in and out on the intensity axis. Use the 1:1 button to return to the default scale.



Chapter 5. Peptide *De Novo* Sequencing

1. Overview

De novo sequencing derives a peptide sequence directly from the tandem mass spectrum without the need of a sequence database. It is the preferred method for identifying novel peptides and the study of unsequenced organisms.

PEAKS Online provides a reliable and comprehensive solution for automated peptide *de novo* sequencing. It features:

Accurate de novo sequencing

The PEAKS Online *de novo* sequencing algorithm and scoring functions are constantly tuned for each instrument type to ensure optimal accuracy.

• Fast sequencing speed

PEAKS Online can perform *de novo* sequencing on over hundreds of spectra per second on servers.

• Easy-to-use user interface

PEAKS Online generates a comprehensive result summary and provides interactive views of annotated spectra.

• Support of all major instrument types

PEAKS Online supports *de novo* sequencing of spectra generated by all instrument types (Ex. Orbitrap, FT-MS, Ion Trap, Time-of-Flight, Quadrupole, timsTOF) in common fragmentation modes (Ex. CID, HCD, ETD, ETHCD).

Automated result validation

PEAKS Online assigns a *local confidence* score to each amino acid in a *de novo* sequence. The local confidence score ranges from 0 to 99%, indicating how confident the algorithm considers a particular amino acid to be the correct assignment. Moreover, the peptide sequence is evaluated by an ALC (Average of Local Confidence) score, which is the average of the local confidence scores of all the amino acids in the sequence.

Note

For details about the PEAKS *de novo* sequencing algorithm, please refer to the initial publication: "PEAKS: Powerful Software for Peptide De Novo Sequencing by Tandem Mass Spectrometry", Rapid Communication in Mass Spectrometry, 17(20): 2337-2342 (2003).

1.1. Standalone DeepNovo/De Novo Analysis

In the PEAKS 11 upgrade release, our standalone De Novo sequencing workflow has been modified to include an optional DeepNovo search engine. Deep Novo will only be available if the user has GPU enabled. If GPU is enabled, it will be used as default. Users can change their default De Novo engine under the Account > Algorithm Parameters menu. For more information, refer to Chapter 2, Section 9.

Select an Analysis



Other PEAKS Online tools, including PEAKS DB for database search; PEAKS PTM for unspecified PTM search; and SPIDER for homology search depend on the regular *de novo* sequencing result.

2. *DeepNovo/De Novo* Analysis Workflow and Parameter Settings

To conduct a PEAKS de novo sequencing analysis, follow the steps below:

- 1. Select "New Project" from the Project drop-down menu, or the Create New Project icon from the home page. This will prompt the "Project Creation" page where the project can be named, a description can be added, and the user can choose to create a new workflow or select a saved workflow.
- 2. Name the new project with optional description, select new or saved workflow, and click the Next button.
- 3. Select "Local" tab and click to bring up a File Explorer or Finder window where files can be selected for uploading. If the administrator has specified folders on the server that can be accessed for data uploading, they will be listed under the "Remote" tab. Click "In Project" tab to bring in data from the existing project if this is a new analysis of an existing project. To add data using the Instrument tab please refer to the Instrument Daemon chapter of the user manual.
- 4. Newly added files will be selected and highlighted in grey. Use the icons to add the group of files as fractions or individual samples. Alternately, use the Select All box to de-select the group of files and then select individual files to add to the project using the different options. To place the selected data from the list into samples: use

 $\Rightarrow \bot$ to place all files in an existing sample; use \Rightarrow^{2} to create a new sample for selected files (i.e., add the

files as fractions); put each file into an individual sample using 2 ; or use 2 to add samples by regular expression.

SUPPORT, STANDALONE W	ORKFLOW (D	DATA	ANALYSIS DEEPNOVO/DE NOVO		ERS		
						CANCEL	NEXT
Select Data	In Parisan	Sample Name	Enzyme Trypsin	Activation Method	Instrument Orbitrap (Orbi-Orbi)	Acquisition	*
Select All	in Project	Sample 1	SS1040HCD_L_T01.raw	✓ HCD	 Orbitrap (Orbi-Orbi) 	• DDA • ↑ ↓	×
Click to upload data	÷°≟						
	⇒ji						
	⇒≞						
	→ ^{c7} <u>A</u>						
					UPLOAD	SAMPLES RE	MOVE ALL

- 5. For each sample, specify the name and the implicit parameter details: "Enzyme", "Activation Method", and "Instrument" type, or specify the project details using the drop-down menus at the top of the list if all samples have been prepared and acquired with the same method. If the required enzyme is not listed, it can be created at this page by selecting Create a New Enzyme from the drop down menu, then adding the pattern (cut set, restrict set, and cut side).
- 6. Click the Next button to continue choosing the analysis workflow.
- 7. For a DDA dataset, select the "De Novo" workflow.



8. Set the Denovo Parameters. See Section 2.1-2.3 below for details.

DEEPNOVO CHECK	DATA	ANALYSIS DEEPNOVO/DE		TERS			
AVE WORKFLOW AS				Priori 1	CANCEL	BACK	SUBMIT
Data Refinement Param	eters						
Mass Correction 🛛 🗹	Associate feat	ures with Chimera Scan					
DeepNovo/De Novo Par	ameters						
Precursor Mass Error Tolerance: 10		Tolerance Unit: PP	M	Fragment Mass Error Tolerance: 0.	02 Toleran	ce Unit: DA	
Enzyme: Spec	ified by each sample	•					
Fixed Modifications:			Variable Modifications:				
Carbamidomethylatic	n		Deamidation (NQ)Oxidation (M)				
	•						
				Max Variable DTM Des Destid			

9. Click SUBMIT button to start the search.

2.1. Precursor and Fragment Error Tolerance

Precursor Mass Error Tolerance: The maximum mass difference between the *de novo* peptide and the monoisotopic mass of the precursor ion.

Tolerance unit: PEAKS *de novo* generates peptides within the precursor mass error tolerance. Precursor mass error tolerance can be specified in either Daltons or ppm.

Fragment Mass Error Tolerance: PEAKS *de novo* uses this value when scoring *de novo* peptide sequences. PEAKS Online considers a fragment ion to be matched if the calculated m/z is within the specified error tolerance. Units are Daltons.

Both precursor and fragment error tolerance parameters need to be set consistently with the mass accuracy of the instrument.

2.2. Enzyme Settings

Select the enzyme used for protein digestion from the enzyme drop-down menu. PEAKS *de novo* respects the enzyme specificity at both ends when generating peptides.

Note

When the selected data node is a project of multiple samples, "Specified by each sample" allows samples to be analyzed separately, using their respective specified enzymes during project creation.

Note

"None" is a special enzyme allowing non-specific cleavage at both ends of the peptide. It is the recommended option when no digestion enzyme was used or when the digestion enzyme exhibits a high degree of non-specificity.

2.3. PTM settings

Click "ADD/REMOVE MODIFICATION" button to open the "Select Modifications" dialogue and specify the fixed PTMs and a few common variable PTMs expected in the sample. Once the PTMs are set, select confirm. Once confirmed, the maximum number of variable PTMs per peptide can be specified and a value of less than 4 is recommended. A fixed modification forces all instances of applicable residues to be modified. A variable modification gives the option for the residues to be modified.

Note

For a specific residue, only one fixed modification is allowed.

Note

The use of variable modifications increases the size of the computational search space for the *de novo* sequencing algorithm. It is recommend not to select too many variable modifications in PEAKS *de novo*. See PEAKS PTM for alternate search options.

3. Understanding PEAKS De Novo Sequencing Results

When *de novo* sequencing is done, you can access the PEAKS Online project from the "My Projects" page. The analysis report for *de novo* search tool is presented in two tabs:

- Summary: This shows an outline of the PEAKS *de novo* search results with key statistics. The overall quality of the experiment can be examined and the filters for *de novo* ALC can be adjusted at the top of the page. This page is displayed after the search is done.
- **De Novo:** This shows a list of peptide sequences detected by *de novo* sequencing tool, which passed the de novo ALC filters set in the Summary page.

The *de novo* view displays the *de novo* sequencing results. The table on the top displays *de novo* sequences and their characteristics (length, m/z etc.), while the bottom section provides additional information about the peptide-spectrum match. The results can be filtered and exported.



3.1. Result Filtration

In the summary tab, the PEAKS Online *de novo* sequencing result can be filtered using Average Local Confidence (ALC) score. Low quality *de novo* sequences can be filtered out by specifying a minimal threshold of the ALC score. The purpose of filtering is to remove poor sequences in which residues are incorrectly identified. By default,

the ALC threshold is set to 50%. After changing the threshold, click **APPLY** to apply the new filter. The result in the De Novo result view will be updated accordingly.

In the De Novo tab, result for different samples can be viewed by selecting the sample using the "Show Results for Specified Sample" drop-down menu. The purpose of this filtering is to limit the quantity of *de novo* sequencing results displayed in the table. Results can also be filtered using the Filtered Peptides icon at the top left of the De Novo window. Here, results can be filtered by an amino acid sequence, scan #, precursor ID, m/z, or RT.

Click APPLY to apply the new filter. The result will be updated accordingly.

3.2. Exporting De Novo Results

The *de novo* results can be exported from the De Novo results tab or from the Export page of the analysis. In the De Novo tab, above the peptide table there are two export options: "Denovo", and "Denovo with all Candidates". In the Export page, there are three files which can be exported: "Summary Table", "Denovo CSV", and "Denovo CSV with all Candidates".

- **Denovo CSV**: the filtered *de novo* results shown are exported. This file can also be downloaded from the Export page as shown in the following figure.
- **Denovo CSV with all candidates**: for each MS2 spectrum, denovo results of the 5 highest ALC scores are generated. *De novo* sequences that pass the ALC score filter are all included in this file. This file can also be downloaded from the Export page as shown in the following figure.
- Summary Table: shows the overview number of each sample for the fields MS2, PSM, PSM (alc>30), PSM (alc>50), PSM (alc>70), and notes.

Analysis 1: De Novo	S / I	Export SELECT	ALL CLEAR ALL
🛄 🚵 De Novo			
👃 Export			
		All Search Parameters	
		Data Refinement	Summary Table
		MGF File	Denovo CSV
		Mzxml	Denovo CSV with all Candidates
		DOWNLOAD DOWNLOAD BY	Y SAMPLE

3.3. Summary View

The "Summary" view reports key statistics as an overview of the result.

M PEAKS On	line 🔒	6	- 5 🛨						PROJECT 🗸 SET	TINGS 🗸 ADMIN 🗸 ACCOUNT 🗸
User Manual Denovo + Analysis 1: De Novo Analysis De Novo	0/1	< Summ		Filters De Novo ALC (6)≥ 50.0	APPLY Created:	Mar 05 2021, 03:58:16 pm			
🖊 Export		ary De Novo					Summary	Statistics	4	
				Sample Name	# MS2	# PSM	# PSM (alc > 30)	# PSM (alc > 50)	# PSM (alc > 70)	# PSM / # MS2
				All	22415	34105	21323	15206	9676	152%
				Sample 1	10458	15697	10139	7398	4733	150%
				Sample 2	11957	18408	11184	7808	4943	154%
E Data Search	۹ × 🖍									
Sample 1	9 1									
😋 Sample 2	Ø 🚺									

3.3.1. Result Statistics

This table of the summary page shows the summary date for each sample for the following fields:

- **# MS2**: The number of MS/MS scans from the data.
- **# PSM**: The number of peptide-spectrum matches.
- # PSM (alc>30): The number of peptide-spectrum matches with an ALC score greater than 30.
- # PSM (alc>50): The number of peptide-spectrum matches with an ALC score greater than 50.
- **# PSM (alc>70)**: The number of peptide-spectrum matches with an ALC score greater than 70.

Note

ALC score is the average local confidence score. ALC is calculated as the total of the residue local confidence scores in the peptide divided by the peptide length.

3.3.2. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

2 data files produced no results. (Expand to see list of zero result data files)

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.

Potentially Problematic Data Sets	~
Sample 3 All fractions have no result.	
Sample 4 All fractions have no result.	

If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics

No Results Found

3.4. Peptide Table

The peptide table shows the filtered *de novo* sequencing results. For each scan, the best candidate is displayed.

The following list describes the contents in each column:

- Scan: The scan number is a unique index for tandem mass spectra in the data. If the scan numbers were read from the data file, the scan numbers that match with the instrument data file will be shown.
- **Peptide**: The amino acid sequence of the peptide as determined by *de novo* sequencing. A modified residue is followed by a pair of parentheses enclosing the mass of that modification.
- ALC (%): The average local confidence. ALC is calculated as the total of the residue local confidence scores in the peptide divided by the peptide length.

- Length: The number of amino acids in the peptide sequence.
- **m/z**: The precursor mass-to-charge ratio.
- **z**: The precursor charge.
- **RT**: The retention time (elution time) at which the spectrum was recorded in the data.
- Area: The area under the curve of the peptide feature found at the same m/z and retention time as the MS/MS scan. This can be used as an indicator of the abundance of the peptide.
- Mass: The monoisotopic mass of the peptide.
- ppm: The precursor mass error, calculated as 10E6 × (precursor mass peptide mass) / peptide mass.
- PTM: The types and the numbers of the modifications present in the peptide shown as color-coded icons.
- Mode: The fragmentation mode in which *de novo* sequencing is performed by the algorithm.

Confidence Scores: Amino acids in *de novo* sequences are color-coded according to their local confidence scores. Red represents a very high confidence (greater than 90%); purple represents a high confidence (80 to 90%); blue represents a medium confidence (60 to 80%); and black represents low confidence (less than 60%). Hover the cursor over the sequence to view local confidence scores of each individual amino acid.

Mass Tags: The low confidence residues can be displayed as mass tags by adjusting the local confidence score threshold by selecting the Set Residue Confidence Threshold icon and then using the model button to adjust the threshold. If the score is set at 0, all of the amino acids in the peptide sequences will be displayed. Increasing the threshold will reduce residues below the threshold to mass tags. When low confidence residues are reduced to mass tags, the remaining residues become sequence tags. The length of the longest tag is called the tag length and reported in the export files.



Filter Peptides: The peptide table can be filtered by selecting the **o** button and prompting the "Denovo Filter" window. The peptides can be searched by sub-sequence, scan number, precursor Id, m/z, and retention time.

Denovo Filter			
Denovo sequence contains			
Scan =			
Precursor Id =			
m/z 2			
RT 2			
	RESET	CANCEL	APPLY

3.5. Spectrum Annotation

The spectrum annotation displays a graphical representation of the peptide-spectrum match.

Moving the cursor over the peptide sequence in the spectrum will show the mass transitions for a particular amino acid residue.

The spectrum annotation panel provides convenient ways to zoom and navigate within the spectrum:

- Zoom in/out on m/z: Place the cursor on the point in the spectrum you would like to zoom in on. Alternatively, place the cursor in the Error (da) figure and scroll the mouse wheel.
- Increase/Decrease peak intensity: Place the cursor on the Intensity axis and scroll the mouse wheel.
- See the whole spectrum, or reset a zoomed in panel: click the 1:1 button.

Click Sto open the Ion Annotation Settings dialogue and change the ions to be annotated in the spectrum. To reset the settings to the PEAKS Online defaults, use "RESET DEFAULT" button.



The solution will switch the y-axis to relative intensity. Once selected, scrolling over peaks will display the relative intensity instead of the absolute intensity.

The *#* button will switch the y-axis to absolute intensity. Once selected, scrolling over peaks will display the absolute intensity instead of the relative intensity.

3.6. Error Map

The "Error Map" shows the mass errors of the matched fragment ions. The m/z ratio is displayed on the x-axis and the error is listed on the y-axis in Daltons. Each matched fragment ion is represented by a dot. The ion types displayed can be configured in "Ion Annotation Settings".

3.7. Ion Table

The "Ion Table" shows the calculated mass of possible fragment ions based on the ion types selected in the Ion Annotation Settings. If a fragment ion is found in the spectrum, its mass value is displayed in color. N-terminal ions are shown in blue and C-terminal ions are shown in red. A fragment ion is found when there is a matching peak within the mass error tolerance, as defined in the *de novo* sequencing parameters.

4. FAIMS Data

Field asymmetric waveform ion mobility spectrometry (**FAIMS**) analyzes complex mixtures and separates the ions according to their mobilities. PEAKS Online has the ability to load FAIMS data and analyze it using *de novo* sequencing, database search (PEAKS DB), unexpected modification (PEAKS PTM) and sequence variants (SPIDER) search, and labelled and label free quantification.

Note

There is no special procedure required to load FAIMS data. Refer to Chapter 1, *Overview* for more details on data loading.

The analysis result for FAIMS dataset contains the additional information for CV values compared to the non-FAIMS dataset. The CV information can be found at the following places for a Denovo analysis.

• Additional CV column in *de novo* table

de novo table contains one additional column for CV values

• Additional **CV** filter in *de novo* table

The filter for *de novo* table contains one additional option for CV, which selects the target scan based on the CV value

• Additional CV column in the exported Denovo CSV

The exported Denovo CSV contains one additional column for CV values

• Additional CV column in the exported Denovo with all candidates CSV

The exported Denovo with all candidates CSV contains one additional column for CV values

1. Overview

PEAKS Online provides a complete set of database search tools for in-depth protein analysis. With PEAKS DB, PEAKS PTM, and SPIDER, proteins in the sample can be identified with high sensitivity and accuracy. It is also possible to find potential sites for modification and mutation. PEAKS Online supports multiple enzyme digestion by which almost full sequence coverage can be achieved for single protein studies. PEAKS Online database search tools have built-in result validation using an enhanced target-decoy approach. The false discovery rate (FDR) is estimated to ensure that only valid results are reported.

PEAKS DB is a database search tool assisted by PEAKS *de novo* sequencing to achieve high sensitivity and accuracy.

Note

For more details, refer to the paper: "PEAKS DB: De Novo sequencing assisted database search for sensitive and accurate peptide identification", Mol Cell Proteomics, 2011, Dec 20.

PEAKS PTM is a dedicated search tool for peptides with unspecified modifications and mutations. It can search with unlimited number of modifications, allowing searches with all modifications in the Unimod database.

Note

For more details, refer to the paper: "PeaksPTM: Mass Spectrometry Based Identification of Peptides with Unspecified Modifications", Journal of Proteomics Research, 2011, 10(7): 2930-2936.

SPIDER is a dedicated search tool for finding novel peptides that are homologous to peptides in a given protein database.

Note

For more details, refer to the paper: "SPIDER: Software for Protein Identification from Sequence Tags Containing De Novo Sequencing Error", J Bioinform Comput Biol, 2005, Jun;3(3):697-716.

2. Database Search Workflow and Parameter Settings

PEAKS DB, PEAKS PTM, and SPIDER can be launched together as a work flow in a PEAKS Online search. An analysis report is generated to show the combined database search result. Users can also do PEAKS DB only, PEAKS DB plus PEAKS PTM, PEAKS DB plus SPIDER or PEAKS DB plus PEAKS PTM plus SPIDER.

To conduct a database search workflow, follow the steps below:

- 1. Refer to Chapter 5, Peptide De Novo Sequencing, Section 2, "*DeepNovo/De Novo* Analysis Workflow and Parameter Settings", Step 1-6 to create a new project and add data.
- 2. Select the "Database Search" workflow.

DEEPNOVO HLA DATA ANALYSIS PEAKS DB	
Select an Analysis	CANCEL BACK NEXT
DDA DIA	
De Novo	
DeepNovo Peptidome	
O Data Deep Novo Peptide Database Search	
PEAKS DB(In-depth de novo assisted search)	
Data De Novo Database Search PEAKS PTM SPIDER QC	
PEAKS O(de novo assisted Quantification)	
O Data De Novo Database Search Quantification QC	
Fractionation Assisted ID Transfer Quantification	
O 💓 Data De Novo Database Search Quantification	

3. Set the DB Search Parameters. Enable Deep Learning Boost algorithm for an enhanced number of PSMs identified (optional). Enable PEAKS PTM search to find more PTMs (optional). Enable SPIDER search to find more mutations (optional). See details in Section 2.1-2.6 below.

Data Refinement P	arameters	5								
Mass Correction	\checkmark	Associate featu	res with Chimera	Scan 🔽	I					
Database Search R	Parameter	5								
ecursor Mass Error Tolerance:	10		Tolerance Unit:	PPM	¥	Fragment	Mass Error Tolerance: 0.02	Tolerar	ce Unit: DA	
Enzyme:	Specified	by each sample*	Digest M	ode: Ser	ni-Specific	•	Missed Cleavag	e: 3]	
Target Database:	UniProt	~	Тахопо	ny: Hom	o sapiens (human) ×	\sim	Contamina Databas	nt e: N/A	~	
Peptide Length:	6	to 45	Deep Le	arning Bo	ost 🔽					
ixed Modifications: • Carboxyethyl					Variable Modifications: • Deamidation (NQ) • Oxidation (M) • Acetylation (Protei	n N-term)				
DD/REMOVE MODIFIC	ATION					Max Variable	PTM Per Peptide:	3		
De Novo Paramete	ers 🔽 U	lse Database Sea	rch Parameters							
Find Unspecified F	TMs with	PEAKS PTM								

eport Filter		
I/Peptide		
O PSN	۸-10LgP ٤ (%) PSM FDR (%)	1 Peptide FDR (%)
ein		
Prot	teins -10LgP ≥ 20 : O Protein	Group FDR (%) : Proteins Unique Peptides 2 1
lovo		
Denovo Or	nly ALC (%)≥ 50 : Show De N	iovo Only Tags sharing . Amino Acids with Protein
(C		
Data QC Att	ributes:	Identification QC Attributes:
	# MS1	#PSM
	# MS2	#MS2
	MS2/MS1 Rate (%)	# Identified Precursors
_	Detected Features	🗹 # Peptides
	Full Width (min)	# Sequences
	FWHM (sec)	# Protein Groups
	BPC	# Top Proteins
~	TIC Correlation	# All Proteins
		D Rate (%)
		# PSM/# Peptides Rate (%)
		MS1 Error Mean (S.D)
		Missed Cleavage Ratio (%)
		Enzyme Specificity Ratio (%)
Input Stan LLDMGET DHSFFIPD	gnotes. dard Peptides Seperated by commas space or tab, 4 DLMLAALR 2 IEYLSDIK,3	example below

4. Click SUBMIT button to start the search.

2.1. Data Refinement Parameters

Mass Correction: The precursor m/z values given by some instruments are often not of the monoisotopic ion. This creates problems in downstream analysis. By enabling mass correction, PEAKS will examine the isotope shapes in the corresponding MS scans and accurately adjust the precursor m/z to equal the monoisotopic ion m/z.

Associate Features with Chimera Scan: If enabled, PEAKS will assume that a tandem scan may contain fragments from two or more peptides, which will allow PEAKS to identify co-eluting peptides found within the acquisition window of the scan. PEAKS accomplishes this by searching for peptides that match the monoisotopic mass of any peptide feature in the acquisition window.

2.2. Precursor and Fragment Error Tolerance

Precursor mass : This enables the selection of a precursor mass error tolerance. PEAKS Online DB Search identifies peptides within the precursor mass tolerance, which is the allowable m/z shift between the theoretical value of the peptide versus the observed value of the precursor ion. Precursor mass tolerance can be specified in either Da or ppm.

Fragment ion: This enables the selection of a fragment ion m/z tolerance, which is the allowable m/z shift between the theoretical value of the fragment ions of the peptide versus the observed value. PEAKS Online DB Search uses this value when scoring peptide-spectrum matches. PEAKS Online considers that a fragment ion is matched if the calculated m/z is within the tolerance. Fragment mass tolerance can be specified in Da.

Error tolerance parameters need to be set consistently with the mass accuracy of the instrument.

2.3. Enzyme Settings

Enzyme : PEAKS Online digests the protein database *in silico* to generate peptide candidates. It is necessary to specify the enzyme for protein digestion from the Enzyme drop-down menu.

Note

When the selected dataset is digested with different enzymes, "Specified by each sample" allows samples to be analyzed separately using their respective enzymes specified during project creation.

Note

"None" is a special enzyme digest rule allowing non-specific cleavage at both ends of the peptide. The "None" digest rule can cut at every residue, generating peptides with lengths up to 65 amino acids. Its usage is recommended when no digestion enzyme was used or when the digestion enzyme exhibits a high degree of non-specificity.

Missed Cleavages : This specifies the number of missed cleavage sites allowed in a peptide.

Digest Mode : This specifies the type of *in silico* cleavages allowed at a peptide termini based on the method used for protein digestion. If "Specific" is selected, the specificity of the selected enzyme is strictly enforced at both termini of a peptide. "Semispecific" allows one terminus to disobey the enzyme specificity rules.

Note

If the enzyme is specified as "None", then no matter which mode is selected for "Digest Mode", the "None" enzyme digest rule will be applied.

2.4. Database Settings

A protein database must be added first before creating a new project to enable its selection in the workflow. In the Target Database drop-down menu, select the appropriate database that have been configured in PEAKS Online and set the taxonomy, if applicable. A contaminant database can also be selected from the same list of databases that have been configured in PEAKS Online. The length for a peptide can be set when creating an analysis, where the default peptide length range is between 6 and 45. Enable Deep Learning Boost algorithm for an enhanced number of PSMs identified.

2.5. PTM Settings

Click "ADD/REMOVE MODIFICATION" button to open the "Select Modifications" dialogue and specify the fixed PTMs and a few (less than 11) common variable PTMs expected in the sample. If the required modification is not listed, the modification can also be created here. Additionally, the maximum number of variable PTMs per peptide can be specified and a value of less than 4 is recommended. A fixed modification forces all instances of applicable residues to be modified. A variable modification gives the option for the residues to be modified.

Note

For a specific residue, only one fixed modification is allowed.

Note

To prevent long search times, select only the most frequent PTMs in the sample for PEAKS DB, and check the PEAKS PTM option to search for unspecified modifications.

2.6. Find Unspecified PTMs with PEAKS PTM

Select this option to enable a PEAKS PTM search after a PEAKS DB search finishes. PEAKS PTM analyzes spectra with good *de novo* sequences that remain unidentified by PEAKS DB. The default setting for PEAKS PTM is to search with all the built-in modifications in the "Common" and "Uncommon" lists, which include all of the natural modifications in the Unimod database.

Alternatively, PEAKS PTM can search with a list of preferred modifications. Once "Search with preferred modifications" has been selected, click the "ADD/REMOVE MODIFICATION" button to select desired PTMs for the PTM search. Importantly, PEAKS PTM allows an unlimited number of variable modifications to be searched, however, limiting the number can improve search accuracy. By default, PEAKS PTM considers the same number of maximum number of variable PTMs per peptide as set in the DB search parameters. A threshold on the de novo ALC score can be specified so that only the spectra with good *de novo* sequences are analyzed by PEAKS PTM. In the PEAKS PTM result page, search results from PEAKS DB and PEAKS PTM are combined and shown.

Find Unspecified P	Ms with PEAKS PTM			^
Perform PTM search	on spectra satifying the follow	ving condition:		
De Novo ALC(%) >:	15 recommend 15%	Search with all built-in modifications	O Search with select list of modifications	
✓ Find Potential	Signature lons			

2.6.1. PTM signature lons

Some modified peptides give rise to signature ions (e.g. immonium ions of the modified residue) during fragmentation. These signature ions are included in PEAKS online to boost confidence of PTM identification. PEAKS uses the identification of these ions to perform re-scoring of identified peptide sequences. Users can specify expected signature ions for any custom modification. Alternatively, users can select "Find Potential Signature Ions" to identify potential signature ions in fragment scans.

PEAKS provides a summary of expected and potential signature ion identification on the PEAKS PTM summary page.

Table 4. Signature Ions Detected

						\leftarrow	1 - 2 of 2 ▼ →
Modification	Signature Ion name	Mass	#PSM with modification	#PSM with modification with signature ion	#PSM without modification	#PSM without modification with signature ion	Specificity (Expected/Potential)
ADP-Ribose_PXD017417-Sig	ADP m10 Adenine AMP m10-H20 ADPr-carbdiimide Adenosine-H20	428.0366 542.0683 136.0623 348.0703 524.0578 584.0902 250.0940	44477	3396 1946 15463 3456 2108 22 11239	55520	224 90 4755 299 198 30 428	Expected
Acetylation (Protein N-term)	Ace_136	136.0600	2383	527	97614	16245	Potential

PSMs with PTM = PSMs without PTM



2.7. Find More Mutations with SPIDER

Select this option to enable a SPIDER search. SPIDER performs homology searches on spectra with good *de novo* sequences (i.e. ALC>15%) that remain unidentified by either PEAKS DB and/or PEAKS PTM. SPIDER identifies novel peptides that are homologous to peptides in the searched protein database. SPIDER is good for cross-species searches and for finding point mutations of the protein. In the SPIDER result page, search results from PEAKS DB, PEAKS PTM (if applicable), and SPIDER are combined and shown.

2.8. PSM Filter

PSM filter section is a mandatory section to filter out some results under the user-defined filter. Any changes to the filter will create a new analysis with old parameters and update the results and statistics.

- **PSM -10lgP:** The PEAKS peptide score (-10lgP) is calculated for every peptide-spectrum match (PSM) reported by PEAKS DB, PEAKS PTM, and SPIDER. The score is derived from the p-value that indicates the statistical significance of the peptide-spectrum match. A peptide may be matched to many spectra, resulting in multiple PSMs. In that case, the peptide's score is calculated as the maximum among all PSM scores. For details of the scoring algorithm, please refer to the publication, "*PEAKS DB: De Novo sequencing assisted database search for sensitive and accurate peptide identification*", *Mol Cell Proteomics, 2011 Dec 20*. A minimum requirement can be set and all identifications must pass this filter.
- **PSM FDR (%):** The PSM filter can also be set to use the false discovery rate. FDR is estimated using a "decoy-fusion" approach.

Note

Decoy fusion is an enhanced target-decoy method for result validation with FDR. Decoy fusion appends a decoy sequence to each protein as the "negative control" for the search. See BSI's web tutorial (http://www.bioinfor.com/fdr-tutorial/) for more details.

- **De Novo ALC (%):** Refer to Chapter 5, Peptide *De Novo* Sequencing, Section 3.4, "Peptide Table" for the full ALC description. This filter controls the minimum ALC score required for display in the *de novo* only result. A default of 50% is set.
- **De Novo Only Tags:** The "Denovo Only Tags" tab shows a table of "de novo only" peptides with sequence tags matched to the protein.

3. Understanding PEAKS Search Results

After a PEAKS Search is complete, all analyses selected during the project creation will be combined under the workflow drop-down menu. One is the *de novo* sequencing result, which includes *de novo* peptide sequences listed, spectrum annotation and other detailed information. The other result nodes are identification results from the database search tools: PEAKS DB, PEAKS PTM, and SPIDER. If more than one database search tool is enabled, the results from previous search tools are automatically merged.

The analysis results for database search tools are presented in four tabs:

- **Summary :** This shows an outline of the PEAKS Online database search results with key statistics. The overall quality of the experiment can be examined and the filters for peptide and protein identifications can be adjusted. This page shows after the search is done.
- **Protein :** This shows a list of protein identifications. This view also visualizes protein sequence coverage and helps with protein characterization.
- **Peptide :** This shows a list of peptide identifications. This view also provides spectrum annotation and other detailed information for peptide precursor spectrum matches.
- **De novo only :** This shows a list of quality peptide sequences detected by *de novo* sequencing that remain unidentified by the PEAKS Online database search.

3.1. Result Filtration

At the top of each summary result page, filters can be set. Identifications that fall below the filters will not be visible in the result pages or exports. After changing any of these filter parameters, click the **set button** to apply and save the new filters. The results and statistical information will be updated accordingly.

Filters	0.00	Protein Group FDR (%)	22.2	:	Proteins Unique Peptides ≥	1	:	APPLY	Created:	Jul 14 2023, 02:31:14 pm
Confident PTM Site AScore ≥	15	lon Intensity (%)≥	0	:						

- **Proteins -10lgP:** The PEAKS protein score (-10lgP) is calculated as the weighted sum of the -10lgP scores of the protein's supporting peptides. After removing any redundant peptides, the supporting peptides are sorted by -10lgP scores in descending order, and the k-th ranked peptide contributes to the weighted sum with a weight of 1/k. A default threshold of 20 is set.
- **Proteins Unique Peptides:** The minimum number of unique supporting peptides for a protein identification. A unique peptide is defined as a peptide that can be mapped to only one protein group.

3.1.1. Confident PTM Site Filter

PEAKS PTM identification is site-specific allowing users to determine confident modification sites based on two parameters: minimal ion intensity and AScore.

- **Minimal Ion Intensity:** Minimum ion intensity of each fragment ion pair, which define the modification within the peptide, compared to the base peak ion intensity. A default of 2% means that at least one pair of ions (e.g. 'y' ions) defining the modified amino acid has intensity of 2% or higher for both ions within the pair.
- **AScore:** A -10LgP score relating to positional confidence of the PTM. The p-value indicates the likelihood that the peptide is matched by chance. Therefore the higher the AScore the better.

Confident PTM Site filters are applied to the protein coverage, PTM profiling view and PTM-profiling CSV export. Modifications will only be annotated above the protein sequence if it passes the filters set by the user.

		Accession	-10LgP ↓	Coverage	#Peptides	#Unique	PTM	Avg. Mass	Descr
	26	Q01082 SPTB2_HUMAN	292.64	30.25%	51	47	C D O P	274609	Spectrin beta chain, non-erythrocytic 1 OS=Homo sap 🔺
	27	P53396 ACLY_HUMAN	292.50	51.68%	50	50	C D O	120839	ATP-citrate synthase OS=Homo sapiens OX=9606 GN
	28	Q09666 AHNK_HUMAN	287.87	16.52%	71	71	CO	629101	Neuroblast differentiation-associated protein AHNAK
	29	P13797 PLST_HUMAN	287.53	70.48%	44	33	ACDO	70811	Plastin-3 OS=Homo sapiens OX=9606 GN=PLS3 PE=1
	30	Q08211 DHX9_HUMAN	285.44	36.69%	39	39	C D O	140958	ATP-dependent RNA helicase A OS=Homo sapiens O)
	31	P22314 UBA1_HUMAN	284.89	48.30%	44	43	C D O	117849	Ubiquitin-like modifier-activating enzyme 1 OS=Homo
	32	P14625 ENPL_HUMAN	279.79	50.31%	44	34	C D O P	92469	Endoplasmin OS=Homo sapiens OX=9606 GN=HSP9(
	33	P12956 XRCC6_HUMAN	279.36	57.31%	39	38	C D O	69843	X-ray repair cross-complementing protein 6 OS=Home
C	34	P0DMV8 HS71A_HUMAN	278.61	56.01%	42	21	ACDOP	70052	Heat shock 70 kDa protein 1A OS=Homo sapiens OX= 🗸
	1			i				i i	•
C P1	Coverage	Peptide PTM Profiling	ns OX=9606 GN=	PLS3 PE=1 SV=	-4				COVERAGE LEGEND PTM FILTER TOOLS
	1	MDEMARGATE VDEIDEIV	EN ENVOIN		PTUPT PV		VENDETTONI	MIDGDDNKD	VICEDEEXVI
	1	MDEMAIIQIS KDELDELK.	D Deem	dation (NO)(10.09)	ETHET LV	EANMPEPO	TRAKPITŐUP	MIDGDKWKD	AISTDETVII
S	Sample 1		AScore: N	16:26.3144				149	
	81	FQEVKSSDIA KTFRKAIN	RK EGICALG	GTS ELSSE	GTQHS YS	EEEKYAFV	NWINKALEND	PDCR hvipm	I PNTDDLFKAV
S	Sample 1	167 169				209			

3.1.2. Show Results for All Samples

For the protein and the peptide tables, "All Samples" can be selected to view the combined search results from all the samples. In the Protein page, selecting "All Samples" will change the coverage panel displayed when a specific protein is highlighted. Supporting peptides from all samples instead of just a single sample will be shown.

Show Results for All Samples 	O Specified Sample	Sample 1	\sim

3.1.3. Show Results for Specific Sample

In the Protein, Peptide and De Novo page results, the filter can also be set to "Specified Sample" which will only show the results a single sample. By clicking in the text you can search for a specific sample by name.

● Specified Sample 1 X V	1 27 11		
	Specified Sample	le Sample 1	x ~

Selecting "Specified Sample" will change the protein coverage view that is visible by clicking on a protein in the protein table. Peptides in the Protein, Peptide and De Novo Only pages will only be visible if they belong to the sample chosen. Once selected, the peptide tables will automatically add the "Area" column and the "# Spec" values will display the spectral count for the specified sample. For more information, refer to Section 3.5.2, "Peptide Table". Similarly, the protein table will also include the columns "Coverage" and "Area" for the specified sample. For more information refer to Section 3.4.2, "Protein Table".

3.2. Result Exporting

SELECT ALL CLEAR ALL Export All Search Parameters Data Refinement Denovo Database Search MGF File Summary Table Summary Table Mzxml Denovo CSV PTM Profile Summary CSV Denovo CSV with all Candidates Protein CSV Protein-Peptide CSV PTM Profiling CSV Protein Fasta Peptide CSV PSM CSV Peptides-Pepxml Peptides-MzIdentML Denovo Only CSV Denovo Only CSV with all Candidates DOWNLOAD BY SAMPLE DOWNLOAD

PEAKS Online database search results can be exported in the Export page:

- Summary Table: Content from the PEAKS DB/PTM/Spider summary statistics table saved in text CSV format.
- **PTM Profile Summary:** Content from the PEAKS DB/PTM/Spider summary PTM profile table (Table 3) saved in text CSV format.
- **Protein CSV:** A comma separated file containing all identified proteins and associated details from the protein table for all samples.
- **Protein-Peptide CSV:** A comma separated file containing all identified peptides and associated details from the peptide table found in the Protein page. Further details are provided regarding the protein the peptide supports. If a peptide supports multiple proteins, multiple rows are included for each supported protein for all samples.
- **PTM Profiling CSV:** Detailed information for the abundance of confidently identified PTMs versus unmodified peptides for all identified proteins. The PTM profile data are grouped together by protein, then by modification type, and then by the modification site on the protein sequence. The headers of the CSV column are similar to the PTM Profile Table.

- **Peptide CSV:** A comma separated file containing all identified peptides and associated details from the peptide table for all samples.
- **PSM CSV:** A comma separated file containing all identified peptide spectrum matches and associated details from the peptide table for all samples.
- **Pepxml:** An XML export of the identification results matching the pepxml schema. This can be uploaded to software that accept data in this format.
- **MzIdentML:** An XML export of the identification results matching the MZIdentML schema. This can be uploaded to software that accept data in this format.
- **Denovo Only CSV:** Top scans from De Novo results that do not have an identification result by PEAKS DB, PEAKS PTM, or SPIDER that pass the set filters. Refer to Chapter 5, Peptide *De Novo* Sequencing, Section 3.2, "Exporting *De Novo* Results" for detailed definition. A separate CSV file is generated for each sample.
- **Denovo Only CSV with all Candidates:** De Novo results from scans that do not have an identification result by PEAKS DB, PEAKS PTM, or SPIDER that pass the set filters. Refer to Chapter 5, Peptide *De Novo* Sequencing, Section 3.2, "Exporting *De Novo* Results" for detailed definition. A separate CSV file is generated for each sample.
- **Protein Fasta** : A list of protein identifications will be saved to proteins.fasta.

3.3. Summary View

The Summary view reports key statistics as an overview of the result. In the report, several statistical charts can be examined to assess result quality.





3.3.1. False Discovery Rate (FDR) Curve

Figure 1 shows the FDR curve with respect to the number of PSMs being kept after filtration by the peptide -10lgP score. By lowering the score threshold, more PSMs are kept in the filtered result. Conversely, the FDR increases because more false positives are kept. In this figure, the vertical line indicates the current score threshold. The number of PSMs and the corresponding FDR by the current score threshold are shown in the top-left corner.

PEAKS Search estimates FDR using the "decoy-fusion" approach. Decoy-fusion is an enhanced target-decoy approach that makes more conservative FDR estimations. For details of the "decoy-fusion" approach, please refer to the publication, "PEAKS DB: De Novo sequencing assisted database search for sensitive and accurate peptide identification", Mol Cell Proteomics, 2011 Dec 20.

It is commonly recommended to set FDR at <1% by adjusting the score threshold. If a rapid growth of the FDR around the 1% FDR threshold is noticed, some PSMs may be sacrificed to significantly reduce the FDR.

Note

When counting the number of PSMs, PEAKS Online keeps one peptide per spectrum at most. Thus, the number of PSMs is actually the number of spectra with assigned peptides.

Note

Decoy matches are not counted in the number of PSMs. Unless otherwise specified, decoy matches are also excluded from the other statistical numbers shown in the Summary view.

3.3.2. Result Statistics Tables

The first chart displays the summary information for each sample.

- **Sample Name:** This column lists the samples names with the exception of the first row, which shows the combination of all the samples.
- # MS2: The total MS/MS scans in the sample summed across all fractions.
- **# PSM:** Total number of peptide spectrum matches (PSMs) found in the sample.
- **# Peptides:** Total number of peptides found ignoring differences with leucine (L) and isoleucine (I) amino acids. Peptides with the same primary sequence but different PTMs are counted separately.
- **# Sequences:** Total number of different peptide backbone sequences found ignoring differences with leucine (L) and isoleucine (I) amino acids. Peptides with the same primary sequence but different PTMs are counted as one sequence.

- **# Proteins:** Provides the number of proteins given the parameters set for Protein Score (-10lgP) and number of unique peptides.
- **# Protein Groups:** PEAKS Online groups proteins identified by a common set of peptides. This number shows the number of protein groups in the filtered result based on All proteins.

Tables 1-4 show the statistical numbers of the data and results.

- Summary Statistics: Presents the total number of MS and MS/MS scans from the data.
- **Table 1. Statistics of filtered results:** Presents the statistics from the data analysis and the results after setting the desired filters. See below for further explanation.
 - **Peptide-Precursor Spectrum Match:** The number of peptide features that had an associated spectrum which matched to a database sequence.
 - **Peptide Sequences:** This represents the number of distinct peptides in the filtered result. Peptides with the same primary sequence but different PTMs are counted separately. However, if several peptides differentiated only by leucine (L) and isoleucine (I), they are counted as one peptide. Since the same peptide may be identified from multiple spectra due to data redundancy and differing charge states, this number is usually smaller than the number of peptide-spectrum matches.
 - **Protein Groups:** PEAKS Online groups proteins identified by a common set of peptides. This number shows the number of protein groups in the filtered result based on All proteins.
 - **Top Proteins:** The number of proteins that are supported by the most unique peptides in the group. Proteins in the group that share a subset of the unique peptides that support the top protein (sub-proteins) will not be included here.
 - **Proteins:** The number of proteins given the parameters set for PEAKS Protein Score(-10lgP) and number of unique peptides.
 - **Proteins (#Unique Peptides):** In Table 2, this entry counts the number of proteins identified with unique peptides. A unique peptide is a peptide with a -10lgP score above the peptide filtering threshold that can be mapped to only one protein group. Peptides are counted based on their backbone sequence while considering I/L amino acids to be the same.
 - **FDR (Peptide-Spectrum Matches):** The total number of decoy database assignments to spectra relative to the total number of target database assignments to spectra represented as a percentage.
 - **FDR** (**Peptide Sequences**): The total number of decoy database assignments to unique peptide sequences relative to the total number of target database assignments to unique peptide sequences represented as a percentage.
 - **FDR** (**Protein Sequences**): The total number of decoy protein groups where the top hit in the protein group is a decoy database protein relative to the total number of target database protein groups where the top hit is from the target database.
 - **-10lgP Cutoff:** The value is determined by the estimated PSM FDR. Only the peptides with PSMs above the -10lgP score threshold are listed in the Peptide view.
 - **De Novo only Spectra:** Corresponds to the number of existing De Novo sequencing results without a positive protein identification, given the filters set for the project.
- Table 2. Number of identified peptides in each sample by the number of missed cleavages. Presents the number of identified peptides in each sample with the number of missed cleavages, indicating the enzyme digestion efficiency.
- **Table 3. PTM Profile:** shows modifications identified in the filtered result and the number of PSMs containing each modification. Confident PTM site filters do not apply here.
 - Name: The given name of the PTM.

- Δ Mass: The monoisotopic change in mass caused by the PTM.
- **Position:** The amino acids modified by the PTM.
- **# PSM:** The number of PSMs with the given PTM.
- **-10lgP:** The highest -10lgP score achieved by a peptide modified by the PTM.
- Area: The area achieved by the modified peptide with highest -10lgP score.
- AScore: The AScore achieved by the modified peptide with highest -10lgP score.

3.3.3. Summary Figures

PSM Score Distribution: Figure 2 displays a histogram of the PSM -10LgP score distribution for target and decoy PSMs. This figure helps visually examine the FDR control in your search results.

Precursor mass error of peptide-spectrum matches: Figures 3(a) and 3(b) show the precursor mass errors of PSMs in filtered results. Figure 2(a) shows the distribution of the precursor mass errors in a histogram with and without software calibration. Figure 2(b) shows precursor m/z versus precursor mass error in a scatter plot after software calibration. Figures 2(a) and 2(b) help examine whether or not the mass spectrometer is properly calibrated. For a well-calibrated instrument, precursor mass errors should center at 0 ppm across the range of m/z.

Data Completeness: Figure4 displays a histogram of data completeness based on the percentage of replicate protein groups identified across samples in the analysis. Overall data completeness is an average of the % replicates identified for all protein groups.

3.3.4. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

2 data files produced no results. (Expand to see list of zero result data files)

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.



If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics

No Results Found

3.4. Protein View

The Protein view lists protein identifications that have been filtered by the current settings set for the project. It also visualizes the protein sequence coverage for identified proteins. After clicking the "Peptide" tab we can view the supporting peptide list, double click any peptide can lead to that peptide sequence in the peptide page.

2 Analysis 1: Da Novo + Databasa 🖉 🔮 🗮	<				Show Results for						
- Analysis I: De Novo + Database 🖉 🖊	2 III 🖸 S	howing Top Proteins			 All Samples 	O Specified Sample	Sample 1	- E	port Protei	n Protein-Pep	tide PTM-Pr
Search A De Novo	3	Access	on	-10LoP Cove	EPentides	#I Inique	PTM	Avr. Mass	÷	1 - 100 of	/003 -
DB Search	a7 1 F	21333 FLNA_HUMAN		691.20	62.90%	156 149	C D O	280739	Filamin-A OS=H	lomo sapiens OX	=9606 GN=FL
Export	2 F	35579(MYH9_HUMAN		677.12	57.65%	141 117	C D O	226532	Myosin-9 OS=H	omo sapiens OX	=9606 GN=MY
•	3 (15149 PLEC_HUMAN		673.33	35.67%	139 128	CDO	531791	Plectin OS=Hon	no sapiens OX=9	606 GN=PLEC
	5 4 C)75369/FLNB_HUMAN		664.04	60.99%	122 112	CDO	278164	Filamin-B OS+H	lomo sapiens OX	=9606 GN=FL
	oept 6	100610ICLH1 HUMAN		615.96	20.70% 54.69%	84 84		409009	Clathrin beauvio	hain 1 OS=Hom	atalytic suburi o saniens OX+
	1 0 0	99Y490ITLN1 HUMAN		614.80	51.16%	84 76		269767	Talin-1 OS=Hon	no sapiens OX=9	606 GN=TLN1
	0 8 F	14618 KPYM_HUMAN		610.22	76.46%	86 82	C D O	57937	Pyruvate kinase	PKM OS=Homo	sapiens OX=
	2 9 C)43707 ACTN4_HUMAN		608.98	71.13%	83 54	CDO	104854	Alpha-actinin-4	OS=Horno sapie	ns OX=9606 0
	0V0 (
	Coverage	Peptide PTM Profil	ing 🛄								
	P21333 FLN/	LHUMAN Filamin-	A OS=Homo sapiens	OX=9606 GN=FLN	A PE=1 SV=4				COVERA	IGE LEGEND PT	M FILTER TO
	,	MCCCUCDACC	CARCANDOCO	23 O VDTDDDDDDDD	TENDI SED SD	HEFTOONEE	DENERLYCY	SVD TANT OF	TODOTR		
	Sample 1	MSSSHSKAGY	SANGAAFGGG	VDIRDALMPA	IERDEREDAP	WKKIQQNIFI	WCNEHLKCV	SULTHINGTO	LODGIK	LIAL	
ta Search 🔍 🗙 🧪	81	LEVLSQKKMH	RKHNQRPTFR	QMQLENVSVA	LEFLDRESIK	LVSIDSKAIV	DGNLKLILGL	IWTLILHYSI	SMPMWD	EEED	
ple 1 🥥 🔳	Sample 1					205 210					
	161 Sample 1	EEAKKQTPKQ	RLLGWIQNKL	PQLPITNFSR	DWQSGR ALGA	LVDSCAPGLC	PDWDSWDASK	PVTNAR EAMQ	QADDWL	GIPQ	
	241	VITPEEIVDF	NVDEHSVMTY	LSQFPKAKLK	PG APLRPK LN	PKKAR AYGPG	IEPTGNMVKK	RAEFTVETRS	AGQGEV	LVXV	
	Sample 1 321	EDPAGHOEEA	K VTANNDKNR	TESVWYVPEV	TGTHKVTVLF	AGOHIAKSPF	EVYVDKSOGD	ASKVTAOGPG	LEPSGN	IANK	
	Sample 1	-									
	401	TTYFEIFTAC	AGTGEVEVVI	QDPMGQK GTV	EPQLEAR GDS	TYR CSYQPTM	EGVHTVHVTF	AGVPIPRSPY	TVTVGQ	ACNP	
	Sample 1										
		403									
		200 0 0 0 0 0 0 0 0 0 0 0 0									
Peptide	/R	Unique	-10LgP 1	Mass _ Leng	th _ ppm _	m/z :	RT	Scan :	#Spec	Start 2368	End 2391
Peptide VHSPSGALEEC(+57.02)VVTEIDQDKYAV LRN(+0.98)GHVGISFVPK	/R	Unique	-10LgP 1	Mass Leng 2765.2966 1423.7936	th ppm - 24 3.1 13 2.1	m/z 692.3336 475.6061	RT	Scan	#Spec _	Start	End 2391 2014
Peptide VHSPSGALEEC(+37.02)VTEIDODKYAV LRN(+0.98)GHVGISFVPK FGGEHVPNSPFQVTALAGDQPSVQPPLR	/R	Unique	-10LgP 1 . 124.54 124.50 124.49	Mass Leng 2765.2966 1423.7936 2944.4832	th ppm 24 3.1 13 2.1 28 3.0	m/z 692.3336 475.6061 982.5046	RT 5182:e151 24.45 2782:e151 62.92 8580:e151	Scan	#Spec1 1 2	Start	End 2391 2014 1753
Peptide VHSPSGALEEC(+57.02)YYTEIDODKYAV LRN(+0.98)GHVGISFVPK FGGEHVPNSFP(VTLALGDQPSVQPPLR FVPAEMGTHVSVK	/R	Unique .	-10LgP 1 . 124.54 124.50 124.49 124.49	Mass Leng 2765.2966 1423.7936 2944.4832 1501.7599	th : ppm : 24 3.1 13 2.1 28 3.0 14 2.4	m/z 692.3336 475.6061 982.5046 501.5951	RT 40.37 5182:e151 24.45 2782:e151 62.92 8580:e151 21.48 2345:e151	Scan ; 108_SS1040HCD 108_SS1040HCD 108_SS1040HCD 108_SS1040HCD	#Spec1 1 2 2	Start 2368 2002 1726 2202	End 239 2014 1753 2213
Peptide VHOPSGALEECK-57.02YVTEIDODKYAV LRN(+0.99)GHVGISFVPK FGGEHVPNSPFQVTALAGOQPSVQPPLR FVPAEMGTHTVSVK EGPYSISVUSDEVPRSPFK	/R	Unique :	-10LgP 1 , 124.54 124.50 124.49 124.49 124.48	Mass Leng 2765.2966 1423.7936 2944.4832 1501.7599 2368.1587	th ppm 24 3.1 13 2.1 28 3.0 14 2.4 21 2.1	m/z 692.3336 475.6061 982.5046 501.5951 790.3951	RT 40.37 5182:e151 24.45 2782:e151 62.92 8580:e151 21.48 2345:e151 61.11 8323:e151	Scan	#Spec	Start 2368 2002 1726 2202 1516	End 239 201 175 221 153
Peptide VHSPSGALEEC(+57.02)/VTE DOD/VAV LRN(+0.38)GHV0SFVPK FOGEHVMPSF071LA80QF9VQPPLR FVPEMOTHTVSVK EGPVSISVLYODEEVPRSPK AWG96LEG9VV6K	/R	Unique . ···· ···· ···· ···· ···· ···· ···· ···· ···· ···· ···· ···· ···· ···· ···· ····	-10LgP 1 124.54 124.50 124.49 124.48 124.48	Mass Leng 2765.2966 1423.7936 2944.4832 1501.7599 2368.1587 1225.6455	th ppm 24 3.1 13 2.1 28 3.0 14 2.4 21 2.1 13 2.6	m/z . 692.3336 475.6061 982.5046 501.5951 790.3951 613.8316	RT 40.37 5182:e151 24.45 2782:e151 62.92 8580:e151 61.11 8232:e151 33.59 4157:e151	Scan	#Spec 1 1 2 2 1 1	Start 2368 2002 1726 2202 1516 581	End 239 201 175 221 153 59
Peptide (VHSPBGALEEC(+57,02VV/EDQDK/AV LEN(+0.08)(HOVSFVPK FGGEHVPNSFQVTALABDQPSVQPFLR FVPABMGTHTVSVK EGYSISVIVGEVPRSPK AVGPGLEGGVVGK AONIM(+159)LUVGVFGR	/R 2	Unique ,	-10LgP 1 124.54 124.50 124.49 124.49 124.48 124.48 124.48	Mass Leng 2765 2966 1423 7936 2944 4832 1501 7599 2368 1587 1225 6455 1449 7510	th ppm 24 3.1 13 2.1 28 3.0 14 2.4 21 2.1 13 2.6 14 2.8	m/z 692.3336 475.6061 982.5046 501.5951 790.3951 613.8316 484.2590	RT 40.37 5182:e151 24.45 2782:e151 62.92 8580:e151 21.48 2345:e151 61.11 8323:e151 33.59 4157:e151 20.24 2138:e151	Scan	#Spec . 1 2 2 1 1 1	Start 2368 2002 1726 2202 1516 581 2585	End 239 201- 175 221: 153 59: 259:
Peptide VH3P9GALEEC(-37.02WTEIDORKAW ERK(-0.98)GHVGISFVPK FGGEHVRISFPCVTAL60D(PSVOPPLR FVPABMGTHYGVK EGPYSISVLYGDEEVPRSPK AWGPGLEGOVVGK AGNIM(-13.99)LLVGVHGPR VWPGEAGEVAHVC(-57.02WSEDIR	/R	Unique	-10LgP 124.54 124.49 124.49 124.48 124.48 124.48 124.47 124.46	Mass Leng 2765/2966 1423.7936 2944/4832 1501.7599 2368.1587 1225.6455 1449.7510 2535.1489	th ppm 24 3.1 13 2.1 28 3.0 14 2.4 21 2.1 13 2.6 14 2.8 21 2.6	m/z 692.338 475.6061 982.5046 501.5951 790.3951 613.8316 484.2590 846.0591	RT 24.45 2782-157 24.45 2782-157 27.48 2345-157 61.11 8323-157 33.59 4157-157 20.24 2138-157 69.53 9552-151	Scan	#Spec 1 1 2 2 1 1 1 1	Start 2368 2002 1726 2202 1516 581 2585 635	End 239 201 175 221 153 59 259 65
Peptide VHSPGGALEECI-57.02/WYTEIDOKYAV LEN(-0.036)KOISFVPK FGGEWVPNSPF0/TALABOQPSVQPPLR FGGEWVPNSPK EGPYSISVLVGDEEVPRSPFK AWSPGLEGOVVGK AWSPGLEGOVVGK AWSPGLEGOVVGK MUTVGISVPK SOLUCIONAL MUTVGISVPK MUTVGISVPK	/R 8	Unique	-10LgP 1 124.50 124.49 124.49 124.48 124.48 124.48 124.48 124.46 124.46	Mass Leng 2765 2066 1423.7936 2944.4832 1501.7599 2368.1587 1225.6455 1429.7510 2335.1489 1153.6244 153.6244	ppm ppm 24 3.1 13 2.1 28 3.0 14 2.4 21 2.1 13 2.6 14 2.8 21 2.6 11 2.0	m/z 692.3336 475.6061 982.5046 501.5951 790.3951 613.8316 484.2590 846.0591 385.5495	RT 40.37 5182:e151 24.45 2782:e151 62.92 8580:e151 21.48 2345:e151 61.11 823:e151 33.59 4157:e151 20.24 2138:e151 22.01 2415:e151	Scan	#Spec 1 1 2 2 1 1 1 1 1 2	Start 2368 2002 1726 2202 1516 581 585 635 2004	End 239 201 175 221 153 59 259 65 201
Peptide UHBPR04LEEC4-57.02/WTEIRORK/AW LRN(+0.38)GH/V015FVPK F0GEH/VPNSPFQ/TALABODPSVGPLR FVPAEMGTH/V3VK EGPY355U/V00EVPASPK AGNNM(+15.99)LU/SVHOPR NGHV015FVPK AGNNM/LC457.02)DLSK	IR 2	Unique ···	-10LgP 124.54 124.49 124.49 124.49 124.48 124.48 124.47 124.46 124.46	Mass Leng 2765/2966 1423.7936 1423.7936 2944.4832 1501.7599 2368.1587 1225.6455 1449.7510 2555.1489 1153.6244 1653.8145 1653.8145	ppm ppm 24 3.1 13 2.1 28 3.0 14 2.4 21 2.1 13 2.6 14 2.8 21 2.6 14 2.8 21 2.6 14 2.8 21 2.6 14 2.8 21 2.6 14 2.8 21 2.6 14 2.8 2.1 2.6 14 2.8 2.1 2.0 16 3.1	m/z 692.3336 475.6061 982.5046 501.5951 790.3951 618.8316 484.2590 846.0591 385.5495 552.2805	RT 24.5 2782-e151 24.45 2782-e151 24.45 2782-e151 21.48 2345-e151 61.11 8323-e151 20.24 2138-e151 20.24 2138-e151 20.03 9552-e151 21.84 2377-e151	Scan	#Spec1 1 2 2 1 1 1 1 2 1 2 1	Start 2368 2002 1726 2202 1516 581 2585 635 2004 2150	End 239 201 175 221 153 59 259 65 201 216
Peptide WH3PB0ALEEC(-57.02)/WTEID00K/AV LBN(-0.98)GHV015F/PK F0GEHVPNSPFQ/TALAB00PSV0PPLE F0GEHVPNSPFQ/TALAB00PSV0PPLE EGPV315V/V0GEVPR5PFK AWOPBLEGOV/V0K AGNIMI(-15.99)LLVCVH0FR MYPCEAGETAVHVLC(-57.02)NSEDIR MGHV03FVPL APSVANVGSHC(-57.02)NLSLK M(-15.99)CC7.22(M-15.39)DNKDG	IR R ISC(+57.02)SVEYIPYE	Unique ···	-10LgP 1 124.55 124.49 124.49 124.48 124.48 124.48 124.48 124.46 124.46 124.46	Mass Leng 2765 2966 1423 7936 2944 4832 1501 7599 2306 1587 1225 6455 1449 .7510 2535 1489 1153 6244 1653 8145 4476 9282 2	ppm ppm 24 3.1 13 2.1 28 3.0 14 2.4 21 2.1 13 2.6 11 2.0 12 2.6 11 2.0 16 3.1 40 3.8	m/z 692.336 475.6061 982.5046 501.5951 613.8316 484.2590 846.0591 385.5495 552.2805 1120.2437	RT	Scan	#Spec 1 1 2 2 1 1 1 1 1 2 1 1 1 1	Start 2368 2002 1726 2202 1516 581 2585 635 2004 2150 1400	End 239 201 175 221 153 59 259 65 201 216 143
Peptide VH3PB04LEECI-57.02VVTEIDORKVAV EN(+0.39)GHV015FVPK F60EHVPNSPFQVTALABODPSVDPLR FVPAEMOTHVSVK EGPV315VLV00EVPRSPK AGNNM(-15.99)LUVOHOPR NGHV015FVPK AGNNM(-15.99)LUVOHOPR NGHV015FVPK AGNNM(-15.99)LUVOHOPR NGHV015FVPK AGNNM(-15.99)LUVOHOPR NGHV015FVPK AGNNM(-15.99)LUVOHOPR NGHV015FVPK AGNNM(-15.99)LUVOHOPR AGNNM(-15.99)L	IR 8 SSC(+57.02)SVEVIPVE	Unique ····	-10LgP 1 124.50 124.49 124.49 124.49 124.48 124.48 124.46 124.46 124.46 124.46	Mass Leng 2765 2966 1423 7936 1423 7936 2944 4832 1501 7599 2368 1587 1225 6455 1449 7510 2535 1489 1153 6244 1633 8145 1439 6244 1633 8145 1496 9282 1908 9105 1905	ppm ppm 24 3.1 13 2.1 28 3.0 14 2.4 21 2.1 13 2.6 14 2.8 21 2.6 11 2.0 16 3.1 40 3.8 17 2.3	m/z 692.338 475.6061 982.5046 501.5951 790.3951 613.8316 484.2590 846.0591 385.5495 552.2805 5120.2457 637.3123	RT 40.27 5182-815 24.45 2782-815 62.92 8500-815 21.48 2345-815 61.11 8323-815 33.99 4157-815 20.24 2138-815 20.39 9552-815 21.04 2377-815 22.58 8515-815 62.25 8515-815 57.12 7731-815	Scan :	#Spec 1 1 2 2 1 1 1 1 1 2 1 1 2	Start 2368 2002 1726 1726 2202 1516 581 2585 635 2004 2150 1400 1516	End 239 201 175 221 153 59 259 65 201 216 143 153
Peptide VH3PB0ALEEC(-57.02)VVTEID00K/AV LBN(-0.98)GHV015FVPK F0GEHVPNSPFQVTALAB00PSV0PPLB FVPAEMOTHVSVK EGPVSISV/V0GEVPRSPFK AWOPBLEGOVV0K AGNIMA-15.99)LUVCVH0PR NGHV015FVPK AGNIMA-15.99)LUVCVH0PR NGHV015FVPK AGNIMA-15.99)LUVCVH0PR LGPV015FVPK LGPV015FVFK LGPV015F	1R 2 SC(+57.02)SVEYIPYE	Unique , 	-10LgP 1 , 124.50 124.49 124.49 124.49 124.48 124.48 124.48 124.46 124.46 124.46 124.46 124.46 124.46 124.46	Mass Leng 2785.2866 1423.7936 1423.7936 2944.4832 1501.7959 225.4455 1425.7959 1225.6455 1449.7510 2353.1489 1153.8145 1153.8244 11903.9105 2227.0481	ppm ppm 24 3.1 13 2.1 28 3.0 14 2.4 13 2.1 14 2.4 21 2.1 13 2.0 14 2.8 21 2.6 11 2.0 16 3.1 40 3.8 17 2.3 19 2.4	m/z 475.6061 962.5046 501.5951 613.8316 484.2590 846.0591 385.5495 552.2805 1120.2437 637.3123	RT 10.87 51.82.e1.51 24.45 278.24.151 62.92 858.0e1.51 21.48 23.24.151 83.99 41.57.e151 30.99 41.57.e151 9.53 9552.e151 22.01 24.158.e151 22.02 21.88.e151 20.24 21.88.e151 22.01 24.158.e151 20.25 8.515.e152 8.515.e152 8.515.e151 49.24 6.556.e157 49.24 6.556.e157	Scan 108,551040HCD 108,551040HCD 108,551040HCD 108,551040HCD 108,551040HCD 108,551040HCD 108,551040HCD 108,551040HCD 108,551040HCD 108,551040HCD	#Spec 1 1 2 1 1 1 1 2 1 1 2 1 1 2 2 2	Start 2368 2002 1726 2202 1516 581 2585 2004 2150 1400 1516 6556	End 239 2011 175 221 153 59 259 65 201 216 143 153 67
Peptide VHSPSGALEECI-57.029/VTEIDORVAA EM(+0.98)GHVGISFVFK FGGEHVPNSPFQVTALAGODPSVDPLB FVPAEMGTHVSVK EGPVSISVLVGDEEVPASFK AGNNML+15.99JLLVGVHGFR NGHVGISFVFK AGNNML+15.99JLLVGVHGFR NGHVGISFVFK AGNNML+15.99JLLVGVHGFR NGHVGISFVFK AGNNML+15.99JLLVGVHGFR M(+15.99JLLVGVHGFR AGNNML+15.99JLLVGVHGFR AGNNML+15.99JLLVGVHGFR M(+15.99JLLVGVHGFR AGNNML+15.99JLLVGVHGFR M(+15.99JLL	/R 8 SSC(+57.02)SVEYIPYE	Unique . ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ····· ······ ····· ·····	-10LgP : 124.50 124.50 124.49 124.49 124.49 124.48 124.48 124.46 124.56 125.56 125.	Mass Leng 2765.2066 1/423.7936 1/423.7936 2944.4832 1501.7599 2864.1887 1225.6455 1/449.7510 2333.1489 1153.6244 1153.6244 1903.9105 1906.9105 2227.0481 2922.5444 292.5444	ppm ppm 22 3.1 13 2.1 28 3.0 14 2.4 21 2.4 21 2.6 14 2.8 21 2.6 14 2.8 21 2.6 14 2.8 21 2.6 14 2.8 21 2.6 14 2.8 21 2.0 16 3.1 40 3.8 17 2.3 19 2.4 29 2.0	m/z 692.3336 475.6061 982.5046 501.9591 790.3951 790.3951 484.2590 846.0591 385.5495 552.2805 1120.2437 637.3123 557.7706 665.1807 965.1807	RT 40.37 5182:e157 62.92 8350:e157 62.92 8350:e157 61.11 8322:e157 83.59 4157:e157 20.04 2138:e157 20.04 2138:e157 21.04 2377:e157 57.12 7731:e157 57.12 7747 57.12 7747 57.1	Scan 108_S51040HCD_ 108_S5104HCD_ 108	#Spec ; 1 1 2 2 1 1 1 1 1 2 2 1 1 1 1 2 2 3 3	Start 2368 2002 1726 2202 1516 581 2585 635 2004 2150 1400 1516 655 2005	End 239 201 175 221 133 59 259 65 201 216 143 153 67 7 199
Peptide VHSPSGALEEC + 37 02/WYTEIOGXK/AV LRH(+0.98)6H015FVPR FGGEHVPRISPF0/TALABQD/SV0PPLB FVPABMGTHYTVSVK EGPYSISV/VDGEVPRSPFK AWOPGLEGOV/WSK ABINIM(-15.99)LU/CVH0PR VMPCALGEVAHVLC(+57.02)MSEDIR NOHV03FVPK ADIAN/HVLC(+57.02)MSEDIR MV1539/SVC/57.02)MSEDIR LSPFMAD/PM36TL0.5LLTATVVPPSGR DAVY06L.52FSVCMTTEUEL05TC() VGSAAD/PM36TL0.5LLTATVVPPSGR DAVY06L.52FSVCMTTEUEL05TC()	IR 8 ISC(+57.02)SVEYIPYE (+57.02)R	Unique -	-10LpP : 124.50 124.50 124.49 124.49 124.49 124.48 124.48 124.46 124.46 124.46 124.46 124.46 124.46 124.45 124.45 124.45	Mass Leng 2768-2060 1423.798 12423.7986 2944.4832 1501.7599 2866.1887 1225.6455 1449.7510 2335.1489 1153.845 1437.9282 2935.1489 1153.845 4476.9282 2227.0481 2892.544 3107.4583 3107.4583	th ppm : 24 3.1 13 2.1 28 3.0 14 2.4 21 2.4 21 2.6 11 2.6 11 2.0 11 2.0 12 2.0	m/2 692.3938 475.6061 982.5046 501.5951 790.3951 613.8316 484.2590 8846.0591 385.5495 552.2805 1120.2437 120.2437 1120.	AU 27 5182:e151 24.45 2782:e151 222 8506:e151 21.42 2836:e151 222 8506:e151 21.42 2434:e151 223 223 31.59 4157:e151 22.01 2438:e151 22.01 2415:e151 22.01 2415:e151 22.01 2415:e151 22.02 8515:e151 24.22 8515:e151 24.22 8515:e151 24.22 8515:e151 84.02 24.55:e151 34.24 6.556:e152 84.01 24.02 35.41 24.02 8515:e151 84.02 35.42 81.52 81.52 84.01 35.42 81.02 84.01 84.01 35.43 81.02 84.01 84.01	Scan 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_ 108_SS1040HCD_	#Spec ; 1 1 2 2 1 1 1 1 1 2 2 2 3 3	Start 2368 2002 1726 2202 1516 581 2385 635 2004 2150 1400 1516 656 1965 2074	End 239 201 175 221 153 59 259 65 2010 216 143 153 67 7 199 210
Peptide VHSPSGALEECI-57.02W/VTEIDGOKVA1 EM(+0.98)GHVGISFVPK FGGEHVPNDFPCVTALAGODPSVOPLE FVPABMOTHVSVK EGPVSISVUVGDEVPRSPK AMVPGLGEVVVGK AGSINM(+15.99)LLVSVHOER NGHVGISFVPK AGSINM(+15.99)LLVSVHOER NGHVGISFVPK AGSINM(+15.99)LLVSVHOER AGSING(+15.92)LSLK M(+15.99)SC(+57.02)M(+15.99)DKKO EGPVSISVUVGDEVPR LSFFMADRGAPQGFHPB DAGYGGLSLSEGFSVV0NTELEGOTC) TUTVSSLGSGFSVN0NTELEGOTC) TUTVSSLGSGFSVN0NTELEGOTC)	/R 8 ISC(+57.02)SVEVIPVE (+57.02)R	Unique . - - - - - - - - - - - - - - - - - - - - - - - - - - - - - -	-10LgP 124.55 124.49 124.49 124.49 124.49 124.48 124.48 124.45 124.45 124.46 124.46 124.46 124.46 124.46 124.44 124.44	Mass Long 2765.2064 1.427.7036 2844.4982 2844.4982 1501.7599 2246.1897 1225.6455 1.449.7110 1449.7110 1.455.8145 155.8145 1.155.644 1156.2022 2200.522 2402.5444 3167.4965 2407.5444 3167.4965	ppm j 13 2.1 13 2.1 14 2.4 13 2.1 14 2.4 13 2.1 14 2.8 11 2.0 11 2.0 11 2.0 12 2.1 13 2.8 14 2.8 11 2.0 12 2.1 13 2.0 14 2.8 12 2.1 3.8 1.1 3.8 1.1 3.9 1.8 3.9 1.8 3.2 2	m/z 692.3336 475.6061 982.5046 501.5951 790.3951 613.8316 484.2590 846.0591 845.2495 555.2405 555.2405 555.7706 695.3123 557.7706 965.1927 1058.8280 755.1526 755.1526	RT 40.37 5182:e157 24.45 2782:e157 24.45 2782:e157 20.42 2850:e157 20.42 2138:e157 20.42 2138:e157 20.42 2138:e157 20.42 2138:e157 20.42 2138:e157 20.42 2137:e157 21.42 2377:e157 21.42 2377;e157 21.42 2377;e157 21	Scan 108_SS1040HCD_ 108_SS104HCD_ 108_SS104HC	#Spec 1 1 2 2 1 1 1 1 1 2 2 3 1 1 1 1 1 1 1 1	Start 2368 2002 1726 2202 1516 581 2585 635 2004 2150 1400 1516 656 1965 2074 2334 2834	End 239 201 175 221 153 59 259 259 201 216 143 153 67 199 210 226 236
Peptide VHSPBGALEEC + 37.02/WTEIOGKX4A LEN(+0.93)6/WGISP/PK FGGEHVPNSPFO/TALABODPSVOPPLE FVPAEMGTHYTSVK EGPYSISU/VGDEEVPRSPK AWGPGLEGOVYGK AGINIM-13.99/LU/GYHORP WTPGCAEGYAHVIC(+37.02/MSEDIR MGH/03/SFVPK AGINIM-13.99/LU/GYHORP MGH/03/SFVPK AGINIM-13.99/LU/GYHORP MGH/03/SFVPK AGINIMOSFVPK MGALPHINGTUSLLTATVVPS0R DAGY0GLSLSKEGPSV/MITELEDGTC RLTVSSLGESUKWQPLSFX/SLNGAK WASBPGLUTTGVPASLPVSLNGAK	IR 8 SSC(+57.02)SVEVIPVE (+57.02)R	Unique	-10LgP : 124.50 124.49 124.49 124.49 124.48 124.48 124.48 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.45 124.45 124.44 124.44 124.45 124.55 124.	Mass Leng 2768.2066 1423.798 2244.4822 1501.799 2366.1867 1225.645 1424.7510 2333.149 1135.0244 1153.0244 1193.80244 2905.9222 2692.5444 3167.4563 3167.4563 2905.5720 2563.2790 2563.2790	ppm ppm 24 3.1 18 2.1 28 3.0 14 2.4 21 2.6 11 2.6 11 2.6 11 2.6 11 2.0 12 2.6 11 2.0 12 2.6 13 2.6 14 2.8 19 2.4 29 2.0 80 1.8 22 2.0 23 2.2 24 2.0	m/2 692.3036 475.6061 982.5046 501.5951 790.3951 613.8316 484.2590 846.0591 385.5495 552.2805 1120.2437 1020.437 1020.437 1058.8280 665.1526 667.1526 677.1526	AD 37 S182:e151 4.0 37 S182:e151 24.45 2782:e151 22.4 25 2830:e151 11.48 2243:e151 31.94 1475*e151 20.24 238:e151 20.24 2138:e151 20.24 2138:e151 20.21 2415:e151 21.42 2831:e151 22.01 2415:e151 22.01 2415:e151 24.25 8515:e151 24.28 8515:e151 24.28 8710:e151 25.48 8710:e151 25.48 870:e151 24.84 870:e152 25.48 870:e152 25.48 870:e151 24.38 870:e151 25.48 70:e151 25.48 70:e151 25.48 70:e151 26.43 870:e151 27.43 870:e151	Scan 108_551040HCD_ 108_55104HCD_ 108_55104HC	#Spec 1 1 2 2 1 1 1 1 1 2 2 1 1 2 2 3 3 1 1 1 2 2 3 1 1 1 2 2 1 1 1 1	Start 2368 2002 1726 2202 1516 581 2985 2014 2150 1400 1516 655 2074 2334 1548	End 239 201 175 221 133 59 65 201 216 61 143 153 67 210 210 216 210 216 143 153
Peptide VHSPBGALEEC(-37.02W/TEIOGOKVA) ERK(-0.98)GHVGISFVPK FGGENVPNSFFOVTALGOGPSVOPFLE FVPABMGTHYTSVK EGPYSISVLYGDEEVPRSFK AWYGPLGEGVVGK AGNIM(-13.99)LLVGYHGPR WYPCGAEGYANHUC(-57.02)MCHGPR AGNIM(-13.99)LLVGYHGPR AGNIM(-13.99)LLVGYHGPR AGNIM(-13.99)LLVGYHGPR AGNIM(-13.99)LLSF AGNIM(-13.99)LLSF AGNIM(-13.99)LSF AGNIM(-13.99)LSF AGNIM(-13.99)LSF AGNIM(-13.99)LSF AGNIM(-13.99)LSF AGNIM(-13.99)LSF AGNISSLISSE AGNISSLISSLISSLISSLISSLISSLISSLISSLISSLISS	rR 2 ISC(+57.02)SVEYIPYE (+57.02)R	Unique . - - - - - - - - - - - - - - - - - - - - - - - - - -	-10LgP 1 124.55 124.49 124.49 124.49 124.49 124.48 124.48 124.45 124.45 124.45 124.45 124.45 124.45 124.45 124.45 124.45 124.41 124.41 124.41	Mass Long 2265.2065 1.422.7036 2444.4932 2.944.4932 1501.7599 2.944.1937 1225.46455 1.254.4555 1449.7510 2.355.1489 153.6244 1.153.6244 153.6244 1.059.9105 1222.5444 3.167.4563 2005.5720 2.966.3799 2564.195 2.964.195	ppm j 13 2.1 13 2.1 14 2.4 13 2.1 14 2.4 13 2.6 14 2.8 21 2.1 13 2.6 14 2.8 21 2.0 14 2.0 10 3.8 19 2.4 20 1.8 20 1.8 20 2.5 20 2.5 20 2.2	m/2 475.6061 962.5046 963.5046 963.5046 963.5046 963.5046 963.5045 962.5045 962.5045 962.5045 962.5045 962.5045 962.5045 962.5045 962.5045 962.5045 962.5046 962.5046 963.5045 973.5045 9	AU S1500 + 55 24.84 2728-115 24.24 2728-115 24.24 2828-115 21.48 2228-115 20.43 2238-115 20.43 2328-115 20.42 2128-115 20.42 2128-115 20.42 2128-115 20.42 2138-115 20.42 2145-115 20.45 8075-8075 20.45 8075-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 20.45 8076-8075 <td>Scan 100, 551040+CD, 100, 55100+CD, 55100+CD, 55100+CD, 55100+CD, 55100+CD, 55100+CD, 55100+CD, 55100+CD,</td> <td>#Spec 1 1 1 2 2 1 1 1 1 1 2 2 3 3 1 1 1 1 1 1</td> <td>Start 2368 2002 1726 21726 1516 581 2885 635 604 2150 1400 1516 656 1965 2074 1516 656 1965 2074 2334 1548 101 102</td> <td>End 239 201 175 221 133 59 65 201 216 67 143 143 153 67 199 210 236 157 12</td>	Scan 100, 551040+CD, 100, 55100+CD, 55100+CD, 55100+CD, 55100+CD, 55100+CD, 55100+CD, 55100+CD, 55100+CD,	#Spec 1 1 1 2 2 1 1 1 1 1 2 2 3 3 1 1 1 1 1 1	Start 2368 2002 1726 21726 1516 581 2885 635 604 2150 1400 1516 656 1965 2074 1516 656 1965 2074 2334 1548 101 102	End 239 201 175 221 133 59 65 201 216 67 143 143 153 67 199 210 236 157 12
Peptide VHSPBGALEEC(+37.02)/V/TEIOOKX/AV LEN(+0.36)/AV/TEIOOKX/AV LEN(+0.36)/AV/TEIOOKX/AV EG0PV315/V/TGDEVPR8PK AVGPGLEG0V9K AGNNM(+15.99)LLVGVHORP VIPOGLAGYAVHVCL(+37.02)/N/SDBIR NGHV015FVPK AGNNM(+15.99)LLVGVHORP VIPOGLAGYAVHVCL(+37.02)/N/SDBIR AGNNG/SHC(+37.02)/N(+15.99)DNK0G EGPV315/V/SDBCVPR LSPFMADIRDAPOPHPOR VGSAADIPINETOLLLTATVPPS0R DAGY0GLSLSIEGPSK/VII/VPS0R DAGY0GLSLSIEGPSK/VII/VPS0R AGN/H-15.99)LVGVHVDL55/VII/VPS0R VISSADGPLITTGVPS1LEFDESIK VISSADGPLITTGVPS1LEFDESIK VISSADGPLITSVPS1LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDESIK VISSADGPLITSVP51LEFDSVP1LE	IR L SSC(+57.02)SVEYIPYE (+57.02)R (+57.02)R	Unique J <td>-10LgP 1 124.50 124.49 124.49 124.49 124.48 124.48 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.45 124.44 124.45 124.55 124.55 124.55 125.</td> <td>Mass Leng 2765.2064 2044.4822 1223.7936 2044.4832 1301.7599 2366.1847 1225.6455 1.449.7510 2335.1489 1133.6244 1153.8245 2290.522 2900.5720 2290.5720 2564.1995 3994.0552 2900.0572 2264.1995 3904.0552 2700.1011</td> <td>i ppm i 24 3.1 3.1 13 2.1 2.8 14 2.4 2.1 21 2.1 1.1 13 2.6 1.1 14 2.8 1.1 40 3.8 1.1 40 3.8 1.1 40 3.8 1.1 29 2.0 30 30 1.8 2.2 26 2.5 2.0 20 2.2 2.7 20 2.2 2.7 20 2.2 2.7 20 2.2 2.7 20 2.3 2.7</td> <td>m/2 692.333 475.6061 962.5046 501.5551 790.3551 613.8316 484.2590 846.0591 385.5495 552.2805 1120.2437 1056.8280 965.1007 1056.8280 973.1328 857.1360 857.1360 877.1356 80755 1000.5182 733.3756</td> <td>RT 40.37 5162-45 24.45 2782-813 262 8306-15 21.42 24345-815 114 82445-815 21.42 24345-815 202-42 8306-15 20.42 2138-15 202-42 138-16 20.53 9552-815 202-12 2138-15 21.42 2377-815 149-24 6556-15 20.42 6556-15 22.68 8710-615 20.44 6556-15 22.68 8710-615 22.68 8710-615 22.68 870-615 23.68 870-615 32.68 870-615 23.68 870-615 33.93 112-12 24.63 816-615 32.73 1039-11 27.53 1039-11 27.53 1039-11 26.75 707-613 1039-11 12-25 27.53 1039-11 12-25 12-25</td> <td>Scan 108_S51040HCD_ 108_S5104HCD_ 108_S510</td> <td>#Spec , 1 2 2 1 1 1 1 1 2 2 3 3 1 1 1 1 1 1 1 1</td> <td>Start 2288 2002 1726 2202 1516 581 2585 2004 21500 1400 1516 655 2004 21500 1400 1516 555 20204 2334 1548 101 1033 2621</td> <td>End 239 201 1755 2211 133 599 259 65 201 216 143 1533 67 199 210 0 236 1577 121 106 1577 122</td>	-10LgP 1 124.50 124.49 124.49 124.49 124.48 124.48 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.45 124.44 124.45 124.55 124.55 124.55 125.	Mass Leng 2765.2064 2044.4822 1223.7936 2044.4832 1301.7599 2366.1847 1225.6455 1.449.7510 2335.1489 1133.6244 1153.8245 2290.522 2900.5720 2290.5720 2564.1995 3994.0552 2900.0572 2264.1995 3904.0552 2700.1011	i ppm i 24 3.1 3.1 13 2.1 2.8 14 2.4 2.1 21 2.1 1.1 13 2.6 1.1 14 2.8 1.1 40 3.8 1.1 40 3.8 1.1 40 3.8 1.1 29 2.0 30 30 1.8 2.2 26 2.5 2.0 20 2.2 2.7 20 2.2 2.7 20 2.2 2.7 20 2.2 2.7 20 2.3 2.7	m/2 692.333 475.6061 962.5046 501.5551 790.3551 613.8316 484.2590 846.0591 385.5495 552.2805 1120.2437 1056.8280 965.1007 1056.8280 973.1328 857.1360 857.1360 877.1356 80755 1000.5182 733.3756	RT 40.37 5162-45 24.45 2782-813 262 8306-15 21.42 24345-815 114 82445-815 21.42 24345-815 202-42 8306-15 20.42 2138-15 202-42 138-16 20.53 9552-815 202-12 2138-15 21.42 2377-815 149-24 6556-15 20.42 6556-15 22.68 8710-615 20.44 6556-15 22.68 8710-615 22.68 8710-615 22.68 870-615 23.68 870-615 32.68 870-615 23.68 870-615 33.93 112-12 24.63 816-615 32.73 1039-11 27.53 1039-11 27.53 1039-11 26.75 707-613 1039-11 12-25 27.53 1039-11 12-25 12-25	Scan 108_S51040HCD_ 108_S5104HCD_ 108_S510	#Spec , 1 2 2 1 1 1 1 1 2 2 3 3 1 1 1 1 1 1 1 1	Start 2288 2002 1726 2202 1516 581 2585 2004 21500 1400 1516 655 2004 21500 1400 1516 555 20204 2334 1548 101 1033 2621	End 239 201 1755 2211 133 599 259 65 201 216 143 1533 67 199 210 0 236 1577 121 106 1577 122
Peptide VHSPBGALEEC(->700WTEIDODKVA) LBN(-0.98)E0HV0ISFVPK F05BHV0ISFVPK E05PUNDEF0VTAL60D0PSV0PFLB FV7ABMGTHV5VK E07V30VV0K A05NM(-13.99)LUV0VH0FR N0HV0ISFVFK A05NM(-13.99)LUV0VH0FR N0HV0ISFVFK A05NM(-13.99)LUV0VH0FR N0HV0ISFVFK A05NA000EVFR E07V30LV000EVFR LSFVFAADIOAP00FVF 00SAADPINUSTOLSLIATVVPS0R DA6V05LSLSEF9KV0HVFS0R A04(-13.99)LUV0VALEPL0HSNLEAL VV5A0F041T0VPA3LFVFTIDAK VV5A0F041T0VPA3LFVFTIDAK VV5A0F041T0VPA3LFVFTIDAK VV5A0F041T0VPA3LFVFTIDAK VV5A0F041T0VPA3LFVFTIDAK	/# 2 ISC(+57.02)SVEYIPYE (+57.02)R IVAPTKPSK	Unique . - - - - - - - - - - - - - - - - - - - - - - - - - -	104.0P 1 124.54 124.50 124.49 124.49 124.49 124.49 124.48 124.49 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.46 124.41 124.41 124.41	Mass Leng 2785.2065 1.422.7036 1.427.7036 2.244.4932 2.944.4932 1.901.7599 2.366.1897 1.225.64.955 1.449.7510 2.255.1489 1.135.6044 1.935.84.45 1.135.6044 1.935.94.55 1.909.7055 2.227.0481 2.909.5054 2.905.5720 2.909.5720 2.264.1995 2.909.2052 2.200.1011 2.909.2052 2.200.1011 2.964.191.4 1.965.2	ppm j 13 2.1 13 2.1 14 2.4 13 2.1 14 2.4 13 2.6 14 2.8 21 2.1 13 2.6 14 2.8 21 2.1 14 2.8 21 2.1 14 2.8 21 2.1 20 2.0 19 2.4 29 2.0 30 1.8 226 2.5 207 2.1 202 2.0	m/2 475.6061 965.5046 951.5951 790.3951 101.8316 484.0591 102.2490 484.0591 102.0497 105.5495 102.0497 105.6280 975.1526 9	PT	Scan 1 108, 551040+CD, 108, 551040+CD, 551040+CD, 551040+CD, 551040+CD, 551040+CD, 551040+CD, 551040+CD, 551040+CD, 551040+CD, 551040+50, 551040+50, 551040+50, 551040+50, 551040+50, 551040+	#Spec	Start 2368 2002 1726 2202 1516 581 2985 2202 1516 581 2985 2202 1516 5156 2004 1516 556 2074 2334 1011 1033 1033 2521	End 239 201 175 222 259 259 259 201 216 143 153 70 199 210 2266 157 157 12 1066 254

3.4.1. Protein View Filters

By selecting the 👩 button, the Protein Filter will pop up. It controls which protein to display.

Pro	tein Filter	
Prote	Top Proteins O All Proteins	
Prote	in description contains	
Prote	in sample area ≿	
✓	Include Contaminant Data	
Nar	ne to search for CLEA	R
	Deamidation (NQ) 🗌 Oxidation (M) 🗌 Carbamidomethylation	
	RESET CANCEL	PPLY

• Show Top or All proteins in each group: Adjust the protein list based on Protein grouping, top is selected by default. Proteins are grouped based on parsimony.

- **Top:** Show only the top proteins in each group. These proteins are supported by the most unique peptides in the group. Proteins in the group that share a subset of the unique peptides that support the top protein (sub-proteins) will not be displayed.
- All: Show all proteins in each group. Proteins are grouped together if they are supported by the same set or a sub-set of the top protein in the group. If a protein is supported by a peptide not supporting the top protein it will be added to a new group.
- **Protein accession/name contains:** If you add text here, only proteins with accessions that contain the text will be included in the protein table.
- **Protein description contains:** If you add text here, only proteins with descriptions that contain the text will be included in the protein table.
- **Protein sample area:** Filters based on total protein area found in each sample. Sample areas are calculated using the total of all peptide features from unique supporting peptides. If a protein is identified and has areas from multiple samples, only one of the samples must pass this filter to be included.

Note

Only numeric digits can be typed in the protein sample area. For example, instead of 1.00e+4, 10000 must be typed.

- **Spider Mutations:** This filter is only visible in SPIDER result view filter. Check it to view proteins that have mutations.
- **Protein Contains Modifications** : Proteins that are supported by peptides containing selected PTMs will be included.

3.4.2. Protein Table

Each row in the table represents a group of proteins that are supported by a common set of peptides. A 😔 indicates that the group has multiple proteins. To expand the group, click that button.

The following columns are sortable: Accession, -10LgP, Coverage, Peptides, #Unique, Avg Mass, when All Samples are shown.

The columns in the Protein table are:

- Accession: The accession id of the protein as seen in the FASTA database.
- **-10lgP:** The protein confidence score. Refer to Section 3.1, "Result Filtration" for detailed explanation.
- **Coverage:** The percentage of the protein sequence that is covered by supporting peptides. This is the total coverage including results from all samples, unless results are only shown for a specified sample.
- **#Peptides:** The number of high-confidence supporting peptides.
- **#Unique:** The number of high-confidence supporting peptides that are mapped to only one protein group. Unique peptides with the same sequence but different modifications are only counted once in this number.
- **PTM:** The identified modifications displayed with color-coded icons.
- Avg. Mass: The protein mass calculated using the average mass.
- **Description:** The protein's header information as seen in the FASTA database.

Note

For #Peptides and #Unique, two peptides with the same starting and ending positions in the protein are counted as one, regardless of their PTM forms. This is to follow the MCP (Molecular & Cellular Proteomics) guidelines.

3.4.3. Global Protein Comparative View

Global Protein Comparative View provides detailed comparisons among different samples regarding the identified protein. To Show/Hide Columns, click **III** on top-left of the protein table. From the dialog, users can add and remove sample coverage, area, and spectral count columns. In the protein table, the selected columns will be displayed and can be exported.

- **Coverage by Sample:** the coverage broken down by sample can be displayed if multiple samples were used. Only peptides that pass the filters are included in the percentage calculation.
- Area by Sample: total area of peptide features from unique supporting peptides in each sample are used for calculation. Peptides must pass the filters.
- **#Spec by Sample:** total number of spectra identified that support the given protein. Totals are given for each sample included in the search.

3.4.4. Protein Coverage

The Protein Coverage view visualizes the mapping of supporting peptides to the protein selected in the Protein table. It also shows all identified modification or mutation sites to assist with protein characterization at the amino acid level. The coverage view shows different views under different modes. The view shows a sample comparative coverage view in the all sample mode while it shows protein sequence coverage view in the specific sample mode.

3.4.4.1. Sample Comparative Coverage

This area visualizes the frequency of supporting peptide sequences from selected samples in the protein sequence. At the top of the protein sequence coverage view, the header information of the protein is shown. At the topright corner, the COVERAGE LEGEND button will show a continuous color legend for the peptide frequency. Specific modifications can be selected in the PTM FILTER button and the template protein can be copied using the TOOLS button.

M PEAKS Online	e 🏦 🖬	5 -5 ±								PROJECT 🗸 SET	TINGS 🗸 🛛 ADI	MIN 🗸 ACCOU	INT 🗸 HELP 🗸	<i>,</i>
DDA Walkthrough +	<					She	w Results for		Overalla 1					
Analysis 1: De Novo + Database Search	/ T S		Showing Top Proteins			۲	All Samples	 Specified Sample 	Sample I		Export Prot	<u>Protein-Pep</u> - 1 - 100 of	tide PIM-Profil 1003 - →	ing
A De Novo	nmar		Access	ion	-10LgP L Co	verage	#Peptides	#Unique	PTM	Avg. Mass				Descr
DB Search		1	P21333 FLNA_HUMAN		691.20	62.90%		156 149	CDO	280739	Filamin-A OS=	Homo sapiens O	(=9606 GN=FLNA	PE 🔺
Export	fote	2	015149IPLEC HUMAN		673.33	35.67%		141 117		220532	Plectin OS=Ho	nomo sapiens OX=9	606 GN=PI EC PE	PE Fal
	s	4	075369 FLNB_HUMAN		664.04	60.99%		122 112	CDO	278164	Filamin-B OS+	Homo sapiens OX	(=9606 GN=FLNB	PE
	Pe	5	P78527 PRKDC_HUMAN		619.72	25.75%		89 89	0	469089	DNA-depender	nt protein kinase o	catalytic subunit O)S=
	ptid	6	Q00610 CLH1_HUMAN		615.96	54.69%		84 84	CDO	191615	Clathrin heavy	chain 1 OS+Hom	o sapiens OX=960	J6 (
	es	7	Q9Y490 TLN1_HUMAN		614.80	51.16%		84 76	C D O	269767	Talin-1 OS=Ho	mo sapiens OX=9	606 GN=TLN1 PE	=1
	Per	9	043707IACTN4 HUMAN		608.98	71.13%		83 54		104854	Alpha-actinin-	4 OS=Homo sapie	ns OX=9606 GN=4	AC. A
	lovo													F
	Only	Coverage	Peptide PTM Profi	ling 🛄										-
		P21333 FLN	IA_HUMAN Filamin-	A OS=Homo sapiens	0X=9606 GN=FL	NA PE=1 S	V=4				COVER	AGE LEGEND PT	IM FILTER TOOLS	s
					3									
		1 Sample 1	MSSSHSRAG	2 SAAGAAPGGG	VDTR DAEMP	A TEKD	LAEDAP	WKKIQQNTFT	RWCNEHLKCV	SKR IANLQT I	D LSDGL	RLIAL		
🔚 Data Search 🔍	×	81	LEVLSQKK	H RKHNQRPTFR	QMQLENVSV	A LEFLI	DRESIK	LVSIDSKAIV	DGNLKLILGL	IWTLILHYS	I SMPMW	DEEED		
Sample 1	Ø 🔳	Sample 1						205 210						
		161	REAKKOTPK	RILGWIONKI	POLPTTNES		RALGA	LVDSCAPGLC	PDWDSWDASK	PUTNAREAM		LGIPO		
		Sample 1	DD111(1(X 1 1 1))	2 Mazoni gini		0120		21200001020	LD ND ONDINOIN	- · · · · · · · · · · · · · · · · · · ·	x x11000	DOTTX		
		ourripro i							297					
		241 Samula 1	VITPEEIVDI	P NVDEHSVMTY	LSQFPKAKLI	K PG AP	LRPKLN	PKKAR AYGPG	IEPTGNMVKK	RAEFTVETR	AGQGE	/LVYV		
		321	EDPAGHQEE!	KVTANNDKNR	TFSVWYVPE	V TGTH	KVTVLF	AGQHIAKSPF	EVYVDKSQGD	ASKVTAQGP	G LEPSGI	NIANK		
		Sample 1												
					_			0				C		
		401	TTYFEIFTA(G AGTGEVEVVI	QDPMGQK GT	V EPQL	EARGDS	TYR CSYQPTM	EGVHTVHVTF	AGVPIPRSP	TVTVG	DACNP		
		Sample 1	483											
			0											
# . Pi	eptide		Unique	-10LgP	Mass Len	igth p	om ,	m/z	RT .	Scan	#Spec	Start	End	
1 VHSPSGALEEC(+57.02)YVTE	EIDQDKYAVR			124.54	2765.2966	24	3.1	692.3336	40.37 5182:e151	108_SS1040HCD	1	2368	2391	C 🔺
2 LRN(+0.98)GHVGISEVPK			<i>✓</i>	124.50	1423./936	13	2.1	4/5.6061	24.45 2782:e151	108_SS1040HCD	1	2002	2014	D
A EVPAEMGTHTVSVK	FSVQFFLR		~	124.49	1501 7500	1/	2.4	501 5051	21.48 2345:e151	108_331040HCD	2	2202	2215	- 11
5 EGPYSISVLYGDEEVPRSPFK			ý –	124.48	2368.1587	21	2.1	790.3951	61.11 8323:e151	108 SS1040HCD	1	1516	1536	- 11
6 AWGPGLEGGVVGK			~	124.48	1225.6455	13	2.6	613.8316	33.59 4157:e151	108_SS1040HCD	1	581	593	- 11
7 AGNNM(+15.99)LLVGVHGPR	t		~	124.47	1449.7510	14	2.8	484.2590	20.24 2138:e151	108_SS1040HCD	1	2585	2598	0
8 YWPQEAGEYAVHVLC(+57.02	2)NSEDIR		~	124.46	2535.1489	21	2.6	846.0591	69.53 9552:e151	108_SS1040HCD	1	635	655	C
9 NGHVGISFVPK			~	124.46	1153.6244	11	2.0	385.5495	22.01 2415:e151	108_SS1040HCD	2	2004	2014	_
10 APSVANVGSHC(+57.02)DLSI	LK		~	124.46	1653.8145	16	3.1	552.2805	21.84 2377:e151	108_SS1040HCD	1	2150	2165	C
11 M(+15.99)SC(+57.02)M(+15.	99)DNKDGSC(+	+57.02)SVEYIPY	E 🗸	124.46	4476.9282	40	3.8	1120.2437	62.25 8515:e151	108_SS1040HCD	1	1400	1439	C
12 EGPYSISVLYGDEEVPR			~	124.46	1908.9105	17	2.3	637.3123	57.12 7731:e151	108_SS1040HCD	2	1516	1532	
13 LSPFMADIRDAPQDFHPDR			~	124.46	2227.0481	19	2.4	557.7706	49.24 6556:e151	108_SS1040HCD	2	050	674	
14 VGSAADIPINISETDESEETATV	VPPSGR	0000	~	124.45	2892.5444	29	2.0	965.1907	89.40 12416:e15	1108_SS1040HCD	3	1965	1993	
15 DAGYGGLSLSIEGPSKVDINTE	ULEDGIC(+57.	.uzjR	~	124.44	310/.4303	30	1.8	1056.8280	03.54 8/10:e151	108_SS1040HCD	1	2074	2103	
17 VKASGPGI NTTGVPASI PUEC	TIDAK			124.44	2900.3720	28	3.2	/20.1320 857.1360	52.08 /U/0:e151 6/133 8816:o151	108_SS1040HCD	1	2334	2307	
18 OM(+15.00)OLENI/OVALEELE	PESIK		~	124.44	2364 1005	20	2.0	790.0755	75.06 10201-015	1108_SS1040HCD	4	101	120	
10 FLPREGRVEVEVEVTVDGVDVD	GSPEPI FAVAR	TKPSK		124.40	2004.1993	37	2.2	1000 5182	72.53 10026-615	1108_SS1040HCD	2	1022	1069	
20 I VSN(+0.98)HSI HETSSVEVD	SITK		~	124.42	2200 1011	20	2.3	734 3760	39.75 5117-4151	108 SS1040HCD	1	2521	2540	D
21 ENECHIPDSPEVVPVASPSGD	AR		~	124.41	2466.1814	23	1.9	823.0693	57.72 7803 e151	108 SS1040HCD	1	2311	2333	-
22 LLGWIQNKLPQLPITNFSR			~	124.41	2237.2683	19	3.0	746.7657	78.80 10955:e15	1108_SS1040HCD	1	172	190	-
4														

Regions in the protein sequence that are covered by supporting peptides are displayed in bold font with a grey background. Placing the cursor over a amino acid on the sequence shows the position of the amino acid in the sequence and the frequencies from different selected samples.

The frequency of supporting peptide sequences from different samples can be shown as colored bars under the protein sequence. The deeper the color, the higher frequency of that supporting peptide in the sample.

3.4.4.2. Protein Sequence Coverage

This area visualizes the coverage of the protein sequence in the specified sample mode. The header information of the protein, a PTM filter and a tool to copy the protein information are shown on the top of the protein sequence coverage view. Clicking the PTM filter button will pop up a menu to list the modifications identified in supporting peptides on the protein and the checkbox on the left controls whether to show the modification in the protein sequence coverage view above.

E DDA Walkthrough +	<						Show Results for										
🔵 🛃 Analysis 1: De Novo + Database 🖉 🦯 🍵	8	Showi	ng Top Proteins				 All Samples 	Specified Sample	e Sample 1	x ~	Export	Protein	Protein-Peptide	PTM-Pro	filing		
Search	з.													<i>~</i>	1 - 100 of 100	3 × →	
De Novo	3		Accession	n .	-10LgP [Coverage Sample 1	Area Sample 1	#Peptides	#Unique	#Spec Sample 1		PTM	Avg. Ma	19			
DB Search	70	2	P21333/PLNA_HUMAN		677.12	62.907	4.209	7 100	14	7 21				280739 H	Iamin-A USeHomo	sapiens UX=9	
Export	6	2	P33379/MTH9_H0MAN		670.02	37.037	2.420	/ 141	10	21				120332 IN	yosin-9 OS#Homo	sapiens oxea	
	5	3	Q15149[PLEC_HUMAN		0/3.33	33.0/1	7.936	0 135	12	5 IO				331/91 P1	ectin US=Momo sa	piens OX=900	
	-	4	073309/PENB_HUMAN		604.04	00.991	1.316	/ 124		2 13				2/0104 PI	Iamin-B US=Momo	sapiens UX=9	00
	- de	3	P78327 PRKDC_HOMAN		019.72	23.737	4.038	0 01		9 10				109069 D	NA-dependent prot	ein Kinase cata	ery:
	tid.	0	QUUGTUICEHT_HUMAN		015.90	54.097	8./39	0 84	8	4 11				191015 CI	athrin neavy chain	1 US=Homo s	api
	- 8 -	7	Q9Y490[TEN1_HUMAN		614.80	51.161	7.49e	6 84	7	b 12				269767 Ta	alin-1 OS+Homo sa	ciens OX=960	b G
	8	8	P14618[KPYM_HUMAN		610.22	/6.461	5.88e	/ 86	8	2 16				5/93/ P	ruvate kinase PKN	US+Homo sa	P14 🗸
	N	4	043707/ACTN4 HUMAN		60X 9X	/1.139	1 444-	/ H2	с з	4 18				102852 A	inha-actinin-4 OSaP	Iomo sabiens	DX N
		8	1 MSSSHSRAGQ 1 LEVLSQKKMH	SAAGAAPGG RKHNQRPTF	R QMQLEN	NSVA LEFLD	RESIK LVSI	ONTET RWG	CNEHLKCV	SKR ianlqtd Iwtlilhysi	LSDGL SMPMW	RLIAL					
Sample 1		16	1 EEAKKQTPKQ	RLLGWIQNK	L PQLPIT	NFSR DWQSG	RALGA LVDS	APGLC PDV	DSWDASK	PVTNAR EAMQ	QADDW	LGIPQ					
		24	1 VITPEEIVDP	NVDEHSVMT	Y LSQFPH	AKLK PG APL	RPKLN PKKA	AYGPG IEI	TGNMVKK	RAEFTVETRS	AGQGE	VLVYV					
		32	1 EDPAGHQEEA	K VTANNDKN	R TFSVWY	VPEV TGTHK	VTVLF AGQH	AKSPF EV	VDKSQGD	ASKVTAQGPG	LEPSG	NIANK					
		40	1 TTYFEIFTAG	AGTGEVEVV	I QDPMGÇ	KGTV EPQLE	AR GDS TYRC	YQPTM EGV	/HTVHVTF	AGVPIPRSPY	TVTVG	QACNP					

Regions in the protein sequence that are covered by supporting peptides are displayed in bold font with a grey background. Confident modifications and mutations identified in supporting peptides are displayed as icons above the protein sequence. Modifications are represented by colored icons with the initial letter of its modification name. If a residue is modified by more than one modification in the same supporting peptide, "*" is used instead of a letter. Mutations are represented by black-framed icons with a letter indicating the mutated residue. Placing the cursor over a modification icon shows the full name of the modification/mutation. The number above a modification icon indicates the position in the protein sequence. Placing the cursor over an amino acid on the sequence shows the position of the amino acid in the sequence.

The de novo only peptides are the confident de novo sequence tags that remain unidentified by the identification search algorithms. To report a de novo only peptide, the De novo score must be equal to or better than the specified threshold. Meanwhile, the score of the spectrum's best identification search result should be no greater than the specified -10lgP threshold. If we enable the de novo only tag in the parameter setting, we can see the de novo only sequence with grey line below.

The supporting peptides can be shown as colored bars under the protein sequence. Placing the cursor over a bar shows detailed information of the peptide. Left clicking on a bar will pop up the annotated spectrum associated with the supporting peptide.

3.4.5. Supporting Peptides

The Peptide table shows a list of supporting peptides for the selected protein. This table is similar to the peptide table in the Peptide View, except that "Accession" is excluded and the columns, "Unique", "Start", "End", and "Found by" are included.

• Unique: This shows whether the peptide is a unique supporting peptide to the protein group.

- Start: This shows the peptide's starting position in the protein.
- End: This shows the peptide's ending position (inclusive) in the protein.
- **Found by** : The name of the PEAKS search workflow that identifies the peptide. This can be either PEAKS, PEAKS PTM, or SPIDER.

Refer to Section 3.5.2, "Peptide Table" for the descriptions of other columns.

Double click a supporting peptide, it will go to that peptide sequence in the peptide page.



3.4.6. PTM Profiling

This tool calculates the difference in abundance between peptides with confidently identified PTMs versus unmodified peptides. Peptide feature areas are used for this comparison. For the Identification results (DB/PTM/ SPIDER), the PTM Profiling tool provides a direct visualization of quantitative information for comparing modified peptides with unmodified peptides for the all modification sites in a protein across all MS samples.



3.4.6.1. PTM Profiling View Filters

Peptide Filter: The PTM peptide filter controls what peptides contribute to modified and unmodified peptide areas. Users can select either All or Fully Digested Peptides, where fully digested peptides contain no missed cleavages.

PTM-Profiling	All	PTM FILTER
÷	Fully Digested	<i>→</i>

PTM Filter: The drop-down list contains the detected and selected confident PTMs for the protein. Select the type of modification to study a specific PTM or choose "All the PTMs" to visualize the profiling information of all PTM sites at together.

Note

Fixed PTMs are not used for PTM Profiling.



3.4.6.2. PTM Profile Table

The PTM Profile table displays a site-specific list of confident mutations on the selected protein. The table can be sorted by any column. The columns in this table are:

- **Protein Position:** Amino acid residue type and position # of the specified modification.
- **-10LgP:** Confidence score for the top PSM containing that modification that passes the confident site filters. Refer to Section 3.1, "Result Filtration" for detailed explanation.
- **Ion Intensity** (%): Ion intensity of the modification site-determining fragment ion pair, compared to the base peak ion intensity, for the top-scoring PSM that passes the confident site filters.
- AScore: The AScore achieved by the modified peptide with highest -10lgP score that passes the confident site filters.
- **Sample Area columns**: Cumulative area of modified and unmodified forms of the peptide on a per sample basis. Modified PSMs which do not pass the confident site filters will be considered as unmodified for area calculations.

This table is paginated showing 10 modification sites per page.

										PTM-Profiling P	eptide Filter: All	PTM FILTER
										\leftarrow	1-6of6 +	\rightarrow
	Protein Position ↓	PTM	-10lgP	Ion Intensity(%)	Ascore	Sample 1 Modified	Sample 1 Unmodified	Sample 2 Modified	Sample 2 Unmodified			
\sim	M398	Oxidation (M)	122.28	21.00	1001.00	5.43e+4	1.40e+5	2.69e+5	8.76e+5			
\sim	M377	Oxidation (M)	122.92	4.00	1001.00	3.70e+4	8.70e+4	2.70e+5	4.32e+5			
\sim	Q358	Deamidation (NQ)	122.94	4.00	1000.00	0.00e+0	9.47e+5	3.08e+4	4.50e+6			
~	M313	Oxidation (M)	123.12	11.00	1001.00	1.86e+5	1.40e+5	8.58e+5	8.89e+5			
~	M302	Oxidation (M)	123.30	1.00	1001.00	2.35e+5	1.73e+5	1.15e+6	8.76e+5			
~	M154	Oxidation (M)	122.99	2.00	1001.00	7.63e+4	1.24e+4	5.55e+5	2.31e+5			

3.4.6.3. PTM Profile Figures

Coverage Peptide PTM Profiling

PTM Profile Chart: Below the PTM Profile table, a dynamic bar graph provides a visualization of the ratio between the modified vs. unmodified forms of the peptides detected at each modification site. The x-axis indicates

the percentage (%) of modified or unmodified peptide feature areas relative to the total abundance whereas the yaxis indicates the position in the protein sequence where a modification was identified for each individual sample. By default, all modification sites listed in the PTM profiling table are displayed. Users may toggle which mutation sites are displayed by deselecting the check-box beside any modification site or using the PTM filter option.

Spectrum Annotation View:The Spectrum Annotation View provides a graphical representation for the best modified peptide and the best unmodified peptide when "All" is selected from the "Peptides" drop-down list. The top spectrum corresponds to the best modified peptide, whereas the bottom spectrum shows the best unmodified peptide. When "Fully digested" is considered, the best modified peptide and the best unmodified peptide are displayed in a compare view for quick visualization and validation.





3.4.6.4. PTM Profile Export

The PTM profile data can be exported to a text file in CSV format for specific proteins or for all identified proteins (for all samples or single sample).

Single Protein: To export PTM profile information for a single protein, click the PTM-Profiling export link within the PTM Profiling tool. This will export all information displayed within the PTM profile table, meaning it will be filtered by the PTM profiling PTM and Peptide filters.



All Proteins: To export PTM profile information for all identified proteins click the PTM-Profiling export link above the Protein table. The exported file contains all necessary information to construct the PTM profile graph. The PTM profile data are grouped together by protein, then by modification type, and then by the modification site on the protein sequence. The headers of the CSV column are similar to the PTM Profile Table.

This export will adhere to all protein view filters applied to the protein table and can be filtered by specified or all samples.

Show Results for							
 All Samples 	O Specified Sample	Sample 1	\sim	Export	Protein	Protein-Peptide	PTM-Profiling

3.5. Peptide View

The Peptide Table shows the filtered peptide identification results. Click on a peptide row to show the spectrum below the peptide table.

Analysis 2: De Novo + Database 🔗 🖍 📳 Search						Show Results fo	or .						
Search	<	Q				All Sample	es 💿 Specifi	ed Sample	Sample 1	X ~	Export Per	otide PSM	
Search	Sun											← 1 - 100 of 64	112 × ->
MA De Novo	nma		Peptide		-10LgP 📋 🚦	Mass	Length	ppm :	m/z	RT	Area Sample 1	Scan	#Spec P1
DB Search	-V	HIADLAGNS	EVILPVPAFNVINGGSHA	GNK	105.48	3010.562	5 30	1.2	753.6488	66.72	1.60e+8	17262:120315QEx2	. 8
Evport	Pro	. HIADLAGNS	3EVILPVPAFNVIN(+0.98)	GGSHAGNK	99.42	3011.546	4 30	2.2	753.8955	67.37	0.00e+0	17495:120315QEx2_	. 1 D
Caport	teir	NMGGPYGG	JUNYUPUUSUUSUUSUU	4	98.33	2188.897	9 20	1.3	1095.4578	54.17	0.3/0+0	12001:120315QEX2	. 3
😋 🗹 Analysis 1: DB Walkthrough 🛛 🔗 🧪 📋		HTGPGILSN	ANAGENTN(+0.08)GSO	FFIC(+57.02)TAK	97.20	2794.332	7 27	1.4	031 // 38	65.87	2.270+0	17070-1203150Ev2	3 0 0
	Pep	i IAIPGLAGAD	GNSVI I VSNI NPER	110(107.02)1/11	96.30	2274.269	5 23	0.9	1138.1431	72.68	4.44e+6	19498:1203150Ex2	. 4
	tid	/ LC(+57.02))	VALDFEOEMATAASSSS	LEK	95.86	2549.166	5 23	1.4	1275.5923	73.31	1.43e+7	19748:1203150Ex2	4 0
	8	3 TNHIGHTGY	/LNTVTVSPDGSLC(+57.0	2)ASGGK	95.65	2742.303	2 27	1.7	915,1099	55.67	5.68e+6	13012:1203150Ex2	3 0
	Pe	VIHDNFGIV	EGLM(+15.99)TTVHAITA	TQK	95.01	2610.347	4 24	0.8	871.1238	69.08	7.59e+7	18243:120315QEx2	. 7 0
	No												,
	00	FRACTION: 1203150	Ex2 RS1 20nl-min 10Hel a	1h 01 raw Y	RF: m/z: 753 6488 z: 4 R	: 66.72 Area : 2	2 81e+7 PPM : 1 24		M: scap: 17262 -10loP: 1	105 48 RT 66 75	m/z : 753 8992 ¥	Protein: P06733IEN	IOA HUMAN -
	Only												
		🔧 # 1:1	HIADLA	GNSEVIL	PVPAFN	V I N G	GSHA	GNK					
		Intensity (%)											
		100 -							У	15			
		90-											
🚍 Data Search O 🗙 🖍		80 -											
		70 -											
😋 Sample 1 🧔 🖉 🚺		00-											
		50-					H10						
		20-				v10	611						
		20-		ь	s í	60							
		10 -	b2	ыл. II	67 Y8 68	Ĩ	v11	b12	y13		y17		
		0		لمحجب اللغف فالسبوتة	لطبيها ومصافدته	بعلية سيطيله	_ ا ه ۱۲۰ ایل	- k k	613 YI4		yis yis		
		0.0 Error (da)	200.0	400.0 600.0	800.0	1	1000.0	1200.0	1400.0	1600.1	3 1800.	0	
		0.050						• •	1				
		0.050			- 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 199					•			
		°	200	400 600	800		1,000	1,200	1,400	1,600	1,800	°	
		# b	b-H20	b-NH3	b(2+)	Seq	У	y-1	H20 y	-NH3	y(2+)	#	
	-	1	138.07 12	20.06 121.0	4 69.5	3 H						30	
		2	251.15 23	33.14 234.1	2 126.0	8 I	2874	.51	2856.50	2857.48	1437.76	29	
		3	322.19 30	04.18 305.1	6 161.5	9 A	2761	.43	2743.42	2744.40	1381.21	28	
		4	437.21 41	19.20 420.1	9 219.1	D	2690	.39	2672.38	2673.36	1345.69	27	
		-					0.575	36	2557.35	2558.34	1288.18	26	
		5	550.30 53	32.29 533.2	7 275.6	5 L	20/0					20	
		6	550.30 50 621.34 60	32.29 533.2 03.32 604.3	7 275.6 1 311.1	5 L 7 A	2575 2462	.28	2444.27	2445.25	1231.64	4 25	
		6	550.30 5: 621.34 60 678.36 6f	32.29 533.2 03.32 604.3 50.35 661.3	7 275.6 1 311.1 3 339.6	5 L 7 A 8 G	23/3 2462 2391	.28 24	2444.27 2373.23	2445.25 2374 21	1231.64	20 4 25 2 24	
		6	550.30 5: 621.34 60 678.36 66 078.30 00	32.29 533.2 03.32 604.3 50.35 661.3 00.35 001.3	7 275.6 1 311.1 3 330.6 3 339.0	5 L 7 A 8 G	2070 2462 2391 2391	.28 74	2444.27 2373 23 2373.23	2445.25 2374 21 2374.21	1231.64 1196.12	2 20 4 25 2 24	
		5 6 7	550.30 5: 621.34 6L 678.36 6F 0/8.30 00 792.40 77	32.29 533.2 03.32 604.3 50.35 661.3 00.35 001.3 74.39 775.3	7 275.6 1 311.1 3 339.6 3 339.6 3 339.6 3 339.6	5 L 7 A 8 G 9 U	2070 2462 2391 2391 2334	.28 74 .24	2444.27 2373 23 2373.23 2316.21	2445.25 2374 21 2374.21 2317.19	1231.64 1196.12 1190.12 1190.12	20 4 25 2 24 3 23	
		5 6 7 8 9	550.30 5: 621.34 60 678.36 64 075.30 00 792.40 77: 879.43 86	32.29 533.2 03.32 604.3 50.35 661.3 00.35 001.3 74.39 775.3 51.42 862.4	7 275.6 1 311.1 3 339.6 3 339.6 8 396.7 2 440.2	5 L 7 A 8 G 0 N 2 S	2373 2462 2391 2394 2334 2220	.28 74 .24 .22	2444.27 2373.23 2373.23 2316.21 2202.17	2445.25 2374 21 2374.21 2317.19 2203.15	1231.64 1196.12 1190.14 1167.50 1110.50	2 23 4 25 7 24 6 24 8 23 5 22	
		5 6 7 8 9 10	550.30 5: 621.34 6i 678.36 6f 792.40 77: 879.43 86 008.47 99	32.29 533.2 33.32 604.3 50.35 661.3 50.35 001.3 74.39 775.3 51.42 862.4 862.4 991.4	7 275.6 1 311.1 3 339.6 3 339.6 8 396.7 2 440.2 5 504.7	5 L 7 A 8 G 0 N 2 S 4 E	2375 2462 2391 2334 2230 2133	.28 74 .22 .18	2444.27 2373 23 2373.23 2316.21 2202.17 2115.13	2445.25 2374.21 2374.21 2317.19 2203.15 2116.12	1231.64 1196.12 1190.12 1167.58 1110.50 1067.07	2 25 4 25 7 24 6 24 8 23 6 22 7 21	
		5 6 7 8 9 10 1 11 1	550.30 5: 621.34 6: 678.36 6: 792.40 77: 879.43 8: 008.47 99: 107.54 107.54	32.29 533.2 33.32 604.3 50.35 664.3 50.35 001.3 74.39 775.3 61.42 862.4 90.46 991.4 90.52 1090.5	7 275.6 1 311.1 3 339.6 3 399.0 8 396.7 2 440.2 5 504.7 2 554.7	5 L 7 A 8 G 0 N 2 S 4 E	2375 2462 2391 2334 2230 2133 2004	.28 74 .22 .18 .15	2444.27 2373 23 2373 23 2316.21 2202.17 2115.13 1986.09	2445.25 2374.21 2374.21 2317.19 2203.15 2116.12 1987.08	1231.64 1196.12 1190.14 1167.56 1110.56 1067.07 1002.55	2 25 4 25 7 24 6 24 8 23 5 22 7 21 5 20	

3.5.1. Peptide View Filters

Click the **Q** button, the following Peptide Filter will pop up. It controls which peptides to display.

Peptide Filter			
Peptide sequence contains			
Scan =			
Precursor Id =			
m/z =			
RT =			
Peptide sample area ≿			
Include Contaminant Data			
Peptide Contains Modifications			
Name to search for			CLEAR
Deamidation (NQ) Oxidation (M)	Carbamidometh	ylation	
	RESET	CANCE	APPLY

- **Peptide sequence contains:** Enter an amino acid sequence, only peptides that contain that sequence will be displayed in the peptide table.
- Scan: Filter for a single scan number based on the scan numbers read from the raw data file.
- **Precursor Id:** Filter is only relevant for TimsTOF data, for which it will filter based on the precursor id of the peptide.
- **m/z:** Filter based on the expected m/z, the last significant digit is rounded.
- **RT:** Filter based on the expected RT, the last significant digit is rounded.
- **Peptide sample area:** Filters based on total peptide area found in each sample. If a peptide is found in multiple samples, there should be at least one sample pass this filter for the peptide to be included.
- **Spider Mutations:** Filter only available for SPIDER result. When checked, peptides that have mutation identified with SPIDER are shown.
- **Peptide Contains Modifications:** Check boxes are provided for each identified modification. Peptides that contain the selected PTMs will be shown.

3.5.2. Peptide Table

The Peptide Table shows the filtered peptide identification results. Each row in the table is a peptide identification represented by its highest-scoring PSM. The table is sorted by the -10lgP value. In the Protein list drop-down window of each peptide, after clicking any of the protein, the protein page will be opened.

The columns Peptides, -10LgP, m/z, RT, Accession are sortable, i.e. Clicking any of these header can sort the whole table based on that column.

The columns in the Peptide table are:

- **Peptide:** The amino acid sequence of the peptide, as determined in PEAKS Search. A modified residue is followed by a pair of parentheses enclosing the modification mass.
- **-10lgP:** The peptide -10lgP score. The score indicates the scoring significance of a peptide-spectrum match.
- Mass: The monoisotopic mass of the peptide.
- Length: The number of amino acids in the peptide sequence.
- **ppm:** The precursor mass error, calculated as $10^{6} \times (\text{precursor mass} \text{peptide mass}) / \text{peptide mass}$.
- m/z: The precursor mass-to-charge ratio.
- **RT:** The retention time (elution time) of the spectrum as recorded in the data.
- Scan: The scan number of the spectrum that matches the peptide sequence with the highest -10lgP.
- **#Spec:** The number of MS2 spectra assigned to the peptide.
- **PTM:** The types and the numbers of modifications present in the peptide shown in color-coded icons.
- AScore: Localization score assigned to modifications on the peptide. It is the -10 log of a p-value. In this case, the p-value is the probability that the modification occurs at the reported position compared to other possible positions. A -10lgP of 20 is equal to a p-value of 0.01. Scores are listed in the format AminoAcid, peptide position, Ascore.
- Accession: The accession number of the highest-scoring protein containing this peptide.
- **Precursor Id:** This column is only available for TimsTof data, it shows precursor's id from the raw data.
- 1/k0: This column is only available for TimsTof data, it shows the ion mobility range of for the precursor.

Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER)

3.5.3. Peptide Global Comparative View

"Peptide Global Comparative View" provides detailed comparisons among different samples regarding the identified peptides. To Show/Hide Columns, click III on top-left of the peptide table. From the pop-up window, users can add and remove area and spectral count columns. In the peptide table, the selected columns will be displayed and can be exported. If "Specified Sample" is selected these columns are automatically displayed.

Selec	t Display Columns and Sample	S
Select	Columns	
🗸 Are	a 🗸 #Spec	
Select	Samples	01540
	SAMPLE NAME	CLEAR
	S1	
	S2	
\checkmark	S3	
\checkmark	W1	
	W2	
\checkmark	W3	
	CANCE	L SAVE

Note

PEAKS Online provides users the option to calculate peptide abundance using the MS1 feature area calculation or using the MS1 feature intensity. By default, area is used. The default can be changed by the administrator.

- Area by sample: The total area of all peptide features matched to spectra that identified the peptide. Area calculations are separated into multiple columns if multiple samples were included in the run.
- **#Peptide-spectrum matches by sample:** The number of peptide spectrum matches associated with a peptide separated by sample.

3.5.4. Peptide Menu

The peptide menu is only shown when "Show Results for Specified Sample" is selected in the filter section. The peptide menu displays the information of the peptide sequence that is being displayed in the spectrum. If a feature is associated to more than one tandem scan, click the "FRACTION", "FEATURE" or "PSM" button to view the Spectrum Annotation of the other PSMs.

Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER)



3.5.5. Peptide-Spectrum Matches

This section displays the spectrum annotation and other information about the peptide selected in the Peptide table. When multiple spectra are matched to the peptide, only the information about the top-scoring PSM is shown by default. The interface of this section is identical to the Spectrum Annotation in the PEAKS De Novo result. For detailed instructions, refer to Section 3.5, "Spectrum Annotation" in Chapter 5, Peptide De Novo Sequencing.

3.6. De Novo Only View

The De Novo Only view displays the list of quality peptide sequences detected by *de novo* sequencing that remain unidentified by the database search. A *de novo* sequence is "de novo only" if it is from an MS/MS spectrum that is not confidently assigned to any database peptide according to the current filters.

The table shows the filtered "de novo only" sequences by the current filters. "De novo only" sequences may suggest novel peptides, peptides with unknown modifications, or other interesting research subjects. The interface is identical to the "De Novo" view in a De Novo sequencing result page. For detailed instructions, refer to Section 3, "Understanding PEAKS *De Novo* Sequencing Results" in Chapter 5, Peptide De Novo Sequencing. In the setting parameter, the option "De novo only tags" is checked to show the locations of MS/MS spectra with de novo only sequences.

4. Running PEAKS PTM and SPIDER Separately

PEAKS PTM and SPIDER can be launched by going back to the WORKFLOW tab. Click the "MODIFY WORK-FLOW" button and click Next to set parameters of PEAKS DB, PEAKS PTM, or SPIDER. They can also be launched by selecting the PARAMETERS tab and click "MODIFY PARAMETERS". Checking "Find unspecified PTMs with PEAKS PTM" and/or "Find more mutations with SPIDER" will launch PEAKS PTM and/or SPIDER.

The parameters used in a PEAKS PTM search are similar to a PEAKS DB search. Selecting a protein database is not necessary as PEAKS PTM automatically uses the database defined in PEAKS DB. For PTM, the search can be done either with all built-in modifications or with selected list of modifications.

5. FAIMS Data

Field asymmetric waveform ion mobility spectrometry (**FAIMS**) analyzes complex mixtures and separates the ions according to their mobilities. PEAKS Online has the ability to load FAIMS data and analyze it using *de novo* sequencing, database search (PEAKS DB), unexpected modification (PEAKS PTM) and sequence variants (SPIDER) search, and labelled and label free quantification.

Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER)

The analysis result for FAIMS dataset contains the additional information for CV values compared to the non-FAIMS dataset. The CV-related information can be found at the following places for a DB/PEM/Spider analysis.

A Venn diagram on the summary page of FAIMS projects.



• Venn diagram overlaps between different CVs in the search for all proteins, top proteins, and peptides are shown on the summary page.

With the introduction of FAIMS data, a CV column will appear on all peptide tables, as well as select exports:

• Additional CV column on the supporting peptide page

The supporting peptide table on the protein page contains one additional column for CV values

• Additional CV column on the exported *Protein-Peptide CSV*

The exported Protein-Peptide CSV contains one additional column for CV values

• Additional CV column on peptide page

table on the peptide page contains one additional column for CV values

• Additional CV filter on peptide page

The filter for peptide page contains one additional option for CV, which selects the target scan based on the CV value

• Additional CV column on the exported *Peptide CSV*

The exported *Peptide CSV* contains one additional column for CV values

Additional CV column on De Novo Only page

The table on the De Novo Only page contains one additional column for CV values

• Additional CV filter on De Novo Only page

The filter for De Novo Only page contains one additional option for CV, which selects the target scan based on the CV value

• Additional CV column on the exported *De Novo Only CSV*

The exported De Novo Only CSV contains one additional column for CV values

• Additional CV column on the exported De Novo Only with all candidates CSV

The exported De Novo Only with all candidates CSV contains one additional column for CV values

Chapter 7. DIA Streamlined Identification

1. Overview

PEAKS 11 offers a powerful and flexible solution for DIA identification using spectral library search with or without a protein inference database, direct database search, or a combination of both for optimal peptide coverage. In addition, de novo sequencing can be applied to ensure no good spectra remain unidentified.

2. Adding a Spectral Library from the Parameters Page

PEAKS Library Search offers an accurate and sensitive identification method for complex spectra, such as DIA spectra. In addition to annotated spectra along with precursor m/z and charge, PEAKS Library search also uses predicted retention time and ion- mobility collisional cross section (CCS) data for improved sensitivity.

To add a new Spectral Library from the Parameters page, open the "Select Library" drop down and select "**Create New Library**" or "**Import New Library**". See Chapter 2, section 2, *Library Management and Configuration Section 2, "Library Management and Configuration"* for details about creating and importing spectral libraries.

3. Spectral Library Search Workflow and Parameter Settings

PEAKS Spectral Library Search can be launched as a workflow in a PEAKS Online search for DIA data (selected from the DIA section of the workflow selection during analysis creation). An analysis report is generated to show the search result.

To conduct a spectral library search workflow, follow the steps below:

- 1. Refer to Chapter 5, Peptide De Novo Sequencing, Section 2, "*DeepNovo/De Novo* Analysis Workflow and Parameter Settings", Step 1-6 to create a new project and add data.
- 2. Select the "DIA" tab at the top of the Workflow step.
- 3. Select the "Identification" workflow.

Select	an Analysis
DDA	DIA
	Identification
•	Data Spectral Library Search Database Search De Novo QC
o 14	DeepNovo Peptidome
	Data Deep Novo Peptide Database Search
	Quantification
	Data Spectral Library Search Database Search Quantification QC

4. Set the Spectral Library Search Parameters. See details in Section 3.1-3.4 below.

Mass Error Tolerance
Precursor Mass 10 Tolerance Unit: PPM Fragment Mass Error 0.02 Tolerance Unit: DA Optimize Tolerance Error Tolerance: 10 0.02 Tolerance: 0.02 Tolerance Unit: DA Optimize Tolerance Image: Comparison of the compar
✓ Library Search Parameters ^
Select Library: PXD013231_DDA_library #Entry: 15581 Peptide Length: 6 to 30 Precursor m/z: 200 to 1500 Charge: 1 to 5
Protein Inference
Target Database: Benchmarking_Uniprot_Hu Taxonomy: all species × Contaminant Contaminants_from_MQ V 20387 fasta sequences 20387 fasta sequences 20100 fasta sequences Contaminants_from_MQ V
Database Search Parameters
De Novo Parameters Use Database Search Parameters
Report Filter
PSM/Peptide
○ PSM -10LgP ≥
Protein
Proteins -10LgP ≥ 20 Protein Group FDR (%) Proteins Unique Peptides ≥ 1
De Novo Denovo ALC (%)≥ 50 :

5. Click SUBMIT button to start the search.

3.1. Precursor and Fragment Error Tolerance

Precursor mass: This enables the selection of a precursor mass error tolerance. PEAKS Online DB Search identifies peptides within the precursor mass tolerance, which is the allowable m/z shift between the theoretical value of the peptide versus the observed value of the precursor ion. Precursor mass tolerance can be specified in either Da or ppm.

Fragment ion: This enables the selection of a fragment ion m/z tolerance, which is the allowable m/z shift between the theoretical value of the fragment ions of the peptide versus the observed value. PEAKS Online DB Search uses this value when scoring peptide-spectrum matches. PEAKS Online considers that a fragment ion is matched if the calculated m/z is within the tolerance. Fragment mass tolerance can be specified in Da.

Error tolerance parameters need to be set consistently with the mass accuracy of the instrument.

Note

When the samples in a project are generated by different instruments and "default" is chosen from the Predefined parameters drop-down menu, the default parameters of the first sample are loaded. However, it is recommended to set the tolerance parameters according to the least accurate instrument.

3.1.1. Optimize Tolerance

When enabled, PEAKS Online will determine the optimal values to use for Precursor and Fragment Mass Error Tolerance for each fraction in your analysis. These values will replace the values provided in the previous Precursor and Fragment Mass Tolerance sections.

3.1.2. Optimize for Short Gradient

This selection will tune parameters specifically for short gradient runs.

3.2. Spectral Library

Library Selection.: Select a PEAKS-generated library for the search. Select one from the list of libraries that have been configured in PEAKS Online and filter for the peptide length, precursor m/z, and charge when necessary. To configure a new library, refer to Chapter 2 Section 2: Library Management and Configuration.

Library Filters.: The spectra that will be used from the library can be filtered based on peptide length, precursor m/z, and charge. This is optional. Each filter can be enabled by checking the checkbox to the left of the filter inputs.

3.3. Protein Inference Settings

Protein inference can optionally be enabled to search peptides found in the selected library against a provided database.

Target Database.: Select a reference database from the list of configured databases or create a new one, and set the taxonomy if applicable. This database is the database used for protein inference.

Contaminant Database.: A contaminant database can optionally be provided as well. #contam# will be added to the beginning of the accession of all identifications from the contaminant database.

3.4. Report Filter

The report filter section is a mandatory section to filter out some results under the user-defined filter. Any changes to the filter will create a new analysis with old parameters and update the results and statistics.

- **PSM -10lgP:** The PEAKS peptide score (-10lgP) is calculated for every peptide-spectrum match (PSM) reported by PEAKS Spectral Library Search. The score is derived from the p-value that indicates the likelihood that the identification was made by random chance alone.
- **PSM FDR (%):** The PSM filter can also be set to use the false discovery rate. FDR is estimated using a "decoy-fusion" approach. A decoy peptide/spectrum for each target peptide/spectrum in a spectral library and the target and decoy spectra are searched together. The decoy method is based on: https://doi.org/10.1021/ pr900947u
- **Peptide FDR (%):** The filter can also be set to use the false discovery rate of the peptides instead of the PSMs. Redundant peptide hits found in multiple spectra are counted once.

4. Understanding PEAKS Spectral Library Results

After a PEAKS Spectral Library Search is complete, the result node will appear within the Analysis tree in the Analyses section of the left menu.

The analysis results for database search tools are presented in three pages:

• **Summary:** This shows an outline of the PEAKS Online spectral library search results with key statistics. The overall quality of the experiment can be examined and the filters for protein identifications can be adjusted. This page shows after the search is done.

• **Protein:** This shows a list of protein identifications. This view also visualizes protein sequence coverage and helps with protein characterization.

Note

This tab only appears if Protein Inference was enabled for the PEAKS Spectral Library Search.

• **Peptide:** This shows a list of peptide identifications. This view also provides spectrum annotation and other detailed information for peptide precursor spectrum matches.

4.1. Result Filtration

At the top of the summary result page, filters can be set. Identifications that fall below the filters will not be visible in the result pages or exports. After changing any of these filter parameters, click the *wey* button to apply and save the new filters. The results and statistical information will be updated accordingly.

Filters	20	Proteins FDR (%)	0	:	Proteins Unique Peptides ≥ 1	:	APPLY Cre	eated:	May 22 2020, 09:24:58 am

- **Proteins -10lgP:** The PEAKS protein score (-10lgP) is calculated as the weighted sum of the -10lgP scores of the protein's supporting peptides. After removing any redundant peptides, the supporting peptides are sorted by -10lgP scores in descending order, and the k-th ranked peptide contributes to the weighted sum with a weight of 1/k. A default threshold of 20 is set.
- **Proteins FDR (%):** Optionally, the proteins false discovery rate can be selected to filter proteins instead of a -10lgP score.
- **Proteins Unique Peptides:** The minimum number of unique supporting peptides for a protein identification. A unique peptide is defined as a peptide that can be mapped to only one protein group. Proteins are grouped using parsimony.

4.1.1. Show Results for All Samples

For the protein and the peptide tables, "All Samples" can be selected to view the combined search results from all the samples. In the Protein page, selecting "All Samples" will change the coverage panel displayed when a specific protein is highlighted. Supporting peptides from all samples instead of just a single sample will be shown.



4.1.2. Show Results for Specific Sample

In the Protein, Peptide and De Novo page results, the filter can also be set to "Specified Sample" which will only show the results a single sample. By clicking in the text you can search for a specific sample by name.

Specified Sample	Sample 1	$\times \mid$ \vee	
•			

Selecting "Specified Sample" will change the protein coverage view that is visible by clicking on a protein in the protein table. Peptides in the Protein, Peptide and De Novo Only pages will only be visible if they belong to the sample chosen. Once selected, the peptide tables will automatically add the "Area" column and the "# Spec" values will display the spectral count for the specified sample. For more information, refer to Section 3.5.2, "Peptide Table". Similarly, the protein table will also include the columns "Coverage" and "Area" for the specified sample. For more information refer to Section 3.4.2, "Protein Table".

4.2. Result Exporting

PEAKS Online spectral library search results can be exported in the Export page:

Export	SELECT ALL	CLEAR ALL	
All Search	Parameters	Spectral Librar	у
		Summary Tabl	e
		Protein CSV	
		Protein-Peptid	e CS
		Peptide CSV	
		PSM CSV	
		Protein Fasta	
DOWNLOAD	DOWNLOAD BY SA	MPLE	

- **Summary Table:** A comma separated file containing the information from the Summary Statistics table on the Summary page.
- **Protein CSV:** A comma separated file containing all identified proteins and associated details from the protein table for all samples.
- **Protein-Peptide CSV:** A comma separated file containing all identified peptides and associated details from the peptide table found in the Protein page. Further details are provided regarding the protein the peptide supports. If a peptide supports multiple proteins, multiple rows are included for each supported protein for all samples.
- **Peptide CSV:** A comma separated file containing all identified peptides and associated details from the peptide table for all samples.
- **PSM CSV:** A comma separated file containing all identified peptide spectrum matches and associated details from the peptide table for all samples.
- **Protein Fasta:** All identified proteins and associated details from the protein table for all samples, formatted as a fasta file.

4.3. Summary View

4.3.1. Summary Statistics

The first chart displays the summary information for each sample.

- **Sample Name:** This column lists the samples names with the exception of the first row, which shows the combination of all the samples.
- **# MS Runs:** The total MS runs in the sample.
- # MS1: The total MS scans in the sample summed across all fractions.
- **# MS2:** The total MS/MS scans in the sample summed across all fractions.
- **# PSM:** Total number of peptide spectrum matches (PSMs) found in the sample.
- **# Peptides:** Total number of peptides found. Peptides with the same primary sequence but different PTMs are counted separately.
- **# Sequences:** Total number of different peptide backbone sequences found. Peptides with the same primary sequence but different PTMs are counted as one sequence.
- **# Proteins:** Provides the number of proteins given the parameters set for Protein Score (-10lgP) and number of unique peptides.
- **# Protein Groups:** PEAKS Online groups proteins identified by a common set of peptides. This number shows the number of protein groups in the filtered result based on All proteins.

• **# PSM / # MS2:** The percentage of MS2 scans in the sample that produced peptide spectrum matches.

4.3.2. False discovery rate (FDR) curve

Figure 1 shows the FDR curve with respect to the number of PSMs being kept after filtration by the peptide -10lgP score. By lowering the score threshold, more PSMs are kept in the filtered result. Conversely, the FDR increases because more false positives are kept. In this figure, the vertical line indicates the current score threshold. The number of PSMs and the corresponding FDR by the current score threshold are shown in the top-left corner.

PEAKS Spectral library search estimates the FDR using decoy peptides and spectra. A decoy peptide/spectrum for each target peptide/spectrum in a spectral library and the target and decoy spectra are searched together. The decoy method is based on: https://doi.org/10.1021/pr900947u

It is commonly recommended to set FDR at <1% by adjusting the score threshold. If a rapid growth of the FDR around the 1% FDR threshold is noticed, some PSMs may be sacrificed to significantly reduce the FDR.

Note

When counting the number of PSMs, PEAKS Online keeps one peptide per spectrum at most. Thus, the number of PSMs is actually the number of spectra with assigned peptides.

Note

Decoy matches are not counted in the number of PSMs. Unless otherwise specified, decoy matches are also excluded from the other statistical numbers shown in the Summary view.

4.3.3. Statistics of filtered results

Table 1 shows the statistical numbers of the filtered results.

- **Top Proteins:** The number of proteins that are supported by the most unique peptides in the group. Proteins in the group that share a subset of the unique peptides that support the top protein (sub-proteins) will not be included here.
- **Proteins (#Unique Peptides):** In Table 2, this entry counts the number of proteins identified with unique peptides. A unique peptide is a peptide with a -10lgP score above the peptide filtering threshold that can be mapped to only one protein group. Peptides are counted based on their backbone sequence while considering I/L amino acids to be the same.
- **FDR (Peptide-Spectrum Matches):** The total number of decoy database assignments to spectra relative to the total number of target database assignments to spectra represented as a percentage
- **FDR (Peptide Sequences):** The total number of decoy database assignments to unique peptide sequences relative to the total number of target database assignments to unique peptide sequences represented as a percentage.
- **FDR** (**Protein Group**): The total number of decoy protein groups where the top hit in the protein group is a decoy database protein relative to the total number of target database protein groups where the top hit is from the target database.
- **-10lgP Cutoff:** The value is determined by the estimated PSM FDR. Only the peptides with PSMs above the -10lgP score threshold are listed in the Peptide view.

4.3.4. Precusor mass error

Figures 2(a) and 2(b) show the precursor mass errors of PSMs in filtered results. Figure 2(a) shows the distribution of the precursor mass error in a histogram with and without software calibration. Software calibration uses the observed distribution of mass error after identification to normalize the mass error around zero. Figure 2(b) shows precursor m/z versus precursor mass error in a scatter plot after software calibration. Figures 2(a) and 2(b) help examine whether or not the mass spectrometer is properly calibrated. For a well-calibrated instrument, precursor mass errors should center at 0 ppm across the range of m/z.

4.3.5. RT Calibration

Plots the RT regression function used during the spectral library search. Peptides are plotted against their indexed retention time (iRT) predicted using machine learning.

4.3.6. Box plot for Rt change

A box plot showing the difference between the observed retention time and iRT peptides within set retention time bins across all fractions, showing the accuracy of retention time prediction.

4.3.7. Number of identified peptides in each sample by number of missed cleavages

Presents the number of identified peptides in each sample with the number of missed cleavages, indicating the enzyme digestion efficiency.

4.3.8. PTM Profile

Table 3 shows modifications identified in the filtered result and the number of PSMs containing each modification.

- Name: The given name of the PTM.
- Δ Mass: The monoisotopic change in mass caused by the PTM.
- Position: The amino acids modified by the PTM.
- **# PSM:** The number of PSMs with the given PTM.
- **-10lgP:** The highest -10lgP score achieved by a peptide modified by the PTM.
- Area: The area achieved by the modified peptide with highest -10lgP score.

4.3.9. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

2 data files produced no results. (Expand to see list of zero result data files)

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.



If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics

No Results Found

4.4. Protein View

The Protein view lists protein identifications that have been filtered by the current settings set for the project. It also visualizes the protein sequence coverage for identified proteins.

PEAKS Online	a -	8 ±								PROJECT 🗸 SETT	TINGS 🗸 ADMIN 🗸 /	CCOUNT 🗸	HELP 🗸
B ABRF 2017 DIA +		<		Showing Tap Brateins			Show Results for		Sample Sample 1		Export Protein	Protein-Pentide	
Analysis 1: Spectral Library Search		Sum		Showing top Proteins			() An oumpico	O opeomed	oumpie		← 1-1	00 of 4759 👻	<i>→</i>
🖳 Spectral Library		3		Accession	-10LgP	Coverage	#Peptides	#Unique	PTM	Avg. Mass			Descr
Export		2	1	Q09666[AHNK_HUMAN	2858.11	39.75%	236	236	CDO	629101	Neuroblast differentiation	associated pro	tein AHNAI *
		Pa	2	Q60FE5 Q60FE5_HUMAN	2828.71	56.22%	203	183	COD	278226	Filamin A OS=Homo sapi	ans GN=FLNA P	E=1 SV=1
		teir	3	Q15149 PLEC_HUMAN	2812.04	36.96%	223	2	C O D	531791	Plectin OS=Homo sapien	s GN=PLEC PE=	1 SV=3
		0	4	Q15149-9 PLEC_HUMAN	2809.87	37.83%	222	1	ACOD	514775	Isoform 9 of Plectin OS=F	łomo sapiens G	IN=PLEC
		Pep	5	Q15149-2 PLEC_HUMAN	2807.73	37.47%	222	1	DCO	518473	Isoform 2 of Plectin OS=H	lomo sapiens G	IN=PLEC
		tid	6	P35579 MYH9_HUMAN	2680.43	40.56%	135	108	DOC	226532	Myosin-9 OS=Homo sapie	ins GN=MYH9 F	2E=1 SV=4
		C.	7	P60709 ACTB_HUMAN	2631.45	71.73%	105	6	A D C O	41737	Actin, cytoplasmic 1 OS=	fomo sapiens 0	3N=ACTB P
			8	Q9Y490 TLN1_HUMAN	2615.59	46.16%	142	133	CDO	269767	Talin-1 OS=Homo sapien:	. GN=TLN1 PE=	1 SV=3
			9	P49327 FAS_HUMAN	2614.96	45.12%	145	145	C D A O	273427	Fatty acid synthase OS=H	omo sapiens G	N=FASN PE 🗸
		<											
		Cov	/erage	Peptide									
		Q096	666 AH	INK_HUMAN Neuroblast differen	iation-associated p	rotein AHNAK	OS=Homo sapi	ens GN=AHNA	K PE=1 SV=2		COVERAGE LEGER	ID PTM FILTER	TOOLS
🔚 Data Search Q	×	Sar Sar Sar	1 mple 1 mple 2 mple 3	MEKEETTREL LLPNWQ	GSGS HGLTIA	QR DD GVF	VQEVTQN SI	PAARTGVVH	(EGDQIVGATI	YFDNLQSGEV	/ TQLLNTMGHH		
G Sample 1	01	Sar	mple 4										
Sample 2		186	npie 5			128							
Sample 3	01		81	TVGLKLHRKG DRSPEP	GOTW TREVF	SCSS EVV	LSGDDEE YO	RIYTTKIF	PRLK SEDGVE	GDLGETOSRI	ITVTRRVTAY		
G Sample 4	01	Sor	mole 1		~					~			
Sample 5	01	Sar Sar Sar Sar	mple 2 mple 3 mple 4 mple 5										
		1	61	TVDVTGREGA KDIDIS	SPEF KIKIPR	HELT EIS	NVDVETQ SC	SKTVIR lps	GSGAASPTGS	AVDIRAGAIS	ASGPELQGAG		
		Sar Sar Sar Sar Sar	mple 1 mple 2 mple 3 mple 4 mple 5										
		2	41	HSKLQVTMPG IKVGGS	GVNV NAKGLD	LGGR GGV	QVPAVDI SS	SSLGGRAVE	VQGPSLESGD	HG KIKFPTMF	VPKFGVSTGR		
		Sar Sar Sar	mple 1 mple 2 mple 3										

4.4.1. Protein View Filters

By selecting the 👩 button, the Protein Filter will pop up. It controls which proteins are displayed.

271.07	49.25%	Q 6	30+6	
Protein Filter				
Top Proteins All Proteins Protein accession/name contains	:			
Protein description contains				
Protein sample area ≥				
Include Contaminant Data				
Name to search for			[CLEAR
Deamidation (NQ) Dividation	on (M) 🗌 Carb	amidometh	ylation	
		DECET	CANCEL	

- Show Top or All proteins in each group: Adjust the protein list based on Protein grouping, top is selected by default. Proteins are grouped based on parsimony.
 - **Top:** Show only the top proteins in each group. These proteins are supported by the most unique peptides in the group. Proteins in the group that share a subset of the unique peptides that support the top protein (sub-proteins) will not be displayed.
 - All: Show all proteins in each group. Proteins are grouped together if they are supported by the same set or a sub-set of the top protein in the group. If a protein is supported by a peptide not supporting the top protein it will be added to a new group.
- **Protein accession/name contains:** If you add text here, only proteins with accessions that contain the text will be included in the protein table.
- **Protein description contains:** If you add text here, only proteins with descriptions that contain the text will be included in the protein table.

• **Protein sample area:** Filters based on total protein area found in each sample. Sample areas are calculated using the total of all peptide features from unique supporting peptides. If a protein is identified and has areas from multiple samples, only one of the samples must pass this filter to be included.

Note

Only numeric digits can be typed in the protein sample area. For example, instead of 1.00e+4, 10000 must be typed.

- **Spider Mutations:** This filter is only visible in SPIDER result view filter. Check it to view proteins that have mutations.
- **Protein Contains Modifications**: Proteins that are supported by peptides containing selected PTMs will be included.

4.4.2. Protein Table

Each row in the table represents a group of proteins that are supported by a common set of peptides. A 😁 indicates that the group has multiple proteins. To expand the group, click that button.

The columns in the Protein table are:

- Accession: The accession id of the protein as seen in the FASTA database.
- -10lgP: The protein confidence score. Refer to Section 3.1, "Result Filtration" for a detailed explanation.
- **Coverage:** The percentage of the protein sequence that is covered by supporting peptides. This is the total coverage including results from all samples, unless results are only shown for a specified sample.
- #Peptides: The number of supporting peptides which pass the given peptide filters.
- **#Unique:** The number of supporting peptides which pass the given peptide filters that are mapped to only one protein group. Unique peptides with the same sequence but different modifications are only counted once in this number.
- **PTM:** The identified modifications displayed with color-coded icons.
- Avg. Mass: The protein mass calculated using the average mass.
- **Description:** The protein's header information as seen in the FASTA database.

Note

For #Peptides and #Unique, two peptides with the same starting and ending positions in the protein are counted as one, regardless of their PTM forms. This is to follow the MCP (Molecular & Cellular Proteomics) guidelines.

4.4.3. Global Protein Comparative View

Global Protein Comparative View provides detailed comparisons among different samples regarding the identified protein. To Show/Hide Columns, click **III** on top-left of the protein table. From the dialog, users can add and remove sample coverage, area, and spectral count columns. In the protein table, the selected columns will be displayed and can be exported.

- **Coverage by Sample:** the coverage broken down by sample can be displayed if multiple samples were used. Only peptides that pass the filters are included in the percentage calculation.
- Area by Sample: the total area of peptide features from unique supporting peptides in each sample. Areas from peptides that don't pass the peptide filter won't be included in the total area.
- **#Spec by Sample:** total number of identified peptide spectrum matches which pass the peptide filters that support the given protein. Totals are given for each sample included in the search.

4.4.4. Protein Coverage

The Protein Coverage view visualizes the mapping of supporting peptides to the protein selected in the Protein table. It also shows all identified modification or mutation sites to assist with protein characterization at the amino acid level. The coverage view shows different views under different modes. The view shows a sample comparative coverage view in the all sample mode while it shows protein sequence coverage view in the specific sample mode.

4.4.4.1. Sample Comparative Coverage

This area visualizes the frequency of supporting peptide sequences from selected samples in the protein sequence. At the top of the protein sequence coverage view, the header information of the protein is shown. At the topright corner, the COVERAGE LEGEND button will show a continuous color legend for the peptide frequency. Specific modifications can be selected in the PTM FILTER button and the template protein can be copied using the TOOLS button.



Regions in the protein sequence that are covered by supporting peptides are displayed in bold font with a grey background. Placing the cursor over a amino acid on the sequence shows the position of the amino acid in the protein sequence and the frequency of identification in each selected sample.

The frequency of supporting peptide sequences from different samples can be shown as colored bars under the protein sequence. The deeper the color, the higher frequency of that supporting peptide in the samples.

4.4.4.2. Protein Sequence Coverage

This area visualizes the coverage of the protein sequence in the specified sample. The header information of the protein, a PTM filter and a tool to copy the protein information are shown on the top of the protein sequence coverage view. Clicking the PTM filter button will pop up a menu to list the modifications identified in supporting peptides on the protein and the checkbox on the left controls whether to show the modification in the protein sequence coverage view above.

Coverage	Peptide									
Q09666 AHN	IK_HUMAN Neurobla	ast differentiation-a	ssociated protein A	HNAK OS=Homo s	apiens GN=AHNAK	PE=1 SV=2			PTM FILTER	TOOLS
1	MEKEETTREL	LLPNWQGSGS	HGLTIAQR DD	GVFVQEVTQN	SPAARTGVVK	EGDQIVGATI	YFDNLQSGEV	TQLLNTMGHH		- 1
81	TVGLKLHRKG	DRSPEPGQTW	TREVF SSCSS	EVVLSGDDEE	YQ RIYTTKIK	PRLK SEDGVE	GDLGETQSRT	ITVTRRVTAY		
161	TVDVTGREGA	KDIDISSPEF	KIKIPR HELT	EISNVDVETQ	SGKTVIR lps	GSGAASPTGS	AVDIRAGAIS	ASGPELQGAG		
241	HSKLQVTMPG	I KVGGSGVNV	NAKGLDLGGR	GGVQVPAVDI	SSSLGGRAVE	VQGPSLESGD	HG KIKFPTMK	VPKFGVSTGR		
321	EGQTPKAGLR	VSAPEVSVGH	KGGKPGL TIQ	APQLEVSVPS	ANIEGLEG KL	KGPQITGPSL	EGDLGL KGAK	PQGHIGVDAS		
401	APQIGGSITG	PSVEVQAPDI	DVQGPGSKLN	VPKMKVPKFS	VSGAKGEETG	IDVTLPTGEV	TVPGVSGDVS	LPEIATGGLE		
481	GKMKGTKVK T	PEMIIQKPKI	SMQDVDLSLG	SPKLKGDIKV	SAPGVQGDVK	GPQVALKGSR	VDIETPNLEG	tltgp rlgsp		
561	SGKTGTCR IS	MSEVDLNVAA	p kvk ggvdvt	LP RVEGKVKV	PEVDVRGPK V	DVSAPDVEAH	GPEWNL KMPK	MK mptfstpg		
641	AKGEGPDVHM	TLP KGDISIS	GPKVNVEAPD	VNLEGLGGKL	KGPDVKLPDM	SVKTPK ismp	dvdlhv kgtk	VKGEYDVTVP		
721	KLEGELKGPK	VDIDAPDVDV	HGPDWHLKMP	KMKMPKFSVP	GFKAEGPEVD	VNLPKADVDI	SGPK IDVTAP	DVSIEEPEGK		
801	LKGPKFKMPE	MNIKVPKISM	PDVDLHLKGP	NVK geydvtm	P KVESEIK VP	dvel ksak md	IDVPDVEVQG	PDWHL KMPKM		

Regions in the protein sequence that are covered by supporting peptides are displayed in bold font with a grey background. Confident modifications and mutations identified in supporting peptides are displayed as icons above the protein sequence. Modifications are represented by colored icons with the initial letter of its modification name. If a residue is modified by more than one modification in the same supporting peptide, "*" is used instead of a letter. Mutations are represented by black-framed icons with a letter indicating the mutated residue. Placing the cursor over a modification icon shows the full name of the modification/mutation. The number above a modification icon indicates the position in the protein sequence. Placing the cursor over an amino acid on the sequence shows the position of the amino acid in the sequence.

The supporting peptides can be shown as colored bars under the protein sequence. Placing the cursor over a bar shows detailed information of the peptide. Left clicking on a bar will pop up the annotated spectrum associated with the supporting peptide.

4.4.5. Supporting Peptides

The Peptide table shows a list of supporting peptides for the selected protein. This table is similar to the peptide table in the Peptide View.

- Unique: This shows whether the peptide is a unique supporting peptide to the protein group.
- **Start:** This shows the peptide's starting position in the protein.
- End: This shows the peptide's ending position (inclusive) in the protein.
- **Found by**: The name of the PEAKS search algorithm that identified the peptide. This can be either PEAKS, PEAKS PTM, or SPIDER.

Refer to Section 3.5.2, "Peptide Table" for the descriptions of other columns.

Double-click a row to navigate to the peptide in the peptide tab for more detailed information.

4.5. Peptide View

The Peptide Table shows the filtered peptide identification results. Click on a peptide row to show the spectrum below the peptide table. Information such as the scan number, retention time, isolation window and fragment error tolerance will be shown above the MS2 spectrum.



4.5.1. Peptide View Filters

Click the 👩 button, the following Peptide Filter will pop up. It controls which peptides to display.

Doptido Eiltor		
Peptide Filter		
Peptide sequence contains		
Scan =		
Precursor Id =		
m/z =		
RT =		
Peptide sample area ≿		
Include Contaminant Data		
Peptide Contains Modifications		
Name to search for		CLEAR
Deamidation (NQ) Oxidation (M	Carbamidomethylation]
	RESET CAN	ICEL APPLY

- **Peptide sequence contains:** Enter an amino acid sequence, only peptides that contain that sequence will be displayed in the peptide table.
- Scan: Filter for a single scan number based on the scan numbers read from the raw data file.
- **Precursor Id:** Filter is only relevant for TimsTOF data, for which it will filter based on the precursor id of the peptide.
- m/z: Filter based on the expected m/z, the last significant digit is rounded.
- **RT:** Filter based on the expected RT, the last significant digit is rounded.
- **Peptide sample area:** Filters based on total peptide area found in each sample. If a peptide is found in multiple samples, there should be at least one sample pass this filter for the peptide to be included.
- **Spider Mutations:** Filter only available for SPIDER results. When checked, peptides that have an identified mutation using SPIDER are shown.
- **Peptide Contains Modifications:** Check boxes are provided for each identified modification. Peptides that contain the selected PTMs will be shown.

4.5.2. Peptide Table

The Peptide Table shows the filtered peptide identification results. Each row in the table is a peptide identification represented by its highest-scoring PSM. The table is sorted by the -10lgP value.

The columns in the Peptide table are:

- **Peptide:** The amino acid sequence of the peptide, as determined in PEAKS Search. A modified residue is followed by a pair of parentheses enclosing the modification mass.
- -10lgP: The peptide -10lgP score. The score indicates the scoring significance of a peptide-spectrum match.
- Mass: The monoisotopic mass of the peptide.
- Length: The number of amino acids in the peptide sequence.
- **ppm:** The precursor mass error, calculated as $10^6 \times (\text{precursor mass} \text{peptide mass}) / \text{peptide mass}$.

- m/z: The precursor mass-to-charge ratio.
- **RT:** The retention time (elution time) of the spectrum as recorded in the data.
- **#RT:** Retention time difference between library and query spectrum.
- Scan: The scan number of the spectrum that matches the peptide sequence with the highest -10lgP.
- **#Spec:** The number of MS2 spectra assigned to the peptide.
- PTM: The types and the numbers of modifications present in the peptide shown in color-coded icons.
- Accession: The accession number of the highest-scoring protein containing this peptide.
- Precursor Id: This column is only available for TimsTof data, it shows precursor's id from the raw data.
- 1/k0: This column is only available for TimsTof data, it shows the ion mobility range of the precursor.

4.5.3. Peptide Global Comparative View

"Peptide Global Comparative View" provides detailed comparisons among different samples regarding the identified peptides. To Show/Hide Columns, click **IIII** on top-left of the peptide table. From the pop-up window, users can add and remove area and spectral count columns. In the peptide table, the selected columns will be displayed and can be exported. If "Specified Sample" is selected these columns are automatically displayed.

Select Display Columns and Samples	S
Select Columns	
🗸 Area 🗸 #Spec	
Select Samples Name to search for	CLEAR
SAMPLE NAME	
✓ S1	
✓ S2	
✓ S3	
✓ W1	
✓ W2	
✓ W3	
CANCEL	SAVE

Note

PEAKS Online provides users the option to calculate peptide abundance using the MS1 feature area calculation or using the MS1 feature intensity. By default, area is used. The default can be changed by the administrator.

- Area by sample: The total area of all peptide features matched to spectra that identified the peptide. Area calculations are separated into multiple columns if multiple samples were included in the run.
- **#Peptide-spectrum matches by sample:** The number of peptide spectrum matches associated with a peptide separated by sample.

4.5.4. Peptide Menu

The peptide menu is only shown when "Show Results for Specified Sample" is selected in the filter section. The peptide menu displays the information of the peptide sequence that is being displayed in the spectrum. If a feature is associated to more than one tandem scan, click the "FRACTION", "FEATURE" or "PSM" button to view the Spectrum Annotation of the other PSMs.

FRACTION: 2017-12-4_ABRF_10_DAI42.raw 🗸 🔰 FEATURE: m/z: 468.9138, z: 3, RT : 63.45, Area : 0.00e+0, PPM : 0.00 🗸 💛 PSM: scan: 38103, -10igP: 500.00, RT: 63.64, m/z : 467.1667 🧹

4.5.5. Protein Selection

The protein drop-down in the top-right of the peptide information pane allows for quick navigation to any of the proteins the currently selected peptide supports.

4.5.6. Peptide-Spectrum Matches

This section displays the spectrum annotation and other information about the peptide selected in the Peptide table. When multiple spectra are matched to the peptide, only the information about the top-scoring PSM is shown by default.

4.5.6.1. Spectrum Annotation

The spectrum annotation displays a graphical representation of the peptide-spectrum match.



Moving the cursor over the peptide sequence in the spectrum will show the mass transitions for a particular amino acid residue.

The spectrum annotation panel provides convenient ways to zoom and navigate within the spectrum:

- Zoom in/out on m/z: Place the cursor on the point in the spectrum you would like to zoom in on, or, place the cursor in the Error (da) figure and scroll the mouse wheel.
- Increase/Decrease peak intensity: Place the cursor on the Intensity axis and scroll the mouse wheel.
- See the whole spectrum: click the 1:1 button.

Click S to open the Ion Annotation Settings dialogue and change the ions to be annotated in the spectrum. To reset the settings to the PEAKS Online defaults, use "RESET DEFAULT" button.

Ion Annotation Settings										
	CID	ETD		ETHCD						
		-H2O	-NH3	2+						
а										
b	\checkmark	\checkmark	\checkmark	\checkmark						
С										
х										
у	\checkmark	\checkmark	\checkmark	\checkmark						
Z										
Z'										
C-H										
immonium										
			CANCEL	SAVE	RESET DEFAULT					

The (*) button will switch the y-axis to relative intensity. Once selected, scrolling over peaks will display the relative intensity instead of the absolute intensity.

The *#* button will switch the y-axis to absolute intensity. Once selected, scrolling over peaks will display the absolute intensity instead of the relative intensity.

4.5.6.2. Mirror Plot



This view shows the ions from the query spectrum compared to the spectrum it was matched to from the library.

4.5.6.3. Error Map

The "Error Map" shows the mass errors of the matched fragment ions. The m/z ratio is displayed on the x-axis and the error is listed on the y-axis in Daltons. Each matched fragment ion is represented by a dot. The ion types displayed can be configured in "Ion Annotation Settings".

4.5.6.4. Ion Table

The "Ion Table" shows the calculated mass of possible fragment ions based on the ion types selected in the Ion Annotation Settings. If a fragment ion is found in the spectrum, its mass value is displayed in color. N-terminal ions are shown in blue and C-terminal ions are shown in red. A fragment ion is found when there is a matching peak within the mass error tolerance, as defined in the *de novo* sequencing parameters.

4.5.6.5. Precursor Profile

An eXtracted Ion Chromatogram (XIC) chart that displays the shape of the selected peptides precursor.

4.5.6.6. Fragment Ion XIC

An eXtracted Ion Chromatogram (XIC) chart that displays the shape of each identified fragment ion across all scans for the selected PSM. Below the XIC charts is a slider to adjust the correlation threshold of the fragment ion with the selected peptide precursor. Increasing the slider value will filter out fragment ions that are below the threshold.

4.5.6.7. TimsTOF In Frame Mode

For timsTOF data only, there is an additional "In Frame Mode" button, =, above the spectrum view. Clicking this button will change the annotated spectrum view from "frame mode", the default view, to "in frame mode".

To switch back click the "Frame Mode" button, The standard, "Frame Mode" displays the scans within a frame merged over the CCS dimension. The "In Frame Mode" displays the scans merged over the retention time dimension.

4.6. Manual Search

The Manual Search tab can be used to manually query your data for a specific library entry. If a peptide is found, PEAKS will display its peptide information and spectrum with scan, RT, isolation window and fragment tolerance information.



4.6.1. Performing a Manual Search

1. Select the sample you wish to search in from the Sample Dropdown, this should be the first dropdown displayed.

Sample 1	×	~
Sample 1		^
Sample 2		
Sample 3		•

2. If your data is fractionated, select the fraction you wish to search within the selected sample form the Fraction Dropdown, this should be the second dropdown displayed.



3.

Click the Input Library Entry Button, is to specify the parameters of the library entry to be queried. *These parameters will be outlined in more detail in the following section.*

4.		
	APPLY	
Click the Apply Button,		, to query the selected data file for the input librarD entry

4.6.2. Manual Search Input

- **Peptide:** The peptide sequence that you wish to search for. This should include any modifications, for example LNIISNLDC(+57.02)VNEVIGIR.
- Charge: The precursor charge to search for.

Note

All optional parameters below will be filled in automatically if left blank with the best match found given the input parameters.

- **Fragmentation:** (*Optional*) The ion fragmentation information. This should be a list of ions with their m/z and intensity values in this format "<m/z>:<ion>", for example "327.13446,0.1547,b2". The intensity should be a relative intensity in decimal format, for example 0.1547 for 15.47% relative intensity.
- Library iRT: (Optional) The indexed retention time to search for, entered in minutes.
- **RT** Apex: (*Optional*) The retention time, in minutes, for the apex scan in the selected data file.
- **RT Start:** (*Optional*) The beginning retention time, in minutes, in the selected data file.
- **RT End:** (*Optional*) The end retention time, in minutes, in the selected data file.
- **CCS Start:** (*Optional*) The starting CCS value in the selected data file. Only applicable if searching TimsTOF Data.
- **CCS End:** (*Optional*) The ending CCS value in the selected data file. Only applicable if searching TimsTOF Data.

Pentide				
YYVTIIDAPGHR				
Charge				
2				
Fragmentation				
164.0672:0.0021:b2[2+];312.1779	:0.0278:y2;327.13	45:1.0000:b2;36	9.1993:0.019	99:y3;426
Library IRT				
38.91				
RT Apex				
65.77				8
RT Start				
64.77				
RT End				
66.77				

4.6.3. Viewing Manual Search Results

Once you have submitted your manual search, the results will be displayed after a brief delay.

If there were no results found in the selected data file that match the input spectra an error will appear indicating that no results could be found.



4.6.4. Peptide Table

The Peptide Table will list all possible identification results in the selected data file that meet the supplied criteria.

	Peptide	Mass :	Charge :	Length :	m/z :	Apex RT :	Predicated RT
0	YYVTIIDAPGHR	1403.7197	2	12	702.8671	65.77	65.77

The columns in this peptide table are:

- **Peptide:** The amino acid sequence of the peptide, as input in the search parameters. A modified residue is followed by a pair of parentheses enclosing the modification mass.
- Mass: The monoisotopic mass of the peptide.

- Charge: The charge of the peptide, as input in the search parameters.
- Length: The number of amino acids in the peptide sequence.
- **m/z:** The precursor mass-to-charge ratio.
- Apex RT: The retention time (elution time) of the apex scan as recorded in the data.
- **Predicted RT:** The predicted retention time (elution time) of the input peptide.
- 1/k0: This column is only available for TimsTof data, it shows the ion mobility range of the precursor.

4.6.5. Peptide Information

For more details on the provided figures refer to Section 4.5.6.

5. DIA Database Search Workflow and Parameter Settings

PEAKS DIA DB can be launched as a standalone workflow, or launched on top of a PEAKS Library Search workflow. An analysis report is generated to show the database search result, combined with the library search results if run on top of a PEAKS Library Search. Users can also run an optional PEAKS DIA Denovo search on top of the PEAKS DIA Database Search.

To conduct a database search workflow, follow the steps below:

- 1. Refer to Chapter 7, Section 3, Spectral Library Search Workflow and Parameter Settings, Section 3, "Spectral Library Search Workflow and Parameter Settings", Step 1-3 to create a new project and add data. If you wish to run the Database Search on top of a Library Search, follow Step 4 as well.
- 2. Check the checkbox next to Database Search Parameters beneath Library Search. Uncheck the checkbox next to Library Search Parameters if you do not wish to run a Library Search before the Database Search.

Database Search Parameters

3. Set the DB Search Parameters. See details in Section 5.1-5.6 below.

Enzyme:	Specified I	by eac	h sample √	Digest N	Mode: S	emi-Spec	ific		*	Miss	ed Cleavage:	1			
Target Database:	Benchmark	(HYE	188 🗸	Taxono	omy: all	species >	ĸ		~	_	Contaminant Database:	N/A		~	
Peptide Length:	7	to	30	Precu	ursor arge: 1	/4 rasta seq	to	4		F	Precursor MZ:	300	to	1800	
Fragment MZ:	200	to	1800												
DD/REMOVE MODIFIC	ATION								Max Varia	able PTM Pe	er Peptide:	1			
DD/REMOVE MODIFIC	ATION								Max Varia	able PTM Po	er Peptide:	1]		
DD/REMOVE MODIFIC Report Filter M/Peptide	ATION								Max Varia	able PTM Pr	er Peptide:	1]		
Report Filter M/Peptide	ation LgP ≥		P:	M FDR (%)				Peptic	Max Varia	able PTM Pr	er Peptide:	1			
DD/REMOVE MODIFIC Report Filter M/Peptide O PSM -101 otein	ation LgP ≥		 P\$ 	M FDR (%)	1			Peptic	Max Varia	able PTM P	er Peptide:	1			

4. Click SUBMIT button to start the search.

5.1. Precursor and Fragment Error Tolerance

Precursor mass: This enables the selection of a precursor mass error tolerance. PEAKS Online DB Search identifies peptides within the precursor mass tolerance, which is the allowable m/z shift between the theoretical value of the peptide versus the observed value of the precursor ion. Precursor mass tolerance can be specified in either Da or ppm.

Fragment ion: This enables the selection of a fragment ion m/z tolerance, which is the allowable m/z shift between the theoretical value of the fragment ions of the peptide versus the observed value. PEAKS Online DB Search uses this value when scoring peptide-spectrum matches. PEAKS Online considers that a fragment ion is matched if the calculated m/z is within the tolerance. Fragment mass tolerance can be specified in Da.

Error tolerance parameters need to be set consistently with the mass accuracy of the instrument.

Note

When the samples in a project are generated by different instruments and "default" is chosen from the Predefined parameters drop-down menu, the default parameters of the first sample are loaded. However, it is recommended to set the tolerance parameters according to the least accurate instrument.

5.1.1. Optimize Tolerance

When enabled, PEAKS Online will determine the optimal values to use for Precusor and Fragment Mass Error Tolerance for each fraction in your analysis. These values will replace the values provided in the previous Precursor and Fragment Mass Tolerance sections.

5.2. Enzyme Settings

Enzyme : PEAKS Online digests the protein database *in silico* to generate peptide candidates. It is necessary to specify the enzyme for protein digestion from the Enzyme drop-down menu.

Note

When the selected dataset is digested with different enzymes, "Specified by each sample" allows samples to be analyzed separately using their respective enzymes specified during project creation.

Note

"None" is a special enzyme digest rule allowing non-specific cleavage at both ends of the peptide. The "None" digest rule can cut at every residue, generating peptides with lengths up to 65 amino acids. Its usage is recommended when no digestion enzyme was used or when the digestion enzyme exhibits a high degree of non-specificity.

Missed Cleavages : This specifies the number of missed cleavage sites allowed in a peptide.

Digest Mode : This specifies the type of *in silico* cleavages allowed at a peptide termini based on the method used for protein digestion. If "Specific" is selected, the specificity of the selected enzyme is strictly enforced at both termini of a peptide. "Semispecific" allows one terminus to disobey the enzyme specificity rules.

Note

If the enzyme is specified as "None", then no matter which mode is selected for "Digest Mode", the "None" enzyme digest rule will be applied.

5.3. Database Settings

A protein database must be added first before creating a new project to enable its selection in the workflow. In the Target Database drop-down menu, select the appropriate database that have been configured in PEAKS Online and set the taxonomy, if applicable. A contaminant database can also be selected from the same list of databases that have been configured in PEAKS Online. The length for a peptide can be set when creating an analysis, where the default peptide length range is between 6 and 45.

5.4. PTM Settings

Click "ADD/REMOVE MODIFICATION" button to open the "Select Modifications" dialogue and specify the fixed PTMs and a few (less than 11) common variable PTMs expected in the sample. If the required modification is not listed, the modification can also be created here. Additionally, the maximum number of variable PTMs per peptide can be specified and a value of less than 4 is recommended. A fixed modification forces all instances of applicable residues to be modified. A variable modification gives the option for the residues to be modified.

Note

For a specific residue, only one fixed modification is allowed.

Note

To prevent long search times, select only the most frequent PTMs in the sample for PEAKS DB, and check the PEAKS PTM option to search for unspecified modifications.

5.5. PSM Filter

PSM filter section is a mandatory section to filter out some results under the user-defined filter. Any changes to the filter will create a new analysis with old parameters and update the results and statistics.

• **PSM -10lgP:** The PEAKS peptide score (-10lgP) is calculated for every peptide-spectrum match (PSM) reported by PEAKS DB, PEAKS PTM, and SPIDER. The score is derived from the p-value that indicates the

statistical significance of the peptide-spectrum match. A peptide may be matched to many spectra, resulting in multiple PSMs. In that case, the peptide's score is calculated as the maximum among all PSM scores. For details of the scoring algorithm, please refer to the publication, "*PEAKS DB: De Novo sequencing assisted database search for sensitive and accurate peptide identification*", *Mol Cell Proteomics, 2011 Dec 20*. A minimum requirement can be set and all identifications must pass this filter.

• **PSM FDR (%):** The PSM filter can also be set to use the false discovery rate. FDR is estimated using a "decoy-fusion" approach.

Note

Decoy fusion is an enhanced target-decoy method for result validation with FDR. Decoy fusion appends a decoy sequence to each protein as the "negative control" for the search. See BSI's web tutorial (http://www.bioinfor.com/fdr-tutorial/) for more details.

6. Understanding PEAKS DIA Database Search Results

After a PEAKS Search is complete, all analyses selected during the project creation will be combined under the workflow drop-down menu. One is the *de novo* sequencing result, which includes *de novo* peptide sequences listed, spectrum annotation and other detailed information. The other result nodes are identification results from the database search tools: PEAKS DB, PEAKS PTM, and SPIDER. If more than one database search tool is enabled, the results from previous search tools are automatically merged.

The analysis results for database search tools are presented in four tabs:

- **Summary :** This shows an outline of the PEAKS Online database search results with key statistics. The overall quality of the experiment can be examined and the filters for peptide and protein identifications can be adjusted. This page shows after the search is done.
- **Protein :** This shows a list of protein identifications. This view also visualizes protein sequence coverage and helps with protein characterization.
- **Peptide :** This shows a list of peptide identifications. This view also provides spectrum annotation and other detailed information for peptide precursor spectrum matches.

6.1. Result Filtration

At the top of each summary result page, filters can be set. Identifications that fall below the filters will not be visible in the result pages or exports. After changing any of these filter parameters, click the *wey* button to apply and save the new filters. The results and statistical information will be updated accordingly.

Filters	Proteins -10lgP ≥ 0	Proteins Unique Peptides ≥ 10	:	APPLY Created:	Nov 05 2019, 12:07:34 pm

- **Proteins -10lgP:** The PEAKS protein score (-10lgP) is calculated as the weighted sum of the -10lgP scores of the protein's supporting peptides. After removing any redundant peptides, the supporting peptides are sorted by -10lgP scores in descending order, and the k-th ranked peptide contributes to the weighted sum with a weight of 1/k. A default threshold of 20 is set.
- **Proteins Unique Peptides:** The minimum number of unique supporting peptides for a protein identification. A unique peptide is defined as a peptide that can be mapped to only one protein group.

6.1.1. Show Results for All Samples

For the protein and the peptide tables, "All Samples" can be selected to view the combined search results from all the samples. In the Protein page, selecting "All Samples" will change the coverage panel displayed when a specific protein is highlighted. Supporting peptides from all samples instead of just a single sample will be shown.

 All Samples 	O Specified Sample	Sample 1	\sim

6.1.2. Show Results for Specific Sample

In the Protein, Peptide and De Novo page results, the filter can also be set to "Specified Sample" which will only show the results a single sample. By clicking in the text you can search for a specific sample by name.



Selecting "Specified Sample" will change the protein coverage view that is visible by clicking on a protein in the protein table. Peptides in the Protein, Peptide and De Novo Only pages will only be visible if they belong to the sample chosen. Once selected, the peptide tables will automatically add the "Area" column and the "# Spec" values will display the spectral count for the specified sample. For more information, refer to Section 3.5.2, "Peptide Table". Similarly, the protein table will also include the columns "Coverage" and "Area" for the specified sample. For more information refer to Section 3.4.2, "Protein Table".

6.2. Result Exporting

PEAKS Online database search results can be exported in the Export page:

Export	SELECT ALL		CLEAR ALL	
All Search Parar	meters 🗌	Da	atabase Sea	rch
	C		Summary Tal	ole
	C		Protein CSV	
	C		Protein-Pepti	de CSV
	C		Peptide CSV	
	C		PSM CSV	
	C		Protein Fasta	I
DOWNLOAD DOW	NLOAD BY SAMPL	E		

- **Protein CSV:** A comma separated file containing all identified proteins and associated details from the protein table for all samples.
- **Protein-Peptide CSV:** A comma separated file containing all identified peptides and associated details from the peptide table found in the Protein page. Further details are provided regarding the protein the peptide supports. If a peptide supports multiple proteins, multiple rows are included for each supported protein for all samples.
- **Peptide CSV:** A comma separated file containing all identified peptides and associated details from the peptide table for all samples.
- **PSM CSV:** A comma separated file containing all identified peptide spectrum matches and associated details from the peptide table for all samples.

- **Pepxml:** An XML export of the identification results matching the pepxml schema. This can be uploaded to software that accept data in this format.
- **MzIdentML:** An XML export of the identification results matching the MZIdentML schema. This can be uploaded to software that accept data in this format.
- **Protein Fasta** : A list of protein identifications will be saved to proteins.fasta.

6.3. Summary View

6.3.1. Summary Statistics

The first chart displays the summary information for each sample.

- **Sample Name:** This column lists the samples names with the exception of the first row, which shows the combination of all the samples.
- **# MS Runs:** The total MS runs in the sample.
- # MS1: The total MS scans in the sample summed across all fractions.
- # MS2: The total MS/MS scans in the sample summed across all fractions.
- **# PSM:** Total number of peptide spectrum matches (PSMs) found in the sample.
- **# Peptides:** Total number of peptides found ignoring differences with leucine (L) and isoleucine (I) amino acids. Peptides with the same primary sequence but different PTMs are counted separately.
- **# Sequences:** Total number of different peptide backbone sequences found ignoring differences with leucine (L) and isoleucine (I) amino acids. Peptides with the same primary sequence but different PTMs are counted as one sequence.
- **# Proteins:** Provides the number of proteins given the parameters set for Protein Score (-10lgP) and number of unique peptides.
- **# Protein Groups:** PEAKS Online groups proteins identified by a common set of peptides. This number shows the number of protein groups in the filtered result based on All proteins.
- # PSM / # MS2: The percentage of MS2 scans in the sample that produced peptide spectrum matches.

6.3.2. False discovery rate (FDR) curve

Figure 1 shows the FDR curve with respect to the number of PSMs being kept after filtration by the peptide -10lgP score. By lowering the score threshold, more PSMs are kept in the filtered result. Conversely, the FDR increases because more false positives are kept. In this figure, the vertical line indicates the current score threshold. The number of PSMs and the corresponding FDR by the current score threshold are shown in the top-left corner.

PEAKS Spectral library search estimates the FDR using decoy peptides and spectra. A decoy peptide/spectrum for each target peptide/spectrum in a spectral library and the target and decoy spectra are searched together. The decoy method is based on: https://doi.org/10.1021/pr900947u

It is commonly recommended to set FDR at <1% by adjusting the score threshold. If a rapid growth of the FDR around the 1% FDR threshold is noticed, some PSMs may be sacrificed to significantly reduce the FDR.

Note

When counting the number of PSMs, PEAKS Online keeps one peptide per spectrum at most. Thus, the number of PSMs is actually the number of spectra with assigned peptides.

Note

Decoy matches are not counted in the number of PSMs. Unless otherwise specified, decoy matches are also excluded from the other statistical numbers shown in the Summary view.

6.3.3. Statistics of filtered results

Table 1 shows the statistical numbers of the filtered results.

- **Top Proteins:** The number of proteins that are supported by the most unique peptides in the group. Proteins in the group that share a subset of the unique peptides that support the top protein (sub-proteins) will not be included here.
- **Proteins (#Unique Peptides):** In Table 2, this entry counts the number of proteins identified with unique peptides. A unique peptide is a peptide with a -10lgP score above the peptide filtering threshold that can be mapped to only one protein group. Peptides are counted based on their backbone sequence while considering I/L amino acids to be the same.
- **FDR (Peptide-Spectrum Matches):** The total number of decoy database assignments to spectra relative to the total number of target database assignments to spectra represented as a percentage
- **FDR (Peptide Sequences):** The total number of decoy database assignments to unique peptide sequences relative to the total number of target database assignments to unique peptide sequences represented as a percentage.
- **FDR** (**Protein Group**): The total number of decoy protein groups where the top hit in the protein group is a decoy database protein relative to the total number of target database protein groups where the top hit is from the target database.
- **-10lgP Cutoff:** The value is determined by the estimated PSM FDR. Only the peptides with PSMs above the -10lgP score threshold are listed in the Peptide view.

6.3.4. Precursor mass error

Figures 2(a) and 2(b) show the precursor mass errors of PSMs in filtered results. Figure 2(a) shows the distribution of the precursor mass error in a histogram with and without software calibration. Software calibration uses the observed distribution of mass error after identification to normalize the mass error around zero. Figure 2(b) shows precursor m/z versus precursor mass error in a scatter plot after software calibration. Figures 2(a) and 2(b) help examine whether or not the mass spectrometer is properly calibrated. For a well-calibrated instrument, precursor mass errors should center at 0 ppm across the range of m/z.

6.3.5. RT Calibration

Plots the RT regression function used during the DIA database search. Peptides are plotted against their indexed retention time (iRT) predicted using machine learning.

6.3.6. Box plot for Rt change

A box plot showing the difference between the observed retention time and iRT peptides within set retention time bins across all fractions, showing the accuracy of retention time prediction.

6.3.7. Number of identified peptides in each sample by number of missed cleavages

Presents the number of identified peptides in each sample with the number of missed cleavages, indicating the enzyme digestion efficiency.

6.3.8. PTM Profile

Table 3 shows modifications identified in the filtered result and the number of PSMs containing each modification.

- Name: The given name of the PTM.
- Δ Mass: The monoisotopic change in mass caused by the PTM.
- **Position:** The amino acids modified by the PTM.
- **# PSM:** The number of PSMs with the given PTM.
- **-10lgP:** The highest -10lgP score achieved by a peptide modified by the PTM.

• Area: The highest area achieved by a peptide modified by the PTM.

6.3.9. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

```
2 data files produced no results. (Expand to see list of zero result data files)
```

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.



If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics

No Results Found

6.4. Protein View

The Protein view lists protein identifications that have been filtered by the current settings set for the project. It also visualizes the protein sequence coverage for identified proteins. After clicking the "Peptide" tab we can view the supporting peptide list, double click any peptide can lead to that peptide sequence in the peptide page.

< Sum		٩	Showing Top Proteins			Show Resu All Sa	ults for amples O Specified S	Sample Sample 3	~	Export Protein Protein Contract Protein Protein	<u>>tein-Peptide</u> of 6094 ▼	<i>→</i>
maŋ			Accessio	on -	-10LgP↓ Cov	rerage #Peptide	es #Unique	PTM	Avg. Mass			Descr
-		1	Q09666[AHNK_HUMAN		1278.89	42.65%	188 188	C 0	629101	Neuroblast differentiation-a	ssociated protei	n AHNAI ^
rot		2	Q15149 PLEC_HUMAN		1185.93	35.25%	130 2	CO	531791	Plectin OS=Homo sapiens G	N=PLEC PE=1 S	V=3
eina		3	Q15149-3 PLEC_HUMAN		1184.03	35.65%	129 1	0	518032	Isoform 3 of Plectin OS=Ho	no sapiens GN=	PLEC
77		4	P78527[PRKDC_HUMAN		1123.48	29.55%	98 98	0	469089	DNA-dependent protein kina	se catalytic sub	unit OS=
epi		5	Q14204 DYHC1_HUMAN		1093.09	25.44%	91 91	C 0	532408	Cytoplasmic dynein T neavy	chain I US=Hor	no sapie
ide	0	0 7	OSOFESIOSOFES HUMAN		1091.48	40.10%	75 61	00	220032	Filomin & OS=Homo sapient	GN=WITH9 PE=	1 SV=4
ŝ		8	013813ISDTN1 HUMAN		1074.00	30.60%	75 1	0	284530	Spectrin alpha chain non-er	vtbrocutic 1 OS:	Homo e
		0	013813-3ISPTN1 HUMAN		1072.81	40.01%	75 1		282282	Isoform 3 of Spectrin alpha	chain non-ervth	rocytic 1 -
	4			•	1072.01	40.0110	70 1		LOLLOL	isoloini o or opeetiin alpita	chain, non cryai	+
	Cov Q096	^{verage} 566 AH	Peptide INK_HUMAN Neurobla	ast differentiation-a	associated protein	AHNAK OS=Homo	sapiens GN=AHNA	< PE=1 SV=2		COVERAGE LEGEND	PTM FILTER	TOOLS
		1	MEKEETTR el	LLPNWQGSGS	HGLTIAQRDD	GVFVQEVTQ	N SPAAR TGVVK	EGDQIVGATI	YFDNLQSGEV	/ TQLLNTMGHH		- 1
	Sar	mple 3 81	TVGLKLHRKG	DRSPEPGQTW	TREVFSSCSS	EVVLSGDDER	E YQRIYTTKIK	PRLK sedgve	GDLGETQSR	I ITVTRRVTAY		- 1
	Sar 1	mple 3 61	TVDVTGREGA	KDIDISSPEF	K IKIPR HELT	EISNVDVET	2 SGKTVIRLPS	GSGAASPTGS	AVDIRAGAIS	S ASGPELQGAG		
	Sar 2	mple 3 41	HSKLQVTMPG	IKVGGSGVNV	NAKGLDLGGR	GGVQVPAVDI	I SSSLGGRAVE	VQGPSLESGD	hgk ikfptMk	VPKFGVSTGR		
	Sar 3	mple 3 21	EGQTPKAGLR	VSAPEVSVGH	K <mark>ggkpgltiç</mark>	APQLEVSVPS	S ANIEGLEGKL	KGPQITGPSL	EGDLGLKGAR	C PQGHIGVDAS		
	Sar 4	mple 3 01	APQIGGSITG	PSVEVQAPDI	DVQGPGSKLN	VPKMKVPKF:	5 VSGAKGEETG	IDVTLPTGEV	TVPGVSGDVS	5 LPEIATGGLE		
	Sar 4	mple 3 81	GKMKGTKVK T	PEMIIQKPKI	SMQDVDLSLG	SPKLKGDIK	V SAPGVQGDVK	GPQVALKGSR	VDIETPNLEG	TLTGPR LGSP		
	Sar 5	mple 3 61	SGKTGTCR IS	MSEVDLNVAA	PKVKGGVDVT	LPR VEGKVKV	V PEVDVRGPK v	DVSAPDVEAH	GPEWNLK MPK	MK MPTFSTPG		
	Sar 6	mple 3 41	AKGEGPDVHM	tlpk gdisis	GPK VNVEAPD	VNLEGLGGK	L KGPDVK lpdm	SVKTPK ismp	dvdlhvk gtk	VKGEYDVTVP		
	Sar 7	mple 3 21	K LEGELKGPK	VDIDAPDVDV	HGPDWHLKMP	KMKMPKFSVI	P GFKAEGPEVD	VNLPKADVDI	SGPK IDVTAF	P DVSIEEPEGK		
	Sar	mple 3	•					859 0				
	8	U1	LKGPKFKMPE	MNIKVPKISM	PDVDLHLKGF	NVKGEYDVTN	M PKVESEIKVP	DVELKSAKMD	IDVPDVEVOG	POWHLK MPKM		*

-44	Dentida	Unimus	101-0	14	L an other			DT	Arres Orenals O	0
#		Unique	-TULGP 1	Mass	Length	ppm :	T29 4006	RI 119.74	Area Sample 3	SC8
2			222.11	1802 8051	16	6.1	601.9760	80.14	2.380+7	50322:2017-1
2		· · ·	222.07	2059.0469	10	7.4	1020 4992	95.20	2.300+7	54042-2017-1
4		· · · · · · · · · · · · · · · · · · ·	222.01	2050.9400	18	0.4	1032.0080	05.20	3.010+7	62170-2017-1
5			222.01	2003.9022	10	9.4	1032.9980	70.95	3.01e+7 2.12o+7	50097-2017-1
6	CDTVCCCI DCICVCCI ECNI OMDCIK	~	222.00	2030.0013	26	7.0	016 1146	100.54	2.12017	91996-2017-1
7		×	221.99	2443.3049	20	7.0	440 5716	04.01	2.30010	61070-0017-1
/			221.90	1324.0649	10	6.0	442.3/10	94.91	0.13e+0	612/9:2017-1
8	VDINAPDVDVHGPDWHLK	~	221.98	2025.9907	18	0.2	0/0.341/	80.32	1.900+7	50412:2017-1
9	VDINAPDVEVHGPDWHLK	~	221.98	2040.0065	18	7.2	681.0143	80.21	2.02e+7	50369:2017-1
10	VDISAPDVDVHGPDWHLK	~	221.98	1998.9799	18	10.2	1000.5074	82.29	2.2/e+/	51891:2017-1
11	VDVSAPDVEAHGPEWNLK	~	221.98	1961.9482	18	8.3	981.9895	77.76	2.79e+7	48622:2017-1
12	VGIDTPDIDIHGPEGK	✓ <i>✓</i>	221.98	1661.8260	16	7.5	831.9265	71.61	2.76e+7	44013:2017-1
13	ISMPDVDLHLK	~	221.96	1266.6642	11	7.0	634.3438	80.66	5.57e+7	50797:2017-1
14	MPEMSIKPQK	~	221.95	1187.6042	10	6.6	594.8133	38.80	3.24e+6	19921:2017-1
15	VDIDAPDVEVHDPDWHLK	~	221.95	2098.9958	18	7.4	700.6777	82.06	2.76e+7	51789:2017-1
16	SPSLDVTVPEAELNLETPEISVGGK	~	221.94	2580.3169	25	7.0	861.1190	120.06	4.60e+6	79576:2017-1
17	LPQFGISTPGSDLHVNAK	~	221.93	1879.9791	18	6.9	627.6713	79.68	9.44e+6	49979:2017-1
18	APEVNLNAPDVDVHGPDWNLK	~	221.92	2299.1233	21	8.1	767.3879	92.94	2.50e+7	59834:2017-1
19	ISAPNVDFNLEGPK	~	221.92	1499.7620	14	6.9	750.8934	86.13	7.70e+6	54802:2017-1
20	LEGELQAPDLELSLPAIHVEGLDIK	1	221.90	2698.4429	25	7.8	900.4952	127.00	6.98e+6	84480:2017-1
21	GDVDVSLPEVEGEMK		221.88	1602.7446	15	7.9	802.3859	86.61	1.58e+7	55149:2017-1
22	MPSLEISAPK	1	221.88	1071.5634	10	6.4	536.7924	68.42	1.08e+7	41676:2017-1
23	ADIDISGPNVDVDVPDVNIEGPDAK	1	221.86	2563.2288	25	7.5	855.4233	112.64	3.46e+6	74243:2017-1
24	GGOIGI OAPGI SVSGPOGHI ESGSGK	./	221.86	2417 2297	26	8.3	806 7573	73.35	2 97e+6	45302.2017-1-

6.4.1. Protein View Filters

By selecting the 👩 button, the Protein Filter will pop up. It controls which protein to display.

	271.07	49 3	5%	9.5	30+6	
Protein Filte	er					
Top Prote Protein accession/	ins O All	Proteins				
Protein description	contains					
Protein sample are	8 ≥					
Include Contains	ontaminant Dat Modifications h for	ta				CLEAR
Deamidat	ion (NQ)	Oxidation (M)	Carba	amidometh	ylation	

- Show Top or All proteins in each group: Adjust the protein list based on Protein grouping, top is selected by default. Proteins are grouped based on parsimony.
 - **Top:** Show only the top proteins in each group. These proteins are supported by the most unique peptides in the group. Proteins in the group that share a subset of the unique peptides that support the top protein (sub-proteins) will not be displayed.
 - All: Show all proteins in each group. Proteins are grouped together if they are supported by the same set or a sub-set of the top protein in the group. If a protein is supported by a peptide not supporting the top protein it will be added to a new group.
- **Protein accession/name contains:** If you add text here, only proteins with accessions that contain the text will be included in the protein table.
- **Protein description contains:** If you add text here, only proteins with descriptions that contain the text will be included in the protein table.
- **Protein sample area:** Filters based on total protein area found in each sample. Sample areas are calculated using the total of all peptide features from unique supporting peptides. If a protein is identified and has areas from multiple samples, only one of the samples must pass this filter to be included.

Note

Only numeric digits can be typed in the protein sample area. For example, instead of 1.00e+4, 10000 must be typed.

- **Spider Mutations:** This filter is only visible in SPIDER result view filter. Check it to view proteins that have mutations.
- **Protein Contains Modifications** : Proteins that are supported by peptides containing selected PTMs will be included.

6.4.2. Protein Table

Each row in the table represents a group of proteins that are supported by a common set of peptides. A 😁 indicates that the group has multiple proteins. To expand the group, click that button.

The following columns are sortable: Accession, -10LgP, Coverage, Peptides, #Unique, Avg Mass, when All Samples are shown.

The columns in the Protein table are:

- Accession: The accession id of the protein as seen in the FASTA database.
- -10lgP: The protein confidence score. Refer to Section 3.1, "Result Filtration" for detailed explanation.
- **Coverage:** The percentage of the protein sequence that is covered by supporting peptides. This is the total coverage including results from all samples, unless results are only shown for a specified sample.
- **#Peptides:** The number of high-confidence supporting peptides.
- **#Unique:** The number of high-confidence supporting peptides that are mapped to only one protein group. Unique peptides with the same sequence but different modifications are only counted once in this number.
- **PTM:** The identified modifications displayed with color-coded icons.
- Avg. Mass: The protein mass calculated using the average mass.
- **Description:** The protein's header information as seen in the FASTA database.

Note

For #Peptides and #Unique, two peptides with the same starting and ending positions in the protein are counted as one, regardless of their PTM forms. This is to follow the MCP (Molecular & Cellular Proteomics) guidelines.

6.4.3. Global Protein Comparative View

Global Protein Comparative View provides detailed comparisons among different samples regarding the identified protein. To Show/Hide Columns, click **III** on top-left of the protein table. From the dialog, users can add and remove sample coverage, area, and spectral count columns. In the protein table, the selected columns will be displayed and can be exported.

- **Coverage by Sample:** the coverage broken down by sample can be displayed if multiple samples were used. Only peptides that pass the filters are included in the percentage calculation.
- Area by Sample: total area of peptide features from unique supporting peptides in each sample are used for calculation. Peptides must pass the filters.
- **#Spec by Sample:** total number of spectra identified that support the given protein. Totals are given for each sample included in the search.

6.4.4. Protein Coverage

The Protein Coverage view visualizes the mapping of supporting peptides to the protein selected in the Protein table. It also shows all identified modification or mutation sites to assist with protein characterization at the amino acid level. The coverage view shows different views under different modes. The view shows a sample comparative coverage view in the all sample mode while it shows protein sequence coverage view in the specific sample mode.

6.4.4.1. Sample Comparative Coverage

This area visualizes the frequency of supporting peptide sequences from selected samples in the protein sequence. At the top of the protein sequence coverage view, the header information of the protein is shown. At the top-right corner, the COVERAGE LEGEND button will show a continuous color legend for the peptide frequency. Specific modifications can be selected in the PTM FILTER button and the template protein can be copied using the TOOLS button.

Coverage	Peptide									
Q09666 AHN	NK_HUMAN Neurobla	ast differentiation-a	ssociated protein A	HNAK OS=Homo s	apiens GN=AHNAK	PE=1 SV=2		COVERAGE LEGEND	PTM FILTER	TOOLS
1	MEKEETTR el	LLPNWQGSGS	HGLTIAQRDD	GVFVQEVTQN	SPAAR TGVVK	EGDQIVGATI	YFDNLQSGEV	TQLLNTMGHH		- 1
Sample 3 81	TVGLKLHRKG	DRSPEPGQTW	TREVFSSCSS	EVVLSGDDEE	YQRIYTTKIK	PRLK sedgve	GDLGETQSRT	ITVTRRVTAY		- 1
Sample 3 161	TVDVTGREGA	KDIDISSPEF	K IKIPR HELT	EISNVDVETQ	SGKTVIR lps	GSGAASPTGS	AVDIRAGAIS	ASGPELQGAG		
Sample 3 241	HSKLOVTMPG	IKVGGSGVNV	NAKGLDLGGR	GGVOVPAVDI	SSSLGGRAVE	VOGPSLESGD	HGK IKFPTMK	VPKFGVSTGR		
Sample 3 321	EGOTPKAGLR	VSAPEVSVGH	KGGKPGLTIO	APOLEVSVPS	ANIEGLEGKL	KGPOITGPSL	EGDLGLKGAK	POGHIGVDAS		
Sample 3	APOIGGSITG	PSVEVOAPDT	DVOGPGSKLN	VPKMKVPKFS	VSGAKGEETG	IDVTLPTGEV	TVPGVSGDVS	LPETATGGLE		
Sample 3	GKMKGTKVK	PEMITOKPKT	SMODVDISIG	SPKLKGDIKV	SAPGVOGDVK	GPOVALKGSR	VDIETPNIEG	TLTGPRLGSP		
Sample 3	SCUTCTOP	MCEUDINUAA	PRUVCCUDUT	TREVECTVVVV	DENDUDGDVW	DUCADDUCAU	CDEWNIKMPK	MUNDTESTIC		
Sample 3	JUCECEDUUN	MSEVDENVAA	CDWWWWWW	UNIT DOL CONT	KODDUK	OUTEDUTOND	DUDI UUKOBK	WEEPVDUTUD		
Sample 3	AKGEGPDVHM	TLPKGDISIS	GPRVNVEAPD	VNLEGEGGKL	KGPDVKLPDM	SVRIPRISMP	DVDLHVKGIK	VKGEIDVIVP		
721 Sample 3	KLEGELKGPK	VDIDAPDVDV	HGPDWHLKMP	KMKMPKFSVP	GFKAEGPEVD	VNLPKADVDI	SGPKIDVTAP	DVSIEEPEGK		
801	LKGPKFKMPE	MNIKVPKISM	PDVDLHLKGP	NVK geydvtm	pk veseik vp	DVELKSAKMD	IDVPDVEVOG	PDWHLK MPKM		•

Regions in the protein sequence that are covered by supporting peptides are displayed in bold font with a grey background. Placing the cursor over a amino acid on the sequence shows the position of the amino acid in the sequence and the frequencies from different selected samples.

The frequency of supporting peptide sequences from different samples can be shown as colored bars under the protein sequence. The deeper the color, the higher frequency of that supporting peptide in the sample.

6.4.4.2. Protein Sequence Coverage

This area visualizes the coverage of the protein sequence in the specified sample mode. The header information of the protein, a PTM filter and a tool to copy the protein information are shown on the top of the protein sequence coverage view. Clicking the PTM filter button will pop up a menu to list the modifications identified in supporting peptides on the protein and the checkbox on the left controls whether to show the modification in the protein sequence coverage view above.

Coverage	Peptide									
Q09666 AHN	IK_HUMAN Neurobla	ast differentiation-a	issociated protein A	HNAK OS=Homo s	apiens GN=AHNAK	PE=1 SV=2			PTM FILTER	TOOLS
1	MEKEETTR EL	LLPNWQGSGS	HGLTIAQRDD	GVFVQEVTQN	SPAAR TGVVK	EGDQIVGATI	YFDNLQSGEV	TQLLNTMGHH		- 1
81	TVGLKLHRKG	DRSPEPGQTW	TREVFSSCSS	EVVLSGDDEE	YQRIYTTKIK	PRLK sedgve	GDLGETQSRT	ITVTRRVTAY		
161	TVDVTGREGA	KDIDISSPEF	K IKIPR helt	EISNVDVETQ	SGKTVIR lps	GSGAASPTGS	AVDIRAGAIS	ASGPELQGAG		
241	HSKLQVTMPG	IK VGGSGVNV	NAKGLDLGGR	GGVQVPAVDI	SSSLGGRAVE	VQGPSLESGD	HGK IKFPTMK	VPKFGVSTGR		
321	EGQTPKAGLR	VSAPEVSVGH	K ggkpgltiq	APQLEVSVPS	ANIEGLEGKL	KGPQITGPSL	egdlglk gak	PQGHIGVDAS		
401	APQIGGSITG	PSVEVQAPDI	DVQGPGSKLN	VPKMKVPKFS	VSGAKGEETG	IDVTLPTGEV	TVPGVSGDVS	LPEIATGGLE		
481	GKMKGTKVK	PEMIIQKPKI	SMQDVDLSLG	SPK LKGDIKV	SAPGVQGDVK	GPQVALKGSR	VDIETPNLEG	TLTGPR LGSP		
561	SGKTGTCR IS	MSEVDLNVAA	PKVK ggvdvt	LPR VEGKVKV	PEVDVRGPK V	DVSAPDVEAH	gpewnlk MPK	MK mptfstpg		
641	AKGEGPDVHM	tlpk gdisis	GPK VNVEAPD	VNLEGLGGKL	KGPDVK lpdm	SVKTPK ISMP	dvdlhvk gtk	VKGEYDVTVP		
721	KLEGELKGPK	VDIDAPDVDV	HGPDWHLKMP	KMKMPKFSVP	GFKAEGPEVD	VNLPKADVDI	SGPK IDVTAP	DVSIEEPEGK		
801	LKGPKFKMPE	MNIKVPKISM	PDVDLHLKGP	NVK geydvtm	pk veseik vp	DVELKSAKMD	IDVPDVEVQG	PDWHLK MPKM		
881	KMPKFSMPGF	KAEGPEVDVN	LPKADVDISG	PK VGVEVPDV	NIEGPEGKLK	GPKFKMPEMN	IKAPK ismpd	VDLHMK GPK V		
nalvsis//proteins			989							*

Regions in the protein sequence that are covered by supporting peptides are displayed in bold font with a grey background. Confident modifications and mutations identified in supporting peptides are displayed as icons above the protein sequence. Modifications are represented by colored icons with the initial letter of its modification name. If a residue is modified by more than one modification in the same supporting peptide, "*" is used instead of a letter. Mutations are represented by black-framed icons with a letter indicating the mutated residue. Placing the cursor over a modification icon shows the full name of the modification/mutation. The number above a modification icon

indicates the position in the protein sequence. Placing the cursor over an amino acid on the sequence shows the position of the amino acid in the sequence.

The de novo only peptides are the confident de novo sequence tags that remain unidentified by the identification search algorithms. To report a de novo only peptide, the De novo score must be equal to or better than the specified threshold. Meanwhile, the score of the spectrum's best identification search result should be no greater than the specified -10lgP threshold. If we enable the de novo only tag in the parameter setting, we can see the de novo only sequence with grey line below.

The supporting peptides can be shown as colored bars under the protein sequence. Placing the cursor over a bar shows detailed information of the peptide. Left clicking on a bar will pop up the annotated spectrum associated with the supporting peptide.

6.4.5. Supporting Peptides

The Peptide table shows a list of supporting peptides for the selected protein. This table is similar to the peptide table in the Peptide View, except that "Accession" is excluded and the columns, "Unique", "Start", "End", and "Found by" are included.

- Unique: This shows whether the peptide is a unique supporting peptide to the protein group.
- Start: This shows the peptide's starting position in the protein.
- End: This shows the peptide's ending position (inclusive) in the protein.
- Found by : The name of the PEAKS search workflow that identifies the peptide. This can be either DB Search or Library search.

Refer to Section 3.5.2, "Peptide Table" for the descriptions of other columns.

Double click a supporting peptide, it will go to that peptide sequence in the peptide page

6.5. Peptide View

The Peptide Table shows the filtered peptide identification results. Click on a peptide row to show the spectrum below the peptide table.



6.5.1. Peptide View Filter

Click the 👩 button, the following Peptide Filter will pop up. It controls which peptides to display.

Peptide Filters			
Peptide sequence contains			
m/z ≅			
RT ≅			
Peptide sample area ≥			
Found By =			
No Filter 👻			
Include Contaminant Data			
Peptide Contains Modifications			
Name to search for			CLEAR
Carbamidomethylation Deamidation (NQ)	🗌 Oxida	ation (M)	
	RESET	CANCE	APPLY

- **Peptide sequence contains:** Enter an amino acid sequence, only peptides that contain that sequence will be displayed in the peptide table.
- m/z: Filter based on the expected m/z, the last significant digit is rounded.
- **RT:** Filter based on the expected RT, the last significant digit is rounded.
- **Peptide sample area:** Filters based on total peptide area found in each sample. If a peptide is found in multiple samples, there should be at least one sample pass this filter for the peptide to be included.
- **Peptide Contains Modifications:** Check boxes are provided for each identified modification. Peptides that contain the selected PTMs will be shown.
- Found by: filter peptides found in DB search or Spectral Library search.

6.5.2. Peptide Table

The Peptide Table shows the filtered peptide identification results. Each row in the table is a peptide identification represented by its highest-scoring PSM. The table is sorted by the -10lgP value. In the Protein list drop-down window of each peptide, after clicking any of the protein, the protein page will be opened.

The columns Peptides, -10LgP, m/z, RT, Accession are sortable, i.e. Clicking any of these header can sort the whole table based on that column.

The columns in the Peptide table are:

- **Peptide:** The amino acid sequence of the peptide, as determined in PEAKS Search. A modified residue is followed by a pair of parentheses enclosing the modification mass.
- **-10lgP:** The peptide -10lgP score. The score indicates the scoring significance of a peptide-spectrum match.
- Mass: The monoisotopic mass of the peptide.

- Length: The number of amino acids in the peptide sequence.
- **ppm:** The precursor mass error, calculated as $10^{6} \times (\text{precursor mass} \text{peptide mass}) / \text{peptide mass}$.
- m/z: The precursor mass-to-charge ratio.
- **RT:** The retention time (elution time) of the spectrum as recorded in the data.
- **#RT:** The retention time difference between the query and predicted spectrum.
- Scan: The scan number of the spectrum that matches the peptide sequence with the highest -10lgP.
- **#Spec:** The number of MS2 spectra assigned to the peptide.
- **PTM:** The types and the numbers of modifications present in the peptide shown in color-coded icons.
- Ion Intensity: The relative ion intensity (%) of site-determining fragment ions for modification sites.
- Accession: The accession number of the highest-scoring protein containing this peptide.
- Precursor Id: This column is only available for TimsTof data, it shows precursor's id from the raw data.
- 1/k0: This column is only available for TimsTof data, it shows the ion mobility range of for the precursor.

6.5.3. Peptide Global Comparative View

"Peptide Global Comparative View" provides detailed comparisons among different samples regarding the identified peptides. To Show/Hide Columns, click III on top-left of the peptide table. From the pop-up window, users can add and remove area and spectral count columns. In the peptide table, the selected columns will be displayed and can be exported. If "Specified Sample" is selected these columns are automatically displayed.

Select Display Columns and Sample	es							
Select Columns								
🗸 Area 🗸 #Spec								
Select Samples								
Name to search for	CLEAR							
SAMPLE NAME								
✓ S1								
✓ \$2								
✓ \$3								
✓ W1								
✓ W2								
✓ W3								
CANCE	EL SAVE							
Note

PEAKS Online provides users the option to calculate peptide abundance using the MS1 feature area calculation or using the MS1 feature intensity. By default, area is used. The default can be changed by the administrator.

- Area by sample: The total area of all peptide features matched to spectra that identified the peptide. Area calculations are separated into multiple columns if multiple samples were included in the run.
- **#Peptide-spectrum matches by sample:** The number of peptide spectrum matches associated with a peptide separated by sample.

6.5.4. Peptide Menu

The peptide menu is only shown when "Show Results for Specified Sample" is selected in the filter section. The peptide menu displays the information of the peptide sequence that is being displayed in the spectrum. If a feature is associated to more than one tandem scan, click the "FRACTION", "FEATURE" or "PSM" button to view the Spectrum Annotation of the other PSMs.



6.5.5. Peptide-Spectrum Matches

This section displays the spectrum annotation and other information about the peptide selected in the Peptide table. When multiple spectra are matched to the peptide, only the information about the top-scoring PSM is shown by default. The interface of this section is identical to the Spectrum Annotation in the PEAKS Spectral Library result. For detailed instructions, refer to Section 3.5, "Spectrum Annotation" in Chapter 5, Peptide De Novo Sequencing.

6.5.6. DIA Database Search Mirror Plot

Similar to Library Search, pressing the button will display a mirror plot instead of the traditional spectrum view.



This view will display the ions from the query spectrum compared to the predicted spectrum generated from the database sequence.

6.5.7. Error Map

The "Error Map" shows the mass errors of the matched fragment ions. The m/z ratio is displayed on the x-axis and the error is listed on the y-axis in Daltons. Each matched fragment ion is represented by a dot. The ion types displayed can be configured in "Ion Annotation Settings".

6.5.8. Ion Table

The "Ion Table" shows the calculated mass of possible fragment ions based on the ion types selected in the Ion Annotation Settings. If a fragment ion is found in the spectrum, its mass value is displayed in color. N-terminal ions are shown in blue and C-terminal ions are shown in red. A fragment ion is found when there is a matching peak within the mass error tolerance, as defined in the *de novo* sequencing parameters.

6.5.9. Precursor Profile

An eXtracted Ion Chromatogram (XIC) chart that displays the shape of the selected peptides precursor.

6.5.10. Fragment Ion XIC

An eXtracted Ion Chromatogram (XIC) chart that displays the shape of each identified fragment ion across all scans for the selected PSM. Below the XIC charts is a slider to adjust the correlation threshold of the fragment ion with the selected peptide precursor. Increasing the slider value will filter out fragment ions that are below the threshold.

6.5.11. TimsTOF In Frame Mode

For timsTOF data only, there is an additional "In Frame Mode" button, =, above the spectrum view. Clicking this button will change the annotated spectrum view from "frame mode", the default view, to "in frame mode". To

switch back click the "Frame Mode" button, The standard, "Frame Mode" displays the scans merged over the CCS dimension. The "In Frame Mode" displays the scans merged over the retention time dimension.

7. DIA *De Novo* Analysis Workflow and Parameter Settings

PEAKS DIA *De Novo* can be launched as a standalone workflow, or launched on top of a PEAKS Library Search, PEAKS DIA Database Search, or PEAKS Library Search + PEAKS Database Search workflow. An analysis report is generated to show the *De Novo* search results, combined with both the library search results and database search results if run on top of a PEAKS Library Search or PEAKS Database Search.

To conduct a DIA *De Novo* search workflow, follow the steps below:

- 1. Refer to Chapter 7, Section 3, Spectral Library Search Workflow and Parameter Settings, Section 3, "Spectral Library Search Workflow and Parameter Settings", Step 1-3 to create a new project and add data. If you wish to run the De Novo Search on top of a Library Search, follow Step 4 as well. If you wish to run the De Novo Search on top of a DIA Database Search follow the steps in Section 2, "Database Search Workflow and Parameter Settings".
- 2. Check the checkbox next to De Novo Parameters beneath Database Search. Uncheck the checkbox next to Library Search Parameters if you do not wish to run a Library Search before the De Novo Search.

De Novo Parameters	Use Database Search Parameters

3. Set the De Novo Parameters. See details in Section 7.1-7.6 below.

Enzyme: Specified by each sample*		
Fixed Modifications:	Variable Modifications:	
Carbamidomethylation	Oxidation (M)	
ADD/REMOVE MODIFICATION	Max Variable PTM Per Peptide: 2	
Report Filter		
PSM/Peptide		
○ PSM -10LgP ≥	DR (%) 1 . <td></td>	
Protein		
	Group FDR (%) Proteins Unique Peptides ≥ 1 ;	
Proteins -10LgP ≥ 20 Protein G		
● Proteins -10LgP ≥ 20 ○ Protein G De Novo		

4. Click SUBMIT button to start the search.

7.1. Precursor and Fragment Error Tolerance

Precursor mass: This enables the selection of a precursor mass error tolerance. PEAKS Online DB Search identifies peptides within the precursor mass tolerance, which is the allowable m/z shift between the theoretical value of the peptide versus the observed value of the precursor ion. Precursor mass tolerance can be specified in either Da or ppm. **Fragment ion**: This enables the selection of a fragment ion m/z tolerance, which is the allowable m/z shift between the theoretical value of the fragment ions of the peptide versus the observed value. PEAKS Online DB Search uses this value when scoring peptide-spectrum matches. PEAKS Online considers that a fragment ion is matched if the calculated m/z is within the tolerance. Fragment mass tolerance can be specified in Da.

Error tolerance parameters need to be set consistently with the mass accuracy of the instrument.

Note

When the samples in a project are generated by different instruments and "default" is chosen from the Predefined parameters drop-down menu, the default parameters of the first sample are loaded. However, it is recommended to set the tolerance parameters according to the least accurate instrument.

7.1.1. Optimize Tolerance

When enabled, PEAKS Online will determine the optimal values to use for Precursor and Fragment Mass Error Tolerance for each fraction in your analysis. These values will replace the values provided in the previous Precursor and Fragment Mass Tolerance sections.

7.2. Enzyme Settings

Enzyme : PEAKS Online digests the protein database *in silico* to generate peptide candidates. It is necessary to specify the enzyme for protein digestion from the Enzyme drop-down menu.

Note

When the selected dataset is digested with different enzymes, "Specified by each sample" allows samples to be analyzed separately using their respective enzymes specified during project creation.

Note

"None" is a special enzyme digest rule allowing non-specific cleavage at both ends of the peptide. The "None" digest rule can cut at every residue, generating peptides with lengths up to 65 amino acids. Its usage is recommended when no digestion enzyme was used or when the digestion enzyme exhibits a high degree of non-specificity.

7.3. PTM Settings

Click "ADD/REMOVE MODIFICATION" button to open the "Select Modifications" dialogue and specify the fixed PTMs and a few (less than 11) common variable PTMs expected in the sample. If the required modification is not listed, the modification can also be created here. Additionally, the maximum number of variable PTMs per peptide can be specified and a value of less than 4 is recommended. A fixed modification forces all instances of applicable residues to be modified. A variable modification gives the option for the residues to be modified.

Note

For a specific residue, only one fixed modification is allowed.

Note

To prevent long search times, select only the most frequent PTMs in the sample for PEAKS DB, and check the PEAKS PTM option to search for unspecified modifications.

7.4. De Novo Filter

The report filter section is a mandatory section to filter out some results under the user-defined filter. Any changes to the filter will create a new analysis with old parameters and update the results and statistics.

• **Denovo ALC (%):** This filter will filter out any De Novo candidates with an average confidence less than the specified value.

8. Understanding PEAKS DIA *De Novo* Sequencing Results

When *de novo* sequencing is done, you can access the PEAKS Online project from the "My Projects" page. The analysis report for *de novo* search tools are presented in two tabs:

- Summary: This shows an outline of the PEAKS *de novo* search results with key statistics. The overall quality of the experiment can be examined and the filters for *de novo* ALC can be adjusted. This page shows after the search is done.
- De novo: This shows a list of peptide sequences that pass the ALC threshold detected by *de novo* sequencing.

The *de novo* view displays the *de novo* sequencing results. The table on the top displays *de novo* sequences, while the bottom section provides additional information about the peptide-spectrum match. The results can be filtered and exported.



8.1. Result Filtration

A PEAKS Online *de novo* sequencing result can be filtered by Average Local Confidence (ALC) score. Result for different samples can be specified at "Specified Sample" filter. Low quality *de novo* sequences can be filtered out by specifying a minimal threshold of the ALC score. The purpose of filtering is to remove poor sequences in which residues are incorrectly identified. By default, the ALC threshold is set to 50%. After changing the threshold, click

APPLY to apply the new filter. The result in the DE NOVO result view will be updated accordingly.

A PEAKS Online *de novo* sequencing results display based on the sample selected in the "specified sample" dropdown menu. The purpose of this filtering is to limit the quantity of *de novo* sequencing results displayed in the table. The specified sample can be changed through the drop-down list, click **APPLY** to apply the new filter. The result will be updated accordingly.

8.2. Exporting *De Novo* Results

The *de novo* results can be exported as "Denovo.csv" or "Denovo CSV with all Candidates". These export files can be downloaded from either the Denovo Results page or the export page.

• **Denovo CSV**: the filtered *de novo* results shown are exported. It can also be downloaded from the export page as shown in the following figure.

• **Denovo CSV with all candidates**: for each MS2 spectrum, denovo results of the 5 highest ALC scores are generated. *De novo* sequences that pass the ALC score filter are all included in this file. It can also be downloaded from the export page shown in the following figure.

In the Denovo results page, above the peptide table there are two export options: "Denovo CSV", and "Denovo CSV with all Candidates".

In the export page, there are 3 files which can be exported: "Summary Table", "Denovo CSV", and "Denovo CSV with all Candidates".

Denovo	
Summary Table	
Denovo CSV	
Denovo CSV with all Can	didates

• Summary Table: the overview number of each sample for the fields MS2, PSM, PSM (alc>30), PSM (alc>50), PSM (alc>70), #PSM/#MS2 and notes.

8.3. Summary View

The "Summary" view reports key statistics as an overview of the result.

< Summar	Filters De Novo ALC (%	•)≥ 50.0 ‡	APPLY Created:	Jul 23 2021, 01:49:13 pm			
y De Novo				Summary	Statistics	÷	- <u>1-2 of 2 ▼</u> →
	Sample Name	# MS2	# PSM	# PSM (alc > 30)	# PSM (alc > 50)	# PSM (alc > 70)	# PSM / # MS2
	All	20483	12450	8029	3637	1133	61%
	Sample 1	20492	12450	8030	9697	1199	610

8.3.1. Result Statistics

This table of the summary page shows the overview number of each sample for the following fields:

- **# MS2**: The number of MS/MS scans from the data.
- **# PSM**: The number of peptide-spectrum matches.
- # PSM (alc>30): The number of peptide-spectrum matches with an ALC score greater than 30.
- # PSM (alc>50): The number of peptide-spectrum matches with an ALC score greater than 50.
- # PSM (alc>70): The number of peptide-spectrum matches with an ALC score greater than 70.
- **#PSM / #MS2:** The number of peptide-spectrum matches per MS2 scan.

Note

ALC score is the average local confidence score. ALC is calculated as the total of the residue local confidence scores in the peptide divided by the peptide length.

8.3.2. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

```
2 data files produced no results. (Expand to see list of zero result data files)
```

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.

Potentially Problematic Data Sets	~
Sample 3 All fractions have no result.	
Sample 4 All fractions have no result.	

If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics

No Results Found

8.4. Peptide Table

The peptide table shows the filtered *de novo* sequencing results. For each scan, the best candidate is displayed.

The following list describes the contents in each column:

- Scan: The scan number. The scan number is a unique index for tandem mass spectra in the data. If the scan numbers could be read from the data file, the scan numbers that match with the instrument data file will be shown.
- **Peptide**: The amino acid sequence of the peptide as determined by *de novo* sequencing. A modified residue is followed by a pair of parentheses enclosing the mass of that modification.
- ALC (%): The average local confidence. ALC is calculated as the total of the residue local confidence scores in the peptide divided by the peptide length.
- Length: The number of amino acids in the peptide sequence.
- **m/z**: The precursor mass-to-charge ratio.
- **z**: The precursor charge.
- **RT**: The retention time (elution time) for the spectrum as recorded in the data.
- Area: The area under the curve of the peptide feature found at the same m/z and retention time as the MS/MS scan. This can be used as an indicator of the abundance of the peptide.

- Mass: The monoisotopic mass of the peptide.
- ppm: The precursor mass error, calculated as 10E6 × (precursor mass peptide mass) / peptide mass.
- PTM: The types and the numbers of the modifications present in the peptide shown as color-coded icons.
- Mode: The fragmentation mode in which *de novo* sequencing is performed by the algorithm.
- Accession: The accession number of the highest-scoring protein containing this peptide. If a de novo only results surpasses the ALC% threshold set on the summary page.

Confidence Scores: Amino acids in *de novo* sequences are color-coded according to their local confidence scores. Red represents a very high confidence (greater than 90%); purple represents a high confidence (80 to 90%); blue represents a medium confidence (60 to 80%); and black represents low confidence (less than 60%). Hover the cursor over the sequence to view local confidence scores of each individual amino acid.

Mass Tags: The low confidence residues can be displayed as mass tags by adjusting the local confidence score threshold using the **button**. If the score is set at 0, all of the amino acids in the peptide sequences will be displayed. Increasing the threshold will reduce residues below the threshold to mass tags. When low confidence residues are reduced to mass tags, the remaining residues become sequence tags. The length of the longest tag is called the tag length and reported in the export files.



Filter Peptides: The peptide table can be filtered by selecting the **o** button and prompting the "Denovo Filter" window. The peptides can be searched by sub-sequence, scan number, precursor Id, m/z, and retention time.



8.5. Spectrum Annotation

The spectrum annotation displays a graphical representation of the peptide-spectrum match.

Moving the cursor over the peptide sequence in the spectrum will show the mass transitions for a particular amino acid residue.

The spectrum annotation panel provides convenient ways to zoom and navigate within the spectrum:

• Zoom in/out on m/z: Place the cursor on the point in the spectrum you would like to zoom in on. Alternatively, place the cursor in the Error (da) figure and scroll the mouse wheel.

- Increase/Decrease peak intensity: Place the cursor on the Intensity axis and scroll the mouse wheel.
- See the whole spectrum: click the 1:1 button.

Click 🕙 to open the Ion Annotation Settings dialogue and change the ions to be annotated in the spectrum. To reset the settings to the PEAKS Online defaults, use "RESET DEFAULT" button.



The (*) button will switch the y-axis to relative intensity. Once selected, scrolling over peaks will display the relative intensity instead of the absolute intensity.

The *button* will switch the y-axis to absolute intensity. Once selected, scrolling over peaks will display the absolute intensity instead of the relative intensity.

8.6. Spectrum View

For DIA De Novo results, the annotated spectrum view can be displayed in three different formats.

8.6.1. Raw Scan



Click on the *button to display the raw scan.*

This view will display the raw scan from the data file, with the identified ions annotated.

8.6.2. Mirror Plot

Click on the ^{###} button to display the mirror plot.



This view will display the ions from the de novo candidate compared with the ions from the predicted spectrum.

8.6.3. Pseudo Scan



8.7. Error Map

The "Error Map" shows the mass errors of the matched fragment ions. The m/z ratio is displayed on the x-axis and the error is listed on the y-axis in Daltons. Each matched fragment ion is represented by a dot. The ion types displayed can be configured in "Ion Annotation Settings".

8.8. Ion Table

The "Ion Table" shows the calculated mass of possible fragment ions based on the ion types selected in the Ion Annotation Settings. If a fragment ion is found in the spectrum, its mass value is displayed in color. N-terminal ions are shown in blue and C-terminal ions are shown in red. A fragment ion is found when there is a matching peak within the mass error tolerance, as defined in the *de novo* sequencing parameters.

8.9. Precursor Profile

An eXtracted Ion Chromatogram (XIC) chart that displays the shape of the selected peptides precursor.

8.10. Fragment Ion XIC

An eXtracted Ion Chromatogram (XIC) chart that displays the shape of each identified fragment ion across all scans for the selected PSM. Below the XIC charts is a slider to adjust the correlation threshold of the fragment

ion with the selected peptide precursor. Increasing the slider value will filter out fragment ions that are below the threshold.

8.11. TimsTOF In Frame Mode

For timsTOF data only, there is an additional "In Frame Mode" button, =, above the spectrum view. Clicking this button will change the annotated spectrum view from "frame mode", the default view, to "in frame mode". To

switch back click the "Frame Mode" button, The standard, "Frame Mode" displays the scans merged over the CCS dimension. The "In Frame Mode" displays the scans merged over the retention time dimension.

9. FAIMS Data

Field asymmetric waveform ion mobility spectrometry (**FAIMS**) analyzes complex mixtures and separates the ions based on their mobility. PEAKS Online has the ability to load FAIMS data and analyze it using *de novo* sequencing, database search (PEAKS DB), database search with unexpected modification identification (PEAKS PTM), sequence variant identification (SPIDER) search, spectral library search, and labelled and label free quantification.

The analysis FAIMS mass spectrometry datasets contains the CV values used when collecting the spectra. The CV-related information can be found at the following places for a spectral Library analysis.

• The Venn diagram on the summary page.

Venn diagram overlaps between different CVs in the search for all proteins, top proteins, and peptides are shown on the summary page.

- The additional **CV** column on the supporting peptide page
- The additional CV column in the exported Protein-Peptide CSV
- The additional CV column on the peptide page
- The additional CV filter on peptide page

The filter for peptide page contains one additional option for CV, which selects the displayed scans based on the selected CV values

• The additional CV column in the exported *Peptide CSV*

Chapter 8. Label Free Quantification (LFQ)

1. Overview

Label free quantification can be used as a quick profiling tool to examine the relative abundance of proteins in large proteomic datasets. In PEAKS Online this quantification method is based on the relative abundance of peptide features detected in multiple samples. PEAKS Online performs feature detection separately on each sample, and uses an EM (expectation-maximization) algorithm to detects and deconvolute overlapped features. Importantly, features of the same peptide from different samples are reliably aligned together (i.e. matched between runs) using a high performance retention time alignment algorithm to improve overall protein quantification rates.

Note

For details of the retention alignment algorithm, refer to the paper " A Combinatorial Approach to the Peptide Feature Matching Problem for Label-Free Quantification", Bioinformatics, 2013, 10.1093.

The usage of this workflow is outlined below.

- 1. Refer to Chapter 5, Peptide De Novo Sequencing, Section 2, "*DeepNovo/De Novo* Analysis Workflow and Parameter Settings", Step 1-6 to create a new project and add data.
- 2. Label Free quantification can be run at the start of a new project, or on PEAKS Database Search results or PEAKS Spectral Library Search results.
 - To run at the start of a new project, select the appropriate Peaks Q workflow after you have added your DDA or DIA data
 - To run Label Free Quantification on existing PEAKS Database Search results, choose the Create New Analysis from within the project and then select the "Peaks Q" workflow in the DDA section
 - To run Label Free Quantification on an existing PEAKS Spectral Library Search results, choose the Create New Analysis from within the project and then select the "Quantification" workflow in the DIA section
- 3. Set Search Parameters.
 - If running LFQ on a DDA DB Search analysis, refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 2, "Database Search Workflow and Parameter Settings" for details.
 - If running LFQ on a DIA Spectral Library Search analysis, refer to Chapter 7, DIA Streamlined Identification, Section 3, "Spectral Library Search Workflow and Parameter Settings" for details.
- 4. Set Quantification Parameters: the LFQ-specific parameters will be shown at the bottom of the Project Submission page. Group your samples using the different add options (e.g. add to a new group, add to existing group, etc.). The names and colours for each groups can be edited by clicking on the name or coloured square, respectively. Next, set your Mass Error and Retention Time Shift tolerance, and other settings for match between runs. See Section 2 below for further details. Label-free quantification will be performed on the combined identification results from all samples in the analysis.

Sample Groups	:			
	→°	Group Group 1 Group 2	Color	× ×
	<u>→122</u>			
Match Between Runs	+t <u>n</u>			REMOVE ALL
Mass Error Tolerance: 15 Tolerance Unit: PPM	▼ Reter	ntion Time Shift Tolerance(min):	3	Auto Detect
Feature Intensity ≥: 0 :				
Feature Intensity ≥: 0 ⋮ RT Range: 0 ⋮ ≤ RT ≤	:	Base Sample: Sar	nple 1 🔹	

- 5. Thresholds and analysis parameters: Set Peptide Feature, Protein, and Normalization parameters by selecting the edit icon (pencil icon) next to each category. Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 2.8, "PSM Filter" for details as well as Section 2 below.
- 6. Review project settings and parameters then click Submit button. Name the analysis (or use the default option) and click OK to start the analysis.

2. Setting Parameters

For more information on DB search parameters, refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 2, "Database Search Workflow and Parameter Settings".

The following parameters are available when the Label Free Quantification workflow is selected.

- Mass Error Tolerance: The mass shift of a precursor ion between different runs. This can be set similar to the precursor mass error tolerance used for database search.
- **Retention Time Shift Tolerance:** The retention time shift tolerance is the maximum elution time difference that is considered for the quantification of an identified peptide after RT alignment.
 - Auto Detect: Let PEAKS Online detect the optimal Retention Time Shift Tolerance for your data based on the observed trend.

Retention Time Shift Tolerance (min):	1	Auto Detect 🔽
---------------------------------------	---	---------------

• **CCS Error Tolerance (if applicable):** The maximum collision cross section (CCS) differential for features to transfer IDs.

Note

This parameter is only available when the analysis contains timsTOF data.

- Feature intensity: Set a value for the minimum intensity a peptide feature must exhibit.
- **RT Range:** Feature vectors within the selected retention time range will be included in the results while everything else will be removed.
- **Base Sample:** The base sample will act as the reference. The peptide and protein abundance corresponding to this reference label will be the denominator in all of the abundance ratio calculations.
- Peptide Feature :



- Avg. Area: The minimum average area of all peptide features associated with the peptide. Features with higher abundance have been shown to be more reproducible across replicates. Only peptide features with an average area above this threshold will be used for quantification.
- **Quality:** The quality score for the peptide feature. Factors that affect the quality score include m/z difference, RT difference, XIC shape similarity, and the feature intensities.
- **Peptide ID Count:** This filter sets the minimum limit for how many times a peptide must be identified within a group to be included in the LFQ results.
- **Charge between:** Only peptides with feature vectors that all fall within this range will be used in protein quantification.
- Have at least <u>confident samples per group</u>: This filter sets the minimum number of sample(s) per group that the peptide feature has an abundance value. A peptide is more quantifiable when it is detected in more samples.
- Use in group Coefficient of Variation Filter : For each feature vector (set of peptide features associated with a peptide), at least one group must have a coefficient of variation (CV) less than the CV threshold. The CV threshold is automatically selected by the software based on the current data set.
- Protein Filter:

Ductoin Cilton

	Proteir	Fille	:			
Significance Method	ANOV	A				
Modified Form Exclusion						
Remove Outlier	\checkmark					
Use Top	3	:				
Significance ○ FDF	R(adjusted	p-value	e)			
Significance ≥	0	:				
Fold Change Between	1	:	and	64		
Used Peptides ≥	1	:				
					01.005	0.01/5
					GLUSE	SAVE

- **Significance Method** : Protein abundance is used for the significance testing calculation. When there are replicates in each group, ANOVA is suggested to be used for significance testing. For moredetails on ANO-VA, refer to the following paper: "On the comparison of several mean values: an alternative approach ", Biometrika, 1951, 38(3/4): 330-336
- **Modified Form Exclusion:** The expression level of modified peptides might be different than the overall expression level of a protein. In such cases, including modified peptides for protein ratio calculation might lead to incorrect results. By checking this box, if a peptide has modified and unmodified forms, both will not be used as candidates for protein ratio calculation, if a peptide only has one modified form, it is still a candidate, but if a peptide has more than one modified form, all forms will not be used as candidates
- **Remove Outliers** : This filter removes supporting peptides that have a different variation trend compared to others.
- Use Top # peptides : This is the number of quantifiable supporting peptide that will be used to calculate the protein profile. Three is the default setting.
- **Significance:** Only protein groups with a significance above this threshold will be listed in the "Protein" view. The significance score is calculated as the -10log of the significance testing p-value (e.g. Significance score of 20 is equivalent to a p- value of 0.05). PEAKS provides the significance testing method: ANOVA.
- **FDR:** Adopt the Benjamin-Hochberg method to adjust the p-value to the false discovery rate for all protein groups that have already passed the other filters. Only protein groups with significance scores passing the calculated FDR will be listed in the "Protein" view. Either this or "Significance" can be selected to set a significance threshold.
- Fold Change: Only protein groups below the lower threshold or above the upper threshold value will be listed in the "Protein" view.
- Used Peptides: Only protein groups that were quantified with this number of peptides (or more) will be listed in the "Protein" view.
- Normalization :

Normalization Methods

0	Use TIC
0	Manual Input
0	Use Internal Standard Proteins
۲	No normalization

• Use TIC: By default, PEAKS Online uses a sliding-window method to calculate retention time-dependent normalization factors from the total ion current (TIC) of each sample. Normalized abundance for each sample is calculated from the raw abundance divided by the normalization factor for that sample.

CLOSE

SAVE

- **Manual input:** This option performs the same default calculation as auto normalization ("Use TIC") but also allows for the manual change of each sample's expected ratios. This option should be used if the expected ratio is not 1:1 for each sample as the software attempts to normalize to a 1:1 ratio.
- Use internal standard proteins: This option displays a list of identified proteins (when a database or library search has been completed) that can be selected to create a normalization ratio. A search bar is included to enable the quick location of select proteins. Normalization by internal standard proteins is performed using local retention-time dependent normalization.
- No Normalization: All samples are automatically assigned Factors of 1.0.
- Sample Groups: LFQ requires sample grouping. All available samples are listed in the left window. After selecting a sample, it can be added to a new group by clicking the → button or to an existing group by clicking the → mathematical button and selecting that option from the drop-down menu. If you select multiple files, you can add each one individually to create multiple new group by clicking the → mathematical button. If you want group files by delimiter or regEx, you can click the → button. Samples can be removed individually or removed all at once by clicking the ← button.

3. Understanding the LFQ Results

After a PEAKS Online label free quantification is complete, the LFQ summary page will display automatically. There are also Protein and Peptide result tabs to view.



- **Summary:** The outline of the LFQ search results with statistics. This is the place to examine the overall performance of the experiment and adjust Protein filters.
- **Proteins:** The quantified proteins with a list of supporting peptide features for each protein.
- **Peptides:** The quantified peptides are listed in a table. Extracted ion chromatogram (XIC) and supporting peptide features are also displayed for each peptide.

3.1. Protein Filters

The results can be filtered using the Protein filter at the top of the Summary tab. Adjustments made here will affect the results displayed in the protein and peptide tabs.

3.1.1. Protein Filter

Click the 🧪 button and the following parameter dialogue will pop up.

Protein Filters				
Significance ○ F	DR(adjusted p-value)			
Significance	≥ 0			
Fold Change between	1	and 64		0 0 0
Used Peptides	≥ 1	0 0 0		
			CLOSE	SAVE

Filter settings used in the screenshot above means that no protein filter is applied, thus giving all quantifiable protein.

Alternately, parameters can be adjusted for additional analysis by selecting the analysis from the project list on the left hand side of the window, and then selecting Modify Parameters from the top right hand side of the Analysis Properties view. This will open all parameters for the analysis and allow for the modification of single or multiple settings. If changes are made using this method a new analysis is submitted and performed.

3.2. Summary View

The "LFQ Summary" view includes a result statistics table, a protein volcano plot, and an expression profile with candidate proteins clustered in a heat map. An "Analysis Description" can be added and displayed below the filter section.

3.2.1. Statistics of Filtered Results

Table 1 in the Summary page lists the filtered results of feature vectors and proteins.

3.2.2. The Volcano Plot for Proteins

A volcano plot combines a statistical test with the magnitude of the change enabling quick visual identification of those data-points that display large-magnitude changes that are also statistically significant. The volcano plot is displayed as Figure 2 in the results, and plots significance versus fold-change of the quantified proteins. Volcano plots show two thresholds:

- Horizontal broken grey line: The selected significance threshold.
- Vertical broken grey lines: The selected fold change thresholds.

Plotting points in this way results in two regions of interest in the plot: those points that are found towards the top of the plot that are far to either the left or the right side. These represent values that display large magnitude fold changes (hence being left or right of center) as well as high statistical significance (hence being towards the top). The ratio is the group ratio set with respect to the base group. If more than two groups are present, the max ratio is used instead.

Circles for the proteins that are above the set significance and fold change thresholds will be displayed in colour. Scrolling the cursor over a circle will show the protein info.

3.2.3. Protein Profile Heatmap

The heatmap is shown in Figure 1 of the results, and displays all the protein groups that pass the filters. The relative protein abundance is represented as a heat map of the representative proteins of each protein group. These representative proteins are clustered if they exhibit a similar expression trend across the samples within a group. The hierarchical clustering is generated using a neighbour-joining algorithm with a Euclidean distance similarity measurement of the log2 ratios of the abundance of each sample relative to the average abundance. If the number of proteins is too high, the heat map will not show in this case.

3.2.4. RT Shift & M/Z Shift

In Figure 3. Distribution Graphs: a) RT shift before vs after alignment. It displays the retention time shift before and after alignment

Distribution Graphs: b) plots the distribution of the m/z shift (in ppm) of the data

3.2.5. Percentage of missing values in each sample with or without ID transfer

Figure 4 shows the percentage of missing values before and after ID transfer for each sample. A missing value before ID transfer is defined as a null value in a sample where a peptide was assigned a feature in another sample with an identification. Missing values after ID transfer are null values when no feature could be found in a sample based in identifications found in other samples. This data is influenced by the Match Between Runs settings selected by the user when the workflow was setup.

3.2.6. Protein Area Coefficient of Variance.

Figure 5 shows a histogram of all protein covariance values. For this chart, protein covariance is calculated within a group.

3.2.7. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

```
2 data files produced no results. (Expand to see list of zero result data files)
```

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.

Potentially Problematic Data Sets	×
Sample 3 All fractions have no result.	
Sample 4 All fractions have no result.	

If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics

No Results Found

3.3. Protein View

The LFQ Protein page lists all the quantified proteins that pass the filter set on the LFQ Summary page, characterizes each protein at the amino acid level, and lists the supporting peptide features of each protein.

									2379 👻 🔺		
Accession	Significance	Coverage	#Peptides	#Unique	PTM . Sample	e Profile Group P	Profile Avg. Mass			1	
P0A6C8IARGB_ECOLI	107.67	2.71%	1	1			27160	Acetylglutamate kina:	se OS=Escherichia co 🛧	100	
P02753ups RETBP_HUMAN_UPS	87.86	81.56%	11	11 C	DO		20575	Retinol-binding protei	in 4 (Chain 19-201) - F	80	
P77473 PDEB_ECOLI	84.03	2.13%	1	1			58720	Probable cyclic di-GM	IP phosphodiesterase	8	
P08758ups ANXA5_HUMAN_UPS	73.91	78.06%	24	23 C C	D O		35806	Annexin A5 (Chain 2-	320) - Homo sapiens (2 00	
P25889 PREA_ECOLI	70.14	2.19%	1	1			45069	NAD-dependent dihyd	dropyrimidine dehydro	i∰ 40	
P55957ups BID_HUMAN_UPS	67.60	68.72%	15	14 C (0 0		21995	6 BH3-interacting doma	ain death agonist (Cha		
P69905ups HBA_HUMAN_UPS	64.94	90.07%	20	18 C I	0 0		15126	i Hemoglobin subunit a	alpha (Chain 2-142) - I	20	1.50
P41159ups LEP_HUMAN_UPS	62.19	85.71%	26	25 C C	D O		16158	Eleptin (Chain 22-167)) - Homo sapiens (Hur	0	N.N.
P37760 RMLD_ECOLI	60.75	4.68%	1	1			32694	dTDP-4-dehydrorham	nose reductase OS=E 🗸	184 1/32 1/18 1/8	1/4 1/2 1 2 4 8 18
											Ratios
1	LIDCEENI	EDIECAL	VNY RES	SHORPLVT	VHGGGCVVDF	E LMKGINI.PVK	KKNGLEVTPA	DOIDIITGAL	AGTANKTLLA	WAKKHOTAAU	
I MMNPLIIKLG GV	TTDSFERT	DUDLOUD.		·				- X			, ODITODODOD
1 MMNPLIIKLG GV	GLIOPGSP	KLINSLI	ENG VL	DVVSSTGV	TDEGOLMNVN	A DOALTALAA	TIGADITIIS	DVSGILDGKG	OPIAFMTAAK	AFOLIFOGII	T TDGMIVKVNI
1 MMNPLIIKLG GV	GLAQPGSP	KLINSLL	ENG YLI	PVVSSIGV	TDEGQLMNVN	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	I TDGMIVKVNA
1 MMNPLIIKLG GV 111 KVTQLDEELG HV 221 ALDAARTLGR PV	GLAQPGSP DIASWRHA	KLINSLLH EQLPALFI	ENG YLI NGM PMO	PVVSSIGV GTRILA	TDEGQLMNVN	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	TDGMIVKVNA
1 MMNPLIIKLG GV 111 KVTQLDEELG HV 221 ALDAARTLGR PV	GLAQPGSP DIASWRHA	KLINSLLH EQLPALFI	ENG YLI NGM PMO	PVVSSIGV GTRILA	TDEGQLMNVI	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	I TDGMIVKVNA
1 MMNPLIIKLG GV 111 KVTQLDEELG HV 221 ALDAARTLGR PV	GLAQPGSP DIASWRHA	KLINSLLI EQLPALFI	ENG YLI NGM PMO	PVVSSIGV GTRILA	TDEGQLMNVN	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	I TDGMIVKVNA
11 MMNPJIIKLG GV 111 KVTQLDEELG HV 221 ALDAARTLGR PV	GLAQPGSP DIASWRHA	KLINSLLI EQLPALFI	ENG YLI NGM PMO	PVVSSIGV GTRILA	TDEGQLMNVN	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	E TDGMIVKVN2
1 MMNPHIRLG GV 111 KVTQLDEELG HV 221 ALDAARTLGR PV	GLAQPGSP DIASWRHA	KLINSLLI EQLPALFI	ENG YLI NGM PMO	PVVSSIGV GTRILA	TDEGQLMNVP	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	E TDGMIVKVNI
1 MMNPFIIKLG GV 111 KVTQLDEELG HV 221 ALDAARTLGR PV	GLAQPGSP DIASWRHA	KLINSLLI EQLPALFI	ENG YLI NGM PMO	PVVSSIGV GTRILA	TDEGQLMNVM	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	I TDGMIVKVNI
1 MMNPIIIKLG GV 111 KVTQLDEELG HV 221 ALDAARTLGR PV	GLAQPGSP DIASWRHA	KLINSLLI EQLPALFI	ENG YLI NGM PMO	PVVSSIGV GTRILA	TDEGQLMNV	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	I TDGMIVKVNI
I MMNPIIIKLG GV 111 KVTQLDEELG HV 221 ALDAARTLGR PV	GLAQPGSP DIASWRHA	KLINSLLI EQLPALFI	ENG YLI	PVVSSIGV GTRILA	TDEGQLMNVI	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	C AEQLIEQGII	I TDGMIVKVNA
1 MMNFFIILG GV 111 KVTQLDEELG HV 221 ALDAARTLGR PV	GLAQPGSP DIASWRHA	KLINSLL EQLPALFI	ENG YLI	PVVSSIGV GTRILA	TDEGQLMNVI	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	E TDGMIVKVNA
I MRNFJILLG GV 111 KVTQLDEELG HV 221 ALDAARTLGR FV gen Peptide	GLAQPGSP DIASWRHA	KLINSLLI EQLPALFI	ENG YLI	PVVSSIGV STRILA	TDEGQLMNVI	A ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	E TDGMIVKVNJ
1 MMNFJIILG GV 111 KVTQLDEELG HV 221 ALDAARTIGR PV 90 Peptide	GLAQPGSP DIASWRHA	KLINSLI EQLPALFI	ENG YLI	PVVSSIGV STRILA	TDEGQLMNV	N ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	Childboost TDGMIVKVNI 4 1-10f1 *
I MINFFIILE GV 111 KVTQLDEELG HV 221 ALDAARTLGR FV per Peptide	GLAQPGSP DIASWRHA	KLINSLLI EQLPALFI	ENG YLI NGM PMO	PVVSSIGV STRILA	TDEGQLMNVN	A ADQAATALAA	TLGADLILLS	DVSGILDGKG	QRIAEMTAAK	AEQLIEQGII	← 1-1of1 + Mor Pario

3.3.1. Protein Table

The quantified proteins are listed in the protein table with parsimonious proteins grouped together.

This protein table is the same as the other protein tables found in the PEAKS DB, PEAKS PTM, and SPIDER results except the following columns:

- **Significance:** The significance score is calculated as the -10log of the significance testing p-value (e.g. Significance score of 20 is equivalent to a p- value of 0.05). PEAKS provides the significance testing method: ANOVA.
- **Sample Profile:** The protein abundance among the samples is depicted as a heat map. Hold the cursor on a profile to view the sample channels, abundances, and ratios with respect to the base sample.
- **Group Profile:** The protein abundance among the groups is depicted as a heat map. The Group Profile is determined by calculated the total abundance (area) of supporting peptides within a group. Hold the cursor on a profile to view the group channels, abundances, and ratios compared to the group that contains the base sample.

Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER) Section 3.4.2, "Protein Table" for more details of other entries of the protein table.

3.3.2. Protein Volcano Plot

The volcano plot facilitates identification of the significant proteins. This interactive volcano plot enables the selection of a protein in the table. Proteins located in the top-right and top-left sections of this plot (above the significance threshold and to the right and left of the fold change thresholds) are considered statistically significant. Green points are less abundant versus the reference group while red points are more abundant versus the reference group. Grayed points represent proteins currently filtered out of the result table based on the Peptide Filters set on the Summay tab.

Red and green protein markers can be clicked and selected. Upon clicking a point on the volcano plot, the corresponding protein in the Protein Table will be highlighted and more details are displayed below the Protein Table.

• Zoom in/out from Volcano Plot: Place the cursor on the volcano plot and drag a rectangle over the area of interest to Zoom in. By double clicking on the volcano plot you can zoom out and view the volcano plot with the default dimensions.



3.3.3. Coverage view

The coverage view characterizes the selected protein sequences at the amino acid level. Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 3.4.4, "Protein Coverage" for details.

Q99JP7|GGT7_MOUSE Gamma-glutamyltransferase 7 OS=Mus musculus GN=Ggt7 PE=1 SV=2

1	MAAENEASOE	SALGAYSPVD	YMSTTSFPRI.	PEDEPAPAAP	LEGEKDEDAE	LGDPDTDPDS	FLKSARLORL	PSSSSEMGSO
0.1	DOODIDEEDV	DDDCAAAAC		UEDOLMENME	URURIUMOTU	DODIDIDIDO	THROMADQAD	1 SSSSBIIOSQ
8 T	DGSPLRETRK	DPFSAAAALC	SCRÖDGTIAI	VTACLTFATG	VIVALVMQIY	FGDFØIFØØG	AVVIDASSCI	ALGMEVLSKQ
161	GSSVDAAVAA	ALCLGIVAPH	SSGLGGGGVM	LVHDIRRNES	HLIDFRESAP	GALREEALQR	SWDTKPGLLV	GVPGMVKGLH
241	EAHQLYGRLP	WSQVLAFAAA	VAQDGFNVTH	DLAHALAEQL	PPNASDRFLD	TFLPLGHPPL	PGSLLRRPDL	AEVLDILGTS
321	GPAAFYNGGN	LTLEMVAEAQ	HAGGVITEED	FSNYSALTEK	PVCGVYRGHL	VLSPPPPHTG	PALISALNIL	EGFNLTSLVS
401	REQALHWVAE	TLK ialalas	RLGDPVYDST	ITESMDDMLS	KVEAANFRGH	ISDSQAAPAP	LLPVYELDGA	PTAAQVLVMG
481	PDDFIVAMVS	CINDDECCCI	IMPOCTLLNS	QMLDFSWPNR	TANHSAPSLE	NSVQPGKRPL	SFLLPTVVRP	AEGLCGTYLA
561	LGANGAARGL	Position: 414~421	LSDS	LARGRLHPDL	OSNLLOVDSE	FTEEEIEFLE	ARGHHVEKVD	VLSWVHGSRR
6 4 1	TNTETTOURD	Significance: 123.35			~			
04 I	INTETIGAND	Avg. ppm: 1.0						
		Avg. Area: 1.94e+5						
		Max Ratio: 64.00						
		Group Profile: Group	Area Ratio					
		Group 10	.00e+0					
		Group 25	.83e+564.00					
		#Vector: 1						
		Peptide: IALALASR						

3.3.4. Peptides view

The supporting peptides assigned to the protein are shown in a separate tab beside the Protein Coverage view. The sequence can be clicked and a pop-up window will show up to display the feature details.

This table contains the following information of the quantifiable support peptides for the selected protein:

- **Peptide:** The amino acid sequence of the peptide, as determined in PEAKS Database or Spectral Library search. A modified residue is followed by a pair of parentheses enclosing the modification mass. All non-unique peptides and those not passing the filter will be excluded automatically.
- Used: Shows whether this peptide is used to calculate the protein abundance. The top three peptides with highest abundance are used to calculate the protein abundance.
- **Significance:** The -10lg of a p-value represents the likelihood that the observed change between conditions is caused by random chance. The peptides are first separated into groups based on similar quality score. Significance is then calculated for each quality score group, or bin, using a two-tail T-test that assumes log normal distribution but does not assume equal variance. The -10log p-value is then calculated from this result and then displayed in the significance column.
- Avg. ppm: A peptide can have one or more feature vectors. Each feature vector has its own ppm calculated from the mass error of each feature. The average mass error of a peptide is the average of mass errors of all the feature vectors. This column displays the average mass error in ppm.
- Avg. Area: Each feature in the feature vector has its own area under the XIC curve. The average area of a peptide is the sum of the average area of all features associated with that peptide.
- **Sample Profile:** The peptide abundance among the samples is depicted as a heat map. Hold the cursor on a profile to view the samples, the peptide areas in samples, and the corresponding ratios with respect to the base sample.
- **Group Profile:** The peptide abundance among the groups is depicted as a heat map. Hold the cursor on a profile to view the groups, the peptide areas in groups, and the corresponding ratios with respect to the group which contains the base sample.
- Group Area: This column shows the peptide area for a group. It is calculated by summing up the areas of feature vectors within a group
- Max Ratio: This shows the maximum of the fold change values compared to the reference sample.
- **RT Mean:** Each feature in the feature vector has its own retention time center. This column displays the average retention time center of the feature vector.
- #Vector: This shows the number of quantifiable feature vectors of a peptide.
- Start: This shows the protein position of the first residue of the peptide.
- End: This shows the protein position of the last residue (inclusive) of the peptide.

- **PTM:** The types and the numbers of modifications present in the peptide shown using color-coded icons. Scroll over each icon to see the modification name and mass.
- Feature Detail: Double click on a supporting peptide to go to the peptide page where the eXtracted Ion Chromatogram (XIC) curves are displayed (as seen below).



3.3.5. Protein View Filter

Click the **o** button on top-left of the protein table to open the protein filter and control which proteins to display.

Protein Filter	
Top Proteins All Proteins Protein accession/name contains	
Protein description contains	
Include Contaminant Data	
Protein Contains Modifications Name to search for	CLEAR
Deamidation (NQ) Oxidation (M)	Carbamidomethylation
	RESET CANCEL APPLY

- **Top Proteins**: If this is selected only the proteins with the highest number of unique peptides supporting them (within their protein group) will be displayed.
- All Proteins: If this is selected all proteins that satisfy the parameters set for PEAKS Protein Score(-10lgP) and number of unique peptides will be displayed.
- **Protein Sequence contains**: Enter a protein sequence, only proteins that contain that sequence will be displayed in the protein table
- **Protein description contains:** Enter a protein description, only proteins that contain that text within their description column will be displayed in the protein table
- **Include Contaminant Data:** Check this to display contaminants that were identified during a PEAKS DB search (if a contaminant database was specified).
- Protein Contains Modifications: Check boxes are provided for each identified modification.

3.4. Peptide View

Quantified proteins are displayed in the **Peptide View** along with their detailed information. The peptides are grouped together based on quantifiable feature vectors with the same sequence.

Peotide	Ava. In	tensity Sia	nificance	Avg. ppm	Sample Profile Group Profi	le Group 1	Group 2	Max Ratio	RT mean #Vector	Accession	PTM	ė.
IWHHTFYNELR		3.8e+8	16.25	0.7		4.20e+8	3.45e+8	1.22	30.62 2	P63260 ACTG_MOUSE, P60710 ACT		8
VAPEEHPTLLTEAPLNPK		2.6e+8	33.76	0.6		3.18e+8	2.04e+8	1.55	35.71 2	P68033 ACTC_MOUSE, P62737 ACT		
KPAAAPAPAPAPAPAPAAAKPK		1.5e+8	3.93	0.8		1.55e+8	1.44e+8	1.07	19.02 3	P05977 MYL1_MOUSE		
KKEEEELIALK		1.4e+8	22.84	1.1		1.68e+8	1.21e+8	1.39	25.28 2	Q9QZ47 TNNT3_MOUSE, 088346 T		
TYFPHFDVSHGSAQVK		1.3e+8	26.07	0.7		1.00e+8	1.51e+8	1.51	33.90 2	P01942 HBA_MOUSE		
AVEPSIVGRPR		1.2e+8	40.19	0.8		1.66e+8	6.71e+7	2.48	32.31 2	P63260 ACTG_MOUSE, P60710 ACT		
GYSEVITAER	140101/	1.00+8	30.41	0.4		1.30e+8	7.520+7	1.73	31.29 1	P68033 ACTC_MOUSE, P62737 ACT		
AVEDOWOD	PASLOK	9.28*7	30.90	1.1		0.81e+/	7.01e47	1.09	44.32 4	PU1942[HBA_MUUSE D622601ACTC_MOUSE_D607101ACT	C	
IIADDED		7.40+7	20.22	1.0		0.200+7	5.47e+7	1.00	22.27 1	P63200/ACTG_MODSE, P00710/ACT		
.3 3.75e+8 0.92 ×1 2.84e+8 0.70 ×2 3.17e+8 0.78	3.50+8-					52 53 w1 w2						
w3 2.79e+8 0.68	2.50+8-					w3						
	1.50e+8-											

3.4.1. Peptide Table

The peptide table contains the list of quantifiable peptides along with relevant information including the accession information for the protein they support.

The view is similar to the support peptide table described in Section 3.3.4, "Peptides view". The Peptide table entries may be filtered by using the **Peptide View Filters** on top of the peptide table, for more information refer to Section 3.5.1, "Peptide View Filters" in Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER).

3.4.2. Feature Details

The Feature Details contains the eXtracted Ion Chromatogram (XIC) chart and Ion Match.

• **Peptide Menu:** It is under the peptide table. The peptide menu displays the information for the peptide that is being displayed in the spectrum. If a feature is associated to more than one tandem scan, click the "FRACTION" button to view the Spectrum Annotation of the other PSMs.

3.4.2.1. eXtracted Ion Chromatogram (XIC) chart

The eXtracted Ion Chromatogram (XIC) chart displays the shape of the selected peptide feature vector over the retention time range associated with the identification. The table on the left displays both the area and the ratio of the features in the highest average abundance feature vector. This table also works as a control to select the features that will be used to draw the XIC plot.



3.4.2.2. Ion Match

This section displays the spectrum annotation and other information about the peptide selected in the Peptide table. When multiple spectra are matched to the peptide, only the information about the top-scoring PSM is shown by default. The interface of this section is identical to the Spectrum Annotation in the PEAKS De Novo result. For detailed instructions, refer to Section 3.5, "Spectrum Annotation" in Chapter 5, Peptide De Novo Sequencing.

	Dentide	Aven Indexeditor	Olamifiaam		Ormala Desfite	One of Deadle	0	0	Mary Dates	DT	A) /+	4	010410
TAME	Peptide HTEVNELP	Avg. Intensity	Significan	16.25 0.7	Sample Profile	Group Profile	Group 1	Group 2	Max Ratio	RI mean	#vector	ACCESSION P53260IACTG_MOUSE_P50710IACT	PIM
VAP	EEHPTI I TEAPI NPK	2 68+1		33.76 0.6			3 18e+8	2.04e+8	1.55	35.71 2	-	P68033IACTC MOUSE P62737IACT	
KPA	AAPAPAPAPAPAPAKPK	1.5e+l		3.93 0.8			1.55e+8	1.44e+8	1.07	19.02 3	-	P05977IMYL1 MOUSE	
KKE	EEELIALK	1.4e+l		22.84 1.1			1.68e+8	1.21e+8	1.39	25.28 2		090Z47/TNNT3_MOUSE.088346/T	
TYE	PHEDVSHGSAQVK	1.3e+l		26.07 0.7			1.00e+8	1.51e+8	1.51	33.90 2		P01942IHBA MOUSE	
AVF	PSIVGRPR	1.2e+I		40.19 0.8			1.66e+8	6.71e+7	2.48	32.31 2		P63260 ACTG_MOUSE, P60710 ACT	
GYS	FVTTAER	1.0e+l		30.41 0.4			1.30e+8	7.52e+7	1.73	31.29 1		P68033 ACTC_MOUSE, P62737 ACT	
LLS	HC(+57.02)LLVTLASHHPADFTPAVHASLDK	9.2e+		36.96 1.1			6.81e+7	1.15e+8	1.69	44.32 4	1	P01942 HBA_MOUSE	C
AVE	PSIVGR	7.9e+		0.32 1.1			7.93e+7	7.91e+7	1.00	37.93 1		P63260 ACTG_MOUSE, P60710 ACT	
IIAP	PER	7.4e+		20.38 1.0			9.30e+7	5.47e+7	1.70	22.37 1		P63260 ACTG_MOUSE, Q8BFZ3 ACT	
100 - 00 - 00 - 70 - 60 -	1)	99	24										
100 - 00 - 00 - 00 - 00 - 50 - 10 - 50 - 10 - 00 - 50 - 10 - 00 - 00 - 50 - 10 - 00 -	v 	94 56(2+1) 2-1	2+j 	<u>b6 ¥6</u>	у <mark>7</mark> 67 1960 о 1960 о	, X® 1100.0 120	<u>. 89</u> 0.0 13000						
100- 100- 00- 00- 00- 00- 00- 00-	ν 1921 - 1923 - 1920 -	ور به دور:	2+j b5j o 760.0	500 X 800 0	97 67 seb o 1660 o		. <u>69 .</u> 0.0 1360.0						
100- 00- 00- 00- 00- 00- 00- 00-	ν 1916 - 2 ¹ ,, 1 ² 1960 - ολος ολος	ور المراجع المراجع المراجع مراجع المراجع ا	24j	be ^{1%}	27 67 ecb o rodo.o		_ 69 5.0 						
100 - 00 - 00 - 00 - 00 - 00 - 00 - 00	N 191 - 191 1960 - 1960 - 1960 100 - 200 - 200	54 14 14 14 14 14 14 14 14 14 1	2+1 b5 700 0		97 57 eco o 108.0 edo 1.000		_ b9 0.0 13600 						
100- 100- 00- 50- 50- 40- 50- 10- 00- 50- 10- 00- 50- 10- 00- 50- 10- 00- 50- 10- 00- 50- 10- 10- 10- 10- 10- 10- 10- 1	0 10 10 10 10 10 10 10 10 10 1	در الله الله الله الله الله الله الله الله	24]	860 0 860 0 900 9	97 57 940 5 100 5 000 1,000 5		00 1300 0 1300 0 1300 y(2+)	# ***					
100- 00- 00- 00- 00- 00- 00- 00-	N 10 20 120 20 20 20 20 20 20 20 20 20 20 20 20 2	24 40 40 40 40 40 40 40 40 40 4	2+) 55 55 1 54 55 55 55 55 55 55 55 55 55	200 96 2000 000 9	97 80 806 0 1006 0 900 1000	. 1 76 trabo trab t.ao trab t.ao trab t.ao trab	- b0 0 13000 	# ^ 111					
100- 00- 00- 00- 00- 00- 00- 00-	0 10 10 10 10 10 10 10 10 10 1	24 24 45 45 45 45 45 45 45 45 45 45 45 45 45	24	800 0 800 0 900 900 900 1402.67	ebs 100s		. 59 0.0 13620 	# * 111 6 100					
100- 00- 00- 00- 00- 00- 00- 00-	₩ ₩25 s x25 x x2	73 capp-1 color colo	24 y5 1000 55 1 50 W 12 H	b6 76 800 0 900 9 9 1216.59	97 80 800 1008 9 900 1008 9 9420 1384 65 1384 65		50 0 0 136500 00 136500 1,500 y(2*) 702.1 609.0	# * 6 110 12 9					
100- 100-	5 5 5 5 5 5 5 5 5 5 5 5 5 5	24 46 46 46 46 46 46 46 46 46 4	24 y5 y5 y5 y6 y6 y6 y6 y6 y6 y6 y6 y6 y6	be ³⁶ este e y 1402.67 1216.59 1079.21	97 97 980 3 1000 3 900 1000 3 9420 1394 65 1196 86 1061 63	78 1160 s 120 • 1,50 t 2 7 yNH3 1385.64 1199.56 1062.63	. 89 00 13000 00 1,300 y(2+) 702,1 6090.0 540.4	# 11 6 100 12 09 8 8					
100- 00- 00- 00- 00- 00- 00- 00-	v v v v v v v v v v v v v v	24 24 24 25 25 26 26 26 26 26 26 26 26 26 26	2-4 y5 rob a rob a Seq Seq 12 H 64 H 17 T	500 500 500 500 500 500 500 500 500 500	97 67 edd 9 todd 9 000 t.doo y+H20 1384.65 1198.56 1081.58 924.55	198 1106 8 120 1385 64 11395 56 11092 55 1092 55	00 1300 00 1300 9(24) 702 1 609 0 540 4 471 6	# 1 6 10 12 9 90 8 10 1					
100- 00- 100- 00- 100- 00- 100- 00-	N N N N N N N N N N N N N N	24 40 40 40 40 40 40 40 40 40 4	24 y5 rob 0 rob 0 ro	b6 ³⁶ exte a y 1402.67 1216.59 1079-3 1079-3 841.57		. x ⁸ 11de s → 12d . 100 1 (2 . 100 1	59 50 1300 0 50 1300 0 50 1300 0 50 1300 0 50 1300 0 50 1300 0 50 10 10 0 50 10 10 0 50 10 0	# 111 6 100 12 9 9 10 8 11 7 7					
100- 00- 00- 00- 00- 00- 00- 00-	0 10 10 10 10 10 10 10 10 10 1	24 40 40 40 40 40 40 40 40 40 4	24 y5 rtb 0 rtb 0 55 1 55 1 55 4 H 12 H 64 H 17 7 7 7 7 7 7 7 7 7 7 7 7 7	26 26 2 200 0 200 0	edio todo s todo todo s yH20 405 1091 69 1091 69 1091 69 1091 69 1097 6 34	1 100 ± 1000 ± 100 ± 100 ± 100 ± 100 ± 100 ± 100 ± 100 ± 100 ± 100 ± 100 ± 10	50 50 1300 50 1300 540.4 471.5 477.5 477.5 540.4	#					

3.4.2.2.1. DIA Ion Match

When LFQ analysis are run on results from a DIA Identification search, there are two additional figures included in the Ion Match pane: The Precursor Profile XIC and the Fragment Ion XIC. For detailed instructions on these figures, refer to Section 4.5.6, "Peptide-Spectrum Matches" in Chapter 7, DIA Streamlined Identification.

4. Ion Mobility Spectrometry (IMS) Data

Ion mobility spectrometry (IMS), including Field asymmetric waveform ion mobility spectrometry (**FAIMS**) and Trapped ion mobility spectrometry (**tims**) analyzes complex mixtures and separates the ions based on their mobility. PEAKS Online has the ability to load IMS data and analyze it using *de novo* sequencing, database search (PEAKS DB), database search with unexpected modification identification (PEAKS PTM), sequence variant identification (SPIDER) search, spectral library search, and labelled and label free quantification.

The analysis result for IMS dataset contains the additional information for CV values compared to the non-IMS dataset. The CV-related information can be found at the following places for a LFQ analysis.

• The Venn diagram on the summary page.

Venn diagram overlaps between different CVs in the search for all proteins, top proteins, and peptides are shown on the summary page.

- The additional CV column on the supporting peptide page
- The additional CV column in the exported Protein-Peptide CSV
- The additional CV column on the peptide page
- The additional CV filter on peptide page

The filter for peptide page contains one additional option for CV, which selects the displayed scans based on the selected CV values

• The additional CV column in the exported Peptide CSV

4.1. Result Exporting

PEAKS LFQ results can be exported to various text formats in the Export page

Export	SELECT ALL CLEAR ALL		
All Search Parameters			
Data Refinement	Denovo	Database Search	LFQ
MGF File	Summary Table	Summary Table	Protein CSV
Mzxml	Denovo CSV	PTM Profile Summary CSV	Protein-Peptide CSV
	Denovo CSV with all Candidates	s Protein CSV	Protein with Peptide CSV
		Protein-Peptide CSV	Peptide CSV
		PTM Profiling CSV	Feature CSV
		Protein Fasta	Normalization factor
		Peptide CSV	
		PSM CSV	
		Peptides-Pepxml	
		Peptides-MzldentML	
		Denovo Only CSV	
		Denovo Only CSV with all Candidates	
DOWNLOAD DOWNLOAD BY	SAMPLE		

- **Proteins CSV:** These are the quantified proteins that pass the filters set on the "Summary" page and their associated details.
- Protein-Peptide CSV: These are the supporting peptides of the quantified proteins and their associated details.
- **Peptide CSV:** A complete list of all quantified peptides that pass the peptide feature filters set on the "Summary" page and their associated details.
- Feature CSV: All predicted peptide features and their associated quantification information. Both identified and unidentified peptide features are included. To include unidentified peptide features, set the Peptide Id Count equal to 0 in the Peptide feature filter.

- Protein with Peptide CSV: A list of supporting peptides for the quantified proteins and their associated details.
- Normalization Factor CSV: A list of normalization factors used for LFQ normalization. For analyses using local normalization methods such as TIC or internal standard proteins, global normalization factors will display as N/A.

Chapter 9. PEAKS Quality Control

1. Overview

PEAKS Quality Control offers a sophisticated and systematic QC analysis. In order to help with verifying the integrity and validity of data being processed through PEAKS Online an optional QC (Quality Control) task can be run on top of PEAKS protein/peptide identification and label-free quantification workflows. This task will provide statistical information about your data to allow for informed decisions on the data's quality.

2. QC Parameters

To enable PEAKS QC to be run on top of your PEAKS identification or quantification results, ensure that the QC section is checked at the bottom of the Project Parameters page.

Control Sample:	Sample 1 🔹	Acceptance Tolerance (%)	: 10	
Data QC Attribute	es:	1	dentificati	on QC Attributes:
	Attribute			Attribute
	# MS1			# PSM
	# MS2			# MS2
	MS2/MS1 Rate (%)		\checkmark	# Identified Precursors
	Detected Features		\checkmark	# Peptides
	Full Width (min)			# Sequences
	FWHM (sec)		\checkmark	# Protein Groups
	BPC			# Top Proteins
	TIC Correlation			# All Proteins
				ID Rate (%)
				MS1 Error Mean (S D)
				MS2 Error Mean (S.D)
				Missed Cleavage Ratio (%)
				Enzyme Specificity Ratio (%)
Standard Peptide Input Standard LLDMGETDLMI DHSFFIPDIEYL	IS: Peptides Seperated by commas sp LAALR 2 SDIK,3	ace or tab, example below		

Before submitting the analysis ensure the following parameters are set properly:

- **Control Sample:** Specify the sample to be used as the Control Sample for quality control checking. All other samples will have their quality control values compared to the corresponding value from the control sample and will be considered failing if the variance is outside of the Acceptance Tolerance.
- Acceptance Tolerance (%): This is acceptable amount of variance a sample can have from the control sample (or average if specified). Any quality control metric that falls outside of this tolerance level will be labelled as failed (red) in the quality control result views. If Acceptance Tolerance is set to 0%, then each attribute will use the value from selected control sample as the threshold.
- **Standard Peptides:** Input amino acid sequences with analogue charge of your standard peptides separated by commas, spaces, or tabs.

2.1. Quality Control Attributes

Different quality control attributes can be analyzed in PEAKS online. These attributes are divided into data, identification and quantification sections. Quantification attributes are only visible if a PEAKS Q workflow is selected.

2.1.1. DDA QC Attributes

- Data QC Attributes: #MS1, #MS2, MS2/MS1 Rate (%), Detected Features, Full Width (min). FWHM (sec), BPC, TIC Correlation
- Identification QC Attributes: #PSM, #MS2, # Identified Features, #Peptides, #Sequences, #Protein Groups, #Top Proteins, #All Proteins, ID Rate (%), #PSM/# Peptides Rate (%), MS1 Error Mean (S.D), MS2 Error Mean (S.D), Missed Cleavage Ratio (%), Enzyme Specificity Ratio (%), Ion Mobility Error Mean (timsTOF only).
- Quantification QC Attributes : #Quantified Peptides, #Quantified Protein Groups, #Quantified Features, Missing Values, Feature Correlation.

2.1.2. DIA QC Attributes

- Data QC Attributes: #MS1, #MS2, Detected Precursors, Full Width (min). FWHM (sec), BPC, TIC Correlation
- Identification QC Attributes: #PSM, #MS2, # Identified Precursors, #Peptides, #Sequences, #Protein Groups, #Top Proteins, #All Proteins, ID Rate (%), MS1 Error Mean (S.D), MS2 Error Mean (S.D), Missed Cleavage Ratio (%), Enzyme Specificity Ratio (%), Ion Mobility Error Mean (timsTOF only).
- Quantification QC Attributes : #Quantified Peptides, #Quantified Protein Groups, #Quantified Precursors, Missing Values, Feature Correlation.

3. PEAKS QC Results

PEAKS QC results consist of Summary, Data, Identification, Quantification and Standard Peptides tabs.

3.1. QC Summary Tab

The QC summary tab contains:

- **Table 1: Statistics ofFiltered Results.** Contains the Statistic of filtered results including information of the Control sample, Acceptance tolerance, Total, Passed and Failed sample counts, Ratio of Passed samples and the total number of Failed Attributes. If no failed attributes present, the message will be displayed "All Attributes Pass!" and only Table 1 and 2 are displayed. Specific selection of QC attributes will also affect if sample is considered Passed or Failed.
- **Figure 1: Failed Attributes Chart.** For failed attributes, a bar chart will be displayed indicating the number of samples where specific attributes failed.
- **Table 2-4**: **Statistics of Passed Attributes**. Statistics for passed data, identification and quantification attributes are summarized in separate tables for Control Sample and for the Average of Passed Samples. Click on "Filter Samples" to select Samples and additional QC Attributes to be displayed and considered in average calculations. For a sample to be included in the average of passed samples, it must have no failing attributes in any category.
- **Table 5-7: Statistics of Failed Attributes.** Statistics for failed data, identification and quantification attributes are summarized in separate tables with entries for each failed sample. Click on "Filter Samples" to select Samples and additional QC Attributes to be displayed and considered in failing attributes tables.



3.2. QC Data Tab

This tab shows several figures that visualize the data quality control measures.

The Data QC summary table displays an overview of identification-related quality control measures. Any value in the table that exceeds the acceptance tolerance, set on the parameters page, will be coloured red.

							\leftarrow	_	1 - 20 of 21 🔻	\rightarrow	
SAMPLE NAME	# MS1	# MS/MS	# PRECURSORS	FULL WIDTH (MIN)	FWHM (SEC)	TOTAL BPC					
P1-A10	925	30525	63528	0.15	8.28	2.105e+9					
P1-G4	929	30657	62065	0.15	8.28	3.637e+9					
P1-G6	927	30591	63148	0.15	8.27	3.396e+9					
P2-A6	926	30558	62300	0.15	8.28	2.630e+9					
P2-C11	928	30624	57903	0.15	8.28	2.928e+9					
P2-C5	929	30657	60535	0.15	8.28	3.002e+9					
P3-D10	931	30723	59086	0.15	8.28	4.474e+9					
PG-G10	928	30624	59056	0.15	8.28	3.456e+9					
P3-G3	927	30591	57269	0.15	8.28	3.314e+9					
Sample 11	931	30723	57604	0.15	8.28	3.644e+9					
P4-84	930	30690	55501	0.15	8.28	4.049e+9					
P4-C2	929	30657	53171	0.15	8.28	3.538e+9					
P4-F10	928	30624	49567	0.19	8.29	2.238e+9					
P4-F7	926	30558	49675	0.19	8.29	2.280e+9					
P5-B12	929	30657	47300	0.19	8.29	1.674e+9					

The columns in this table are:

- **Sample Name:** Name of the sample represented by this row. The table will always be sorted by this column (in sample order) with the control sample at the top and the average values at the bottom. The selected control sample will have its name displayed in bold in this column.
- All Attributes: This column displays whether samples either pass or fail quality control. This considers all attributes selected from the summary page filter menu, not only those in the data attribute list.
- # MS1 Scans: Total number of MS1 scans in the sample.
- # MS2 Scans: Total number of MS2 scans in the sample.
- # MS2/MS1 Rate (%): Total number of MS2 scans per MS1 scan from the sample. Displayed in percentage (%). (DDA Only).
- Detected Precursors/Features: Total number of precursors or features for DIA and DDA respectively.
- Full Width (min): Full peak width at base in minutes.

- FWHM (sec): Full width at Half Maximum of peak in seconds.
- **TIC Correlation:** The Pearson correlation coefficient with respect to points on the control sample TIC
- **BPC:** TIC at 80% of other TICs are above it.

3.2.1. Data QC Figures

Figure 1 will display a visualized summary of the selected column in the Data QC summary table, which will be highlighted in blue. By default the Detected Features/Precursors column will be selected, to select a new column click on the header row of the column you wish to select.

The dotted lines displayed on the chart represent the minimum and maximum values of the acceptance range (determined by the acceptance tolerance set on the parameters page). Any bar with a height that falls within these two dotted lines represents a value within the acceptance tolerance. Any bar with a height below or above these lines represents a value outside of the set acceptance tolerance.

Note

For analyses with greater than 30 samples, a line chart will be displayed instead of a bar chart. In this case, users may place mouse over the graph and zoom in and out to show more or less samples along the x-axes.



Figure 2. shows the TIC charts for selected samples side-by-side on the same axis. Up to 5 samples are selectable for display at once from a list beside the figure. By default the first 3 samples are selected.



Figure 3. shows the cumulative precursor count over retention time. Only 2 samples can be displayed at once for pairwise comparison. These samples can be selected from the drop down menus above the figure.

Figure 3. Cumulative Precursor Count. Shows precursor count over retention time



3.3. QC Identification Tab

This tab shows several figures that visualize the data quality control measures.

The Data QC summary table displays an overview of identification-related quality control measures. Any value in the table that exceeds the acceptance tolerance, set on the parameters page, will be coloured red.

															\leftarrow	1 - 5 of 5 🔻	\rightarrow
SAMPLE NAME	ALL ATTRIBUTES	# MS2	# IDENTIFIED FEATURES	# PSM	# PEPTIDES	# SEQUENCES	# PROTEIN GROUP	# TOP PROTEINS	# ALL PROTEINS	MISSED CLEAVAGE RATIO (%)	ENZYME SPECIFICITY RATIO (%)	# PSM/# PEPTIDES RATE (%)	MS1 ERROR MEAN (S.D)	MS2 ERROR MEAN (S.D)	ID RATE (%)		
Thermo_DDA_HE_01	Pass	31611	21880	25710	19340	19207	3201	3280	3704	8.19	97.24	1.33	1.35(1.45)	2.75e-3(3.73e-3)	81.33		
Thermo_DDA_HE_02	Pass	31759	22194	25970	19620	19472	3228	3311	3718	8.27	97.12	1.32	1.35(1.45)	2.76e-3(3.73e-3)	81.77		
Thermo_DDA_HE_03	Pass	32058	22127	25881	19666	19538	3255	3333	3892	8.92	97.06	1.32	1.33(1.45)	2.86e-3(3.73e-3)	80.73		
Thermo_DDA_HE_04	Pass	31767	22299	26025	19803	19665	3283	3350	3773	8.85	97.03	1.31	1.33(1.46)	2.85e-3(3.72e-3)	81.92		
Average	Pass	31798	22125	25896	19607	19470	3241	3318	3771	8.56	97.11	1.32	1.34(1.45)	2.80e-3(3.73e-3)	81.44		

The columns in this table are:

- **Sample Name:** Name of the sample represented by this row. The table will always be sorted by this column (in sample order) with the control sample at the top and the average values at the bottom. The selected control sample will have its name displayed in bold in this column.
- All Attributes: This column displays whether samples either pass or fail quality control. This considers all attributes selected from the summary page filter menu, not only those in the identification attribute list.
- **# MS2** Total number of MS2 scans in the sample.
- **#Identified Features/Precursors:** The number of identified features/precursors in the sample for DDA and DIA respectively.
- **# PSM:** The number of peptide-spectrum matches..
- **#Peptides:** Total number of peptides found ignoring differences with leucine (L) and isoleucine (I) amino acids. Peptides with the same primary sequence but different PTMs are counted separately.
- **#Sequences:** Total number of different peptide backbone sequences found ignoring differences with leucine (L) and isoleucine (I) amino acids. Peptides with the same primary sequence but different PTMs are counted as one sequence.
- **#Protein Groups:** PEAKS Online groups proteins identified by a common set of peptides. This number shows the number of protein groups in the filtered result based on All proteins.
- **#Top Proteins:** The number of proteins that are supported by the most unique peptides in the group. Proteins in the group that share a subset of the unique peptides that support the top protein (sub-proteins) will not be included here.
- #All Proteins: Total number of proteins identified in the sample.
- Missed Cleavage Ratio (%): The percentage of missed cleavages in a sample based on the number of expected cleavage sites as per the selected enzyme.
- Enzyme Specificity Ratio (%): The percentage of cuts that took place at expected cleavage sites as per the selected enzyme.
- MS1 Error Mean (S.D): PSM precursor error mean (ppm). Standard deviation is shown in brackets.
- MS2 Error Mean (S.D): Fragment Ion Error mean (Da). Standard deviation is shown in brackets.
- **ID Rate** (%): The percentage of #MS2 scans which produce a PSM.

Columns which are only present for DDA data:

• **#PSM/#Peptide Rate:** The **#**PSM divided by the number of identified peptides in a sample.

3.3.1. Identification QC Figures

Figure 1 will display a visualized summary of the selected column in the Identification QC summary table, which will be highlighted in blue. By default the # Protein Groups column will be selected, to select a new column click on the header row of the column you wish to select.

The dotted lines displayed on the chart represent the minimum and maximum values of the acceptance range (determined by the acceptance tolerance set on the parameters page). Any bar with a height that falls within these two dotted lines represents a value within the acceptance tolerance. Any bar with a height below or above these lines represents a value outside of the set acceptance tolerance.Place mouse over the graph and zoom in and out to show all or less samples/sample names on the x-axes

Note

For analyses with greater than 30 samples, a line chart will be displayed instead of a bar chart.

Note

For DIA spectral library searches without protein inference, #Peptides will be the default selected column.



Figures 2 and 3. shows sample reproducibility charts. These charts display overlap in Protein and Peptide identifications respectively. By default, only 2 samples are shown for pairwise comparison. Using the drop down menu above the figure, users can select different samples for comparison.



Figure 4. displays violin plots of the PSM precursor (ppm) and fragment ion error (Da) respectively.

Figure 4. MS1 & MS2 Mass Error Distribution.



Figure 5. shows the peptide frequency distributions based on the number of PSMs per peptide identification at the individual sample level.



3.4. QC Quantification Tab

This tab shows several figures that visualize the label-free quantification quality control measures.

The LFQ quantification QC summary table displays an overview of identification-related quality control measures. Any value in the table that exceeds the acceptance tolerance, set on the parameters page, will be coloured red.

SAMPLE NAME	ALL ATTRIBUTES	# IDENTIFIED PRECURSORS	# QUANTIFIED PRECURSORS	MISSING VALUES (%)	PRECURSOR CORRELATION	# IDENTIFIED PEPTIDES	# QUANTIFIED PEPTIDES	# IDENTIFIED PROTEIN GROUP	# QUANTIFIED PROTEIN GROUPS	
Sample 1	Pass	68241	0	0.00	0.00	60224	0	7331	0	
Sample 2	Fail	55541	0	0.00	0.00	49329	0	6975	0	
Sample 3	Fail	68385	0	0.00	0.00	60109	0	7305	0	
Sample 4	Fail	56109	0	0.00	0.00	49400	0	7160	0	
Sample 5	Fail	44625	0	0.00	0.00	40280	0	6313	0	
Sample 6	Fail	65226	0	0.00	0.00	57082	0	7304	0	
Average	Fail	59687	0	0.00	0.00	52737	0	7064	0	

The columns in this table are:

- **Sample Name:** Name of the sample represented by this row. The table will always be sorted by this column (in sample order) with the control sample at the top and the average values at the bottom. If average is selected as the control sample it will be displayed at the top of the table instead of the bottom. The selected control sample will have its name displayed in bold in this column.
- All Attributes: This column displays whether samples either pass or fail quality control. This considers all attributes selected on the summary page, not only those in the quantification attribute list.

- **#Identified Features/Precursors:** The number of identified features/precursors in the sample for DDA and DIA respectively.
- **#Quantified Features/Precursors:** The number of quantified features/precursors in the sample for DDA and DIA respectively.
- Missing Values (%): The percent of peptides where no feature could be found in a sample based on identification in other samples.
- **Precursor Correlation** (%): Measures the pearson correlation coefficient of the MS1 features compared to the control sample.
- #Identified Peptides: Total number of peptides identified in the sample.
- **#Quantified Peptides:** Total number of peptides quantified in the sample.
- #Identified Protein Groups: The number of identified protein groups in the sample.
- #Quantified Protein Groups: Total number of proteins quantified in the sample.

3.4.1. Quantification QC Figures

Figure 1 will display a visualized summary of the selected column in the Quantification QC summary table, which will be highlighted in blue. By default the # Identified Features and #Quantified Features columns will be selected, to select a new column(s) click on the header row of the column you wish to select.

The dotted lines displayed on the chart represent the minimum and maximum values of the acceptance range (determined by the acceptance tolerance set on the parameters page). Any bar with a height that falls within these two dotted lines represents a value within the acceptance tolerance. Any bar with a height below or above these lines represents a value outside of the set acceptance tolerance.Place mouse over the graph and zoom in and out to show all or less samples/sample names on the x-axes.

Note

For analyses with greater than 30 samples, a line chart will be displayed instead of a bar chart.

Note

For DIA spectral library searches without protein inference, #Identified peptides and #Quantified peptides will be the default selected column.



Figure 1. Identified and Quantified Precursors.

Figures 2, 3, and 4. show relative area correlation at the feature, peptide and protein level. Correlation values are displayed in log2 ratio. By default, only 2 samples are shown for pairwise comparison. Using the drop down menu above the figure, users can select different samples for comparison.



Figure 5. displays the LFQ RT alignment chart. Shows sample retention time alignments to the control sample. By default only 2 samples are shown. Using the drop down menu, users can select different samples for comparison to control. If one of the samples selected is the control sample, this chart will only display one line.



3.5. QC Standard Peptides Tab

QC Standard Peptides tab displays RT and M/Z reproducibility graphs. Additionally, 1/k0 reproducibility of standard peptides will be displayed for timsTOF data.

O 1

Note

The eye icon is used to highlight samples with missing values.






3.6. QC Results Exporting

3.6.1. Result Page Exporting

There are two export options available for quick exporting at the top right of the summary page. The exports are:

- Statistics: This export link will download a CSV containing summary of all information displayed in Table 1.
- Failed Samples: This export link will download a CSV containing summary of Failed Samples statistics as per selection made in "Filter Samples".

• **Passed Samples:** This export link will download a CSV containing summary of Passed Samples statistics as per selection made in "Filter Samples" and as displayed in Table 2.

3.6.2. Analysis Export Page

A QC checkbox in the Analysis Export page offers four options for result download. Selecting this option will add the All Samples QC export (as detailed above) to the resulting export zipped file. This option will also add the All Samples QC CSV for both the Download and Download by Sample export options.

< Export SELECT ALL	CLEAR ALL	
All Search Parameters		
Spectral Library	Database Search	QC
Summary Table	Summary Table	VC Summary Table
Protein CSV	Protein CSV	✓ All Samples CSV
Protein-Peptide CSV	Protein-Peptide CSV	Passed Samples CSV
Peptide CSV	Peptide CSV	✓ Failed Samples CSV
PSM CSV	PSM CSV	
Protein Fasta	Protein Fasta	
DOWNLOAD DOWNLOAD BY SAMPLE		

Chapter 10. Isobaric Labelling Quantification (TMT/iTRAQ)

1. Overview

Isobaric labelling quantification with chemical labels at the MS2 (or MS3) level is one of the three quantification modes that are supported by the optional PEAKS Q module of PEAKS Online. In contrast to LFQ Quantification, TMT/iTRAQ Quantification is based on the relative intensities of the fragment peaks at fixed m/z values within an MS2 (or MS3) spectrum. To prepare for this, isobaric chemical labels are introduced into samples during the sample preparation stage. The samples are then combined and analyzed together in an LC-MS/MS experiment. The same peptides from different samples will have the same precursor m/z and retention time, and are fragmented together. In the MS/MS (MS3) scans, labels from different samples will produce different reporter ions, which can then be used to calculate the quantification ratio between samples. Both user-defined and commercial labels (i.e. iTRAQ and TMT) are supported by PEAKS Q.

The usage of this workflow is outlined below.

- 1. Refer to Chapter 5, Peptide De Novo Sequencing, Section 2, "DeepNovo/De Novo Analysis Workflow and Parameter Settings", Step 1-6 to create a new project and add data.
- 2. Select the DDA "Quantification" workflow.
- 3. Set DB Search Parameters. Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 2, "Database Search Workflow and Parameter Settings" for details.

Note

TMT/iTRAQ Modifications will be automatically added to DB search parameters when TMT/iTRAQ is selected.

- 4. Set the Report Filters. Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 2.8, "PSM Filter" for details.
- 5. Set the Quantification parameters by first select "TMT/iTRAQ". This will automatically open the appropriate parameter settings. Set the TMT/iTRAQ specific Quantification settings: Label Q Method, Mass Error and Report Ion Type. See Section 2 below for further details.

uantification 🔘 Label Free	• • • TMT/iTRAQ Label	⊖ silac
Select Methods:	•	Mass Error Tolerance: 15 Mass Error Tolerance: PPM Reporter Ion Type: MS2
Perform Purity Correct	on	~
Spectrum Filter		
Qualityz: 0	: Reporter Ion Intensity ≥:	0 : Number of Channels Present ≥: 1 : Reference Channel Present □
Protein Filter		
Significance ≥	0 : O FE	DR (adjusted p-value)s: 0 : 1 : s Fold change s 64 :
Unique Peptides ≥:	1 : 🗆 M	odified Form Exclusion
Experiment Setting 🧪 All E	xperiment, No inter experiment	t normalization

- 6. Within the Quantification section: Set the Spectrum and Protein Filters. See Chapter 5 and below for further details.
- 7. Within the Quantification section: Edit the Experiment Settings. See below for further details.
- 8. Review project settings and parameters then click Submit button. Name the analysis and click OK to start the analysis.

2. Setting Parameters

For more information on DB search parameters, refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 2, "Database Search Workflow and Parameter Settings".

The following parameters are available when the TMT/iTRAQ Quantification workflow is selected.

- Select Methods: From the Select Methods drop-down menu, choose the appropriate quantification method used in the experiment, e.g. iTRAQ-4plex
- Mass Error Tolerance: This parameter is used to locate the reporter ion peaks in the MS/MS scan. Either Da or ppm can be selected as the mass unit.
- **Report Ion Type:** This parameter allows for the selection of MS2 or MS3 as the reporter ion type for quantification.
- **Perform Purity Correction:** Select the checkbox to edit correction factors for chemical labels (if applicable). Once the checkbox is selected, click the down arrow to open the dialogue that allows users to setup the purity correction. The dialogue is setup to mimic a Certificate of Analysis, where each factor within the table can be modified by double-clicking the particular cell of interest.

2.1. Spectrum Filter:

- **Quality:** A higher quality peptide indicates that the peptide is more quantifiable. Factors that affect the quality score include the identification -10LgP score, the noise around the reporter ions, and the mass error of reporter ions.
- **Reporter Ion Intensity:** Peptides with at least one reporter ion above this value will be kept.
- **Number of Channels Present:** Only quantifiable peptides that contain at least the number of labels selected will be used in protein quantification.
- **Reference Channel Present:** Only quantifiable peptides in which the reference label is present will be used in protein quantification. Note: The reference label is set in the experiment setting pop-up. See Experiment Setting section below

2.2. Protein Filter:

• Refer to Chapter 8, Label-Free Quantification (LFQ), Section 2, "Setting Parameters", for detailed descriptions of quantification protein filters.

2.3. Experiment Setting:

The following parameter dialogue appears when the 🧪 button is clicked.

erform Inter-Experiment N	Iormalization					← 1-1	of1 🔹 $ ightarrow$	
				Experiment	A	lias/Spiked Channel		
ference Experiment Sa	mple 1			Sample 1	Sample 1 TMT	r8-126 ▼		
clude Spike Channel for S	Significance							
ra Sample Normalization								
No Normalization								
) Auto Normalization								
) Manual Input								
Normalize to Spike								
xperiment Sample 1	*							
Channel	Name	Easter	Expected 🔺					
Chaimer	Name	Pactor	Ratios					
TMT8-126	Sample 1 TMT8-126	1.00						
TMT8-127N	Sample 1 TMT8-127N	1.00						
TM18-12/C	Sample 1 IM18-12/C	1.00	Ŧ					
periment Groups								
Select All	۹							
				Group		Color		
			-102	 Group 1 Same 	ale 1 TMT8-126		×	
				Samp	ple 1 TMT8-127C		×	
			→ "E"	Samp	ale 1 TMT8-129N		×	
				💷 Samp	ale 1 TMT8-130		×	
			→ 82	Group 2			×	
				Samp	ale 1 IMI8-12/N		~	
				Same	1 TMT8-128		× •	

- Select Experiment: When running quantification on an identification node that searched data within multiple samples (i.e., replicate experiments within different treatments), this drop-down menu allows for the selection of a single sample or every sample (All Experiments) for analysis. Every label of the sample(s) selected will appear in the screen located below the search bar.
- Setting Sample Channel Alias: The name of the Sample Channel can be changed by using the alias name.
- **Inter-Experiment Normalization:** Check this box if a spike label or normalization channel was used in the experiment. This is used to enable quantification between experiments.
 - Alias/Spike Channel Drop-down Menus: Select the channel in each sample that will be used for normalization. The software assumes that similar peptide features containing this label across experiments will approach a 1:1 ratio. Differences are assumed to be due to systematic error and all labeled channels are normalized to this spike label.
 - Exclude Spike Label for Significance: Since the spike label is assumed to be in a 1:1 ratio between experiments, it should not be included in the significance calculation.
- Intra Sample Normalization:
 - No Normalization: If there are multiple samples in the quantification run and all samples are selected, No Normalization is chosen as the default option. It will be the only normalization option available for selection unless "Perform Inter Experiment Normalization" is check-marked in Experiment Settings.
 - Auto normalization: This option will calculate a global ratio from the total intensity of all labels in all quantifiable peptides. Factor values and Expected Ratios between the channels (i.e., the labels) will be presented. Selecting another sample from the Experiment drop-down menu will display the results of other samples.

- Manual input: This option performs the same default calculation as Auto normalization but also allows for the manual change of each label's Expected Ratios. After changing a ratio and hitting "Enter" on the keyboard, PEAKS will recalculate the Factor for the non-referenced label(s). The Experiment drop-down menu can be opened to display the results of another sample.
- Normalize to spike: This option displays a list of identified proteins that can be selected to create a normalization ratio. Use the search bar to quickly locate specific proteins. Right-clicking in the Protein list will open a pop-up menu to select all highlighted rows or remove all selected proteins instead. The expected ratios can be manually modified. It is assumed that the total protein content of the selected proteins in different labels have the ratios entered in the 'Expected ratio' column. If desired, sample names can be changed by double-clicking the chosen channel below the "Name" column and typing the new name.
- Experiment Groups : TMT/iTRAQ Quantification requires sample grouping. All available samples are listed in the left window. After selecting a sample, it can be added to a new group by clicking the subton or to an existing group by clicking the subton and selecting that option from the drop-down menu that appears when you choose the add to existing option. If you select multiple files, you can add each one to a new group by clicking the subton. Samples groups can be removed by selecting the "X" next to the group name. Individual samples within a group can be removed by using the down arrow next to the group name to show all
- **Reference Label:** Use this drop-down menu to select a reference label. The reporter ion intensity corresponding to this reference label will be the denominator in all ratio calculations. Only a label that is already listed in the Experiment Groups screen can be selected as a reference.

of the samples in a group, and then selecting the "X" next to the sample you want to remove from the group.

3. Understanding the Results

After a PEAKS Online TMT/iTRAQ quantification is complete, the isobaric labelling results will be displayed in a Summary, Protein, and Peptide page. The summary page will be the first page in view, and the others are accessed by clicking on the respective tabs.



- **Summary:** The outline of the isobaric labelling search results with statistics. This is the place to examine the overall performance of the experiment and adjust filters. If there are too many proteins, the heatmap will not show on the summary page.
- **Proteins:** The quantified proteins with a list of supporting peptide features for each protein.
- **Peptides:** The quantified peptides are listed in a table. Annotated spectrum chart, ion match table, and reporter ion spectrum are also displayed for each peptide.

3.1. Result Filtration

On the top of each page, there are different options for users to filter isobaric labelling results that can be accessed by selecting the edit icon (pencil): First choose Significance or FDR % (adjusted p-value), then set your values. Next, enter a Fold Change and the number of Unique Peptides. After changing any of these filter parameters, click

next to the filtration window to apply the new filters and update the results and statistics.

- **Significance:** Only protein groups with a significance above this threshold will be listed in the "Protein" view. The significance score is the -10lg of the significance testing p-value. A significance score threshold of 20 is recommended, which equals to a significance testing p value of 0.01. Either this or "FDR (adjusted p-value)" can be selected to set a significance threshold. Significant is calculated using ANOVA as described in Chapter 8.
- **FDR:** Adopt the Benjamin-Hochberg method to adjust the p-value to the false discovery rate for all protein groups that have already passed the other filters. Only protein groups with significance scores passing the calculated FDR will be listed in the "Protein" view. Either this or "Significance" can be selected to set a significance threshold.
- Fold Change: Only protein groups at or above this fold change threshold will be listed in the "Protein" view.
- Unique Peptides: Only protein groups with at least this many unique peptides will be listed in the "Protein" view.

3.2. Result Exporting

PEAKS TMT/iTRA Q results can be exported to various text formats in the Export option listed under the analysis (all export options), or from the top right of the Protein or Peptide pages (two export options).

Export SELEC	CT ALL CLEAR ALL		
All Search Parameters			
Data Refinement	Denovo	Database Search	Label Q
MGF File	Summary Table	Summary Table	Protein CSV
Mzxml	Denovo CSV	PTM Profile Summary CSV	Protein-Peptide CSV
	Denovo CSV with all Candidates	Protein CSV	Peptide CSV
		Protein-Peptide CSV	PSM CSV
		PTM Profiling CSV	Filter/Parameters
		Protein Fasta	Normalization factor
		Peptide CSV	
		PSM CSV	
		Peptides-Pepxml	
		Peptides-MzIdentML	
		Denovo Only CSV	
		Denovo Only CSV with all Candidates	
DOWNLOAD DOWNLOAD BY SAMPLE			

- **Proteins CSV:** These are the quantified proteins that pass the filters set in the "Summary" page and their associated details.
- Protein-Peptide CSV: These are the supporting peptides of the quantified proteins and their associated details.
- **Peptide CSV:** A complete list of all quantified peptides that pass the peptide feature filters set in analysis parameters and their associated details.

- **PSM CSV:** All predicted peptide features and their associated quantification information. Both identified and unidentified peptide features are included. To include unidentified peptide features, set Peptide Id Count equals to 0 in the Peptide feature filter during selection of parameter settings.
- Filter/Parameters: Filter/Parameters for the current analysis
- Normalization Factor: Normalization Factors used in the current analysis

3.3. Summary View

The isobaric labelling view includes an expression profile with candidate proteins clustered in a heat map, result statistics table, and a protein volcano plot. An "Analysis Description" can be added and displayed below the filter section.

3.3.1. Statistics of Filtered Results

Table 1 in the Summary page lists the filtered results of feature vectors and proteins.

3.3.2. The Volcano Plot for Proteins

A volcano plot (Figure 1) combines a statistical test with the magnitude of the change enabling quick visual identification of those data-points that display large-magnitude changes and that are also statistically significant. The volcano plot in Figure 1 plots significance versus fold-change of the quantified proteins.

- Horizontal broken grey line: The selected significance threshold.
- Vertical broken grey lines: The selected fold change thresholds.

Plotting points in this way results in two regions of interest in the plot: those points that are found towards the top of the plot that are far to either the left or the right side. These represent values that display large magnitude fold changes (hence being left or right of center) as well as high statistical significance (hence being towards the top). The ratio is the group ratio set with respect to the base group. If more than two groups are present, the max ratio is used instead.

Circles for the proteins that are above the set significance and fold change thresholds will be displayed in colour. Scrolling the cursor over a circle will show the protein info.

3.3.3. Protein Profile Heatmap

The heatmap in Figure 2 displays the protein groups that pass the filters. The relative protein abundance data is represented as a heat map with a row for each protein group. Proteins are clustered if they exhibit a similar expression trend across the samples. The hierarchical clustering is generated using a neighbour-joining algorithm with a Euclidean distance similarity measurement of the log2 ratios of the abundance of each sample relative to the average abundance. Similarly, the conditions in different samples are clustered if they exhibit a similar expression trend across the protein groups.

3.3.4. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

2 data files produced no results. (Expand to see list of zero result data files)

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.

Potentially Problematic Data Sets	~
Sample 3 All fractions have no result.	
Sample 4 All fractions have no result.	

If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics

No Results Found

3.4. Protein View

The isobaric labelling Protein page lists all the quantified proteins that pass the filter set on the Summary page, characterizes each protein at the amino acid level, and lists the supporting peptide features of each protein.



Isobaric Labelling Quantification (TMT/iTRAQ)

	Deptide	lload	Quality	101 eD	Aug name	Comple Drofile	Crown Drofile	Intensity Crown 1	Intensit
*	replice	Useu	Quality	-TULGP	Avg. ppm	Sample Prome	Group Profile	Intensity Group 1	Interist
2	L(+144.10)ESTVDDEAFIR		24.70	50.96	2.0			7,710+6	-
2	V(+144.10)ADTEETVR(+144.10)	~	24.03	49.26	1.0			1.54e+7	_
3	W(+144.10)/000 FODVDR	~	34.21	40.20	1.0			1.34647	
- 4	W(1144.10)EVEC(137.02)NPGLACIAER	· · ·	33.00	37.03	1.4			1.11+17	
6	A(+144.10)WEVTVK(+144.10)		33.24	20.34	4.0			0.54.15	
0	L(+144.10)DWDK(+144.10)AWEV1VK(+144.10)	~	33.00	30.49	4.4			6.34e+3	
/	1(+144.10)GEETISDLDQLRR(+144.10)	~	32.32	59.47	1.7			5.200+0	
0	V(+144.10)PINPINSLPDVQVK(+144.10)	~	32.42	51.24	2.0			1.68e+6	
9	F(+144.10)K(+144.10)VFADYEEYVK(+144.10)	~	32.21	42.30	5.0			8.25e+5	
10	L(+144.TU)ITAIGDVVNHDPVVGDR	~	30.80	/3.1/	1.5			7.12e+o	
11	I(+144.10)C(+57.02)GGWQM(+15.99)EEADDWLR	1	30.42	37.98	2.6			1.638+5	
12	V(+144.10)LVDLER	~	30.33	22.48	1.1			1.82e+7	-
13	A(+144.10)RPEFTLPVHFYGR	~	29.40	33.88	1.6			6.0be+5	
14	S(+144.10)LFDVQVK(+144.10)	~	28.70	24.15	4.3			2.93e+6	
15	L(+144.10)PAPDEK(+144.10)IP	~	28.68	23.85	4.3			3.40e+6	
16	A(+144.10)APGYHMAK(+144.10)	~	27.77	33.83	-0.9			6.80e+3	
17	E(+144.10)PNK(+144.10)FVVPR	~	26.64	17.44	3.5			5.98e+5	
18	I(+144.10)GEEYISDLDQLR	~	26.09	49.75	2.9			9.64e+5	
19	L(+144.10)VLC(+57.02)NPGLAEIIAER	~	25.75	26.89	4.0			8.87e+4	
20	Q(+144.10)IIEQLSSGFFSPK(+144.10)	~	25.30	27.26	-4.5			3.28e+5	
21	K(+144.10)(+144.10)EPNK(+144.10)FVVPR	~	25.24	33.88	3.7			1.03e+6	
22	V(+144.10)EDVDRLDQR	~	22.49	22.30	0.2			4.59e+5	
23	I(+144.10)C(+57.02)GGWQMEEADDWLR	~	22.01	63.31	1.4			3.34e+6	
24	L(+144.10)LSYVDDEAFIRDVAK(+144.10)	~	21.58	28.72	10.3			2.20e+5	*

3.4.1. Protein Table

The quantified proteins are listed in the protein table with homologous proteins grouped together. If proteins are grouped, the individual proteins within a group can be displayed by clicking on the plus sign next in the far left of the row.

This protein table is the same as the other protein tables found in the PEAKS DB, PEAKS PTM, and SPIDER results except the following columns:

- Significance: The protein significance is calculated as explained above.
- **Sample Profile:** The protein abundance among the samples is depicted as a heat map. Hold the cursor on a profile to view the sample channels, abundances, and ratios with respect to the base sample. Channel intensities are the sum of all reporter ion intensities from supporting peptides that pass the filters.
- **Group Profile:** The protein abundance among the groups is depicted as a heat map. The Group Profile is calculated by summing up the group areas of its used supporting peptides. Hold the cursor on a profile to view the group channels, abundances, and ratios with respect to the group which contains the base sample. Intensity values are the average of the sample profile intensities based on the selected grouping.

Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER) Section 3.4.2, "Protein Table" for more details of other entries of the protein table.

3.4.2. Protein Volcano Plot

The volcano plot facilitates identification of the significant proteins. This interactive volcano plot enables the selection of a protein in the table. Proteins located in the top-right and top-left sections of this plot (above the significance threshold and to the right and left of the fold change thresholds) are considered statistically significant. Green points are less intense relative to the reference group while red points are more intense relative to the reference group. Grayed points are filtered out as set in the filter. The volcano plot facilitates identification of significant proteins.

Protein markers that are in color can be clicked and selected. Upon clicking a point on the volcano plot, the corresponding protein in the Protein Table will be highlighted and more details are displayed below the Protein Table.

• Zoom in/out from Volcano Plot: Place the cursor on the volcano plot and drag a rectangle over the area of interest to Zoom in. By double clicking on the volcano plot you can zoom out and view the volcano plot with the default dimensions.

Figure 1. The volcano plot for proteins.



3.4.3. Protein Coverage

The coverage view characterizes the selected protein sequences at the amino acid level. Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 3.4.4, "Protein Coverage" for details. Click the supporting peptide sequence to open an XIC pop-up window.



3.4.4. Support Peptides

The supporting peptides assigned to the protein are shown in the tab beside the Coverage tab.

This table contains the following information of the quantifiable support peptides for the selected protein in the protein table.

- **Peptide:** The amino acid sequence of the peptide, as determined in PEAKS Search. A modified residue is followed by a pair of parentheses enclosing the modification mass. All non-unique peptides and ones not passing the filter will be excluded automatically.
- Used: Shows whether this peptide is used to calculate the protein profile. The top three peptides with highest abundance are used to calculate the protein abundance.
- **Quality:** This defines how well the peptide can be quantified. It is calculated by taking the average of the quality scores of the feature vectors.
- Avg. ppm: A peptide can have one or more feature vectors. Each feature vector has its own ppm calculated from the mass error of each feature. The average mass error of a peptide is the average of mass errors of all the feature vectors. This column displays the average mass error in ppm.
- **Sample Profile:** The peptide abundance among the samples is depicted as a heat map. Hold the cursor on a profile to view the samples, the peptide areas in samples, and the corresponding ratios with respect to the base sample. The channel intensities are the sum of intensities from all PSMs that identified the peptide.

- **Group Profile:** The peptide abundance among the groups is depicted as a heat map. Hold the cursor on a profile to view the groups, the peptide areas in groups, and the corresponding ratios with respect to the group which contains the base sample. The given intensities are the average intensities of sample intensities included in each group.
- Group Area: This column shows the peptide area in a group that is calculated by summing up the group areas of its feature vectors
- Max Ratio: This shows the maximum of the fold changes between a the given groups and the reference group.
- Start: This shows the position of the first residue of the peptide that covers the protein.
- End: This shows the position of the last residue (inclusive) of the peptide that covers the protein.
- PTM: The types and the numbers of modifications present in the peptide shown in color-coded icons.
- **Spectrum View:** The spectrum view for any peptide in the table can be displayed by clicking on the supporting peptide of interest, which will open a window displaying its Ion Match or Reporter Ion



3.4.5. Protein View Filter

Click the **Q** button on top-left of the protein table to open the protein filter and control which proteins to display.

Top Proteins O All Proteins Protein accession/name contains			
Protein description contains			
_			
Include Contaminant Data			
Include Contaminant Data Protein Contains Modifications Name to search for			CLEAR
Include Contaminant Data Protein Contains Modifications Name to search for ITRAQ 4plex (K, N-term) Deamidation (NQ)	Oxidat	ion (M)	CLEAR
Include Contaminant Data Protein Contains Modifications Name to search for ITRAQ 4plex (K, N-term) Deamidation (NQ) Carbamidomethylation	Oxidat	ion (M)	CLEAR

• **Top Proteins** : The number of proteins that have the highest number of unique peptides supporting them within their protein group.

- All Proteins : The number of proteins given the parameters set for PEAKS Protein Score (-10lgP) and number of unique peptides.
- **Protein Sequence contains** : Enter a protein sequence, only protein that contain that sequence will be displayed in the protein table
- **Protein description contains:** Enter a protein description, only protein that contain that description will be displayed in the protein table
- Include Contaminant Data: Check this to display contaminants that were identified during a PEAKS DB search.
- **Protein Contains Modifications** : Check boxes are provided for each identified modification.

3.5. Peptide View

The quantifiable peptide spectrum matches with the same sequence are grouped together to get the list of quantifiable peptides and are displayed in the **Peptide View** along with their detailed information.



3.5.1. Peptide Table

The peptide table contains the list of quantifiable peptides along with relevant information including the protein accession.

The view is similar to the support peptide table described in Section 3.4.4, "Support Peptides". The Peptide table entries may be filtered by using the **Peptide View Filters** on top of the peptide table, for more information refer to Section 3.5.1, "Peptide View Filters" in Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER).

3.5.2. Ion Match

The "Ion Match" tab in the bottom pane contains a table with possible fragment ions for the selected MS/MS scan. Each ion in the table shows the calculated mass. If the fragment ion is present in the spectrum, its mass is displayed in color. N-terminal ions are shown in blue and C-terminal ions are shown in red. A fragment ion is found when there is a matching peak within the mass error tolerance, as defined in the de novo sequencing parameters, and when relative intensity of the matching peak is at least 2%. The ion types displayed in the table can be configured in "Spectrum Annotation Settings" (Section 3.2.2, "Spectrum Annotation").

3.5.3. Reporter Ion

This view provides details on characteristics of the reporter ions used in the experiment. The table provides information on the normalized intensity, the raw intensity, and the ratio of each channel with respect to the set reference channel. Additionally, an associated MS2 or MS3 spectrum is included of the reporter ions in the experiment, determined by the selection made in the Quantification parameter settings, see Section 2, "Setting Parameters". This is an interactive spectrum, enabling zooming in and out using the scroll wheel of the mouse. Double-clicking will zoom out and reveal the entire MS3 spectrum.

nsityNormalized intensity	Patio							
	Ratio	Intensity						
3e+7 1.431e+7	1.00	9.00e+7 -		MsLevel:2	RT:42.30	Scan:2121	MZ:539.3414	z :2
2e+7 8.716e+7	6.09	8.00e+7 -						
1e+6 8.510e+6	0.59	7.00e+7 -						
7e+7 5.671e+7	3.96	6.00e+7 -						
		5.00e+7 -						
		4.00e+7 -						
		3.00e+7 -						
		2.00e+7 -						
		1.00e+7 -						
		0.00e+0		te de		at.	ati ata	_
20	9+78,7160+7 2+68,5100+6 0+75,6710+7	2+78,7160+7 6.09 2+68,5100+6 0.59 e+75,6710+7 3.96	8+7 8,716e+7 6,09 8.00+7 - 2+6 8,510+6 0.59 7.00+7 - e+7 5,671e+7 3.96 8.00+7 - 4.00+7 - 3.00+7 - 3.00+7 - 3.00+7 - 3.00+7 - 3.00+7 - 1.00+7 - 1.00+7 - 1.00+7 - 1.00+7 -	2+7 8.716e+7 6.09 8.00e+7 2+6 8.510e+6 0.59 7.00e+7 e+7 5.671e+7 3.96 8.00e+7 6.00e+7 3.96 8.00e+7 1.00e+7 3.00e+7 1.00e+7 2.00e+7 1.00e+7 1.00e+7 1.00e+7 1.00e+7 1.00e+7 1.00e+7 1.00e+7 1.00e+7	<u>8+78,716+7</u> 6,09 <u>8:00+7</u> <u>6:6716+7</u> 8,059 <u>6:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u>8:00+7</u> <u></u>	<u>9+7 8, 716+7 6,09</u> <u>e+6 8, 510e+6 0,59</u> <u>e+7 5, 671e+7 3,96</u> <u>8, 00e+7</u> <u>8, 00e+7</u> <u>8, 00e+7</u> <u>8, 00e+7</u> <u>8, 00e+7</u> <u>8, 00e+7</u> <u>1, 00e+7 <u>1, 00e+7</u> <u>1, 00e+7 <u>1, 00e+7</u> <u>1, 00e+7</u> <u>1, 00e+7 <u></u></u></u></u>	<u>8+78,716+7</u> 6,09 <u>8+68,510e+6</u> 0.59 <u>8+776,671e+7</u> 3,96 <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u> <u>8.00e+7</u>	<u>9+7 8, 17 6+7 6,09</u> <u>e+6 8, 510e+6 0,59</u> <u>e+7 5, 671e+7 3,396</u> 8,00e+7 - 4,00e+7 - 3,00e+7 - 2,00e+7 - 1,00e+7 - 1,00e+7 - 1,00e+7 - 1,10e 7 -

4. Ion Mobility Data

Ion mobility data, including Field asymmetric waveform ion mobility (**FAIMS**), and trapped ion mobility (**tims**) spectrometry analyzes complex mixtures and separates the ions based on their mobility. PEAKS Online has the ability to load IMS data and analyze it using *de novo* sequencing, database search (PEAKS DB), database search with unexpected modification identification (PEAKS PTM), sequence variant identification (SPIDER) search, spectral library search, and labelled and label free quantification.

The analysis result for IMS dataset contains the additional information for CV values compared to the non-IMS dataset. The CV-related information can be found at the following places for a TMT analysis.

• The Venn diagram on the summary page.

Venn diagram overlaps between different CVs in the search for all proteins, top proteins, and peptides are shown on the summary page.

- The additional **CV** column on the supporting peptide page
- The additional CV column in the exported Protein-Peptide CSV
- The additional CV column on the peptide page
- The additional CV filter on peptide page

The filter for peptide page contains one additional option for CV, which selects the displayed scans based on the selected CV values

• The additional **CV** column in the exported *Peptide CSV*

Chapter 11. Stable Isotope Labeling by Amino acids in Cell culture (SILAC)

1. Overview

Precursor ion quantification with isotope labels at the MS level is one of the three quantification modes that are supported by the optional PEAKS Q module for PEAKS Online. Typically, an unlabelled sample is mixed with a sample subjected to stable isotope labelling. The samples are then analyzed together in an LC-MS/MS experiment. The same peptide from different samples is recognized by a set of precursor ion peaks with similar retention time and mass differences equal to the expected difference caused by the stable isotope label. The ratio is calculated from the intensities of those peaks. PEAKS Q supports both user-defined labels and commercial quantification labels. PEAKS supports the analysis of Super-SILAC experiments that uses a mixture of SILAC-labeled cells as a spike-in standard for accurate quantification of other unlabeled or labeled samples.

The usage of this workflow is outlined below.

- 1. Refer to Chapter 5, Peptide De Novo Sequencing Section 2, "*DeepNovo/De Novo* Analysis Workflow and Parameter Settings", step 1-6 to create a new project and add data.
- 2. SILAC quantification can be run on PEAKS Database Search results. Select either "PEAKS Q" workflow in the DDA section of the Workflow step.
 - To run SILAC quantification on PEAKS Database Search results, select the "PEAKS Q" workflow in the DDA section.
- 3. Set the Identification Search Parameters.
 - If running SILAC quantification on a DDA DB Search analysis, refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER) Section 2, "Database Search Workflow and Parameter Settings"
- 4. For a SILAC workflow, the SILAC-specific parameters will be shown at the bottom of the Project Submission page. Set the SILAC sample grouping and Retention Time Error Tolerance. See Section 2 below for more details. SILAC quantification will be performed on the combined identification result.

Select Method	ds: SILAC-2plex (R10,K8)	RT Error Range: 1	min		
ch Between Runs					
Mass Error Tolerar Feature Intensit	nce: 20 y ≥: 0 :	Tolerance Unit: PPM	Retention Time Shift Tolerance(min)	1 Ar	uto Detect 🔽
le Groups	2				
	<u> </u>		Group	Condition	
Select All		≁° ⊡	Group ID Transfer Group Sample 4	Condition	×
	<u> </u>	+9 <u>0000</u> → 9000	Group ID Transfer Group Sample 4 Group 1 Group 1 Light Heavy	Condition × Condition 1 Condition 2	×
Used	Condition List	+°⊡ → %±	Group UD Transfer Group Group Group 1 Group 1 Group 1 Group 2 Group 2	Condition X Condition 1 Condition 2 X	×

- 5. Set PSM Filters. Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER) Section 2.8, "PSM Filter", for details.
- 6. Set the match between runs (MBR) parameters. MBR allows for identified MS/MS scans from one sample to be applied to LC-MS peptide features from other samples within the mass error and RT error tolerances.
 - Mass Error Tolerance: Controls the allowable difference in mass between the identified MS/MS scan in one sample versus LC-MS peptide features in other samples. Either Da or ppm can be selected as the mass unit.
 - Retention Time Shift Tolerance: The maximum allowable difference between the retention time of the identified MS/MS scan and the LC-MS peptide feature considered for MBR.
 - Feature intensity: Set a value for the minimum intensity a peptide feature must exhibit to be considered for MBR.
- 7. Review project settings and parameters, then click the Submit button at the top of the page.
- 8. Name the analysis, then click OK to start the analysis.

2. Setting Parameters

For more information on DB search parameters, refer to Chapter 5, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 2, "Database Search Workflow and Parameter Settings".

The following parameters are available when the SILAC Quantification workflow is selected.

- Select Method: From the Select Methods drop-down menu, choose the appropriate quantification method used in the experiment.
- Match Between Runs (MBR):

Match Between Runs	^
Mass Error Tolerance: 20 Tolerance Unit: PPM Retention Time Shift Tolerance(min): 1 Auto Detect	
CCS Error Tolerance: 0.05 Feature Intensity ≥: 0 :	

- Mass Error Tolerance: Controls the allowable difference in mass between the identified MS/MS scan in one sample versus features in other samples. This can be set similar to the precursor mass error tolerance used for database search.
- **Retention Time Shift:** The maximum allowable difference between the retention time of the identified MS/ MS scan and the peptide feature considered for MBR.
 - Auto Detect: Let PEAKS Online detect the optimal Retention Time Shift Tolerance for your data. An initial quantification run will be performed to determine the retention time shift trend. The optimal error tolerance will then be used for the full search.
- CCS Error Tolerance: The maximum collisional cross section (CCS) differential between an identified MS/ MS scan and peptide features in other samples to allow MBR.

Note

This parameter is only available when the analysis contains TimsTOF data.

- **Feature Intensity:** Set a value for the minimum intensity a peptide feature must exhibit to be considered for MBR. By default, the integrated area under the curve of the peptide feature will be used to represent feature intensity.
- **RT Error Range:** The retention time shift tolerance is the maximum elution time difference that is considered for the quantification of an identified peptide after RT alignment.

ID Transfer Group: If the "Fraction Assisted ID transfer" method was selected use the ***** button to add the ID transfer sample to the ID Transfer group

- **Groups**: The samples can be arranged into one or more groups. A group represents a state of the experiment where quantitative ratios between groups are investigated. Replicate MS samples should be put together in one group. Double-click on the group node to rename the group. Display all samples in each group or just the individual groups by selecting the Expand/Collapse buttons, respectively. Samples can be removed either individually, or all at once by clicking the 'X' next to the sample or the Remove All button, respectively.
- **Conditions**: The number of conditions depends on the quantification method selected. For example, SILAC-2plex(R10,K8) has 2 conditions and SILAC-3plex(R10,K8|R6,K4) has 3 conditions to study. Each label in a sample represents one unique condition. To remove a label from analysis, uncheck the corresponding condition. Condition names can be renamed.
- **Reference Condition**: The condition which is the denominator of the calculated ratios. The reference condition can be chosen from the selected conditions only.
- Peptide Feature Filter:

Peptide Feature Filter				
Avg. Area ≥:	0	1	10	Quality ≥: 1
Number of labels with Id present ≥:	1	Number of labels with feature present \geq :	1	Reference label present

- Average Area: The minimum average area of all peptide features associated with the peptide to be included in the result. Features with higher abundance have been shown to be more reproducible across replicates.
- Charge: Only peptides with feature vectors that all fall within this range will be used in protein quantification.
- **Quality:** A quality value will be applied to each quantified peptide to represent the likely reproducibility of the quantification result. Set the minimum quality score cut-off. Factors that affect the quality score include m/z difference, RT difference, XIC shape similarity, and the feature intensities.
- **Number of labels with ID present**: Only peptides with an identified MS/MS scan in the number of given labels will be included in the peptide table.
- Number of labels with feature present: Only peptides with a predicted LC-MS peptide features in the number of given labels will be included in the peptide table.
- **Reference label present:**Only quantifiable peptides in which the reference label peptide feature is present will be included in the peptide table if this is checked.
- Protein Filters:

Protein Filter

Significance ≥ 0	⋮ FDR (adjusted p-value)≥: 0 ⋮ 1 ⋮ ≤ Fold change ≤ 64 ⋮
Unique Peptides ≥: 1	: O Modified Form Exclusion

- **Significance:** Only protein groups with a significance above this threshold will be listed in the "Protein" view. The significance score is calculated as the -10lg of the significance testing p-value. Significance is calculated using welch's ANOVA.
- **FDR%** (**adjusted p-value**): Adopt the multiple hypothesis testing Benjamin-Hochberg method to adjust the p-value to the given false discovery rate. Only protein groups with significance scores passing the calculated FDR will be listed in the "Protein" view. Either this or "Significance" can be selected to set a significance threshold.
- Fold Change: The required magnitude of change of abundance between conditions. The minimum value is one because the fold change is calculated as the magnitude of change where the lower value is always the denominator.
- Unique Peptides: Only protein groups with unique peptides at or above this number will be listed in the "Protein" view. Peptides are considered unique if they are only matched to one protein group.
- **Modified Form Exclusion:** When checked, peptides that are identified as both a modified and unmodified form will be excluded from the protein ration calculation. This does not apply to the quantification labels. When a peptide is split between an unmodified and modified form, the ratio of both forms may differ from the protein ratio. Therefore, including modified peptides in the protein ratio calculation might lead to incorrect results. By checking this box, if a peptide has modified and unmodified forms, both will not be used as candidates, if a peptide only has one modified form, it is still a candidate, but if a peptide has more than one modified form, all forms will not be used as candidates.
- Normalization :

Normalization Methods

O Use TIC			
O Manual Input			
🔘 Use Internal Standar	d Proteins		
 No normalization 			
Select Experiment Sam	ple 5	*	
Labels Name	Factor	Expected Ratios	
Light Condition 1	-	1	
Heavy Condition 2	-	1	
		CLOSE	SAVE

- Use TIC: Using the total intensity of all quantifiable peptides from each condition, a global ratio is created. The global ratio is used as a normalization factor for all proteins and peptides.
- **Manual input:** This option performs the same default calculation as auto normalization ("Use TIC") but also allows for the manual change of each sample's expected ratios. This should be used whenever the expected ratio between conditions isn't 1:1.
- Use internal standard proteins: This option displays a list of identified proteins, which can be selected to create a normalization factor. A search bar is included to enable the quick location of select proteins.
- No Normalization: All samples are automatically assigned Factors of 1.0.

3. Understanding the SILAC Results

After a PEAKS Online SILAC quantification is complete, the SILAC summary page will display automatically. Beneath the Summary tab are the Protein and Peptide result tabs which are available to view.



- **Summary:** The outline of the SILAC search results with statistics. This is the place to examine the overall performance of the experiment and adjust filters.
- Proteins: The quantified proteins with a list of supporting peptide features for each protein.

• **Peptides:** The quantified peptides are listed in a table. Extracted ion chromatogram (XIC) and supporting peptide features are also displayed for each peptide.

3.1. Result Filtration

Click the analysis name of a SILAC analysis, then click Modify Parameters, in the Quantification section, the following parameters are available.

3.1.1. Protein Filter

Click the 🧪 button and the following parameter dialogue will pop up.

Protein Filters		
Significance O FDR(adjusted p-value)		
Significance ≥ 0 *		
Fold Change between 1 * and 64 *		
Used Peptides ≥ 1 :		
	CLOSE	SAVE

These filters determine what will be included in the protein group significance calculation, as well as what significance score thresholds will be set. The parameters used in the screenshot above indicate that no protein filter is applied, thus giving all quantifiable proteins.

- **Significance:** Only protein groups with a significance above this threshold will be listed in the "Protein" view. The significance score is calculated as the -10lg of the significance testing p-value. The significance is calculated using Welch's ANOVA.
- **FDR:** Adopt the Benjamin-Hochberg method for multiple hypothesis testing to adjust the p-value to the false discovery rate for all protein groups that have already passed the other filters. Only protein groups with significance scores passing the calculated FDR will be listed in the "Protein" view. Either this or "Significance" can be selected to set a significance threshold.
- **Fold Change:** Only protein groups at or above this fold change threshold will be listed in the "Protein" view. Fold changes are always listed where the more intense sample is the denominator. So the fold change is always greater than 1.
- Unique Peptides: Only protein groups with unique peptides at or above this number will be listed in the "Protein" view. Unique peptides are peptides that only support one protein group.

3.2. Summary View

The "SILAC Summary" view includes an expression profile with candidate proteins sorted in a heat map, result statistics table, and protein volcano plots. An "Analysis Description" can be added and displayed below the filter section.

3.2.1. Statistics of Filtered Results

Table 1 in the Summary page lists the filtered results of feature vectors and proteins.

3.2.2. Protein Profile Heatmap

The heatmap in Figure 1 displays the protein groups that pass the filters. The relative protein abundance is represented as a heat map of the representative proteins of each protein group. These representative proteins are clustered if they exhibit a similar expression trend across the samples. The hierarchical clustering is generated using a neighbour-joining algorithm with a Euclidean distance similarity measurement of the log2 ratios of the abundance of each sample relative to the average abundance. Similarly, the conditions in different samples are clustered if they exhibit a similar expression trend across the protein groups. If the number of proteins is too high, the heatmap will not be displayed.

3.2.3. Ratio Distribution

Figure 2 displays a histogram of the distribution of ratios relative to the reference condition for a specific sample. The sample displayed can be changed using the drop down menu.

3.2.4. The Volcano Plot for Proteins

A volcano plot (Figure 3 & Figure 4) combines a statistical test with the magnitude of the change enabling quick visual identification of those data-points that display large-magnitude changes and that are also statistically significant. The volcano plot plots significance versus fold-change of the quantified proteins.

- Horizontal dashed lines: The selected significance threshold.
- Vertical dashed lines: The selected fold change thresholds.

Plotting points in this way results in two regions of interest in the plot: those points that are found towards the top of the plot that are far to either the left or the right side. These represent values that display large magnitude fold changes as well as high statistical significance. The ratio is the group ratio set with respect to the base group.

Circles for the proteins that are above the set significance and fold change thresholds will be displayed in colour. Scrolling the cursor over a circle will show the protein info.

Note

Volcano plot cannot be displayed when multiple groups are selected.

3.2.5. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

2 data files produced no results. (Expand to see list of zero result data files)

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.

Potentially Problematic Data Sets	~
Sample 3 All fractions have no result.	
Sample 4 All fractions have no result.	

If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics

No Results Found

3.3. Protein View

The SILAC Protein page lists all the quantified proteins that pass the filter set on the SILAC Summary page, characterizes each protein at the amino acid level, and lists the supporting peptide features of each protein.

the OHO	and top trotonio											Export	LIMEN EI	<u>vientrep</u>
								<	1 - 100	of 4378 ▼ →		Condition 2 -	,	
	Accession	Condition 2: S	ignificance	-10LaP	#Pentides	#Unique	PTM .	Condition 2 : Ratio	Profile .	Group 1: Condition 2 F	Ratio		-	
1	009666IAHNK_HUMAN	Condition 2. C	328.92	437,43	#1 eptides	168 168	C D O	Condition 2 . Ratio	Tome	1.58		260	1	1
2	P63261 ACTG_HUMAN		244.86	423.52		83 1	C D O			1.39		500	•	
3	Q15149IPLEC_HUMAN		194.17	464.22		177 169	C D O			1.58		300		
4	P10809 CH60_HUMAN		174.70	368.81		52 52	CDO			1.52	-	250		
5	P21333 FLNA_HUMAN		170.70	454.60		118 108	CDO			1.33	901	200	2	
6	P54652 HSP72_HUMAN		137.06	283.45		22 1	CD			1.62	fica	150	·	
7	P35579 MYH9_HUMAN		129.47	408.62		84 77	CD			1.58	.ee	100		
8	P07237 PDIA1_HUMAN		129.45	311.48		30 29	C			2.10	0	50	1	-
9	P02545 LMNA_HUMAN		124.48	322.14		44 43	CDO			1.83	*	0	Section of the local division of the local d	
											•	-50	2 1 2 4 8 16	3 32 64
Coverage	e Peptide											Rati	ios	
)9666 <mark> </mark> /	AHNK_HUMAN Neuroblas	t differentiation-a	ssociated prote	in AHNAK (OS=Homo s	apiens GN=AHNA	K PE=1 SV=2						PTM FILT	ER TOOI
	1 MEKEETTR EL I	LLPNWQGSGS	HGLTIAQRI	DD GVFV	QEVTQN	SPAAR TGVVK	EGDQIVG	ATI YFDNL(QSGEV	TQLLNTMGHE				
	81 TVGLKLHRKG I	DRSPEPGQTW	TR EVFSSC	SS EVVL	SGDDEE	YQR IYTTKIK	PRLK SED	GVE GDLGE	TQSRT	ITVTRR VTAY	1			
1	61 TVDVTGREGA H	DIDISSPEF	KIKIPR HE	LT EISN	VDVETQ	SGKTVIRLPS	GSGAASP	TGS AVDIR	AGAIS	ASGPELQGAG				
2	41 HSKLOVTMPG	IKVGGSGVNV	NAKGLDLG	GR GGVO	VPAVDI	SSSLGGRAVE	VOGPSLE	SGD HGKIKI	FPTMK	VPKFGVSTGF				
_														
3	21 EGQTPKAGLR	VSAPEVSVGH	KGGKPGLT	IQ APQL	EVSVPS	ANIEGLEGKI	KGPQITG	PSL EGDLG	LKGAK	PQGHIGVDAS				
4	01 APQIGGSITG I	PSVEVQAPDI	DVQGPGSKI	LN VPKM	IKVPKFS	VSGAK GEETG	IDVTLPT	GEV TVPGV	SGDVS	LPEIATGGLE	1			
4	81 GK MKGTKVK T	PEMIIQKPKI	SMQDVDLSI	LG SPKL	KGDIKV	SAPGVQGDVK	GPQVALK	GSR VDIET	PNLEG	TLTGPR LGS H	2			
5	61 SGKTGTCR IS	MSEVDLNVAA	PKVKGGVD	VT LPRV	EGKVKV	PEVDVRGPK	DVSAPDV	EAH GPEWN	LKMPK	MKMPTFSTPG	1			
6	41 AKGEGPDVHM	flpk gdisis	GPK VNVEA	PD VNLE	GLGGKL	KGPDVKLPDM	SVKTPK	SMP DVDLH	VKGTK	VKGEYDVTVE				
7	21 KLEGELKGPK	VDIDAPDVDV	HGPDWHLK	MP KMKM	IPK FSVP	GFKAEGPEVD	VNLPK AD	VDI SGPK I	DVTAP	DVSIEEPEGF	1			
0	Dentida											1/04 1/32 1/10 1/8 1/4 1/2 Datic	1 2 9 0 10	22 04
coverage	Teplide												1 - 14 of 14	• →
# :	Peptide		Used .	-10lgP 📋 :	Avg. Area	Condition 2 : Ratio .	Condition 3 : Ra	atio : Group 1 : C	Group 1 : C	#Vector	Start :	End :	PTM	
1 V	/NAGHGLTYHNVK(*)		1	81.07	5.87	e+6		0.00	0	1,94 9	189	201		
2 A	AIAAIPEM(+15.99)HELNIGHAIIGR(*)		~	72.11	9.04	e+6		0.00	0	.97 6	202	221 0		
3 A	AIAAIPEMHELNIGHAIIGR(*)		~	65.23	6.82	e+6		0.00	1	.02 6	202	221		
4 N			1	50.00	2.18	e+0		0.00	1	05 4	190	201		
6 T	DAEOAOELAR(*)		~	59.84	7.24	e+6		0.00	1	.05 3	164	174		
7 A	ATFAASLGLK(*)		~	55.58	8.92	e+6		0.00		1.97 3	178	188		
8 A	AIPEMHELNIGHAIIGR(*)		1	52.43	2.90	e+6		0.00	C	1.99 3	204	221		
9 Q	EVTTEGGLDVAGQR(*)		 V 	48.84	5.33	e+6		0.00	1	.94 2	99	113		
10 A	AGHGLTYHNVK(*)		~	43.69	9.96	e+5		0.00	ç	.22 1	191	201		
11 G	STAYPDPVQAAFIAEQAGADGITVHLR	t(*)	~	43.36	1.28	e+7		0.00	1	.05 3	21	47		
12 L	GVNIDHIATER(*)	DHEC(1E7.02)C(1E7.0	~	41.50	3.56	e+5		0.05	0	1.50 2	6	1/	0	
10 N	WINLEW(T10.99)AVTEEMLAIAVETK(*	·)=nru(+07.02)0(+07.0.		34.72	3.30	et0		0.00		1.00 1	222	97 O C 1	C	
14 19	sentinen(")		~	34.13	2.37	erg	-	0.00	L L	new I	222	220		

3.3.1. Protein Table

The quantified proteins are listed in the protein table with homologous proteins grouped together.

This protein table is the same as the other protein tables found in the PEAKS DB, PEAKS PTM, and SPIDER results except the following columns:

• Significance: The protein significance is calculated as explained in Section 1, "Overview" in this chapter.

- **Condition Profile:** The protein abundance among the samples is depicted as a heat map. Hold the cursor on a profile to view the sample channels, abundances, and ratios with respect to the base sample.
- **Condition Ratio:** The ratio columns show the peptide abundance in each condition for all groups. A peptide's abundance in each condition is the sum of all supporting feature vectors abundance within the filters. Then the peptide abundance in each condition is compared to the reference condition to obtain the peptide condition ratio in each sample separately. The peptide condition ratio of a group is computed by calculating the median from the peptide condition ratios in the samples in that group.

Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER) Section 3.4.2, "Protein Table" for more details of other entries of the protein table.

3.3.2. Protein Volcano Plot

The volcano plot facilitates identification of the significant proteins. This interactive volcano plot enables the selection of a protein in the table. Proteins located in the top-right and top-left sections of this plot (above the significance threshold and to the right and left of the fold change thresholds) are considered statistically significant. Green points are more abundant in the reference sample while red points are less abundant in the reference sample. Grayed points were filtered out.

Protein markers that are in color can be clicked and selected. Upon clicking a point on the volcano plot, the corresponding protein in the Protein Table will be highlighted and more details are displayed below the Protein Table.

- Selecting a Different Condition: Click the condition dropdown above the volcano plot to display all volcano plot options, one for each experimental condition. Each volcano plot will show a different set of proteins that are relevant to that condition.
- Zoom in/out from Volcano Plot: Place the cursor on the volcano plot and drag a rectangle over the area of interest to Zoom in. By double clicking on the volcano plot you can zoom out and view the volcano plot with the default dimensions



3.3.3. Protein Coverage

The coverage view characterizes the selected protein sequences at the amino acid level. Refer to Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 3.4.4, "Protein Coverage" for details.

 ${\tt Q09666} | {\tt AHNK_HUMAN \ Neuroblast \ differentiation-associated \ protein \ {\tt AHNAK \ OS=Homo \ sapiens \ GN=AHNAK \ PE=1 \ SV=2}$

1	MEKEETTR EL	LLPNWQGSGS	HGLTIAQRDD	GVFVQEVTQN	SPAAR TGVVK	EGDQIVGATI	YFDNLQSGEV	TQLLNTMGHH
81	TVGLKLHRKG	DRSPEPGQTW	TR EVFSSCSS	EVVLSGDDEE	YQR IYTTKIK	PRLK sedgve	GDLGETQSRT	ITVTRR VTAY
161	TVDVTGREGA	K didisspef	K IKIPR HELT	EISNVDVETQ	SGKTVIRLPS	GSGAASPTGS	AVDIRAGAIS	ASGPELQGAG
241	HSKLQVTMPG	IKVGGSGVNV	NAK (R@186)R	GGVQVPAVDI	SSSLGGRAVE	VQGPSLESGD	HGK IKFPTMK	VPKFGVSTGR
321	EGQTPKAGLR	VSAPEVSVGH	KGGKPGLTIQ	APQLEVSVPS	ANIEGLEGKL	KGPQITGPSL	EGDLGLK GAK	PQGHIGVDAS
401	APQIGGSITG	PSVEVQAPDI	DVQGPGSKLN	VPKMKVPKFS	VSGAK GEETG	IDVTLPTGEV	TVPGVSGDVS	LPEIATGGLE
481	gk mkgtkvk t	PEMIIQKPKI	SMQDVDLSLG	SPK LKGDIKV	SAPGVQGDVK	GPQVALKGSR	VDIETPNLEG	tltgpr Lgsp
561	SGKTGTCR IS	MSEVDLNVAA	PK VKGGVDVT	LPRVEGKVKV	pevdvrgpk v	DVSAPDVEAH	GPEWNLK MPK	MK mptfstpg
641	AKGEGPDVHM	TLPK GDISIS	GPK vnveapd	VNLEGLGGKL	KGPDVKLPDM	SVKTPK ismp	dvdlhvk GTK	VKGEYDVTVP
721	KLEGELKGPK	VDIDAPDVDV	HGPDWHLKMP	KMKMPK fsvp	GFKAEGPEVD	vnlpk advdi	SGPK IDVTAP	DVSIEEPEGK
801	LKGPKFK mpe	MNIK VPKISM	PDVDLHLKGP	NVK GEYDVTM	PK VESEIKVP	DVELKSAK MD	IDVPDVEVQG	PDWHLK MPKM

3.3.4. Support Peptides

The supporting peptides assigned to the protein are shown in a separate tab beside the Protein Coverage. The sequence can be clicked and a pop-up window will show up to display the feature details.

This table contains the following information of the quantifiable support peptides for the selected protein in the protein table.

- **Peptide:** The amino acid sequence of the peptide, as determined by the PEAKS Search. A modified residue is followed by a pair of parentheses enclosing the modification mass. All non-unique peptides and ones not passing the filter will be excluded automatically.
- Used: Shows whether this peptide is used to calculate the protein profile. All peptides which pass the peptides filters are used to calculate the protein abundance.
- **-10LgP:** The peptide -10LgP score. The score indicates the likelihood that the peptide-spectrum match identification is correct.
- Avg. Area: Each peptide feature associated with the peptide has its own area under the XIC curve. The average area of is the average area of all the features associated with the peptide.
- **Condition Profile:** The relative peptide abundance for each condition compared to the reference condition among the samples is depicted as a heat map. Hold the cursor on a profile to view the samples, the areas, and the ratios with respect to the reference condition.
- **Condition Ratio:** The ratio columns show the relative peptide abundance for each condition with respect to the reference condition for all groups. A peptide's abundance in each condition is the sum of its supporting peptide feature abundances within the peptide filters. Then the peptide abundance in each condition is compared to the reference condition to obtain the peptide condition ratio in each sample separately. The peptide condition ratio of a group is computed by calculating the median from the peptide condition ratios of the samples in that group.
- **#Vector:** This shows the number of quantifiable feature vectors of a peptide. A feature vector is a full set of peptide features used to calculate a ratio.
- Start: This shows the protein position of the first residue of the peptide that covers the protein.
- End: This shows the protein position of the last residue (inclusive) of the peptide that covers the protein.
- **PTM:** The types and the numbers of modifications present in the peptide shown in color-coded icons. Scroll over the modification icon to see the modification name and mass.
- Accession: The proteins supported by the peptide, separated by ';' if the peptide supports multiple proteins. If none of the proteins that the peptide supports pass the protein filters, the accession field will be empty.

3.3.5. Protein View Filter

Click the **Q** button on top-left of the protein table to open the protein filter and control which proteins to display.

Protein Filter	
Top Proteins All Proteins	
Protein accession/name contains	
Protein description contains	
Protein Contains Modifications	
Name to search for	EAR
Deamidation (NQ) Oxidation (M) Carbamidomethylation	
RESET CANCEL	APPLY

- **Top Proteins**: The number of proteins that have the highest number of unique peptides supporting them within their protein group.
- All Proteins: The number of proteins given the parameters set for PEAKS Protein Score(-10lgP) and number of unique peptides.
- **Protein Sequence contains**: Enter a protein sequence, only proteins that contain that sequence will be displayed in the protein table
- **Protein description contains:** Enter a protein description, only proteins that contain that description will be displayed in the protein table
- **Include Contaminant Data:** Check this to display identifications from the contaminant database that were identified during a PEAKS DB search.
- Protein Contains Modifications: Check boxes are provided for each identified modification.

3.4. Peptide View

The quantifiable feature vectors with the same sequence are grouped together to get the list of quantifiable peptides and are displayed in the **Peptide View** along with their detailed information.



3.4.1. Peptide Table

The peptide table contains the list of quantifiable peptides along with relevant information including the protein accession.

The view is similar to the support peptide table described in Section 3.3.4, "Support Peptides", with some new columns.

- Quality Score: The quality score represents the predicted reproducibility of the quantification result. Factors that affect the quality score include m/z difference, RT difference, XIC shape similarity, and the feature intensities.
- Accession: The proteins supported by the peptide, separated by ',' if the peptide supports multiple proteins. If none of the proteins that the peptide supports pass the protein filters, the accession field will be empty.

The Peptide table entries may be filtered by using the **Peptide View Filters** on top of the peptide table, for more information refer to Section 3.5.1, "Peptide View Filters" in Chapter 6, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER).

3.4.2. Peptide View Filters

Click the button on the upper left corner of the peptide table to open the peptide filter and control which peptides to display.

- **Peptide Sequence Contains:** Enter a peptide sequence, only peptides that contain that sequence will be displayed in the peptide table
- **Protein Accession Contains:** Enter a peptide accession, only peptides that contain that accession will be displayed in the peptide table
- Show Transferred IDs Only: Only display peptides where match between runs was used to include at least one peptide feature in the result.
- **Ratio Filter:** Only peptides where the ratio for the selected group and condition fall within the specified range will be displayed in the peptide table.
- **Peptide Contains Modifications:** Use the checkboxes to indicate which modified peptides to include in the result.

3.4.3. Feature Details

The Feature Details contains the eXtracted Ion Chromatogram (XIC) chart, Ion Match, and the Reporter Ion Spectrum.

- **Feature Menu:** It is under the peptide table. The feature menu displays the information of the feature that is being displayed in the spectrum.
 - **Samples:** If a peptide is quantified in more than one sample, click the "SAMPLE" drop-down menu to view the Spectrum Annotation of the other PSMs.
 - **Feature:** Click the Feature drop down menu to show all peptide features included in the result: -10lgP, Avg. Area, Condition Intensity, Condition Ratio, Id Count, m/z, z, RT, and ppm for each feature are given.
 - If the transferred column for the feature displays "true" that means the ID for at least one feature was transferred.
 - **PSM Label:** Click the PSM Label drop-down menu to show other identified MS/MS scans associated with the peptide.

3.4.3.1. eXtracted Ion Chromatogram (XIC) chart

This map presents the changes in feature intensity over the retention time range of the peaks for all labels of the selected peptide. Additionally, the coloured isotopic distribution plot -M0, M+1, M+2, etc. - of each label is included, which shows the relative composition of each label.



3.4.3.2. Ion Match

This section displays the spectrum annotation and other information about the peptide selected in the Peptide table. When multiple spectra are matched to the peptide, only the information about the top-scoring PSM is shown by default. The interface of this section is identical to the Spectrum Annotation in the PEAKS De Novo result. For detailed instructions, refer to Section 3.5, "Spectrum Annotation" in Chapter 5, Peptide *de novo* Sequencing.



3.4.3.3. MS1 Spectrum

This section provides an MS1 spectrum zoomed in on the reporter ions in the experiment. This is an interactive spectrum, enabling zooming in and out using the scroll wheel of the mouse. Double-clicking will zoom in on an area of interest. Clicking and dragging will slide the spectrum to the left or right.



3.5. Result Exporting

PEAKS SILAC results can be exported to various text formats in the Export page. The protein and peptide results can also be exported from the Protein and Peptide page, respectively.

Export SELECT	ALL CLEAR ALL		
All Search Parameters	Denovo	Database Search	SILAC
Data Refinement	Summary Table	Summary Table	Summary Table
MGF File	Denovo CSV	Protein CSV	Protein CSV
Mzxml	Denovo CSV with all Candidates	Protein-Peptide CSV	Protein-Peptide CSV
		Protein Fasta	Peptide CSV
		Peptide CSV	Feature CSV
		PSM CSV	
		Peptides-Pepxml	
		Peptides-MzIdentML	
		Denovo Only CSV	
		Denovo Only CSV with all Candidates	
DOWNLOAD DOWNLOAD BY	SAMPLE		

- **Proteins CSV:** These are the quantified proteins that pass the filters set in the "Summary" page and their associated details.
- Protein-Peptide CSV: These are the supporting peptides of the quantified proteins and their associated details.
- **Peptide CSV:** A complete list of all quantified peptides that pass the peptide feature filters set in the "Summary" page and their associated details.
- Feature CSV: All predicted peptide features and their associated quantification information. Both identified and unidentified peptide features are included. To include unidentified peptide features, set Peptide Id Count equals to 0 in the Peptide feature filter.

Note

Exporting by sample will begin downloading the same .zip folder, since this does not apply to SILAC quantification.

Chapter 12. Fraciton Assisted ID Transfer SILAC

1. Overview

Precursor ion quantification with isotope labels at the MS level is one of the three quantification modes that are supported by the optional PEAKS Q module for PEAKS Online. In this mode, the isotope labels with different mass values are introduced as two or more samples. The samples are then analyzed together in an LC-MS/MS experiment. The same peptide from different samples is recognized by a set of precursor ion peaks with similar retention time and mass differences within the retention time window. The ratio is calculated from the intensities of those peaks. PEAKS Q supports both user-defined labels and commercial quantification labels. PEAKS supports the analysis of Super-SILAC experiments that uses a mixture of SILAC-labeled cells as a spike-in standard for accurate quantification of other unlabeled or labeled samples.

Compared to the regular SILAC, ID transfer can provide quantification workflow with high throughput and coverage. One group is designated as the ID Transfer group. It is used to associate peptide features from other samples to identified MS/MS scans in the ID transfer group. It is recommended that the ID Transfer group be a highly fractionated sample. The experimental design is outlined below.

- To get high coverage
 - 1. SILAC sample -> fractionation -> LC-MS runs (only once)
 - 2. Get a list of SILAC pairs by combining all fractions
- To get high throughput
 - 1. SILAC sample -> LC-MS (routine)
 - 2. Transfer SILAC pairs from pre-run fractions while performing data analysis

In the ID transfer SILAC, the analysis steps are outlined below.

- Get expected SILAC pairs from the fractionated sample
 - 1. Search against a target protein database using PEAKS DB.
 - 2. Generate a SILAC library for future ID transfer
- Routine SILAC quantification
 - 1. Find SILAC pairs directly from LC-MS/MS
 - 2. For remaining unidentified precursors/pairs, SILAC pairs are transferred from the ID transfer group run previously
 - 3. Combine 1 and 2 to get the final quantification results

The usage of this workflow is outlined below.

- 1. Refer to Chapter 4, Peptide De Novo Sequencing Section 2, "*DeepNovo/De Novo* Analysis Workflow and Parameter Settings", step 1-6 to create a new project and add data.
- 2. SILAC quantification can be run on PEAKS Database Search results. Select the "Fractionation Assisted ID Transfer Quantification" workflow in the DDA section of the Workflow step.
 - To run SILAC quantification on PEAKS Database Search results, select the "Fractionation Assisted ID Transfer Quantification" workflow in the DDA section of the Workflow step.

- 3. Set Search Parameters.
 - If running SILAC quantification on a DDA DB Search analysis, refer to Chapter 5, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), LINK, for details. Section 2, "Database Search Workflow and Parameter Settings"
- 4. For an ID Transfer SILAC workflow, the SILAC-specific parameters will be shown at the bottom of the Project Submission page. Choose an ID Transfer Group and set the SILAC sample grouping as well as the Retention Time Error Tolerance. See Section 2 below for more details. ID Transfer SILAC quantification will be performed on the combined identification result.

Select Methods:	SILAC-2plex (R10,K8)	RT Error Range: 1	min		
ch Between Runs					
Mass Error Tolerance Feature Intensity ≥	20 :: 0 :	Tolerance Unit: PPM	 Retention Time Sh Tolerance(min 	lift 1 Auto De	etect 🔽
le Groups □ Select All _Q			Group	Condition	
le Groups		-92	Group ID Transfer Group Sample 4 Court 1	Condition	×
le Groups ☐ Select All <u>Q</u>		→ ⁰ → ⁰ 11 11	Group ID Transfer Group Sample 4 Group 1 Sample 5 Light Heavy	Condition X Condition 1 Condition 2	×
le Groups □ Select All <u>Q</u> Used	Condition List	+900 → %±	Group Group Group Group 1 Group 1 Group 2 Group 2 Group 2 Group 6	Condition × Condition 1 • Condition 2 •	×

- 5. Set the PSM Filters. Refer to Chapter 5, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER) Section 2.8, "PSM Filter", for details.
- 6. Review project settings and parameters, then click the Submit button at the top of the page.
- 7. Name the analysis, then click OK to start the analysis.

2. Setting Parameters

For more information on DB search parameters, refer to Chapter 5, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 2, "Database Search Workflow and Parameter Settings".

The following parameters are available when a SILAC Quantification workflow is selected.

- Select Method: From the Select Methods drop-down menu, choose the appropriate quantification method used in the experiment.
- **RT Error Range:** The retention time shift tolerance is the maximum elution time difference that is considered for the quantification of an identified peptide after RT alignment.
- Match Between Runs Match Between Runs is enabled for all ID Transfer SILAC. For more information on Match Between Runs, refer to Chapter 10, Stable Isotope Labeling by Amino acids in Cell culture (SILAC), Section 2, "Setting Parameters".

• Samples: The MS samples are listed for selection. After selecting a sample, it can be added to a new group by

clicking the * button or to an existing group by clicking the

button and selecting that option from the \rightarrow

drop-down menu. To add multiple samples into their own separate groups, click the butt

- **Groups**: The samples can be arranged into one or more groups. A group represents a state of the experiment where quantitative ratios between groups are investigated. Replicate MS samples should be put together in one group. Double-click on the group node to rename the group. Display all samples in each group or just the individual groups by selecting the Expand/Collapse buttons, respectively. Samples can be removed either individually, or all at once by clicking the 'X' next to the sample or the Remove All button, respectively.
- **Conditions**: The number of conditions depends on the quantification method selected. For example, SILAC-2plex(R10,K8) has 2 conditions and SILAC-3plex(R10,K8|R6,K4) has 3 conditions to study. Each label in a sample represents one unique condition. To remove a label from analysis, uncheck the corresponding condition.
- **Reference Condition**: The condition which is the denominator of the calculated ratios. The reference condition can be chosen from the selected conditions only.
- Peptide Feature Filter:

Peptide	Feature	Filter	

Avg. Area ≥:	0	:	1	*	≤ Charge ≤	10	:	Quality ≥:	1	:
Number of labels with Id present \geq :	1	Number o	f labels with	feature	e present ≥:	1	* *	Reference	ce label prese	ent

- Average Area: The minimum average area of all peptide features associated with the peptide to be included in the result. Features with higher abundance have been shown to be more reproducible across replicates.
- Charge: Only peptides with feature vectors that all fall within this range will be used in protein quantification.
- **Reference Label:** Only quantifiable peptides in which the reference label is present will be displayed if this is checked.
- Number of labels with ID present: Only protein groups with a number of labels with ID present at or above this number will be listed in the "Protein" view.
- Number of labels with feature present: Only protein groups with a number of labels with feature present at or above this number will be listed in the "Protein" view.
- Protein Filters:

Protein Filter

Significance ≥ 0	FDR (adjusted p-value)≥: 0 i 1 i ≤ Fold change ≤ 64 i
Unique Peptides ≥: 1	: D Modified Form Exclusion

- **Significance:** Only protein groups with a significance above this threshold will be listed in the "Protein" view. The significance score is calculated as the -10lg of the significance testing p-value. PEAKS provides the significance testing method: ANOVA.
- **FDR%** (**adjusted p-value**): Adopt the Benjamin-Hochberg method to adjust the p-value to the false discovery rate for all protein groups that have already passed the other filters. Only protein groups with significance scores passing the calculated FDR will be listed in the "Protein" view. Either this or "Significance" can be selected to set a significance threshold.

- Fold Change: Only protein groups at or above this fold change threshold will be listed in the "Protein" view.
- Unique Peptides: Only protein groups with unique peptides at or above this number will be listed in the "Protein" view. Peptides are considered unique if they are only matched to one protein group.
- **Modified Form Exclusion:** When checked, peptides that are modified with a variable PTM will not be included in the significance calculation for a protein group. This does not apply to the quantification labels. The expression level of modified peptides might be different than the overall expression level of a protein. Therefore, including modified peptides in the protein ratio calculation might lead to incorrect results. By checking this box, if a peptide has modified and unmodified forms, both will not be used as candidates, if a peptide only has one modified form, it is still a candidate, but if a peptide has more than one modified form, all forms will not be used as candidates.

• Normalization :

Normalization Methods

🔿 Use T	IC			
🔿 Manu	al Input			
🔿 Use Ir	nternal Standard	Proteins		
No no	ormalization			
Select Ex	periment Samp	ole 5	*	
Labels	Name	Factor	Expected Ratios	
Light	Condition 1	-	1	
Heavy	Condition 2	-	1	
			CLOSE	SAVE

- Use TIC: Using the total intensity of all quantifiable peptides from each condition, a global ratio is created. The global ratio is used as a normalization factor for all proteins and peptides.
- **Manual input:** This option performs the same default calculation as auto normalization ("Use TIC") but also allows for the manual change of each sample's expected ratios. This should be used whenever the expected ratio between conditions isn't 1:1.
- Use internal standard proteins: This option displays a list of identified proteins, which can be selected to create a normalization factor. A search bar is included to enable the quick location of select proteins.
- No Normalization: All samples are automatically assigned Factors of 1.0.

3. Understanding the SILAC Results

After a PEAKS Online SILAC quantification is complete, the SILAC summary page will display automatically. Beneath the Summary tab are the Protein and Peptide result tabs which are available to view.



- **Summary:** The outline of the SILAC search results with statistics. This is the place to examine the overall performance of the experiment and adjust filters.
- **Proteins:** The quantified proteins with a list of supporting peptide features for each protein.
- **Peptides:** The quantified peptides are listed in a table. Extracted ion chromatogram (XIC) and supporting peptide features are also displayed for each peptide.

3.1. Protein Filter

Click the 🧪 button and the following parameter dialogue will pop up.

Protein Filters		
Significance O FDR(adjusted p-value)		
Significance ≥ 0 *		
Fold Change between 1 and 64		
Used Peptides ≥ 1		
	CLOSE	SAVE

These filters determine what will be included in the protein group significance calculation, as well as what significance score thresholds will be set. The parameters used in the screenshot above indicate that no protein filter is applied, thus giving all quantifiable proteins.

- **Significance:** Only protein groups with a significance above this threshold will be listed in the "Protein" view. The significance score is calculated as the -10lg of the significance testing p-value. PEAKS provides significance testing method: ANOVA.
- **FDR:** Adopt the Benjamin-Hochberg method to adjust the p-value to the false discovery rate for all protein groups that have already passed the other filters. Only protein groups with significance scores passing the calculated FDR will be listed in the "Protein" view. Either this or "Significance" can be selected to set a significance threshold.
- **Fold Change:** Only protein groups at or above this fold change threshold will be listed in the "Protein" view. Fold changes are always listed where the more intense sample is the denominator. So the fold change is always greater than 1.
- Used Peptides: Only protein groups with unique peptides at or above this number will be listed in the "Protein" view. Unique peptides are peptides that only support one protein group.

3.2. Summary View

The "SILAC Summary" view includes an expression profile with candidate proteins sorted in a heat map, result statistics table, and protein volcano plots. An "Analysis Description" can be added and displayed below the filter section.

3.2.1. Statistics of Filtered Results

Table 1 in the Summary page lists the filtered results of feature vectors and proteins.

3.2.2. Protein Profile Heatmap

The heatmap in Figure 1 displays the protein groups that pass the filters. The relative protein abundance is represented as a heat map of the representative proteins of each protein group. These representative proteins are clustered if they exhibit a similar expression trend across the samples. The hierarchical clustering is generated using a neighbour-joining algorithm with a Euclidean distance similarity measurement of the log2 ratios of the abundance of each sample relative to the average abundance. Similarly, the conditions in different samples are clustered if they exhibit a similar expression trend across the protein groups. If the number of proteins is too high, the heatmap will not be displayed.

3.2.3. Protein Profile Heatmap

A volcano plot (Figure 2 & Figure 3) combines a statistical test with the magnitude of the change enabling quick visual identification of those data-points that display large-magnitude changes and that are also statistically significant. The volcano plot in Figure 1 plots significance versus fold-change of the quantified proteins.

- Horizontal dashed lines: The selected significance threshold.
- Vertical dashed lines: The selected fold change thresholds.

Plotting points in this way results in two regions of interest in the plot: those points that are found towards the top of the plot that are far to either the left or the right side. These represent values that display large magnitude fold changes (hence being left or right of center) as well as high statistical significance (hence being towards the top). The ratio is the group ratio set with respect to the base group. If more than two groups are present, the max ratio is used instead.

Circles for the proteins that are above the set significance and fold change thresholds will be displayed in colour. Scrolling the cursor over a circle will show the protein info.

3.2.4. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

```
2 data files produced no results. (Expand to see list of zero result data files)
```

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.

Potentially Problematic Data Sets	~
Sample 3 All fractions have no result.	
Sample 4 All fractions have no result.	

If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics

No Results Found

3.3. Protein View

The SILAC Protein page lists all the quantified proteins that pass the filter set on the SILAC Summary page, characterizes each protein at the amino acid level, and lists the supporting peptide features of each protein.

et onor	wing rop roteins											Export	Flotein F	Totentrepti
								←	1 - 100	of 4378 ▼ →		Condition 2	·	
	Accession	Condition 2: S	ignificance	-101 aP	#Pentides	#Inique	PTM .	Condition 2 · Ratio	Profile	Group 1: Condition 2	Ratio		-	
1	009666IAHNK HUMAN	Condition 2. C	328.92	437.43	wi epilded	168 16		Condition 2 . Hallo	Tomic	1.58	-		1	
2	P63261 ACTG_HUMAN		244.86	423.52	2	83				1.39		300	•	
3	Q15149 PLEC_HUMAN		194.17	464.22	2	177 16				1.58		300		
4	P10809 CH60_HUMAN		174.70	368.81		52 5	2 C D O			1.52	0	250		
5	P21333 FLNA_HUMAN		170.70	454.60)	118 10	BCDO			1.33	2	200	2	
6	P54652 HSP72_HUMAN		137.06	283.45	5	22	1 C D			1.62	fice	150	·	
7	P35579 MYH9_HUMAN		129.47	408.62	2	84 7	7 C D 🛛			1.58	B	100	- See	
8	P07237 PDIA1_HUMAN		129.45	311.48	3	30 2	9 C			2.10	0	50	1 · · ·	•
9	P02545 LMNA_HUMAN		124.48	322.14	1	44 4	3 C D O			1.83	*	0	and the second s	
											+	-50	2 1 2 4 8 1	6 32 64
Coverage	Peptide											Rat	ios	
)9666 <mark> </mark> A	HNK_HUMAN Neurobl	ast differentiation-a	ssociated prot	ein AHNAK	OS=Homo s	apiens GN=AHNA	K PE=1 SV=2						PTM FILT	TER TOOLS
	1 MEKEETTR EL	LLPNWQGSGS	HGLTIAQR	DD GVFV	QEVTQN	SPAAR T GVVI	K EGDQIVGA	TI YFDNL	QSGEV	TQLLNTMGH	н			
8	81 TVGLKLHRKG	DRSPEPGQTW	TR evfssc	SS EVVI	SGDDEE	YQRIYTTKI	PRLK SEDG	VE GDLGE	rqsr T	ITVTRR VTA	Y			
1 (61 TVDVTGREGA	KDIDISSPEF	K IKIPR HE	LT EISN	VDVETQ	SGKTVIRLP:	GSGAASPI	GS AVDIR	AGAIS	ASGPELQGA	G			
24	41 HSKLOVTMPG	IKVGGSGVNV	NAKGLDLG	GR GGVC	VPAVDI	SSSLGGRAV	VOGPSLES	GD HGKIK	FPTMK	VPKFGVSTG	R			
_						,					_			
32	21 EGQTPKAGLR	VSAPEVSVGH	KGGKPGLT	IQ APQI	EVSVPS	ANIEGLEGK	KGPQITGE	SL EGDLG	GAK	PQGHIGVDA	5			
4 (01 APQIGGSITG	PSVEVQAPDI	DVQGPGSK	LN VPKN	IKVPKFS	VSGAK GEET	J IDVTLPTO	EV TVPGV	SGDVS	LPEIATGGL	E			
48	81 GK MKGTKVK T	PEMIIQKPKI	SMQDVDLS	LG SPKI	LKGDIKV	SAPGVQGDVI	GPQVALKO	SR VDIET	PNLEG	TLTGPR LGS	P			
5 (61 SGKTGTCR IS	MSEVDLNVAA	PK VKGGVD	VT LPRV	VEGKVKV	PEVDVRGPK	/ DVSAPDVE	AH GPEWN	L K MPK	MKMPTFSTP	G			
64	41 AKGEGPDVHM	TLPK GDISIS	GPK VNVEA	PD VNLE	GLGGKL	KGPDVKLPD	1 SVKTPK IS	MP DVDLH	/KGTK	VKGEYDVTV	P			
72	21 KLEGELKGPK	VDIDAPDVDV	HGPDWHLK	MP KMKN	IPK FSVP	GFKAEGPEVI	VNLPKADV	/DI SGPK I	OVTAP	DVSIEEPEG	К			
Coverage	Pentide											104 132 110 10 14 12 Rati	1 4 4 0 10 DS	1 22 04
oorenage	1 option											÷	1 - 14 of 14	• >
#	Peptid	e	Used :	-10lgP 📋 :	Avg. Area	Condition 2 : Ratio	Condition 3 : Rati	o Group 1 : C	Group 1 : C.	#Vector	Start	End	PTM	
1 VI	NAGHGLTYHNVK(*)		1	81.07	5.87	'e+6		0.00	0.	94 9	189	201		
2 AI	IAAIPEM(+15.99)HELNIGHAIIGF	!(*)	~	72.11	9.04	le+6		0.00	0.	97 6	202	221 0		
3 AI	AGHGITYHNI/K(*)		· · ·	65.23	6.82	(e+0		0.00	1.	02 6	202	221		
-+ N	ELLLGVNIDHIATLR(*)		~	59.90	3.41	e+6		0.00	1	05 4	2	17		
6 TI	DAEQAQELAR(*)		~	59.84	7.24	le+6		0.00	1.	05 3	164	174		
7 A	ATFAASLGLK(*)		1	55.58	8.92	le+6		0.00	0.	97 3	178	188		
8 A.	AIPEMHELNIGHAIIGR(*)		<i>✓</i>	52.43	2.90	le+6		0.00	0.	99 3	204	221		
9 QI	EVTTEGGLDVAGQR(*)		1	48.84	5.33	ie+6		0.00	1.	94 2	99	113		
10 A	GHGLTYHNVK(*)		~	43.69	9.96	e+5		0.00	9.	22 1	191	201		
11 G	TAYPDPVQAAFIAEQAGADGITVH	HLR(*)	~	43.36	1.28	le+7		0.00	1.	05 3	21	47		
12 LC	GVNIDHIATLR(*)	W(4) DUED(- 57.00) O(- 57.0	~	41.56	3.56	ie+5		0.05	0.	50 2	6	17		
13 M	INLEM(+15.99)AVTEEMLAIAVET	K(*)PHPG(+57.02)C(+57.0.		34.72	3.30	ie+3		0.00	0.	25 1	222	9/ 0 C	C	
14 10	***** GER(")		~	34.13	2.37		_	0.00	υ.	en 1	222	220		

3.3.1. Protein Table

The quantified proteins are listed in the protein table with homologous proteins grouped together.

This protein table is the same as the other protein tables found in the PEAKS DB, PEAKS PTM, and SPIDER results except the following columns:

• Significance: The protein significance is calculated as explained in Section 1, "Overview" in this chapter.

- **Condition Profile:** The protein abundance among the samples is depicted as a heat map. Hold the cursor on a profile to view the sample channels, abundances, and ratios with respect to the base sample.
- **Condition Ratio:** The ratio columns show the peptide abundance in each condition for all groups. A peptide's abundance in each condition is the sum of its feature vector(s)' abundance. Then the peptide abundance in each condition is compared to the reference condition to obtain the peptide condition ratio in each sample separately. The peptide condition ratio of a group is computed by calculating the median from the peptide condition ratios in the samples in that group.

Refer to Chapter 5, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER) Section 3.4.2, "Protein Table" for more details of other entries of the protein table.

3.3.2. Protein Volcano Plot

The volcano plot facilitates identification of the significant proteins. This interactive volcano plot enables the selection of a protein in the table. Proteins located in the top-right and top-left sections of this plot (above the significance threshold and to the right and left of the fold change thresholds) are considered statistically significant. Green points are down regulated while red points are up-regulated. Grayed points are filtered out as set in the filter. The volcano plot facilitates identification of significant proteins.

Protein markers that are in color can be clicked and selected. Upon clicking a point on the volcano plot, the corresponding protein in the Protein Table will be highlighted and more details are displayed below the Protein Table.

- Selecting a Different Condition: Click the condition dropdown above the volcano plot to display all volcano plot options, one for each experimental condition. Each volcano plot will show a different set of proteins that are relevant to that condition.
- Zoom in/out from Volcano Plot: Place the cursor on the volcano plot and drag a rectangle over the area of interest to Zoom in. By double clicking on the volcano plot you can zoom out and view the volcano plot with the default dimensions



3.3.3. Protein Coverage

The coverage view characterizes the selected protein sequences at the amino acid level. Refer to Chapter 5, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER), Section 3.4.4, "Protein Coverage" for details.
Q09666|AHNK_HUMAN Neuroblast differentiation-associated protein AHNAK OS=Homo sapiens GN=AHNAK PE=1 SV=2

1	MEKEETTR EL	LLPNWQGSGS	HGLTIAQRDD	GVFVQEVTQN	SPAAR TGVVK	EGDQIVGATI	YFDNLQSGEV	TQLLNTMGHH
81	TVGLKLHRKG	DRSPEPGQTW	TR EVFSSCSS	EVVLSGDDEE	YQR IYTTKIK	PRLK sedgve	GDLGETQSRT	ITVTRR VTAY
161	tvdvtgr EGA	KDIDISSPEF	K IKIPR helt	EISNVDVETQ	SGKTVIRLPS	GSGAASPTGS	AVDIRAGAIS	ASGPELQGAG
241	HSKLQVTMPG	IKVGGSGVNV	NAK (R@186 R	GGVQVPAVDI	SSSLGGRAVE	VQGPSLESGD	HGK IKFPTMK	VPKFGVSTGR
321	EGQTPKAGLR	VSAPEVSVGH	KGGKPGLTIQ	APQLEVSVPS	ANIEGLEGKL	KGPQITGPSL	egdlglk gak	PQGHIGVDAS
401	APQIGGSITG	PSVEVQAPDI	DVQGPGSKLN	VPKMKVPKFS	VSGAK GEETG	IDVTLPTGEV	TVPGVSGDVS	LPEIATGGLE
481	gk mkgtkvk t	PEMIIQKPKI	SMQDVDLSLG	SPK LKGDIKV	SAPGVQGDVK	GPQVALKGSR	VDIETPNLEG	TLTGPR LGSP
561	SGKTGTCR IS	MSEVDLNVAA	PK VKGGVDVT	LPRVEGKVKV	pevdvrgpk v	DVSAPDVEAH	GPEWNLK MPK	MK mptfstpg
641	AKGEGPDVHM	TLPK GDISIS	gpk vnveapd	VNLEGLGGK L	KGPDVKLPDM	SVKTPK ismp	dvdlhvk GTK	VKGEYDVTVP
721	KLEGELKGPK	VDIDAPDVDV	HGPDWHLK MP	KMKMPK fsvp	GFKAEGPEVD	vnlpk advdi	SGPK IDVTAP	DVSIEEPEGK
801	LKGPKFK mpe	MNIK VPKISM	PDVDLHLKGP	NVK GEYDVTM	PKVESEIKVP	DVELKSAK MD	IDVPDVEVQG	PDWHLK MPKM

3.3.4. Support Peptides

The supporting peptides assigned to the protein are shown in a separate tab beside the Protein Coverage. The sequence can be clicked and a pop-up window will show up to display the feature details.

This table contains the following information of the quantifiable support peptides for the selected protein in the protein table.

- **Peptide:** The amino acid sequence of the peptide, as determined by the PEAKS Search. A modified residue is followed by a pair of parentheses enclosing the modification mass. All non-unique peptides and ones not passing the filter will be excluded automatically.
- Used: Shows whether this peptide is used to calculate the protein profile. All peptides which pass the peptides filters are used to calculate the protein abundance.
- **-10LgP:** The peptide -10LgP score. The score indicates the likelihood that the peptide-spectrum match identification is correct.
- Avg. Area: Each peptide feature associated with the peptide has its own area under the XIC curve. The average area of is the average area of all the features associated with the peptide.
- **Condition Profile:** The relative peptide abundance for each condition compared to the reference condition among the samples is depicted as a heat map. Hold the cursor on a profile to view the samples, the areas, and the ratios with respect to the reference condition.
- **Condition Ratio:** The ratio columns show the relative peptide abundance for each condition with respect to the reference condition for all groups. A peptide's abundance in each condition is the sum of its feature vector(s)' abundance. Then the peptide abundance in each condition is compared to the reference condition to obtain the peptide condition ratio in each sample separately. The peptide condition ratio of a group is computed by calculating the median from the peptide condition ratios in the samples in that group.
- #Vector: This shows the number of quantifiable feature vectors of a peptide.
- Start: This shows the protein position of the first residue of the peptide that covers the protein.
- End: This shows the protein position of the last residue (inclusive) of the peptide that covers the protein.
- **PTM:** The types and the numbers of modifications present in the peptide shown in color-coded icons. Scroll over the modification icon to see the modification name and mass.
- Accession: The proteins supported by the peptide, separated by ';' if the peptide supports multiple proteins. If none of the proteins that the peptide supports pass the protein filters, the accession field will be empty.

3.3.5. Protein View Filter

Click the **Q** button on top-left of the protein table to open the protein filter and control which proteins to display.

Protein Filter
Top Proteins O All Proteins Protein accession/name contains
Protein description contains
Protein Contains Modifications Name to search for CLEAR Deamidation (NQ) Oxidation (M) Carbamidomethylation
RESET CANCEL APPLY

- **Top Proteins**: The number of proteins that have the highest number of unique peptides supporting them within their protein group.
- All Proteins: The number of proteins given the parameters set for PEAKS Protein Score(-10lgP) and number of unique peptides.
- **Protein Sequence contains**: Enter a protein sequence, only proteins that contain that sequence will be displayed in the protein table
- **Protein description contains:** Enter a protein description, only proteins that contain that description will be displayed in the protein table
- **Include Contaminant Data:** Check this to display identifications from the contaminant database that were identified during a PEAKS DB search.
- Protein Contains Modifications: Check boxes are provided for each identified modification.

3.4. Peptide View

The quantifiable feature vectors with the same sequence are grouped together to get the list of quantifiable peptides and are displayed in the **Peptide View** along with their detailed information.



3.4.1. Peptide Table

The peptide table contains the list of quantifiable peptides along with relevant information including the protein accession.

The view is similar to the support peptide table described in Section 3.3.4, "Support Peptides", with some new columns.

- **Quality Score:** The quality score for the peptide feature. Factors that affect the quality score include m/z difference, RT difference, XIC shape similarity, and the feature intensities.
- Accession: The proteins supported by the peptide, separated by ',' if the peptide supports multiple proteins. If none of the proteins that the peptide supports pass the protein filters, the accession field will be empty.

The Peptide table entries may be filtered by using the **Peptide View Filters** on top of the peptide table, for more information refer to Section 3.5.1, "Peptide View Filters" in Chapter 5, Peptide, PTM, and Mutation Identification (PEAKS DB, PEAKS PTM, and SPIDER).

3.4.2. Peptide View Filters

Click the button on the upper left corner of the peptide table to open the peptide filter and control which peptides to display.

- **Peptide Sequence Contains:** Enter a peptide sequence, only peptide that contain that sequence will be displayed in the peptide table
- **Peptide Accession Contains:** Enter a peptide accession, only peptides that contain that accession will be displayed in the peptide table
- Protein Contains Modifications: Check boxes are provided for each identified modification.

3.4.3. Feature Details

The **Feature Details** contains the eXtracted Ion Chromatogram (XIC) chart, Ion Match, and the Reporter Ion Spectrum.

- **.Feature Menu:** It is under the peptide table. The feature menu displays the information of the feature that is being displayed in the spectrum.
 - **Samples:** If a peptide is quantified in more than one sample, click the "SAMPLE" drop-down menu to view the Spectrum Annotation of the other PSMs.
 - **Feature:** Click the Feature drop down menu to show all peptide features included in the result: -10lgP, Avg. Area, Condition Intensity, Condition Ratio, Id Count, m/z, z, RT, and ppm for each feature are given.
 - If the transferred column for the feature displays "true" that means the ID for at least one feature was transferred.
 - **PSM Label:** Click the PSM Label drop-down menu to show other identified MS/MS scans associated with the peptide.

3.4.3.1. eXtracted Ion Chromatogram (XIC) chart

This map presents the changes in feature intensity over the retention time range of the peaks for all labels of the selected peptide. Additionally, the coloured isotopic distribution plot -M0, M+1, M+2, etc. - of each label is included, which shows the relative composition of each label.

Fraciton Assisted ID Transfer SILAC



3.4.3.2. Ion Match

This section displays the spectrum annotation and other information about the peptide selected in the Peptide table. When multiple spectra are matched to the peptide, only the information about the top-scoring PSM is shown by default. The interface of this section is identical to the Spectrum Annotation in the PEAKS De Novo result. For detailed instructions, refer to Section 3.5, "Spectrum Annotation" in Chapter 4, Peptide *de novo* Sequencing.



3.4.3.3. MS1 Spectrum

This section provides an MS1 spectrum zoomed in on the reporter ions in the experiment. This is an interactive spectrum, enabling zooming in and out using the scroll wheel of the mouse. Double-clicking will zoom in on an area of interest. Clicking and dragging will slide the spectrum to the left or right.



3.5. Result Exporting

PEAKS SILAC results can be exported to various text formats in the Export page. The protein and peptide results can also be exported from the Protein and Peptide page, respectively.

Fraciton Assisted ID Transfer SILAC

Export SELEC	CLEAR ALL		
All Search Parameters	Denovo	Database Search	
Data Refinement	Summary Table	Summary Table	Summary Table
MGF File	Denovo CSV	Protein CSV	Protein CSV
Mzxml	Denovo CSV with all Candidates	Protein-Peptide CSV	Protein-Peptide CSV
		Protein Fasta	Peptide CSV
		Peptide CSV	Feature CSV
		PSM CSV	
		Peptides-Pepxml	
		Peptides-MzIdentML	
		Denovo Only CSV	
		Denovo Only CSV with all Candidates	
DOWNLOAD DOWNLOAD B	Y SAMPLE		

- **Proteins CSV:** These are the quantified proteins that pass the filters set in the "Summary" page and their associated details.
- Protein-Peptide CSV: These are the supporting peptides of the quantified proteins and their associated details.
- **Peptide CSV:** A complete list of all quantified peptides that pass the peptide feature filters set in the "Summary" page and their associated details.
- Feature CSV: All predicted peptide features and their associated quantification information. Both identified and unidentified peptide features are included. To include unidentified peptide features, set Peptide Id Count equals to 0 in the Peptide feature filter.

Note

Exporting by sample will begin downloading the same .zip folder, since this does not apply to SILAC quantification.

Chapter 13. DeepNovo Peptidome Workflow

1. Overview

The DeepNovo peptidome workflow takes advantage of the latest *de novo* technology *DeepNovo* developed using deep learning. It's designed for identification of undigested peptides from peptidomic datasets. The algorithm was trained with an immunopeptidomic dataset for accurate and sensitive identification of undigested peptides. PEAKS online now supports DeepNovo Peptidome workflows for both DDA and DIA data acquisition modes.

Note

For more details, refer to the paper: "Deep learning enables de novo peptide sequencing from data-independent-acquisition mass spectrometry", Nat Methods 16, 63–66 (2019).

2. DDA Peptidome Workflow and Parameter Settings

DeepNovo and Peptide Database Search can be launched together as a DeepNovo Peptidome workflow in a PEAKS Online search. An analysis report is generated to show the combined results. DeepNovo peptides are used to assist the following protein database search using PEAKS DB allowing for accurate and sensitive identification of undigested peptides present in the protein sequencing database.

Note

For more details, refer to the paper: "PEAKS DB: De Novo Sequencing Assisted Database Search for Sensitive and Accurate Peptide Identification", Mol Cell Proteomics. 2012 Apr;11(4):M111.010587.

Spectra unmatched by database search are re-analyzed using the *de novo* peptide sequence homology search SPIDER to allow for accurate identification of peptides with a single amino acid variation from the target database.

Note

For more details, refer to the paper: "SPIDER: software for protein identification from sequence tags with de novo sequencing error", J Bioinform Comput Biol. 2005 Jun;3(3):697-716.

The combination of the results are then presented in the peptide table where the most confident identification that passes the given filters is given for each scan in the dataset.

To conduct a database search workflow, follow the steps below:

- 1. Refer to Chapter 5, Peptide De Novo Sequencing, Section 2, "*DeepNovo/De Novo* Analysis Workflow and Parameter Settings", Step 1-6 to create a new project and add data.
- 2. Select the "DeepNovo Peptidome" workflow.



3. Set the DeepNovo and Peptide Search parameters. See more information regarding the parameter settings in section 2.1-2.5 below.

SAVE WORKFLOW AS	Priority 3 CANCEL BACK SUBMIT									
Data Refinement Parameters	^									
Mass Correction 🗹 Associate features with Chimera Scan 🗹										
Mass Error Tolerance	^									
Precursor Mass Error Tolerance: 10 Tolerance Unit: PPM * Fragment Mass Error Tolerance: 0.02										
Deep Novo Parameters	^									
Enzyme: None 👻										
Fixed Modifications: Variable Modifications: • Acetylation (Protein N-term) • Oxidation (M)										
ADD/REMOVE MODIFICATION Max Variable PTM P Peptide Search Parameters	er Peptide: 2									
Target Database: PXD022950_UniProt Taxonomy: all species × Contaminant Contaminants_from 42148 fasts sequences Peptide Length: 8 to 14										
Report Filter	^									
PSM/Peptide										
○ PSM -10LgP ≥ ○ PSM FDR (%) ⋮ ● Peptide FDR (%) 0.1 ⋮ Confident Amino Acid Threshold (%) 2 ⋮ ⋮ ●										
Deep Novo										
Deep Novo Score (%)≥ 60 ‡ Deep Novo Protein Association Tag Sharing 5 ‡ Ar	nino Acids									

4. Click SUBMIT button to start the search.

2.1. Data Refinement Parameters

Mass Correction: The precursor m/z values given by some instruments are often not of the monoisotopic ion. This creates problems in downstream analysis. By enabling mass correction, PEAKS will examine the isotope shapes in the corresponding MS scans and accurately adjust the precursor m/z to equal the monoisotopic ion m/z.

Associate Features with Chimera Scan: If enabled, PEAKS will assume that a tandem scan may contain fragments from two or more peptides, which will allow PEAKS to identify co-eluting peptides found within the acquisition window of the scan. PEAKS accomplishes this by searching for peptides that match the monoisotopic mass of any peptide feature in the acquisition window.

2.2. Precursor and Fragment Error Tolerance

Precursor mass : This enables the selection of a precursor mass error tolerance. PEAKS Online DB Search identifies peptides within the precursor mass tolerance, which is the allowable m/z shift between the theoretical value of the peptide versus the observed value of the precursor ion. Precursor mass tolerance can be specified in either Da or ppm.

Fragment ion: This enables the selection of a fragment ion m/z tolerance, which is the allowable m/z shift between the theoretical value of the fragment ions of the peptide versus the observed value. PEAKS Online DB Search uses this value when scoring peptide-spectrum matches. PEAKS Online considers that a fragment ion is matched if the calculated m/z is within the tolerance. Fragment mass tolerance can be specified in Da.

Error tolerance parameters need to be set consistently with the mass accuracy of the instrument.

2.3. Enzyme Settings

Enzyme : The Deep Novo Peptidome workflow is designed for both undigested and Trypsin digested samples. The enzyme may be either none or Trypsin.

2.4. PTM Settings

Click "ADD/REMOVE MODIFICATION" button to open the "Select Modifications" dialogue and specify the fixed PTMs and common variable PTMs expected in the sample. It's recommended that no more than 11 variable modifications are selected. If the required modification is not listed, the modification can also be created here. Additionally, the maximum number of variable PTMs per peptide can be specified and a value of less than 4 is recommended. A fixed modification forces all instances of applicable residues to be modified. A variable modification gives the option for the residues to be modified.

Note

For a specific residue, only one fixed modification is allowed.

Note

To prevent long search times, select only the most frequent PTMs in the sample for PEAKS DB, and check the PEAKS PTM option to search for unspecified modifications.

2.5. Peptide Search Parameters

A peptide database must be added first before creating a new project to enable its selection in the workflow. In the Target Database drop-down menu, select the appropriate database that have been configured in PEAKS Online and set the taxonomy, if applicable. A contaminant database can also be selected from the same list of databases that have been configured in PEAKS Online. The length for a peptide can be set when creating an analysis, where the default peptide length range is between 6 and 45

2.6. Report Filter

PSM filter section is a mandatory section to filter out some results under the user-defined filter. Any changes to the filter will create a new analysis with old parameters and update the results and statistics.

• **PSM -10lgP:** The PEAKS peptide score (-10lgP) is calculated for every peptide-spectrum match (PSM) reported by PEAKS DB, PEAKS PTM, and SPIDER. The score is derived from the p-value that indicates the statistical significance of the peptide-spectrum match. A peptide may be matched to many spectra, resulting in multiple PSMs. In that case, the peptide's score is calculated as the maximum among all PSM scores. For details of the scoring algorithm, please refer to the publication, "*PEAKS DB: De Novo sequencing assisted database*

search for sensitive and accurate peptide identification ", Mol Cell Proteomics, 2011 Dec 20. A minimum requirement can be set and all identifications must pass this filter.

• **PSM FDR (%):** The PSM filter can also be set to use the false discovery rate. FDR is estimated using a "decoy-fusion" approach.

Note

Decoy fusion is an enhanced target-decoy method for result validation with FDR. Decoy fusion appends a decoy sequence to each protein as the "negative control" for the search. See BSI's web tutorial (http://www.bioinfor.com/fdr-tutorial/) for more details.

• **Peptide FDR (%):** Peptide filter can also be set to use the false discovery rate. FDR is estimated using a "decoy-fusion" approach.

Note

Decoy fusion is an enhanced target-decoy method for result validation with FDR. Decoy fusion appends a decoy sequence to each protein as the "negative control" for the search. See BSI's web tutorial (http:// www.bioinfor.com/fdr-tutorial/) for more details.

- **Confident Amino Acid Threshold (%):** Minimum ion intensity of each fragment ion pair, which define the amino acid position within the peptide, compared to the base peak ion intensity. A default of 2% means that at least one pair of ions (e.g. 'y' ions) defining the amino acid has intensity of 2% or higher for both ions within the pair..
- **Deep Novo ALC (%):** This filter controls the minimum ALC score required for display of *deep novo* results. A default of 50% is set.

3. DIA Peptidome Workflow and Parameter Settings

DeepNovo and Peptide Database Search can be launched together as a DeepNovo Peptidome workflow in a PEAKS Online search. An analysis report is generated to show the combined results. DeepNovo peptides are used to assist the following protein database search using PEAKS DB allowing for accurate and sensitive identification of undigested peptides present in the protein sequencing database. As with our streamlined DIA workflow, DIA DB search can be performed library-free or following an optional spectral library search.

Note

For more details please refer to the paper "A streamlined platform for analyzing tera-scale DDA and DIA mass spectrometry data enables highly sensitive immunopeptidomics", Nat Commun. 2022 Jun, 13(3108):10.1038/s41467-022-30867-7.

The combination of the results are then presented in the peptide table where the most confident identification that passes the given filters is given for each scan in the dataset.

To conduct a database search workflow, follow the steps below:

- 1. Refer to Chapter 5, Peptide De Novo Sequencing, Section 2, "*DeepNovo/De Novo* Analysis Workflow and Parameter Settings", Step 1-6 to create a new project and add data.
- 2. Select the "DeepNovo Peptidome" workflow.



3. Set the DeepNovo and Peptide Search parameters. See more information regarding parameter settings in sections 3.1-3.5 below.

SAVE WORKFLOW AS UPDATE WORKFLOW 3 CANCEL BACK SUBMIT
Mass Error Tolerance
Precursor Mass Error Tolerance: 10 Tolerance Unit: PPM Fragment Mass Error O.02 Tolerance Unit: DA
Deep Novo Parameters
Enzyme: None 🔹
Fixed Modifications: Variable Modifications: Oxidation (M)
ADD/REMOVE MODIFICATION Max Variable PTM Per Peptide: 1
Peptidome Search Parameters
Target Library:
Target Database: Benchmarking_Unip Taxonomy: all species × Contaminant contaminants_from V 20007 fasts sequences Precursor Database: Database: Contaminant
10 1800
Report Filter
PSM/Peptide
○ PSM -10LgP ≥ ● PSM FDR (%) 1 ⋮ ○ Peptide FDR (%) ⋮ Confident Amino Acid Threshold (%) 2 ⋮
Deep Novo
Deep Novo Score (%)≥ 70 : Deep Novo Protein Association Tag Sharing 5 : Amino Acids

4. Click SUBMIT button to start the search.

3.1. Precursor and Fragment Error Tolerance

Precursor mass : This enables the selection of a precursor mass error tolerance. PEAKS Online DB Search identifies peptides within the precursor mass tolerance, which is the allowable m/z shift between the theoretical value of the peptide versus the observed value of the precursor ion. Precursor mass tolerance can be specified in either Da or ppm.

Fragment ion: This enables the selection of a fragment ion m/z tolerance, which is the allowable m/z shift between the theoretical value of the fragment ions of the peptide versus the observed value. PEAKS Online DB Search uses this value when scoring peptide-spectrum matches. PEAKS Online considers that a fragment ion is matched if the calculated m/z is within the tolerance. Fragment mass tolerance can be specified in Da.

Error tolerance parameters need to be set consistently with the mass accuracy of the instrument.

3.2. Enzyme Settings

Enzyme : The Deep Novo Peptidome workflow is designed for both undigested and Trypsin digested samples. The enzyme may be either none or Trysin.

3.3. PTM Settings

Click "ADD/REMOVE MODIFICATION" button to open the "Select Modifications" dialogue and specify the fixed PTMs and common variable PTMs expected in the sample. It's recommended that no more than 11 variable modifications are selected. If the required modification is not listed, the modification can also be created here. Additionally, the maximum number of variable PTMs per peptide can be specified. For DIA workflows, setting this value greater than 1 will significantly increase search times. A fixed modification forces all instances of applicable residues to be modified. A variable modification gives the option for the residues to be modified.

Note

For a specific residue, only one fixed modification is allowed.

Note

To prevent long search times, select only the most frequent PTMs in the sample for PEAKS DB.

3.4. Peptide Search Parameters

3.4.1. Target Library

Target library can be enabled to enhance identification of expected (targeted) peptides.

Library Selection: Select a PEAKS-generated library for the search. Select one from the list of libraries that have been configured in PEAKS Online and filter for the peptide length, precursor m/z, and charge when necessary. To configure a new library, refer to Chapter 2 Section 2: Library Management and Configuration.

3.4.2. Target Database

A peptide database must be added first before creating a new project to enable its selection in the workflow. In the Target Database drop-down menu, select the appropriate database that have been configured in PEAKS Online and set the taxonomy, if applicable. A contaminant database can also be selected from the same list of databases that have been configured in PEAKS Online. The length for a peptide can be set when creating an analysis, where the default peptide length range is between 6 and 45. Peptidome results can also be filtered by precursor charge, m/z and fragment m/z.

3.5. Report Filter

PSM filter section is a mandatory section to filter out some results under the user-defined filter. Any changes to the filter will create a new analysis with old parameters and update the results and statistics.

- **PSM -101gP:** The PEAKS peptide score (-101gP) is calculated for every peptide-spectrum match (PSM) reported by PEAKS DB, PEAKS PTM, and SPIDER. The score is derived from the p-value that indicates the statistical significance of the peptide-spectrum match. A peptide may be matched to many spectra, resulting in multiple PSMs. In that case, the peptide's score is calculated as the maximum among all PSM scores. For details of the scoring algorithm, please refer to the publication, "*PEAKS DB: De Novo sequencing assisted database search for sensitive and accurate peptide identification* ", *Mol Cell Proteomics, 2011 Dec 20*. A minimum requirement can be set and all identifications must pass this filter.
- **PSM FDR (%):** The PSM filter can also be set to use the false discovery rate. FDR is estimated using a "decoy-fusion" approach.

Note

Decoy fusion is an enhanced target-decoy method for result validation with FDR. Decoy fusion appends a decoy sequence to each protein as the "negative control" for the search. See BSI's web tutorial (http://www.bioinfor.com/fdr-tutorial/) for more details.

• **Peptide FDR (%):** Peptide filter can also be set to use the false discovery rate. FDR is estimated using a "decoy-fusion" approach.

Note

Decoy fusion is an enhanced target-decoy method for result validation with FDR. Decoy fusion appends a decoy sequence to each protein as the "negative control" for the search. See BSI's web tutorial (http:// www.bioinfor.com/fdr-tutorial/) for more details.

- **Confident Amino Acid Threshold (%):** Minimum ion intensity of each fragment ion pair, which define the amino acid position within the peptide, compared to the base peak ion intensity. A default of 2% means that at least one pair of ions (e.g. 'y' ions) defining the amino acid has intensity of 2% or higher for both ions within the pair.
- **Deep Novo ALC (%):** This filter controls the minimum ALC score required for display of *deep novo* results. A default of 50% is set.

4. Understanding Peptidome Results

After a PEAKS Search is complete, all analyses selected during the project creation will be combined under the workflow drop-down menu. One is the *deep novo* sequencing result, which includes *de novo* peptide sequences listed, spectrum annotation and other detailed information. The other result node contains DeepNovo, PEAKS DB, and optional spectral library search results.

The analysis results for database search tools are presented in two tabs:

- **Summary :** This shows an outline of the PEAKS Online peptidome search results with key statistics and the overall quality of the experiment can be examined. This page shows after the search is done.
- **Peptides :** This shows a list of peptide identifications. This view also provides spectrum annotation and other detailed information for peptide precursor spectrum matches.

4.1. Result Filtration

4.1.1. Show Results for All Samples

For the peptide tables, "All Samples" can be selected to view the combined search results from all the samples.

Show Results for All Samples 	O Specified Sample	Sample 1	~

4.1.2. Show Results for Specific Sample

In the Peptides and DeepNovo page results, the filter can also be set to "Specified Sample" which will only show the results for a single sample. By clicking in the text you can search for a specific sample by name.

Specified Sample	Sample 1	× ~
------------------	----------	-------

Peptides will only be visible if they belong to the sample chosen. Once selected, the peptide tables will automatically add the "Area" column and the "# Spec" values will display the spectral count for the specified sample. For more information, refer to Section 3.5.2, "Peptide Table".

4.2. Result Exporting

PEAKS Online peptidome search results can be exported in the Export page:

Export	SELECT ALL	CLEAR ALL						
All Search Parameter	S							
Deep Novo		Peptide DB Search						
Summary Table		Peptide DB Summary Table						
DeepNovo CSV		Peptic	le CSV					
DeepNovo CSV with a	ll Candidates	PSM (CSV					
DOWNLOAD DOWNLOAD	BY SAMPLE							

- All search parameters: A comma separated file containing all used search parameters.
- **DeepNovo Summary Table**: A comma separated file containing a summary of the number of peptides *de novo* sequenced in each sample.
- DeepNovo CSV: A comma separated file containing the details from the DeepNovo table
- **DeepNovo CSV with all Candidates:** A comma separated file containing all details from the DeepNovo table including the top 5 *de novo* candidates from each spectrum within the ALC threshold set in the search parameters.
- **Peptide CSV :** A comma separated file containing all identified peptides and associated details from the peptide table for all samples.
- **PSM CSV :** A comma separated file containing all identified peptide spectrum matches and associated details from the peptide table for all samples.

4.3. DeepNovo Summary View

The DeepNovo summary view reports the key statistics from DeepNovo peptide sequencing

4.3.1. Result Statistics

Details about the filtered result:



- Sample Name: The name of the sample given during the project setup
- #MS2: Number of MS2 scans in the sample
- **#PSM:** The 'precursor spectrum match' is the number of peptide features with matching DeepNovo results that pass the filters set in the search parameters
- **#PSM(alc > 30):** The is the number of peptide features with matching DeepNovo results above an average local confidence of 30
- **#PSM(alc > 50)**: The is the number of peptide features with matching DeepNovo results above an average local confidence of 50
- **#PSM(alc > 70)**: The is the number of peptide features with matching DeepNovo results above an average local confidence of 70
- **#PSM/#MS2**: The number of PSMs with DeepNovo results above the ALC filter set in the search parameters divided by the number of MS2 scans in the dataset. (DDA only).

Table 1 provides the total number of MS2 scans and PSMs total as well as the number of MS2 scans and PSMs with different ALC thresholds for each sample. For DDA acquisition, the #PSM / #MS2 provides a de novo sequencing identification rate metric for each sample.

Figures 1 - 3 are used to visualize the distribution of the DeepNovo results

- Figure 1: Average local confidence score versus precursor PPM: The distribution of DeepNovo result average local confidence versus the precursor mass error in parts per million. A color gradient from green (low density) to orange (high density) is used to represent the density of results at each position on the dot plot
- Figure 2: Residue Local confidence histogram: Using the percentage local confidence assigned to each amino acid, a histogram is reported with the number of amino acids with a local confidence at the given local confidence score.
- Figure 3a: Average local confidence score versus the delta retention time: Compares the average local confidence to the difference between the predicted retention time of a peptide versus the actual retention time in minutes. This figure validates the quality of the retention time prediction

- Figure 3b: Scatterplot of retention time versus the predicted retention time for each DeepNovo result: Each point represents the retention time from the data file versus the predicted retention time. The regression function used to predict retention time is plotted against the data.
- Figure 4a: Histogram of peptide $\Delta 1/k0$ for timsTOF data.
- Figure 4b: Regression plot of peptide 1/k0 versus predicted 1/k0 for timsTOF data.



4.4. DeepNovo Sequence Table

<	٩			Show Res Specified S	ults for ample	Sample 2	2	x 🗸	Export	<u>)eep Novo</u>	<u>Deep Novo with a</u>	II Candidates		
umn		Show mass tag for confidence less than: 0%									\leftarrow	1 - 100 of 33	321 👻	\rightarrow
hary		Scan	Precursor	Peptide	SC	ORE(%)	ALC(%) ↓ .	Length	m/z	. z .	1/k0	∆1/k0	RT	. 4
0	1	1550341:Sara_Ligandome_murine_CT26_12	13829:Sara_Ligando	AGPRLELHHR		0.0	98.9	10	395.889	99 3	0.7408-0.7472	0.0243	10.50	0 _*
ee	2	2910859:Sara_Ligandome_murine_CT26_12	29899:Sara_Ligando	DYLADKSYL		0.0	98.7	9	544.268	37 2	0.8905-0.8969	-0.0065	19.65	5
No	3	2968487:Sara_Ligandome_murine_CT26_12	30508:Sara_Ligando	SPGPSRLADYL		0.0	98.6	11	588.308	33 2	0.9071-0.9134	-0.0070	20.03	3
0VO	4	2080212:Sara_Ligandome_murine_CT26_12	21526:Sara_Ligando	PRYEYHW		0.0	98.5	7	350.830)7 3	0.7212-0.7277	0.0262	14.04	4
	5	560233:Sara_Ligandome_murine_CT26_120	5048:Sara_Ligando	TGEKGFGYK		0.0	98.4	9	493.751	0 2	0.8564-0.8628	0.0198	3.81	1
	6	1888451:Sara_Ligandome_murine_CT26_12	18828:Sara_Ligando	VGPELHHKLL		0.0	98.1	10	571.835	58 2	0.9463-0.9527	0.0305	12.75	5
	7	2569893:Sara_Ligandome_murine_CT26_12	28046:Sara_Ligando	TGAAEQLKHLLA		0.0	98.1	12	626.355	52 2	0.9723-0.9786	-0.0083	17.37	7
	8	2484753:Sara_Ligandome_murine_CT26_12	26934:Sara_Ligando	HEWKKLLLT		0.0	98.0	9	584.347	1 2	0.9753-0.9817	0.0305	16.80	0
	9	3035122:Sara_Ligandome_murine_CT26_12	31121:Sara_Ligando	TDFGEALVRHDEF		0.0	98.0	13	768.359	98 2	1.1070-1.1133	0.0384	20.48	8 .
	4		1	1	1									E.





4.4.1. Peptide View Filters

Click the 👩 button, the following De Novo Filter will pop up. It controls which peptides to display.

Denovo Filter			
Denovo sequence contains			
Scan -			
Precursor Id =			
m/z =			
RT a			
I∆RT) s			
Tag Length 2	Tag Length s D		
		RESET	CANCEL APPLY

- **Denovo sequence contains:** Enter a string of amino acids. The filter will return any result containing the string. Modification masses can be included as their mass within circle brackets, i.e (+15.99)
- Scan: Filter for a single scan number based on the scan numbers read from the raw data file.
- Precursor Id: Filter is only relevant for TimsTOF data, for which it will filter based on the precursor id of the peptide.
- m/z: Filter based on the expected m/z, the last significant digit is rounded.
- **RT:** Filter based on the expected RT, the last significant digit is rounded.
- |**#RT**| <=: Only results with an absolute difference in retention time between the predicted retention time and the observed retention time greater than or equal to the given value will be displayed.
- **Tag Length** >= : Only results with a tag length greater than the given value will be displayed.
- Tag Length <= : Only results with a tag length less than the given value will be displayed.

4.4.1.1. Local Confidence Slider

The local confidence slide accessed by clicking the

40

30



Color code: .>90% .80-90% .60-80% .<60%

The slider is used to set the minimum local confidence for all amino acids. Amino acids with a local confidence score below the threshold will be reported as their mass in daltons. If multiple amino acids are found in a row below the cut-off the mass of each amino acid will be added to create a mass tag. This slider also controls the de novo tag that appears in the tag column of the export.

4.4.2. DeepNovo Table

The DeepNovo table shows the filtered DeepNovo results. Each row in the table is a precursor identification represented by its highest-scoring *de novo* sequencing result. The table is sorted by the ALC value.

button to bring up the local confidence slider.

The columns Scan, peptide, ALC, m/z, z, RT, and Delta RT, are sortable, i.e. Clicking any of these header can sort the whole table based on that column.

The columns in the Peptide table are:

- Scan: The scan number, followed by the precursor ID (for timsTOF data), from the raw file followed by the raw file name
- Peptide: The predicted de novosequence color coded by the local confidence score of each amino acid. The

color legend can be found by clicking the local confidence slider button

- Score (%): A peptide confidence score which adds penalty to ALC (%) based on the the difference between predicted RT and observed RT. Typically Score(%) will be smaller than ALC(%) for the same peptide.
- ALC(%): The average local confidence of the peptide. It is the average of the individual local confidence scores. Local confidence scores are calculated by permutation of the possible peptide sequences that can match the precursor mass and comparing the sequence alignments. The local confidence represents the likelihood that the amino acid reported at the position is correct compared to the other possible matches.
- Length: The number of amino acids in the peptide sequence.
- m/z: The precursor mass-to-charge ratio.
- **z:** The charge of the peptide retrieved from the raw file
- **RT:** The retention time (elution time) of the spectrum as recorded in the data.
- Delta RT: The difference between the observed retention time and the retention time predicted during analysis.
- Area: The area of the peptide feature associated with the *de novo* result. If no area could be associated from feature detection, an empty cell is give.
- Mass: The theoretical mass of the predicted de novo sequencing result
- **ppm:** The precursor mass error, calculated as $10^{6} \times (\text{precursor mass} \text{peptide mass}) / \text{peptide mass}$.
- **PTM:** The types and the numbers of modifications present in the peptide shown in color-coded icons.
- Mode: The fragmentation mode of the scan associated with the de novo result

4.4.3. Annotated Spectrum

The annotated spectrum view by default will give a mirror plot comparing the DeepNovo result to the spectrum predicted by the deep learning model. To the right of the mirror plot, the precursor profile displays the extracted ion chromatogram of the MS1 signal associated with the M0, M+1, and M+2 ions of the intact peptide. Below the mirror plot, the ion details are given:

Fragment ion details table:

- Label: The type of predicted fragment ion
- Predicted M/Z: The theoretical mass to charge ratio of the amino acid predicted at the position
- Predicted Intensity (%): The predicted relative intensity of the fragment ion from the deep learning model
- Query M/Z: The observed mass to charge ratio of the fragment ion
- Query Intensity: The observed intensity of the fragment ion



Click the button to switch to the annotated spectrum view. The annotated spectrum view is described in the Section 3.5, "Spectrum Annotation" section.

4.5. Peptide Search Summary View

The Summary view reports key statistics as an overview of the result. In the report, several statistical charts can be examined to assess result quality.



Figure 3. RT figures for Sara_Ligandome_murine_CT26_120320_R02_10of10ul_Slot2-20_20-04-19_1169.d with outliers removed (a) Histogram of peptide ΔRT. (b) Scatterplot of peptide RT versus Predicted RT.



Figure 4. 1/K0 figures for Sara_Ligandome_murine_CT26_120320_R02_100f10ul_Stor2-00_20-04-19_1169 d with outliers and charge 1 peptides removed (a) Histogram of peptide Δ1/K0. (b) Scatterplot of peptide Real 1/K0 versus Predicted 1/K0.



Figure 5. Boxplot for median ΔRT across all MS runs



4.5.1. Result Statistics

The first chart displays the summary information for each sample.

- **Sample Name:** This column lists the samples names with the exception of the first row, which shows the combination of all the samples.
- **#MS Run:** The number of data files included in the sample
- **#MS1:** The number of MS scans in the sample
- # MS2: The total MS/MS scans in the sample summed across all data files.
- # Precursor: Total number of precursors (DIA only).
- **# PSM:** Total number of precursor spectrum matches (PSMs) found in the sample.
- **# Peptides:** Total number of peptides found ignoring differences with leucine (L) and isoleucine (I) amino acids. Peptides with the same primary sequence but different PTMs are counted separately.
- **# Sequences:** Total number of different peptide backbone sequences found ignoring differences with leucine (L) and isoleucine (I) amino acids. Peptides with the same primary sequence but different PTMs are counted as one sequence.
- **#PSM/#MS2:** The precursor spectrum matches identified divided by the total number of MS2 in the sample

4.5.2. Summary View Figures

Figure 1 shows the FDR curve with respect to the number of PSMs being kept after filtration by the peptide -10lgP score. By lowering the score threshold, more PSMs are kept in the filtered result. Conversely, the FDR increases because more false positives are kept. In this figure, the vertical line indicates the current score threshold. The number of PSMs and the corresponding FDR by the current score threshold are shown in the top-left corner.

PEAKS Search estimates FDR using the "decoy-fusion" approach. Decoy-fusion is an enhanced target-decoy approach that makes more conservative FDR estimations. For details of the "decoy-fusion" approach, please refer to the publication, "PEAKS DB: De Novo sequencing assisted database search for sensitive and accurate peptide identification", Mol Cell Proteomics, 2011 Dec 20.

It is commonly recommended to set FDR at <1% by adjusting the score threshold. If a rapid growth of the FDR around the 1% FDR threshold is noticed, some PSMs may be sacrificed to significantly reduce the FDR.

Note

When counting the number of PSMs, PEAKS Online keeps one peptide per spectrum at most. Thus, the number of PSMs is actually the number of spectra with assigned peptides.

Note

Decoy matches are not counted in the number of PSMs. Unless otherwise specified, decoy matches are also excluded from the other statistical numbers shown in the Summary view.

Figure 2: Distribution of precursor mass error of filtered PSM in ppm: Shows a histogram of mass error of identifications bundled by the number of precursor spectrum matches in each group.

Table 1. Statistics of filtered results.

- DB Peptides: The number of peptides matched by PEAKS DB database search
- Homolog Peptides: The number of peptides matched by SPIDER homology search (DDA Only)
- Targeted Peptides: The number of peptides matched using the optional target library search (DIA Only)
- DeepNovo Peptides: The number of peptides matches by DeepNovo de novo sequencing

- **FDR (Peptide-spectrum matches):** The false discovery rate of the full list of target hits from the final round search versus hits from the decoy database at the precursor spectrum match level.
- **FDR** (**Peptide sequences**): The false discovery rate of the full list of target hits from the final round search versus hits from the decoy database at the peptide level.
- -10LgP Cutoff (Peptides): The score where the FDR cut-off is reached selected by the search parameters.

Figure 3a. Histogram of peptide delta RT: The difference in RT between the observed RT versus the predicted RT bundled by the number of peptides in each range for each individual algorithm that detected them eg. DeepNovo, DB, SPIDER (DDA), Target Library (DIA)

Figure 3b. Scatterplot of peptide RT versus Predicted RT: Each point represents a predicted peptide sequence. The distribution of predicted RT versus observed RT is shown for peptides for each individual algorithm that detected them eg. DeepNovo, DB, SPIDER (DDA), Target Library (DIA). The predicted regression function is also given.

Figure 4a. Histogram of peptide $\Lambda_1/k0$ for timsTOF data.

Figure 4b. Regression plot of peptide 1/k0 versus predicted 1/k0 for timsTOF data.

Figure 5. Boxplot of median delta RT across all MS runs: The retention time plotted against delta RT showing the trend of different in predicted RT versus observed RT over time. This chart highlights times across all runs where RT prediction may be different from the average.

4.5.3. Low Quality Data Warnings

If PEAKS Online detects any data files producing very few or no results a warning section will appear at the top of the Summary Page indicating the number of data files producing poor or no results.

2 data files produced no results. (Expand to see list of zero result data files)

This section can be expanded to display a list of data files in the analysis that are producing poor or no results.

Potentially Problematic Data Sets								
San All	nple 3 fractions have no result.							
San All	nple 4 fractions have no result.							

If there were no results found in any of your data files, a warning message stating "No Results Found" will be displayed instead of the usual Summary Page figures and tables.

Summary Statistics



4.6. Peptide View

The Peptide Table shows the filtered peptide identification results. Click on a peptide row to show the spectrum below the peptide table.

< 2	Sh	Q Iow mass tag	g for confi	dence less t	han: 0%							Show Results for All Samples	O sp	pecified Sample	Sample 2		 Export
			Peptide	e		10LgP ↓	ADI(%)	Mass	Length	ppm	m/z	RT	ΔRT	MS2 Correla	1/k0	∆1/k0	
2 5	92	GAQNRNYS	SKLL			120.01	0	.0 1262.67	31 1	1 -1.1	632.3431	11.99	-0.10	0.00	0.9812-0.9876	-0.0093	1779917:Sara_
	93	TPATAAAE	EERQA			120.01	0	.0 1343.63	17 1:	3 1.5	672.8241	10.68	0.24	4 0.00	0.9614-0.9678	0.0125	1581869:Sara_
di g	94	TATTRGSP	VGGNDN			120.01	0	.0 1345.62	22 1-	4 -3.5	673.8160	3.75	-3.42	2 0.00	0.9449-0.9513	0.0008	556004:Sara_L
PS Q	95	QYAKDIGFI	IKLD			120.01	0	.0 1409.75	54 1	2 -1.3	705.8841	20.67	-0.37	7 0.00	1.0483-1.0547	-0.0212	3066711:Sara_
ç	96	HLPVDTGT	ENRTY			120.01	0	.0 1501.71	61 1	3 -0.6	751.8649	13.09	0.18	3 0.00	1.0396-1.0460	-0.0102	1938373:Sara_
ç	97	TYTATQIGI	EW			120.01	0	.0 1281.62	41 1	1 2.0	641.8206	26.93	-1.64	4 0.00	0.9482-0.9546	-0.0140	3995858:Sara
ç	98	STM(+15.9	9)KPVQK\	/LE		120.01	0	.0 1274.69	04 1	1 -5.4	638.3491	10.17	-0.76	5 0.00	0.9425-0.9489	0.0189	1508652:Sara_
ç	99	TQM(+15.9	9)SVPEQA	AELEKLK		120.01	0	.0 1745.88	70 1	5 2.3	582.9709	15.74	0.54	4 0.00	0.8971-0.9035	-0.0219	2336970:Sara_
1	100	TKYIVDADI	DRSGE			120.01	0	.0 1467.68	41 1	3 2.1	734.8509	11.90	0.47	7 0.00	1.0163-1.0227	-0.0071	1767745:Sara_
	100 900 80 50 50 50 50 50 50 50 50 50 50 50 50 50	ensity (%)	b2 0 200.0	b3 b 300.0	4 b5	b6 b7 b6 b7 y	y b8 5 5 700.0	y3 y7 6 5 y7 y8 soo.0 e00.0	y9 y10 y9 y10 y10	Homolo y12 y11 y12 y11 Predicte	ed	X U U U U U U U U U U U U U U U U U U U	2.84 10	Precursor P	rofile	N N N	10 1+1 1+2
		# Label	Predi	cted M/Z 199,1083	Pi	redicted Intensit	y(%) 2.8	Query M/Z	Query Int	ensity	*					RI .	
		2 b3	3	270.145			14.0										
		3 b4		371.1931			20.3	371.1980		2.55e+1(6.6%)							
	ļ	4 y 3	3	374.214			7.7										
		5 b5		442.2302			21.8	442.2290		.16e+2(30.1%)							
	}	7 54		513 267			23.0	503.2700	4	56e+2(40.5%)							
		8 b7		584 304			13.3	584 3070		3 78e+1(9.8%)							
	ł	9 v5	5	632.2998			28.3	554.5070		55011(5.0%)							
		10 b8	3	713.3470			6.1	713.3550	4	.51e+1(11.7%)							
		11 y6	5	761.3424			49.1	761.3390	1	.44e+2(37.5%)	-						

4.6.1. Peptide View Filters

Click the **o** button, the following Peptide Filter will pop up. It controls which peptides to display.

Peptide Filters	
Peptide sequence contains	
Protein Accession contains	
CAA(%) ≥	
Scan =	
Precursor Id =	
m/z ≅	
RT ≅	
Peptide sample area ≥	
MS2 Correlation ≥	
I∆RTI ≤	
IΔ1/K0I ≤	
Peptide Length ≥	Peptide Length ≤
Found By =	
Include Contaminant Data	_
Peptide Contains Modifications	5
Name to search for	CLEAR
Carbamidomethylation Oxid	ation (M) SILAC-V-6.01Da
	RESET CANCEL APPLY

- **Peptide sequence contains:** Enter an amino acid sequence, only peptides that contain that sequence will be displayed in the peptide table.
- **Peptide Accession contains:** Enter an keyword, only peptides that are associated with protein accessions containing that keyword will be displayed in the table.
- CAA (%): Enter a percentage. This represents the percentage of residues which pass the CAA threshold set by the user in the project parameters. Only peptides with a percentage of passing residues greater than this will be displayed in the table.
- Scan: Filter for a single scan number based on the scan numbers read from the raw data file.

Note

Scan filter is only available for DDA samples.

- **Precursor Id:** Filter is only relevant for TimsTOF data, for which it will filter based on the precursor id of the peptide.
- m/z: Filter based on the expected m/z, the last significant digit is rounded.
- **RT:** Filter based on the expected RT, the last significant digit is rounded.
- **Peptide sample area:** Filters based on total peptide area found in each sample. If a peptide is found in multiple samples, there should be at least one sample pass this filter for the peptide to be included.
- **MS2 Correlation:** Enter a percentage.Only peptides with MS2 Correlation greater than this will be displayed in the table.

- |**#RT**| <=: Only results with an absolute difference in retention time between the predicted retention time and the observed retention time less than the given value will be displayed.
- **Peptide Length** >=: Only results with a difference in retention time between the predicted retention time and the observed retention time greater than the given value will be displayed.
- **Peptide Length** <=: Only results with a difference in retention time between the predicted retention time and the observed retention time less than the given value will be displayed.
- Found By: The algorithm used to find the peptide sequences
- **Include Contaminant Data:** Check this to display contaminants that were identified during the search (if a contaminant database was specified). Leaving this unchecked will only display peptides with at least one non-contaminant accession.
- **Peptide Contains Modifications:** Check boxes are provided for each identified modification. Peptides that contain the selected PTMs will be shown.

4.6.2. Peptide Table

The Peptide Table shows the filtered peptide identification results. Each row in the table is a peptide identification represented by its highest-scoring PSM. The table is sorted by the -10lgP value. In the Protein list drop-down window of each peptide, after clicking any of the protein, the protein page will be opened.

The columns Peptides, -10LgP, m/z, RT, Accession are sortable, i.e. Clicking any of these header can sort the whole table based on that column.

The columns in the Peptide table are:

- **Peptide:** The amino acid sequence of the peptide, as determined in PEAKS Search. A modified residue is followed by a pair of parentheses enclosing the modification mass.
- **-10lgP:** The peptide -10lgP score. The score indicates the scoring significance of a peptide-spectrum match.
- CAA (%): The percentage (%) of residues which have a CAA score passing the CAA threshold set by the user in the peptide view filters.
- Mass: The monoisotopic mass of the peptide.
- Length: The number of amino acids in the peptide sequence.
- **ppm:** The precursor mass error, calculated as $10^{6} \times (\text{precursor mass} \text{peptide mass}) / \text{peptide mass}$.
- m/z: The precursor mass-to-charge ratio.
- **RT:** The retention time (elution time) of the spectrum as recorded in the data.
- Delta RT: The difference in retention time between the observed retention time and the predicted retention time.
- **MS2 Correlation:** Correlation between the predicted and query MS2 spectra (including ion match and intensity).
- Scan: The scan number of the spectrum that matches the peptide sequence with the highest -10lgP.
- **#Spec:** The number of MS2 spectra assigned to the peptide.
- **PTM:** The types and the numbers of modifications present in the peptide shown in color-coded icons.
- **AScore:** Localization score assigned to modifications on the peptide. It is the -10 log of a p-value. In this case, the p-value is the probability that the modification occurs at the reported position compared to other possible positions. A -10lgP of 20 is equal to a p-value of 0.01. Scores are listed in the format AminoAcid, peptide position, Ascore (DDA Only).

- **Ion Intensity:** Ion intensity of the modification site-determining fragment ion pair, compared to the base peak ion intensity, for the top-scoring PSM. Used as a measure of PTM site confidence (DIA Only).
- Found By: The algorithm the peptide was found by. For DDA the options are PEAKS DB, DeepNovo, or Spider (shown as homolog). For DIA the options are DeepNovo, Targeted Peptide (if optional spectral library search is enabled), or DB Search.
- Accession: The accession number of the highest-scoring protein containing this peptide.
- 1/k0: This column is only available for TimsTof data, it shows the ion mobility range of for the precursor.

4.6.3. Peptide Global Comparative View

"Peptide Global Comparative View" provides detailed comparisons among different samples regarding the identified peptides. To Show/Hide Columns, click **III** on top-left of the peptide table. From the pop-up window, users can add and remove area and spectral count columns. In the peptide table, the selected columns will be displayed and can be exported. If "Specified Sample" is selected these columns are automatically displayed.

Select	t Display Columns and Samp	oles	
Select	Columns		
🗸 Are	ea 🗸 #Spec		
Select	Samples		
Name t	o search for	0	LEAR
~	SAMPLE NAME		
\checkmark	S1		
\checkmark	S2		
\checkmark	S3		
\checkmark	W1		
	W2		
~	W3		
	CAN	ICEL	SAVE

Note

PEAKS Online provides users the option to calculate peptide abundance using the MS1 feature area calculation or using the MS1 feature intensity. By default, area is used. The default can be changed by the administrator.

- Area by sample: The total area of all peptide features matched to spectra that identified the peptide. Area calculations are separated into multiple columns if multiple samples were included in the run.
- **#Peptide-spectrum matches by sample:** The number of peptide spectrum matches associated with a peptide separated by sample.

4.6.4. Annotated Spectrum

The annotated spectrum view by default will give a mirror plot comparing the result to the spectrum predicted by the deep learning model. To the right of the mirror plot, the precursor profile displays the extracted ion chromatogram of the MS1 signal associated with the M0, M+1, and M+2 ions of the intact peptide. Below the mirror plot, the ion details are given:

Fragment ion details table:

- Label: The type of predicted fragment ion
- Predicted M/Z: The theoretical mass to charge ratio of the amino acid predicted at the position
- Predicted Intensity (%): The predicted relative intensity of the fragment ion from the deep learning model
- Query M/Z: The observed mass to charge ratio of the fragment ion
- Query Intensity: The observed intensity of the fragment ion



Click the button to switch to the annotated spectrum view. The annotated spectrum view is described in the Section 3.5, "Spectrum Annotation" section.

Chapter 14. Instrument Daemon

The PEAKS Online Instrument Daemon is a separate process included with your PEAKS Online server. It can be installed on an Instrument or Data machine to allow for an automatic upload and the analysis of raw data as the mass spectrometric acquisitions are complete.

This section of the manual covers the use of an Instrument Daemon that has already been installed and registered with your PEAKS Online server. For help with installing or configuring an Instrument Daemon please refer to the Instrument Daemon section of the Admin manual.

1. What is an Instrument Daemon?

An Instrument Daemon allows for real-time processing of data as it comes off the instrument. Instrument Daemons are separate processes run alongside PEAKS Online, typically on the data acquisition machine. Once installed, the Instrument Daemon will be pointed towards the directory on the machine that new data is written to. The Instrument Daemon will then scan the directory periodically for new data.

From the PEAKS Online Web Interface, projects can be scheduled using these Instrument Daemons. When the Instrument Data scans its directory and locates new data that fits the requirements for a scheduled project, it will automatically upload the data and submit it for processing.

Notes on Instrument Daemons:

- Your PEAKS Online server can have any number of Instrument Daemons connected to it.
- A single project can request data from multiple Instrument Daemons at the same time.
- Data can be uploaded using an Instrument Daemon either pro-actively (before your experiment has run through the Mass Spectrometer) or retro-actively (after your experiment has been run through the Mass Spectrometer).

2. How to add Data using an Instrument Daemon

Adding data to a project, either a new project or an existing project, is very similar to adding data either from a Local or Remote data source. To review how to add data to a project refer to Chapter 1 Section 5.2 - Creating a PEAKS Online Project.

2.1. Data Page

To add data from an Instrument Daemon, instead of using existing data either Locally or from a Remote Data Repository, follow these steps:

USER MANUAL PROJECT DATA								
						CANCEL	ВАСК	NEXT
		Enzyme		Activation Meth	hod	Instrument		
Select Data Local Remote Instrument In Project	Sample Name	Trypsin	*	HCD	•	Orbitrap (Orbi-Orbi)	•	
Select All								
Example Instrument Daemon								
		PARSING RULE						
	Instrument Daemon							
	L							

1. On the Data Page, select the "Instrument" tab instead of the "Local" or "Remote" tabs.

2.

Select one, or more, Instrument Daemons from the list and click the button to move them to the right side of the page. This will add them to the "Instrument Daemon" section at the bottom of the page, instead of the "Samples" section above.

Instrument Daemon	PARSING R	ULE	DATA COPY			
Example Instrument	Trypsin	*	CID	*	Orbitrap (Orbi-Orbi)	• ×

3. Once added, you will need to specify the criteria for data on the selected Instrument Daemons to be added to



Instrument	t Daemon Parsi	ng Rule									
Delimiters:	✔ Period (.)	Hyphen (-)	Semicol	on (;)	Space	e 🔽 (Inderscor	e (_) [Co	omma (,)	
	Others:										
ExampleProje	ect_Sample01_Fract	tion01.raw									
Project:	ExampleProject	Sample01	Fraction01	raw							
Sample:	ExampleProject	Sample01	Fraction01	raw							
Enzyme:	ExampleProject	Sample01	Fraction01	raw							
Activation:	ExampleProject	Sample01	Fraction01	raw							
Samples i	in Project: 12		Schedule Max	Waitin	ig Time for	Sample	(hours) :	12			
Fractions pe	er Sample: 1		Schedule Max	k Waitir	ng Time for	Project	(hours) :	72			
										CLOSE	SAVE

- 4. In this new dialog you will need to complete the following steps:
 - i. Type in, or copy and paste, an example filename that follows the format of files from the experiment you wish to analyze.

Note

It is best if this is an actual filename from your experiment (either from a file that has already been generated or an example of one that will be generated).

- ii. Select the delimiters, at the top of the dialog, that should be used to split your example filename. This will be used to cut your filename into sections, and these sections will then be used to determine if data should be added to this project or not as well as how to group files into samples within the project. For the example above, period (.) and underscore (_) were selected as delimiters, which breaks the example filename into 4 sections, "ExampleProject", "SampleO1", "FractionO1", and "raw".
- iii. Beneath your example file name, select the sections of the filename that should be used to do the following grouping:
 - a. Project. This should be a portion of the filename that is common across every file that will be generated for this project, but that is not used for any other files on that Instrument Daemon. In this example, the first section, "ExampleProject" was selected. This means any data on the selected Instrument Daemon(s) that starts with the words "ExampleProject" will be added to this project. This would include existing data on the Instrument Daemon.
 - b. Sample. This should be a portion of the filename that is common across all fractions of a single sample. In this example, the second section, "Sample01", was selected. This means any data that shares the same text in this section will be grouped together in a sample. So all files with "Sample01" will be grouped as "Sample01" and all files with "Sample02" will be grouped as "Sample02".
 - c. Enzyme. This should be a portion of the filename that relates to the Enzyme used for that Mass Spec run.

Important

This must match an Enzyme that exists on your PEAKS Online server exactly in order to work. For example, Trypsin.

d. Activation. This should be a portion of the filename that relates to the Activation Method used for that Mass Spec run.

Important

This must match one of the built-in Activation Methods on PEAKS Online exactly in order to work. For example, HCD.

- e. Samples in Project. Here you should specify the number of samples you expect to run, or have already run, for this experiment.
- f. Fractions per Sample. Here you should specify the number of fractions that you are expecting for each sample you are running as part of this experiment.

Important

All samples added to a project using Instrument Daemons must have the same number of fractions.

- g. Schedule Max Waiting Time for Sample. This is an optional parameter that will set a max wait time for each sample. Once a sample has received its first fraction (in the case of fractionated samples) if it has not received all of its fraction files within the specified time period, that sample will be considered failed. This is to prevent files that may have run into issues on the instrument from causing projects to hang indefinitely.
- h. Schedule Max Waiting Time for Project. This is an optional parameter that will set a max wait time for the entire project. After the specified time has elapsed, any samples that have not received all of their fractions will be considered failed.
- iv.

Once this has all been set, click the SAVE button to save the parsing rule.

5.

Once the parsing rule has been set, you can either click the

NEXT

parameter settings or the

UPLOAD SAMPLES

button to load the data without performing an analysis.

2.2. Data Copy

And additional function that the Instrument Daemon has is the ability to copy data into a Data Repository. This is an optional step that can be requested when adding new data to a new or existing project using an Instrument Daemon.

In order to utilize this function, follow these steps:

1.

After the Instrument Daemon has been added to the project on the Data Page, click the button at the top of the Instrument Daemon section. This will pop up a new dialog.



button to continue to the analysis

Sources
Sources (Click to open folder)
Data Center
Scratch
Regression
Test Data

CANCEL CLEAR OK

- 2. This new dialog will contain a list of all Remote Data Repositories attached to your PEAKS Online server, similar to selecting the "Remote" tab for data upload.
- 3. Select the Data Repository you wish to copy the data to and navigate to the location within that repository where you wish to save the data.
- 4.

Click the ^{OK} button.

After a location has been specified, any data that is sent to the master from an Instrument Daemon as a part of this analysis will then also be copied to the specified Data Repository location.

Note

Copying data to a Remote Repository will not modify or affect the original copy of the data on the Instrument Daemon machine.

2.3. Instrument Daemon Samples

Once added to a project, samples from an Instrument Daemon will appear like any other sample on the project data page.

Before the Instrument Daemon starts the upload for a sample, placeholder samples will be created and displayed on the data page.

Search CLEAN	R Done	0 Trocessing	1 📋 Pending 1 📋	Failed 0 📋	Actions C
Sample Name	No. of Fraction	Enzyme	Activation Method	Instrument	Progress
<pre>e151108_Orbitrap (Orbi-Orbi)_e151108_Orbitrap /</pre>	1 Fraction	Trypsin	CID	Orbitrap (Orbi-Orbi)	Processing
e151108_Orbitrap (Orbi-Orbi)_e151108_Orbitrap fraction1	1 Fraction	Trypsin	CID	Orbitrap (Orbi-Orbi)	Pending

When a sample finishes loading, the sample names and file names from the placeholder sample will be updated with the actual names.

Search CLEAR	Done	2 Processing 0	Pending 0	A Failed 0	ctions C
Sample Name	No. of Fraction	Enzyme	Activation Method	Instrument	Progress
🗢 T03 🎤	1 Fraction	Trypsin	CID	Orbitrap (Orbi-Orbi)	Done
e151108_SS1040HCD_L_T03.raw MS1 Scan Count:	6128 MS2 Scan Co	unt: 8876 Elution time:	122 min 🛛 🔋 🛨		
🗢 B03 🧪	1 Fraction	Trypsin	CID	Orbitrap (Orbi-Orbi)	Done
e151108_SS1040HCD_R_B03.raw MS1 Scan Count:	5012 MS2 Scan Co	ount: 10928 Elution time	e: 123 min 🛛 🔋 🛨		